Problems with a lot of Potential: Energy Optimization on Compact Spaces

A DISSERTATION SUBMITTED TO THE FACULTY OF THE UNIVERSITY OF MINNESOTA BY

Ryan William Matzke

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

Dmitriy Bilyk

May 2021

© Ryan William Matzke 2021 ALL RIGHTS RESERVED

Acknowledgments

I do not think that I can ever completely express my gratitude towards my adviser, Professor Dmitriy Bilyk. When I was applying for the NSF Graduate Research Fellowship in my first year of graduate school, with a project in Additive Combinatorics, a subject no one at the University of Minnesota particularly studies, Dmitriy was the only one who was willing to advise the project. When I asked him what he did for research, he offered for me, someone he barely knew, to join his current research, letting me learn about Discrepancy Theory and Energy Optimization by working on new problems, and getting me interested in and excited about the two subjects. He helped me work through the culture shock I had as I switched from undergraduate math courses and research to the much harder work of being a professional mathematician and the impostor syndrome I dealt with as I faced, for the first time in my life, academic challenges I wasn't sure I could overcome. I am deeply grateful for Dmitriy's consistent, unyielding support, his thoughtful feedback and advice, and the independence he encouraged, all of which has helped me grow into the mathematician I couldn't imagine being in my first year. I also appreciate him as a person, bringing life with him wherever he goes, caring about others, and making research such a fun and exciting experience. So, to my mentor, collaborator, role model, and friend, Dmitriy Bilyk, thank you so much. You've helped me make my dream job a reality. It has been an honor and a blast being your first student and I hope I continue to make you proud.

I don't think I can fully express my appreciation of Professor Béla Bajnok either. He put a great deal of work, thought, and fun into the Abstract Math and research courses he created at Gettysburg College, and by far these courses, as well as Béla himself, are most responsible for my decision of going into mathematics, a subject I never considered pursuing when I began college. Through these courses, I learned how much more there was to mathematics that just using formulas or procedures, how it could be so much more fun and challenging, how to discuss and present mathematical ideas, and what it meant to do research. Course by course, I grew immensely as a mathematician and fell more in love

with the subject. Beyond the courses, Béla has been a phenomenal and fun collaborator and a pillar of support and mentorship, encouraging my continued growth, giving me guidance on how to do that, and believing in me. When I was stuck deciding between pursuing Physics or Mathematics, Béla helped me see that I would benefit most from focusing more on one subject rather than an even split between the two. By letting go of Physics, I was able to do a semester abroad in the Budapest Semesters in Mathematics program, participate in several research projects and programs, and attend more conferences as an undergraduate, all of which helped me get accepted into an excellent doctorate program. So, to my mentor, collaborator, role model, and friend Béla Bajnok, thank you so much. You helped me realize what my dream job is and have helped me all along the way.

To the many teachers of Neshaminy School District and professors and staff of Gettysburg College and the University of Minnesota, as well as the various mathematicians worldwide, who have given me several positive experiences as a student and supported me on my journey to this point, thank you. In particular, I want to thank Randi Danziger, who taught a class called Academic Enrichment (A.E.) at my elementary school. A.E. was a course that covered a wide variety of topics not normally covered in standard classes and designed for students that needed something more challenging and academically engaging. It was a class I desperately needed, but for a time it seemed that I couldn't take it. I have had Attention Deficit Hyperactivity Disorder my entire life, something that made my first few years of school very difficult, and near the end of second grade I was diagnosed and began taking medication that drastically improved my ability to concentrate. However, this seemingly came too late, as I had already failed the test to get into the class for third grade, due to my struggle with focus, and the school's administration said that it didn't matter that I had been diagnosed and medicated, I couldn't join A.E. since I didn't pass the test. Having had me as a student for A.E. in first and second grade, Randi knew that I needed the class and that I had been diagnosed and medicated. When she found out what we had been told by the administration, she told my parents how I could retake the test with the

school psychologist, and I passed with flying colors. It's a good thing too, because while I was bursting with curiosity as a child, A.E. classes helped me develop a true love for learning and critical thinking, qualities that have stuck with me into my adult life. I want to thank Darren Glass, who was my undergraduate adviser and my first math professor at Gettysburg College. Darren and his Honors Calculus 2 course, which focused on only sequences, series, and differential equations, instead of all the tricks for integration, convinced me to try the college's introductory proof-writing course (Abstract Math 1), which was where I truly started to fall in love with mathematics. Darren was the first person to show me how mathematics could involve a higher level of critical thinking and how it was more than just finding answers to problems. He guided me through undergrad, helping me find and partake in many opportunities that furthered my development as a mathematician and supported me throughout and beyond, and I am very grateful to him. I am also very thankful of Professors Pete Johnson and Overtoun Jenda, who gave me my first research experience outside of Gettysburg and outside of Additive Combinatorics through their REU at Auburn University, as well as the many opportunities they gave me to present my work or participate in more research collaborations through the Masamu program with SAMSA. They have been incredibly supportive throughout my mathematical career, and have given me several wonderful experiences with mathematics which have certainly played a role in my growth as a mathematician.

I am grateful as well to my collaborators of the work in this dissertation beyond my adviser: Feng Dai, Damir Ferizović, Alexey Glazyrin, Josiah Park, and Oleksandr Vlasiuk. I always have more fun working on research with others, and I appreciate so very much how great of an experience it has been to work with you all so far and how much you've helped me learn and grow over the past six years. I would like to thank my Preliminary and Final Oral Exam committee members, Max Engelstein, Svitlana Mayboroda, Andrew Odlyzko, and Vic Reiner for all of their feedback and everything they helped me learn through my two exams. I would also like to thank Greg Blekherman, Henry Cohn, David de Laat, Axel Flinth, Michael Lacey, Tongseok Lim, Galyna Livshyts, Svitlana Mayboroda, Robert McCann, Bruno Poggi, Danylo Radchenko, Bruce Reznick, Alexander Reznikov, Daniela Schiefeneder, Alexander Shapiro, Stefan Steinerberger, and Yao Yao for the helpful discussions my collaborators or I have had with them during the course of our work.

I asked many friends, family members, and colleagues to proofread this dissertation, and for their effort and incredibly helpful feedback, I would like to thank Angeliki Matzke, Dmitriy Bilyk, Megan Aslanian, Esther Banaian, Katie Behling, Olivia Cannon, Mary Clarke, Jennifer Dorey, Alexey Glazyrin, Taylor Jacovich, Michelle Janossy, Steve Kenyon, Scott Magers, Jessica Matzke, Julie Matzke, Lauren McCabe, Brian McEvoy, Jackie McMillen, Lisa Naples, Amanda Plimpton, Maddie Price, Cecily Santiago, Josie Spencer, Stefan Steinerberger, Eric Stucky, and Oleksandr Vlasiuk. I am also grateful to Adrienne Sands for helping me with the template for this dissertation.

Finally, I am incredibly grateful to all of my friends and family for making my life so much brighter and loving and supporting me throughout everything, on both good days and rough days, and helping me make it though all of my struggles, especially my impostor syndrome. To my siblings and parents, thank you so much for helping me become who I am today, and believing in me, whether or not you think you know what I do. Thank you for being there for me when I didn't think I could become a doctor, and making sure I knew you all were proud of me no matter what, but you still thought I could make it through the program. Thank you for helping me realize what I needed to change in order to tackle the biggest challenge of my life and helping me believe that I could do it again. Thank you for the love and support you show me everyday. To my parents, thank you for always trying to do what was best for me. Thank you for getting me help when I needed it, whether it was speech therapy or getting me tested for ADHD. Thank you for providing me with all the opportunities you could, while never pushing me to do anything I didn't want to. Thank you for teaching me a good work ethic and how to be independent. Thank you for teaching me how to work for my goals and to believe in my ability to do so. Thank you

for being such great role models. Thank you for being such phenomenal parents. To my wife, Angeliki, thank you for reading through my entire dissertation, something only my committee members and I did, and telling me all the things I did wrong. Thank you for loving me and supporting me through all the madness of grad school. Thank you for being there for me each and everyday no matter what that day brings. Thank you for calming me down when I've been stressed and close to panicking, and for checking in to make sure I haven't forgotten to eat or drink when I've been working for too long. Thank you for supporting me and my dreams and getting so excited for me. It means the world to me and I love you so much.

This work was supported in part by the NSF Graduate Research Fellowship 00039202 and the University of Minnesota Doctoral Dissertation Fellowship.

Dedication

To Angeliki, Mom, Dad, Jackie, and Ben,

You make the road rise up to meet my feet, make the wind always at my back, make the sun shine warm upon my face, make the rain fall soft upon my field, and bring equilibrium to my life.

Abstract

In this dissertation, we provide a survey of the author's work in energy optimization on compact spaces with continuous potentials. We present several new results relating the positive definiteness of a potential, convexity of its energy functional, and properties of the minimizing measures of the energy, first in general spaces, then specifically on two-point homogeneous spaces, and especially on spheres. We also obtain sufficient conditions for the existence, and in some cases uniqueness, of discrete minimizers for a large class of energies. We discuss the Stolarsky Invariance Principle, which connects discrepancy and energy, as well as some analogues and generalizations of this phenomenon. In addition, we investigate some particularly interesting optimization problems, such as determining the maximum sum of pairwise angles between N points on the sphere \mathbb{S}^{d-1} and the maximum sum of angles between N lines passing through the origin, both of which are related to conjectures of Fejes Tóth. We also study the *p*-frame energies, which are related to signal processing and quantum mechanics. We show that on the sphere, the support of any minimizer of the *p*-frame energy has empty interior whenever *p* is not an even integer, and, moreover, that tight designs are the unique minimizers for certain values of p, among other results. We complete this paper by developing the theory of minimization for energies with multivariate kernels, i.e. energies for which pairwise interactions are replaced by interactions between triples, or more generally, *n*-tuples of particles. Such objects arise naturally in various fields and present subtle difference and complications when compared to the classical two-input case. We introduce appropriate analogues of conditionally positive definite kernels, establish a series of relevant results in potential theory, and present a variety of interesting examples, including some problems in probabilistic geometry which are related to multivariate versions of the Riesz s-energies.

viii

Table of Contents

	Ack	nowledgments	i
	Dedi	ication	vi
	Abst	ract	vii
1	Intr	oduction	1
	1.1	Summary of Main Results	5
	1.2	Notation	6
2	Bacl	kground	11
	2.1	Energy	11
	2.2	Positive Definite Kernels	16
	2.3	Mercer's Theorem	20
	2.4	Two-Point Homogeneous Spaces	26
		Jacobi Polynomials	30
		Antipodal Symmetry	32
	2.5	The Sphere	33
		Gegenbauer Polynomials	35
	2.6	Designs	38
3	Elen	nentary Aspects of Energy Optimization	47
	3.1	Positive Definite Functions	48

		Positive Definiteness and Inequalities for Mixed Energies	48
		Positive Definiteness and Convexity of the Energy Functional	51
		Minimizing Measures: Basic Potential Theory	53
		Energy Minimizers and Hilbert–Schmidt Operators	54
	3.2	Invariant Measures	56
		Definition, Examples, and Comments	56
		A Crucial Identity	58
		Conditional Positive Definiteness and Energy Minimization	60
		Conditional Positive Definiteness vs. Positive Definiteness up to an Addi-	
		tive Constant	61
		Local and Global Minimizers	62
	3.3	Invariant Measures and Minimizers with Full Support	65
	3.4	Energies on Two-point Homogeneous Spaces	72
	3.5	Energy on the Sphere	74
4	Sup	port of Minimizers	77
	4.1	Extreme Points for Sets of Moment-constrained Measures	81
		Applications of Karr's Theorem: Existence of Discrete Minimizers	83
	4.2	Minimizers of Energies with Analytic Kernels	86
		Applications to Polynomial Energies	88
	4.3	Linear Programming and Optimality of Tight Designs	90
		Linear Programming	91
		Properties of Orthogonal Polynomials	93
		Hermite Interpolation	95
		Optimality of Tight Designs	98
		Causal Variational Principle	103

5	Stol	arsky-type Principles	107
	5.1	Stolarsky Invariance Principle	109
	5.2	Hemispheric Stolarsky Principle and Geodesic Riesz s-Energy	114
		Hemispheric Stolarsky Principle	115
		Sum of Geodesic Distances	116
		Geodesic Distance Energy Integral	121
		Geodesic Riesz <i>s</i> -Energy	125
	5.3	The Generalized Stolarsky Principle for the Sphere	127
	5.4	The Generalized Stolarsky Principle on Compact Metric Spaces	130
6	P-fr	ame Energy	133
	6.1	Frame Energy	136
	6.2	P-frame Energy	139
	6.3	Optimality of the 600-cell	144
	6.4	Empty Interior of <i>p</i> -frame Energy Minimizers	146
	6.5	<i>p</i> -frame Energies in Non-compact Spaces	155
	6.6	Mixed Volume Inequalities	158
7	Acute Angle Energy 16		161
	7.1	New Results in all Dimensions	165
		New Bound	165
		Dimension Reduction Argument	167
	7.2	The Case of \mathbb{S}^1 Revisited	168
		Chebyshev Polynomial Expansion	169
		Fourier Series	170
		Stolarsky Principle	172
8	Ene	rgy with Multivariate Potentials	175
	8.1	Background and Definitions	180

8.2	First Principles
	Bounds on Mutual Energies
	Convexity
8.3	Minimizers of the Energy Functional
8.4	Multi-input Energy on the Sphere
8.5	Positive Definite Kernels
	General Classes of (Conditionally) <i>n</i> -positive Definite Kernels
	Three-positive Definite Kernels on the Sphere
8.6	Some Counterexamples
8.7	Semidefinite Programming
8.8	Results in Probabilistic Geometry
	Volume of the Tetrahedron/Parallelepiped
	Area of the Triangle
	Discrete Maximizers

225

Bibliography

List of Figures

- 3.1 Equivalences in Theorem 3.3.1: double line (cyan) arrows are implications that hold without additional assumptions; single line (blue) ones require *K*-invariance, but not full support; the dashed (black) arrows represent the implications which do require the assumption of full support. 67

List of Tables

2.1	A list of known tight spherical designs (with the 600-cell). Here <i>M</i> denotes	
	the strength of the design, d the dimension of the ambient space \mathbb{R}^d , and N	
	is the size of the design	45
2.2	A list of parameters for the known to exist projective tight designs (besides	
	designs in \mathbb{FP}^1 for $\mathbb{F} \neq \mathbb{R}$). Here <i>M</i> denotes the strength of the design, <i>d</i>	
	the dimension of the ambient space \mathbb{F}^d , and N is the size of the design. For	
	SIC-POVMs, the $(*)$ indicates that these exist for certain values of d , and	
	may or may not exist for all values	46

xvi

Chapter 1

Introduction

In numerous areas of mathematics and other sciences, one is faced with problems that can be reformulated as a question of minimizing the discrete or continuous pairwise interaction energy, i.e. expressions of the type

$$E_K(\omega_N) := \frac{1}{N^2} \sum_{x, y \in \omega_N} K(x, y) \text{ or } I_K(\mu) := \int_{\Omega} \int_{\Omega} K(x, y) d\mu(x) d\mu(y)$$
(1.1)

where ω_N is a discrete set of *N* (not necessarily distinct) points in Ω , μ is a Borel probability measure on the domain Ω , and *K* is the potential function describing the pairwise interaction. Perhaps one of the most famous examples of such a problem is the 1904 Thomson Problem, asking for the minimum electrostatic potential energy configuration(s), i.e. equilibrium distributions (according to Coulomb's Law), of *N* electrons on the unit sphere, which is notoriously still open for most values of *N*. Note that in this setting, where $K(x,y) = \frac{1}{||x-y||}$, as well as others where $K(x,x) = \infty$ for all $x \in \Omega$, we remove the diagonal terms from the sum in (1.1). In an abstract sense, one can generally view ω_N as a collection of particles (and μ as a charge distribution) which repel or attract according to the potential *K*.

The interactions described by (1.1) model many natural phenomena beyond electrostatics: swarm behavior, self-assembly in computational chemistry, patterns of pores on spherical pollen grains, and the structure of molecules, to name a few. The problem of determining optimal measures or point distributions for (1.1) also appears in a wide variety of more abstract settings, such as signal processing, coding theory, optimal transport, discrepancy theory, and discrete geometry, among others. In particular, energy optimization provides a method of distributing points on a manifold (i.e. the discretization of a manifold), which arises in several contexts that are of interest to the scientific community: quadrature rules, information theory, interpolation schemes, finite element tessellations, statistical sampling, etc. The vast quantity of applications and connections to other areas has stimulated much study into energy optimization, which has been developed into a full-blown theory whose state of the art is well presented in [BHS19].

In the present work, we add to the current theory by providing new methods to determine minimizers for certain classes of energies. While characterizing optimizers for energies (1.1) is of particular interest, this can often prove to be a difficult problem to address completely, so we also present methods to determine certain general properties, such as the discreteness or concentration of a minimizing measure's support. We also apply our methods to a number of interesting potentials.

We collect all of the necessary preliminary material in Chapter 2. In Chapter 3 we explore, and show connections between, a variety of properties of kernels K, their energy integrals I_K , and local and global minimizers of such energies. In particular, we provide numerous necessary and sufficient conditions in Section 3.3 to determine whether or not a measure μ with full support is a minimizer of I_K . In Sections 3.4 and 3.5, we provide some additional results specific to the compact, connected, two-point homogeneous spaces and the sphere, respectively. The new results in this chapter come predominantly from [BMV].

It shall be shown in Chapter 3 that if *K* is not positive definite (modulo an additive constant) on Ω then any minimizer of I_K cannot have full support. This naturally begs the question of what the support of minimizing measures could be in this situation. In Chapter 4, we provide some answers to this question, particularly on the compact, connected,

two-point homogeneous spaces. Section 4.1 uses the structure of extreme points of sets of moment-constrained measures to provide a sufficient condition for I_K to have a discrete minimizer on the sphere. In Section 4.2, we show that for analytic kernels that are not positive definite (modulo a constant), the support of any minimizer must have empty interior. In Section 4.3, we make use of a linear programming method to show that tight designs are (possibly unique) minimizers of certain energies. This chapter is based on work from [BGM⁺a, BGM⁺b].

In Chapter 5, we present results that appear in [BDM18, BMV]. The problem of determining optimal configurations for E_K for a fixed number N of points on the sphere is often more delicate than finding the optimal measures for I_K . In the case that σ , the normalized Lebesgue measure on \mathbb{S}^{d-1} , is a minimizer of I_K , one might expect that as N increases, minimizers of E_K should be "uniformly distributed," in some sense, in order to approximate σ . Chapter 5 discusses using discrepancy, a popular means of measuring uniformity of distribution of points on the sphere, to determine minimizers of I_K and E_K for positive definite functions K via a Generalized Stolarsky Principle. Section 5.1 provides a simple proof of the classic Stolarsky Invariance Principle, and in Section 5.2, we present an analogue that addresses a conjecture of Fejes Tóth about the maximum sum of geodesic distances on the sphere. The Generalized Stolarsky Principle on spheres, and more generally in compact metric spaces, is then given in section 5.3 and 5.4, respectively.

In Chapter 6 we discuss the *p*-frame energies, introduced by Ehler and Okoudjou in [EO12] as a generalization of the frame energy, which Benedetto and Fickus used in [BF03] to classify finite unit norm tight frames (FUNTFs). We discuss the frame energy in Section 6.1 and the more general *p*-frame energy in Section 6.2, in particular applying the results of Section 4.3 to show that for some values of *P*, tight designs minimize the *p*-frame energy, when they exist. Section 6.3 shows that the 600-cell is also a minimizer for certain *p*-frame energies. In Section 6.2, we conjecture that for $p \notin 2\mathbb{N}$, all minimizers of the *p*-frame energy are discrete. Though a proof of this conjecture remains elusive, in Section 6.4, we

show that for these *p*-frame energies on the real sphere, the support of any minimizer must have empty interior. Section 6.5 extends some of our results to a non-compact setting, and Section 6.6 connects our results to a problem from Convex Geometry. The results in this chapter are based on the work in $[BGM^+a, BGM^+b]$.

In Chapter 7, we discuss a conjecture of Fejes Tóth about the maximum sum of acute angles of points on the sphere, and present new results in this direction in Section 7.1, as well as new proofs for the conjecture on S^1 in Section 7.2. The results presented were obtained in [BM19].

Chapter 8 addresses a different type of problem than the other sections. In the last few decades, many-body interactions (i.e. kernels with three or more inputs instead of two) have become a greater subject of interest in material science, quantum mechanics, and discrete geometry. While a great deal of theory has been developed to address energy optimization problems that form from two-body interactions, i.e., those described in (1.1), very little exists for optimizing energies of the form

$$\frac{1}{N^n} \sum_{j_1=1}^N \cdots \sum_{j_n=1}^N K(x_{j_1}, ..., x_{j_n}) \text{ or } \int_{\Omega} \cdots \int_{\Omega} K(x_1, ..., x_n) d\mu(x_1) ... d\mu(x_n).$$
(1.2)

To address these deficiencies, we present the first steps in developing a general theory for such energies. In Section 8.1, we introduce some relevant notation and definitions, in particular defining *n*-positive definiteness, a generalization of positive definiteness. In Sections 8.2, 8.3, and 8.4, we provide various necessary and sufficient conditions for a measure μ to be a minimizer of the *n*-input energy. Many of these results are analogues, for our multivariate setting, of results in Chapter 3. In Sections 8.5 and 8.6, we present examples of *n*-input kernels which are (conditionally) *n*-positive definite and which are not, respectively. In Section 8.7, we adapt the semidefinite programming method from [BV08] to our setting, and apply that method, as well as others, to answer certain questions from probabilistic discrete geometry. This chapter is based on work from [BFG⁺a, BFG⁺b].

1.1 Summary of Main Results

The following is a list of the main results of this thesis.

- A list of properties equivalent to a kernel *K* being conditionally positive definite when there exists a *K*-invariant probability measure of full support (Theorem 3.3.1). Similar characterizations are provided for positive definiteness (Theorem 3.3.2) and conditionally strict positive definiteness (Theorem 3.3.3).
- The existence of discrete minimizers for the energy of any rotationally invariant kernel on the sphere with finitely many positive Gegenbauer coefficients (Theorem 4.1.3).
- A proof that the energy of a real-analytic kernel that is not positive definite (modulo a constant) on the sphere can only be minimized by a measure whose support has empty interior (Theorem 4.2.1).
- Characterizations of large classes of energies on spheres and projective spaces for which tight designs are minimizers (Theorem 4.3.1) and for which tight designs are unique minimizers (Theorem 4.3.12).
- An explicit connection between L^2 discrepancy with respect to hemisphere and the sum of geodesic distances of any point set on the sphere (Theorem 5.2.1). This is used to provide a complete solution of a conjecture of Fejes Tóth, which characterized all configurations on a sphere that maximize the sum of geodesic distances between points on the sphere (Theorem 5.2.10).
- Generalizations of the Stolarsky Invariance Principle for positive definite kernels on the sphere (Theorem 5.3.1) and positive definite kernels on general compact metric spaces (Theorem 5.4.1).

- A classification of *p*-frame energies for which tight designs are the unique minimizers (Theorem 6.2.3).
- A proof that the minimizers of the *p*-frame energy on the sphere necessarily have empty interior, whenever *p* is not a even integer (Theorem 6.4.1).
- The best known bound on the acute angle energy, which comes from a conjecture of Fejes Tóth (Theorem 7.1.1).
- Development of an initial theory for energy optimization with multivariate kernels (Chapter 8). In particular:
 - Necessary (Theorem 8.3.2) and sufficient (Theorem 8.3.3) conditions for a measure to minimize an multivariate energy.
 - The relationship between the local minimizers of a multivariate energy, and the corresponding two-input potential (Theorem 8.3.8).
 - The determination of a large class of multivariate energies on the sphere for which the uniform measure is a minimizer, based on a semidefinite programming method (Theorem 8.7.1).
 - A characterization of the maximizing measures of the expected value of the squared area of a triangle with i.i.d. vertices on the sphere (Theorem 8.8.4).

1.2 Notation

We shall use the following notation throughout the text.

- \mathbb{R} The set of real numbers
- \mathbb{C} The set of complex numbers
- \mathbb{H} The set of quaternions

\bigcirc	The set of octonions
\mathbb{F}	$\mathbb{R}, \mathbb{C}, \mathbb{H}, $ or \mathbb{O}
$\langle x, y \rangle_H$	The inner product for some Hilbert space H
$\langle\cdot,\cdot angle$	The inner product on \mathbb{R}^d , \mathbb{C}^d , or \mathbb{H}^d
$ x _H$	The norm for some space <i>H</i>
x	The Euclidean norm on \mathbb{R}^d , \mathbb{C}^d , or \mathbb{H}^d
$\dim_{\mathbb{R}}(\mathbb{F})$	The dimension of \mathbb{F} as a real manifold
\mathbb{N}	The set of positive integers
\mathbb{N}_0	The set of nonnegative integers
$\Omega = (\Omega, ho)$	A metric space, compact unless stated otherwise
$\Phi = \Phi^{(\alpha,\beta)}$	A compact, connected, two-point homogeneous space
\mathbb{FP}^{d-1}	The projective space over $\mathbb F$
\mathbb{S}^{d-1}	The $(d-1)$ -dimensional unit sphere in \mathbb{R}^d
α	For \mathbb{S}^{d-1} or \mathbb{FP}^{d-1} , $\frac{(d-1)\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1$
β	α for $\Phi = \mathbb{S}^{d-1}$ and $\frac{\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1$ for $\Phi = \mathbb{FP}^{d-1}$
$\mathbb{S}^{d-1}_{\mathbb{F}}$	The $(d-1)$ -dimensional unit sphere in \mathbb{F}^d
$L^2(\Omega,\mu)$	The space of real-valued square-integrable functions on Ω
$L^2(\Omega,\mu,\mathbb{C})$	The space of complex-valued square-integrable functions on Ω
$V_n = V_n^{(\alpha,\beta)}$	The eigenspaces of the Laplace-Beltrami operator on $\Phi^{(\alpha,\beta)}$
$Y_{n,k} = Y_{n,k}^{(\alpha,\beta)}$	The basis elements of $V_n^{(\alpha,\beta)}$
\mathscr{H}_n^d	The space of spherical harmonics of degree n on \mathbb{S}^{d-1}
λ	For the sphere \mathbb{S}^{d-1} , $\frac{d-2}{2}$
$\gamma_{lpha,eta}$	$2^{\alpha+\beta+1}B(\alpha+1,\beta+1)$, where <i>B</i> is the beta function
A_{d-1}	The surface area of \mathbb{S}^{d-1}
Vol _d	The volume of the unit ball in \mathbb{R}^d
η	The normalized uniform surface measure on Φ
σ	The normalized $(d-1)$ -dimensional Hausdorff measure on \mathbb{S}^{d-1}

$d\mathbf{v}^{(\pmb{lpha},\pmb{eta})}(t)$	$\frac{1}{\gamma_{\alpha,\beta}}(1-t)^{\alpha}(1+t)^{\beta}dt$
$dw_{\lambda}(t)$	$\frac{\Gamma(\lambda+1)}{\sqrt{\pi}\Gamma(\lambda+\frac{1}{2})}(1-t^2)^{\lambda-\frac{1}{2}}dt$
$\mathbb{P}(\Omega)$	The set of Borel probability measures on Ω
$\mathbb{P}^*(\Omega)$	The subset of $\mathbb{P}(\Omega)$ satisfying $\int_{\Omega} x ^2 d\mu(x) = 1$
$\widetilde{\mathbb{P}}(\Omega)$	The set of signed probability measures on Ω
$\mathscr{M}(\Omega)$	The set of finite signed Borel measures on Ω
$\mathscr{Z}(\Omega)$	The set of finite signed Borel measures v such that $v(\Omega) = 0$
$\mathscr{B}(\Omega)$	The set of finite positive Borel measures on Ω
θ	The geodesic (Riemannian) metric on Φ
ϑ^*	The normalization of ϑ , i.e. $\frac{\vartheta}{\pi}$
D	The Euclidean metric, i.e. $\mathscr{D}(x, y) = x - y $
ρ	The chordal metric on Φ
$\tau(x,y)$	$\cos(\vartheta(x,y))$ for $x, y \in \Phi$
Z	The cardinality of a finite set Z
$P_n^{(\alpha,\beta)}$	The Jacobi polynomials, scaled so that $P_n^{(\alpha,\beta)}(1) = \dim(V_n^{(\alpha,\beta)})$
$C_n = C_n^{(\alpha,\beta)}$	The Jacobi polynomials, scaled so that $C_n(1) = 1$
$Q_n = Q_n^{(\alpha,\beta)}$	The monic Jacobi polynomials, i.e. scaled so the leading coefficient is 1
C_n^{λ}	The Gegenbauer polynomials
$\widehat{F}_n = \widehat{F}_n^{(\alpha,\beta)}$	The Jacobi coefficient of F with respect to $C_n^{(\alpha,\beta)}$
$\widehat{F}(n, \lambda)$	The Gegenbauer coefficients of F
C	A code (finite point configuration) on Φ
$\mathscr{A}(\mathscr{C})$	The "distance set" of \mathscr{C} , $\{\tau(x, y) : x, y \in \mathscr{C}\}$
C(z,h)	The spherical cap of height <i>h</i> centered at <i>z</i> , i.e. $\{x \in \mathbb{S}^{d-1} : \langle x, z \rangle > h\}$
$p_{\mathbb{F}}$	The projection from the sphere $\mathbb{S}^{d-1}_{\mathbb{F}}$ to the projective space \mathbb{FP}^{d-1}
$e_1,, e_d$	The standard orthonormal basis in \mathbb{R}^d
\overline{Y}	The complex-conjugate of a function <i>Y</i>
$diam(\Omega)$	The diameter of Ω , i.e. $\sup_{x,y\in\Omega} \rho(x,y)$

$\operatorname{supp}(\mu)$	The support of μ .
\overline{A}	The closure of a set A
U°	The interior of a set U

Chapter 2

Background

We begin with some important background material. Section 2.1 contains the discussion of background for our study of energy, including an introduction to the Riesz *s*-energies. Sections 2.2 and 2.3 introduce positive definiteness and various consequences of this property, including Mercer's Theorem and resulting characterization of positive definite kernels. In Section 2.4, we discuss compact, connected, two-point homogeneous spaces and covers the relevant properties of such spaces and the corresponding system of orthogonal, isometry invariant polynomials. Section 2.5 does the same for the real unit sphere. In Section 2.6, we provide an exposition on designs and, in particular, tight designs.

2.1 Energy

Unless otherwise noted, in what follows, we assume that (Ω, ρ) is a compact metric space and that the kernel (also referred to as potential) $K : \Omega^2 \to \mathbb{R}$ is continuous and symmetric, i.e. for all $x, y \in \Omega$, K(x, y) = K(y, x). We denote by $\mathscr{M}(\Omega)$ the set of finite signed Borel measures on Ω , and by $\mathbb{P}(\Omega)$ the set of Borel probability measures on Ω . Given $\mu, \nu \in \mathscr{M}(\Omega)$, we define their mutual/mixed *K*-energy as

$$I_{K}(\mu, \nu) = \int_{\Omega} \int_{\Omega} K(x, y) d\mu(x) d\nu(y), \qquad (2.1)$$

and the (continuous) K-energy of μ to be

$$I_K(\mu) := I_K(\mu, \mu) = \int_{\Omega} \int_{\Omega} K(x, y) d\mu(x) d\mu(y).$$
(2.2)

We are interested in finding the optimal (maximal or minimal, depending on *K*) values of $I_K(\mu)$ over all $\mu \in \mathbb{P}(\Omega)$, as well as extremal measures for which these values are achieved, i.e. equilibrium measures with respect to *K*. Note that the minimization of I_K is equivalent to the maximization of I_{-K} , so we will often discuss optimization problems in terms of minimization, but switch to energy maximization when it is more convenient. In various parts of the text, we will be interested in local minimizers or global minimizers over different sets of measures, so it is worth specifying here that whenever we say a measure μ is a "minimizer" of I_K without any additional conditions, we mean that μ is a global minimizer of I_K over $\mathbb{P}(\Omega)$, i.e. for all $v \in \mathbb{P}(\Omega)$, $I_K(\mu) \leq I_K(v)$. We denote the minimal continuous *K*-energy, i.e. the Wiener constant, by

$$\mathscr{I}_{K}(\Omega) = \inf_{\mu \in \mathbb{P}(\Omega)} I_{K}(\mu).$$
(2.3)

A naturally related question is that of discrete energy optimization. Let $\omega_N = \{z_1, z_2, ..., z_N\}$ be an *N*-point configuration (multiset) in Ω for $N \ge 2$. We define the discrete *K*-energy of ω_N to be

$$E_K(\omega_N) = \frac{1}{N^2} \sum_{x,y \in \omega_N} K(x,y)$$
(2.4)

and the minimal discrete N-point K-energy of Ω as

$$\mathscr{E}_{K}(\Omega, N) := \inf_{\omega_{N} \subseteq \Omega} E_{K}(\omega_{N}).$$
(2.5)

Note that since we are working with continuous K, the energy I_K is well-defined for all finite signed Borel measures. In addition, the compactness of Ω implies that $\mathbb{P}(\Omega)$ is weak^{*} compact and guarantees the existence of global minimizers for both discrete and continuous energies.

The definitions of discrete (2.4) and continuous (2.2) energies are compatible in the sense that

$$E_K(\omega_N) = I_K(\mu_{\omega_N}), \text{ where } \mu_{\omega_N} = \frac{1}{N} \sum_{x \in \omega_N} \delta_x$$
 (2.6)

and due to the weak^{*} density of the linear span of Dirac masses in $\mathbb{P}(\Omega)$, we have the following lemma [Cho58, FN08].

Lemma 2.1.1. For any continuous kernel K on Ω ,

$$\lim_{N \to \infty} \mathscr{E}_K(\Omega, N) = \mathscr{I}_K(\Omega).$$
(2.7)

Moreover, suppose that $\{\omega_N\}_{N=2}^{\infty}$ is a sequence of N-point configurations on Ω satisfying

$$\lim_{N\to\infty}E_K(\omega_N)=\mathscr{I}_K(\Omega),$$

and define, for each $N \ge 2$,

$$\mu_{\omega_N} = \frac{1}{N} \sum_{x \in \omega_N} \delta_x.$$

If $\mathcal{N} \subset \mathbb{N}$ is a sequence of integers such that μ_{ω_N} weak^{*} converges to a probability measure μ as $N \to \infty$, $N \in \mathcal{N}$, then

$$I_K(\mu) = \mathscr{I}_K(\Omega)$$

i.e. μ is a minimizer of I_K .

For a measure $\mu \in \mathbb{P}(\Omega)$, we define the *potential* U_K^{μ} of μ with respect to *K* as

$$U_K^{\mu}(x) := \int_{\Omega} K(x, y) d\mu(y), \ x \in \Omega.$$
(2.8)

Notice that this meaning of the term "potential" is consistent with us calling *K* a potential, since the function K(x, y) is simply the potential generated by a unit point charge at *y*, i.e. $K(x, y) = U_K^{\delta_y}(x).$

Though this work predominantly focuses on energy minimization with compact domains Ω and continuous kernels K, energy optimization need not be restricted to such settings. For instance, if (Ω, ρ) is an arbitrary metric space, energy minimization is sensible so long as we can guarantee the existence of optimal discrete sets or probability measures. This clearly occurs when Ω is compact and K continuous, but for non-compact Ω , continuity of the kernel K is no longer sufficient. Instead, the compactness of the support of minimizers can be achieved by placing certain additional restrictions on our kernel K, as will be discussed in the introduction of Chapter 4, or by placing some conditions on the set of measures we optimize over, as will be discussed in Section 6.5.

Another, more common, generalization is to consider lower semi-continuous kernels $K : \Omega^2 \to (-\infty, \infty]$. In this setting, one often removes the diagonal terms from 2.4 (and must if $K(x,x) = \infty$ for all $x \in \Omega$), and can only consider the mutual energy of finite signed measures $\mu, \nu \in \mathcal{M}(\Omega)$ if $I_K(\mu, \nu)$ is well-defined. Many of the results we present in Section 3.3 have analogues or generalizations to lower semi-continuous kernels, though the proofs are often more technically demanding, by necessity.

The most famous family of lower semi-continuous kernels in Potential Theory are known as the Riesz *s*-kernels, defined by

$$D_s(x,y) = ||x - y||^{-s} \quad s \in \mathbb{R} \setminus \{0\}.$$
(2.9)

The optimization of Riesz s-energies is related to several other well-known problems

in mathematics and other sciences and has been the subject of study for over a century, with the physicist J.J. Thomson posing the problem of finding the minimizers of E_{D_1} on \mathbb{S}^2 in 1904. Though exact minimizing configurations for the Thompson problem are known only for a few values of N [And96, Föp12, Sch, Sch20, Yud92], the asymptotic behavior of minimizers of the discrete Riesz *s*-energies are better understood. In particular, on the sphere \mathbb{S}^{d-1} , it is known that for all s > 0, minimizing point configurations are asymptotically uniformly distributed with respect to the normalized (d-1)-dimensional Hausdorff measure on the sphere, σ [KS98, Lan72]. For $s \ge d$, this result, known as the "Poppy-Seed Bagel Theorem," has been generalized to *d*-rectifiable manifolds [HS05]. This is a natural consequence of the fact that as $s \to \infty$ with N fixed, the discrete Riesz *s*-energy is increasingly dominated by the term(s) involving the smallest pairwise distances, a property that, on the sphere, leads to the limiting case (i.e. $s = \infty$) being the best-packing problem [CS99, BHS19].

In the case where s < 0, we see that the Riesz *s*-kernels are continuous, and the optimization problem becomes one of maximization instead of minimization. The maximum value of I_{D_s} for s < 0 was first studied by Pólya and Szegö on spheres and balls in [PS31]. In [Bjö56], Björck determined properties of maximizing measures on general compact sets.

Theorem 2.1.2. Let $d \ge 1$, s < 0, and $\Omega \subseteq \mathbb{R}^d$ be compact. If μ is a maximizer of I_{D_s} over $\mathbb{P}(\Omega)$, then the following hold:

- 1. $\operatorname{supp}(\mu)$ is a subset of the boundary of Ω , i.e. $\overline{\Omega} \setminus \Omega^\circ$, where $\overline{\Omega}$ and Ω° are the closure and interior of Ω in \mathbb{R}^d , respectively.
- 2. *if* s < -1, then supp (μ) is a subset of the extreme points of the convex hull of Ω .
- 3. If s < -2, then μ is a discrete measure consisting of no more than d + 1 point masses.
- 4. If 0 > s > -2, then μ is the unique maximizer of I_{D_s} .

Moreover, on the sphere, Björck was able to completely characterize the maximizer of these Riesz *s*-energies on the unit sphere \mathbb{S}^d , see Theorem 5.2.9.

We finish this section with a fairly basic and intuitive idea, but one which will play an important role in our results.

Lemma 2.1.3. If $\mu \in \mathbb{P}(\Omega)$ is a minimizer of I_K , then for all $C \in \mathbb{R}$, μ is a minimizer of I_{K+C} . Likewise, if ω_N is a minimizer of E_K over all N-point configurations in Ω then for all $C \in \mathbb{R}$, ω_N is also a minimizer of E_{K+C}

Proof. Let $C \in \mathbb{R}$. For all $v \in \mathbb{P}(\Omega)$,

$$I_{K+C}(v) = I_K(v) + (v(\Omega))^2 C = I_K(v) + C \ge I_K(\mu) + C = I_{K+C}(\mu).$$

The discrete case follows immediately from the continuous case.

For a more comprehensive exposition on Energy Optimization, especially one that discusses lower semi-continuous kernels and Riesz *s*-energy in greater detail, we refer the reader to [BHS19].

2.2 **Positive Definite Kernels**

We now recall the classical notion of positive definite kernels, which play an extremely important role in various areas of mathematics, such as Partial Differential Equations, Machine Learning, and Probability. Here, we will focus on their relation to energy minimization problems, but an exposition on their role in other areas can be found in [Fas11]. We state the definition in the form most relevant to our work.

Definition 2.2.1. A kernel $K : \Omega^2 \to \mathbb{R}$ is called **conditionally positive definite** if for every $v \in \mathscr{Z}(\Omega)$ (i.e. finite signed Borel measures satisfying $v(\Omega) = 0$), $I_K(v) \ge 0$.

If, moreover, $I_K(v) \ge 0$ for every finite signed Borel measure, i.e. $v \in \mathcal{M}(\Omega)$, then we call K positive definite.

We call a kernel strictly positive definite or conditionally strictly positive definite if it is positive definite or conditionally positive definite, respectively, and $I_K(v) = 0$ only if v(A) = 0 for all Borel sets $A \subseteq \Omega$.

If there exists some $C \in \mathbb{R}$ such that K + C is a (strictly) positive definite kernel, we call K (strictly) positive definite modulo an additive constant (or up to an additive constant).

A more standard way of defining positive definiteness of a kernel $K : \Omega^2 \to \mathbb{R}$ is by requiring, for every $n \in \mathbb{N}$ and $\{z_i\}_{i=1}^N \subset \Omega$, the matrix $[K(z_i, z_j)]_{i,j=1}^N$ is positive semidefinite, i.e. for any sequence $\{c_i\}_{i=1}^N \subset \mathbb{R}$, the kernel K satisfies the inequality

$$\sum_{i,j=1}^{N} c_i c_j K(z_i, z_j) \ge 0.$$
(2.10)

Since our kernel *K* is continuous, this is clearly equivalent to Definition 2.2.1 due to the weak^{*} density of discrete measures in $\mathcal{M}(\Omega)$.

Occasionally in this text, especially when focusing on energy maximization, we will also want to make use of the negative definiteness of a kernel. We shall call a kernel *K* negative definite if the kernel -K is positive definite.

A constant, positive kernel, i.e. K(x,y) = c > 0 for all $x, y \in \Omega$, is clearly positive definite, so it is clear such kernels always exist. Moreover, we can construct continuous positive definite kernels in the following way:

Lemma 2.2.2. For $j \in \mathbb{N}_0$, let $\lambda_j \ge 0$ and $\phi_j : \Omega \to \mathbb{R}$ be continuous. Then if the series converges absolutely and uniformly, the kernel

$$K(x,y) = \sum_{j=0}^{\infty} \lambda_j \phi_j(x) \phi_j(y)$$
(2.11)

(which is continuous, due to uniform convergence) is positive definite.

Proof. Let $\mu \in \mathcal{M}(\Omega)$. Then

$$I_{K}(\mu) = \sum_{j=0}^{\infty} \lambda_{j} \int_{\Omega} \int_{\Omega} \phi_{j}(x)\phi_{j}(y)d\mu(x)d\mu(y)$$
$$= \sum_{j=0}^{\infty} \lambda_{j} \left(\int_{\Omega} \phi_{j}(x)d\mu(x)\right)^{2}$$
$$\geq 0.$$

In Section 2.3, we will show that this actually provides a characterization of positive definite kernels.

In addition to a relatively simple characterization, positive definite kernels have several other useful properties, many of which will be discussed in Chapter 3. To begin with, they are closed under addition, multiplication (a result known as the Schur product theorem), and limits of uniformly convergent sequences.

Lemma 2.2.3. If K and L are positive definite kernels on Ω , then so are K + L and KL. If $K_1, K_2, ...,$ are positive definite and $\lim_{n\to\infty} K_n = K$ uniformly, then K is positive definite. The statements regarding the sum and limit (but not the product) hold if we replace positive definiteness with conditional positive definiteness.

Though the theory tends to focus on the (strict) positive definiteness of kernels, Lemma 2.1.3 shows us that adding a constant does not affect the minimizer, so we often consider kernels that are (strictly) positive definite modulo a constant. However, adding a constant never changes conditional (strict) positive definiteness, as for all $C \in \mathbb{R}$ and $v \in \mathscr{Z}(\Omega)$,

$$I_{K+C}(\mathbf{v}) = I_K(\mathbf{v}) + (\mathbf{v}(\Omega))^2 C = I_K(\mathbf{v}).$$

Since (strict) positive definiteness implies conditional (strict) positive definiteness, we arrive at the following lemma.
Lemma 2.2.4. *If K is (strictly) positive definite modulo a constant, then K is conditionally (strictly) positive definite.*

In Section 3.3, we will demonstrate that the converse of Lemma 2.2.4 can hold under certain conditions. However, it does not hold in general:

Example 2.2.5. Consider $K : (\mathbb{S}^{d-1})^2 \to \mathbb{R}$ defined by $K(x, y) = x_1 + y_1$, where $x_1 = \langle x, e_1 \rangle$ here. Then K is conditionally positive definite, but not positive definite modulo a constant.

Proof. For all $v \in \mathscr{Z}(\mathbb{S}^{d-1})$,

$$I_K(\mathbf{v}) = 2 \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} x_1 d\mathbf{v}(x) d\mathbf{v}(y) = 0,$$

so *K* is conditionally positive definite.

Now we show there is no constant C such that K + C is positive definite. If C < 0, then

$$I_{K+C}(\boldsymbol{\sigma}) = 2 \int_{\mathbb{S}^{d-1}} x_1 d\boldsymbol{\sigma}(x) + C = C < 0.$$

Suppose that $C \ge 0$ and let $\mu = (C+1)\delta_{-e_1} - C\delta_{e_1} \in \mathscr{M}(\mathbb{S}^{d-1})$. Then

$$I_{K+C}(\mu) = 2\mu(\mathbb{S}^{d-1}) \int_{\mathbb{S}^{d-1}} x_1 d\mu(x) + C(\mu(\mathbb{S}^{d-1}))^2$$
$$= 2(-2C-1) + C$$
$$= -3C - 2 < 0.$$

Our proof is now complete.

We complete this section by presenting an interesting result that allows one to construct positive definite kernels from conditionally positive definite ones, a result we shall use in Chapter 8.

Lemma 2.2.6 (Chp. 3, Lemma 2.1, [BCR84]). Let $x_0 \in \Omega$, $K : \Omega^2 \to \mathbb{R}$ be a kernel, define

$$K'(x,y) := K(x,y) + K(x_0,x_0) - K(x,x_0) - K(x_0,y)$$

Then K' is positive definite if and only if K is conditionally positive definite. If $K(x_0, x_0) \le 0$ and

$$K'_0(x,y) := K(x,y) - K(x,x_0) - K(x_0,y),$$

then K'_0 is positive definite if and only if K is conditionally positive definite.

2.3 Mercer's Theorem

Let μ be a Borel probability measure on Ω and let K be a continuous kernel on Ω . We shall consider the operator $T_{K,\mu}$ associated to K on the space of real-valued functions on Ω that are square-integrable with respect to μ , $L^2(\Omega,\mu)$. This is a linear integral operator, with kernel K, defined by

$$T_{K,\mu}\psi(x) = \int_{\Omega} K(x,y)\psi(y)d\mu(y).$$
(2.12)

Lemma 2.3.1. Let $\widetilde{\Omega} = \operatorname{supp}(\mu)$. The operator $T_{K,\mu}$ is self-adjoint and Hilbert-Schmidt, and the eigenfunctions of $T_{K,\mu}$ corresponding to non-zero eigenvalues are continuous on $\widetilde{\Omega}$. The kernel K is positive definite on $\widetilde{\Omega}$ if and only if $T_{K,\mu}$ is a positive operator on $L^2(\Omega,\mu)$.

Proof. Self-adjointedness immediately follows from K(x, y) being symmetric. Since Ω is compact (and hence $\widetilde{\Omega}$ is also) and *K* continuous, we know that

$$\int_{\Omega} \int_{\Omega} |K(x,y)|^2 d\mu(x) d\mu(y) < \infty,$$

which implies that $T_{K,\mu}$ is Hilbert-Schmidt.

Now, suppose that $T_{K,\mu}\phi = \lambda\phi$ for $\lambda \neq 0$. Then the representation

$$\phi(x) = \frac{1}{\lambda} \int_{\Omega} K(x, y) \phi(y) d\mu(y)$$

implies that ϕ is continuous on Ω .

We now show that the positive definiteness of *K* and the positivity of $T_{K,\mu}$ are equivalent. If *K* is positive definite on $\widetilde{\Omega}$, then for any $\psi \in L^2(\Omega, \mu)$,

$$\langle \psi, T_{K,\mu}\psi\rangle_{L^2(\Omega,\mu)} = \int_{\Omega} \int_{\Omega} K(x,y)\psi(x)\psi(y)d\mu(x)d\mu(y) = I_K(\psi(\cdot)\mu) \ge 0,$$

so $T_{K,\mu}$ is indeed positive.

Assume instead that $T_{K,\mu}$ is positive. Observe that measures, which are absolutely continuous with respect to μ and have bounded density, i.e. measures of the form $d\nu = f d\mu$, where f is a bounded Borel measurable function on $\tilde{\Omega}$, are weak* dense in $\mathscr{M}(\tilde{\Omega})$. To show this, notice that for each ball B(z,r) of radius r > 0 centered at the point $z \in \tilde{\Omega}$, we have $\mu(B(z,r)) \neq 0$, and therefore, the functions $f_r(x) = \frac{1}{\mu(B(z,r))} \mathbb{1}_{B(z,r)}(x)$ are well-defined and bounded. Obviously, the measures ν_r defined by $d\nu_r = f_r d\mu$ converge weak* to δ_z as $r \to 0$, which suffices due to weak* density of discrete measures.

Then for all such measures of the form $dv = f d\mu$, since bounded functions are in $L^2(\Omega, \mu)$, we have

$$I_K(\mathbf{v}) = \langle T_{K,\mu}f, f \rangle \geq 0$$

and by weak^{*} density, it follows that K is positive definite on Ω .

Moreover, since $T_{K,\mu}$ is a Hilbert-Schmidt operator, it is in fact a compact operator. Hence, we may apply the Spectral Theorem.

Theorem 2.3.2 (Spectral Theorem for Compact Operators). Suppose that H is a Hilbert space and $T: H \rightarrow H$ is a compact, self-adjoint operator. Then there exists an orthonor-

mal basis $\{\phi_j\}_{j=1}^{\dim(H)}$ of H consisting of eigenvectors of T and corresponding eigenvalues $\{\lambda\}_{j=1}^{\dim(H)}$ such that $|\lambda_j| \ge |\lambda_{j+1}|$ for $1 \le j < \dim(H)$. If $\dim(H) = \infty$, then $\lim_{j\to\infty} \lambda_j = 0$.

Thus, there exists an orthonormal basis $\{\phi_j\}_{j=1}^{\dim(L^2(\Omega,\mu))}$ of $L^2(\Omega,\mu)$ consisting of eigenfunctions of $T_{K,\mu}$, i.e. $T_{K,\mu}\phi_j = \lambda_j\phi_j$, where the sequence of eigenvalues satisfies $|\lambda_j| \ge |\lambda_{j+1}|$ and $\lim_{j\to\infty} \lambda_j = 0$ if $\dim(L^2(\Omega,\mu)) = \infty$. Moreover, the Spectral Theorem tells us that, in the L^2 sense,

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x) \phi_j(y)$$

For the rest of this section, we will assume that $\operatorname{supp}(\mu) = \Omega$, in other words, every non-empty open subset of Ω has strictly positive measure (if $\operatorname{supp}(\mu) \neq \Omega$, then the results of this section apply for $\widetilde{\Omega} = \operatorname{supp}(\mu)$), and that K, $\{\phi_j\}_{j=1}^{\dim(L^2(\Omega,\mu))}$, and $\{\lambda\}_{j=1}^{\dim(L^2(\Omega,\mu))}$ are as above. If K is positive definite, we know that $T_{K,\mu}$ is a positive operator, so for each $j \geq 1$,

$$\lambda_j = \langle \phi_j, \lambda_j \phi_j \rangle_{L^2(\Omega, \mu)} = \langle \phi_j, T_{K, \mu} \phi_j \rangle_{L^2(\Omega, \mu)} \ge 0.$$

Mercer's theorem then provides a series representation for any positive definite kernel K.

Theorem 2.3.3 (Mercer's Theorem). If K(x, y) is positive definite, then

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x) \phi_j(y), \qquad (2.13)$$

where the sum converges absolutely and uniformly.

It is worth noting that it would suffice to assume that *K* is continuous and that all eigenvalues of $T_{K,\mu}$ are nonnegative (which is in fact equivalent to *K* being positive definite).

Proof. As mentioned above, the fact that (2.13) holds in the L^2 sense follows from the Spectral Theorem for compact operators, hence, only uniform and absolute convergence need to be proven. They immediately follow if dim $(L^2(\Omega, \mu)) < \infty$, so we assume that $L^2(\Omega, \mu)$ is infinite dimensional.

Consider the remainder of the series, i.e. the continuous function

$$R_N(x,y) = K(x,y) - \sum_{j=1}^N \lambda_j \phi_j(x) \phi_j(y).$$

The corresponding Hilbert-Schmidt operator defined by $T_{R_N,\mu}\psi(x) = \int_{\Omega} R_N(x,y)\psi(y)d\mu(y)$ is clearly bounded and positive: if $\psi = \sum_{j=1}^{\infty} \widehat{\psi}_j \phi_j$, then

$$\langle T_{R_N,\mu}\psi,\psi\rangle = \sum_{j=N+1}^{\infty} \lambda_j |\widehat{\psi}_j|^2 \ge 0.$$

For a positive Hilbert-Schmidt operator, the kernel is non-negative on the diagonal, i.e. $R_N(x,x) \ge 0$ for each $x \in \Omega$. Indeed, assume that for some $x \in \Omega$ we have $R_N(x,x) < 0$. Then we can choose a neighborhood U of x so that R_N is negative on $U \times U$, so

$$\langle T_{R_N,\mu} \mathbb{1}_U, \mathbb{1}_U \rangle = \int\limits_U \int\limits_U R_N(x,y) d\mu(x) d\mu(y) < 0,$$

which is a contradiction.

Thus, $R_N(x,x) \ge 0$ for each $x \in \Omega$, i.e. for any $N \ge 1$ we have $\sum_{j=1}^N \lambda_j \phi_j^2(x) \le K(x,x)$, and thus

$$\sum_{j=1}^{\infty} \lambda_j \phi_j^2(x) \le K(x,x).$$

Invoking the Cauchy-Schwarz inequality, we find that

$$\left|\sum_{j=1}^{\infty} \lambda_j \phi_j(x) \phi_j(y)\right| \le \sum_{j=1}^{\infty} \lambda_j \left|\phi_j(x) \phi_j(y)\right| \le \left(\sum_{j=1}^{\infty} \lambda_j \phi_j^2(x)\right)^{1/2} \left(\sum_{j=1}^{\infty} \lambda_j \phi_j^2(y)\right)^{1/2}$$
$$\le K^{1/2}(x, x) K^{1/2}(y, y) \le \sup_{x \in \Omega} K(x, x) < \infty.$$

Therefore, the series in (2.13) converges absolutely. Combining this with the Cauchy Criterion for Uniform Convergence (which we may use, since Ω is compact, and therefore complete), we find that the series converges uniformly.

We now provide a few simple corollaries of Mercer's Theorem. From the absolute convergence, we obtain

$$\int_{\Omega} K(x,x) d\mu(x) = \sum \lambda_j \int \phi_j^2 d\mu = \sum \lambda_j,$$

and thus

$$\sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j < \infty.$$
(2.14)

By combining Mercer's Theorem with Lemma 2.2.2 we arrive at a characterization of all positive definite kernels.

Corollary 2.3.4. The kernel K is positive definite if and only if for some orthonormal basis $\{\psi_j\}_{j=1}^{\dim(L^2(\Omega,\mu))}$ of $L^2(\Omega,\mu)$ and sequence of nonnegative real numbers $\{\kappa_j\}_{j=1}^{\dim(L^2(\Omega,\mu))}$,

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \kappa_j \psi_j(x) \psi_j(y)$$

where the sum converges absolutely and uniformly, $\kappa_j \ge 0$ for each j, and ψ_j is continuous whenever $\kappa_j > 0$.

We can also obtain a clear generalization of Mercer's Theorem. Consider the sets $N_+(K) = \{n \ge 1 : \lambda_j > 0\}$ and $N_-(K) = \{n \ge 1 : \lambda_j < 0\}$. If $|N_-(K)| < \infty$, then the kernel

$$K(x,y) - \sum_{n \in N_{-}(K)} \lambda_{j} \phi_{j}(x) \phi_{j}(y)$$

is continuous and positive definite. The absolute and uniform convergence of this new kernel then guarantees the following Corollary.

Corollary 2.3.5. Suppose the operator $T_{K,\mu}$ has finitely many negative eigenvalues. Then (2.13) holds, with the series converging uniformly and absolutely.

Finally, we supply another way to characterize positive definite functions.

Proposition 2.3.6. A kernel K is positive definite if and only if there exists some $k \in L^2(\Omega \times \Omega, \mu \times \mu)$ such that k can be represented as

$$k(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \kappa_j \phi_j(x) \phi_j,$$

in the L^2 sense, and for all $x, y \in \Omega$,

$$K(x,y) = \int_{\Omega} k(x,z)k(z,y)d\mu(z).$$
(2.15)

As will be shown in the proof below, the coefficients κ_j satisfy $\kappa_j^2 = \lambda_j$ for all j.

Proof. For any choices of real κ_j 's, we have, in the L^2 sense, the following equalities:

$$\int_{\Omega} k(x,z)k(z,y)d\mu(z) = \langle k(x,\cdot), k(y,\cdot) \rangle_{L^2(\Omega,\mu)} = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \kappa_j^2 \phi_j(x)\phi_j(y).$$

If we assume that *K* is positive definite, and choose $\kappa_j = \sqrt{\lambda_j}$ for each *j*, then clearly $k \in L^2(\Omega \times \Omega, \mu \times \mu)$, and, since

$$\int_{\Omega} k(x,z)k(z,y)d\mu(z) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x)\phi_j(y) = K(x,y)$$

in the L^2 sense, we have pointwise equality due to Mercer's Theorem.

Alternatively, if we assume that

$$K(x,y) = \int_{\Omega} k(x,z)k(z,y)d\mu(z),$$

then for any finite point configuration $\omega_N = \{z_1, ..., z_N\}$ in Ω and $c_1, ..., c_N \in \mathbb{R}$, we have

$$\sum_{j=1}^{N} \sum_{i=1}^{N} K(z_j, z_i) c_j c_i = \sum_{j=1}^{N} \sum_{i=1}^{N} c_j c_i \int_{\Omega} k(z_j, z) k(z, z_i) d\mu(z)$$
$$= \int_{\Omega} \left(\sum_{j=1}^{N} c_j k(z_j, z) \right)^2 d\mu(z) \ge 0.$$

Thus, *K* is clearly positive definite.

2.4 **Two-Point Homogeneous Spaces**

In this paper, we will predominantly focus on energy optimization problems for compact, connected, two-point homogeneous spaces. A metric space (Ω, ρ) is said to be *two-point homogeneous*, if for every two pairs of points x_1, x_2 and y_1, y_2 such that $\rho(x_1, x_2) = \rho(y_1, y_2)$ there exists an isometry of Ω mapping x_i to y_i , i = 1, 2. It is known [Wan52] that any such compact connected space is either a real sphere \mathbb{S}^{d-1} , a real projective space \mathbb{RP}^{d-1} , a complex projective space \mathbb{CP}^{d-1} , a quaternionic projective space \mathbb{HP}^{d-1} , or the Cayley projective plane \mathbb{OP}^2 . Note that it suffices to consider \mathbb{FP}^d for d > 2 only, as \mathbb{FP}^1 is isomorphic to the sphere $\mathbb{S}^{\dim_{\mathbb{R}}(\mathbb{F})}$ [Bae02, p. 170], and so it will not be separately considered in what follows. Naturally, other two-point homogeneous spaces exist, such as the Euclidean spaces \mathbb{R}^d , which are not compact, and the Hamming Cube, which is not connected, but here we will only consider the spaces listed above.

Below, Φ always refers to a compact connected two-point homogeneous space, equipped with the geodesic distance ϑ , normalized to take values in $[0, \pi]$. We let η denote the unique probability measure invariant under the isometries of Φ , i.e. the normalized uniform surface measure.

The first three types of projective spaces $\{\mathbb{FP}^{d-1}:\mathbb{F}=\mathbb{R},\mathbb{C},\mathbb{H}\}$ have a simple descrip-

tion: they may be represented as the spaces of lines passing through the origin in \mathbb{F}^d ,

$$x\mathbb{F} = \{x\lambda \mid \lambda \in \mathbb{F} \setminus \{0\}\}.$$
(2.16)

Observe that the isometry groups O(d), U(d), Sp(d) of the corresponding vector spaces \mathbb{F}^d act transitively on each space, and that the stabilizers of a line represented by $x \in \mathbb{F}^d$ are $O(d-1) \times O(1)$, $U(d-1) \times U(1)$, and $Sp(d-1) \times Sp(1)$, respectively. Thus one has [Wol07, p. 28] the following quotient representations:

$$\begin{split} \mathbb{R}\mathbb{P}^{d-1} &= O(d) / \Big(O(d-1) \times O(1) \Big), \\ \mathbb{C}\mathbb{P}^{d-1} &= U(d) / \Big(U(d-1) \times U(1) \Big), \\ \mathbb{H}\mathbb{P}^{d-1} &= Sp(d) / \Big(Sp(d-1) \times Sp(1) \Big) \end{split}$$

where we write O(d), U(d), Sp(d) for the groups of matrices X, over the respective algebras, satisfying $XX^* = I$.

Using the identification (2.16), one can associate each element of \mathbb{FP}^{d-1} ($\mathbb{F} = \mathbb{R}, \mathbb{C}, \mathbb{H}$) with a unit vector $x \in \mathbb{F}^d$, ||x|| = 1, and we shall at times abuse notation by doing so. In addition to the Riemannian metric ϑ , this gives us the *chordal distance* between points $x, y \in \Phi$, defined by

$$\boldsymbol{\rho}(\boldsymbol{x},\boldsymbol{y}) = \sqrt{1 - |\langle \boldsymbol{x},\boldsymbol{y} \rangle|^2},$$

where $\langle x, y \rangle = \sum_{i=1}^{d} x_i \overline{y}_i$ is the standard inner product in \mathbb{F}^d . The chordal distance $\rho(x, y)$ is related to the geodesic distance $\vartheta(x, y)$ by the equation

$$\cos(\vartheta(x,y)) = 1 - 2\rho(x,y)^2 = 2|\langle x,y \rangle|^2 - 1.$$

Since the algebra of octonions is not associative, the line model of (2.16) fails, and instead a model given by Freudenthal [Fre53] is used to describe \mathbb{OP}^{d-1} . It is known [Bae02] that only two octonionic spaces exist: \mathbb{OP}^1 and \mathbb{OP}^2 , however \mathbb{OP}^1 is just \mathbb{S}^8 , as noted above.

 \mathbb{OP}^2 can be described as the subset of 3×3 Hermitian matrices Π over \mathbb{O} , satisfying $\Pi^2 = \Pi$ and Tr $\Pi = 1$ (see, e.g., [CKM16, Skr17]). A metric for \mathbb{OP}^2 is then given by the Frobenius product,

$$\rho(\Pi_1,\Pi_2) = \frac{1}{\sqrt{2}} \|\Pi_1 - \Pi_2\|_F = \sqrt{1 - \langle \Pi_1,\Pi_2 \rangle},$$

where $\langle \Pi_1, \Pi_2 \rangle = \text{Re Tr } \frac{1}{2}(\Pi_1 \Pi_2 + \Pi_2 \Pi_1)$. This is the chordal distance on \mathbb{OP}^2 whereas the geodesic distance can be defined through $\sin \frac{\vartheta(x,y)}{2} = \rho(x,y)$, as in the above projective spaces. All Π given as above may be written in the form

$$\begin{pmatrix} |a|^2 & a\overline{b} & a\overline{c} \\ b\overline{a} & |b|^2 & b\overline{c} \\ c\overline{a} & c\overline{b} & |c|^2 \end{pmatrix},$$

where $|a|^2 + |b|^2 + |c|^2 = 1$ and (ab)c = a(bc). This gives a representation of \mathbb{OP}^2 as the quotient $F_4/\text{Spin}(9)$ [Bae02, p. 189].

One feature of spaces Φ that allows for the application of linear programming methods is the existence of a decomposition of $L^2(\Phi, \eta, \mathbb{C})$, the space of complex-valued squareintegrable functions on Φ . Consider the representation L(g), of the isometry group G of Φ , for $L^2(\Phi, \eta, \mathbb{C})$, defined by

$$L(g)\phi(x) = \phi(g^{-1}x).$$

This representation is decomposable into an orthogonal direct sum of pairwise non-equivalent irreducible representations $L_n(g)$ acting on isometry invariant finite dimensional subspaces $V_n^{(\alpha,\beta)}$, of continuous functions (see [Lev98, Vil]), with $V_0^{(\alpha,\beta)}$ being the space of constant

functions, where $\alpha = (d-1)\frac{\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1$ and

$$\beta = \begin{cases} \alpha, & \text{if } \Phi = \mathbb{S}^{d-1}; \\ \frac{\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1, & \text{if } \Phi = \mathbb{FP}^{d-1} \end{cases}.$$
(2.17)

These subspaces $V_n^{(\alpha,\beta)}$, which satisfy

$$L^{2}(\Phi, \eta, \mathbb{C}) = \overline{\bigoplus_{n \ge 0} V_{n}^{(\alpha, \beta)}}, \qquad (2.18)$$

can be chosen as the eigenspaces of the Laplace–Beltrami operator on Φ corresponding to the *n*-th eigenvalue in the increasing order [Skr, Wol07]. Let $Y_{n,k} = Y_{n,k}^{(\alpha,\beta)}$, for $k = 1, \ldots, \dim(V_n^{(\alpha,\beta)})$, be an orthonormal basis in $V_n^{(\alpha,\beta)}$. Because of the invariance of $V_n^{(\alpha,\beta)}$ and the two-point homogeneity of Φ , the reproducing kernel for $V_n^{(\alpha,\beta)}$ only depends on the distance $\vartheta(x, y)$ between points [CS99, Ven01]. Furthermore, as a function of

$$\tau(x,y) := \cos \vartheta(x,y),$$

the reproducing kernel is a polynomial $P_n^{(\alpha,\beta)}$ of degree *n*, which satisfies

$$P_n^{(\alpha,\beta)}(\tau(x,y)) = \sum_{k=1}^{\dim(V_n^{(\alpha,\beta)})} Y_{n,k}(x) \overline{Y_{n,k}(y)}.$$
(2.19)

Formula (2.19) is known as the *addition formula*, and a slight alteration of the proof of 2.2.2 tells us that the $P_n^{(\alpha,\beta)}$ are positive definite on Φ . We note that on the sphere, $\tau(x,y) = \langle x, y \rangle$. For any kernel *K* that depends only on the distance between points $\vartheta(x, y)$, there is a corresponding function $F : [-1,1] \to \mathbb{R}$ defined by

$$F(\tau(x,y)) := K(x,y).$$
 (2.20)

We call F positive definite if and only if K is positive definite, and may replace K with F in definitions such as the energy integral.

For more details on compact, connected, two-point homogeneous spaces, particularly projective spaces, we refer the reader to [Bes78, Hel78, Hel84, Lev92, Sha01, Skr20, Skr, Wol11, Wol07].

Jacobi Polynomials

The polynomials $C_n = C_n^{(\alpha,\beta)} = \frac{P_n^{(\alpha,\beta)}}{\dim(V_n^{(\alpha,\beta)})}$ satisfy $C_n(1) = 1$ and are orthogonal with respect to the probability measure on [-1,1],

$$d\mathbf{v}^{(\alpha,\beta)} = \frac{1}{\gamma_{\alpha,\beta}} (1-t)^{\alpha} (1+t)^{\beta} dt,$$

where, as above, $\alpha = (d-1)\frac{\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1$,

$$eta = egin{cases} lpha, & ext{if } \Phi = \mathbb{S}^{d-1}; \ rac{\dim_{\mathbb{R}}(\mathbb{F})}{2} - 1, & ext{if } \Phi = \mathbb{FP}^{d-1}, \end{cases}$$

and the normalization factor is given by

$$\gamma_{\alpha,\beta} = 2^{\alpha+\beta+1}B(\alpha+1,\beta+1)$$

where *B* is the beta function. The weight measure $v^{(\alpha,\beta)}$ is related to integration on Φ in the following way: for any $p \in \Phi$,

$$\int_{\Phi} F(\tau(x,p))d\eta(x) = I_F(\eta) = \int_{-1}^{1} F(t)d\mathbf{v}^{(\alpha,\beta)}(t).$$
(2.21)

The normalized Jacobi polynomials, C_n , form an orthogonal basis in $L^2([-1,1], v^{(\alpha,\beta)})$; equivalently, the span of $C_n(\tau(x,y))$, $n \ge 0$, is dense in the subset of $L^2(\Phi \times \Phi, \eta \otimes \eta)$ consisting of functions that depend only on the distance between x and y. In the case that Φ is the sphere, i.e. $\alpha = \beta = \frac{d-3}{2}$, then these polynomials are known as the (normalized) Gegenbauer/ultraspherical polynomials.

This allows expanding functions from $L^2([-1,1], dv^{(\alpha,\beta)})$ in terms of C_n :

$$F(t) = \sum_{i=0}^{\infty} \widehat{F}_n C_n(t), \quad \text{where} \quad \widehat{F}_n = \dim(V_n^{(\alpha,\beta)}) \int_{-1}^1 F(t) C_n(t) \, d\nu^{(\alpha,\beta)}(t). \tag{2.22}$$

For a fixed space Φ we will not indicate the dependence of polynomials $C_n = C_n^{(\alpha,\beta)}$, the spaces $V_n = V_n^{(\alpha,\beta)}$, and the functions $Y_{n,k} = Y_{n,k}^{(\alpha,\beta)}$ on the indices α , β . We refer to \widehat{F}_n as the Jacobi coefficients of the function *F*; the normalization $C_n(1) = 1$ used here is common in the coding theory community [Sze75, Lev92].

We have already seen that Jacobi polynomials C_n are positive definite on Φ , so uniformly convergent series with nonnegative coefficients must be as well. Fortunately, similar to the result of Mercer's Theorem, if $F \in C([-1,1])$ has a Jacobi expansion with nonnegative coefficients, then the sum is uniformly convergent [BD19, Gan67, Lyu09].

Lemma 2.4.1. Let $F \in C([-1,1])$, $F(t) = \sum_{n=0}^{\infty} \widehat{F}_n C_n(t)$ with $\widehat{F}_n < 0$ for finitely many $n \in \mathbb{N}_0$. Then the Jacobi expansion of F converges uniformly and absolutely to F on [-1,1].

With this lemma, we now have the following analogue of Corollary 2.3.4 [Boc41, Gan67, Sch42].

Proposition 2.4.2. A function $F \in C([-1,1])$ is positive definite on Φ if and only if $\widehat{F}_n \ge 0$ for all $n \ge 0$.

Proof. One direction is clear. For the other, let us assume that for some $m \in \mathbb{N}_0$, $\widehat{F}_m < 0$. For a fixed point $p \in \Phi$, we see that $Y_m(x) := C_m(\tau(x, p))$ is in V_m and real-valued. By orthogonality of the spaces V_n in (2.18) and the fact that $P_m(\tau(x, y)) = \dim(V_m)C_m(\tau(x, y))$ is the reproducing kernel of V_m , we have that, for $d\mu(x) = Y_m(x)d\eta(x)$,

$$I_F(\mu) = \int_{\Phi} \int_{\Phi} \int_{\Phi} F(\tau(x, y)) Y_m(x) Y_m(y) d\eta(x) d\eta(y)$$
$$= \frac{\widehat{F}_m}{\dim(V_m)} \int_{\Phi} Y_m(y)^2 d\eta(y) < 0,$$

so F is not positive definite.

Antipodal Symmetry

Let us consider the kernels of the form $K(x,y) = G(\langle x,y \rangle)$ on the unit sphere $\mathbb{S}_{\mathbb{F}}^{d-1}$ for $\mathbb{F} \in \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$, where $\langle x, y \rangle$ is the inner product in the ambient space. We observe that if G(t) = G(|t|), the energy I_G remains the same after averaging over unit multiples of vectors in the support of a measure μ . Let $U(\mathbb{F})$ be the set of units in \mathbb{F} , $U(\mathbb{F}) = \{c \in \mathbb{F} : |c| = 1\}$, and ζ be the uniform measure on $U(\mathbb{F})$. If one defines, for Borel sets $B \subset \mathbb{S}_{\mathbb{F}}^{d-1}$,

$$\mu_{U(\mathbb{F})}(B) = \frac{1}{\zeta(U(\mathbb{F}))} \int_{U(\mathbb{F})} \mu(cB) d\zeta(c)$$
(2.23)

then

$$egin{aligned} &I_G(\mu_{U(\mathbb{F})}) = rac{1}{\zeta(U(\mathbb{F}))^2} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{U(\mathbb{F})} \int\limits_{U(\mathbb{F})} G(|\langle cx, by
angle|) d\mu(x) d\mu(y) d\zeta(c) d\zeta(b) \ &= rac{1}{\zeta(U(\mathbb{F}))^2} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{U(\mathbb{F})} \int\limits_{U(\mathbb{F})} G(|\langle x, y
angle|) d\mu(x) d\mu(y) d\zeta(c) d\zeta(b) \ &= \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} \int\limits_{\mathbb{S}_{\mathbb{F}}^{d-1}} G(|\langle x, y
angle|) d\mu(x) d\mu(y) = I_G(\mu). \end{aligned}$$

Moreover, this tells us that for any $\mu, \nu \in \mathbb{P}(\mathbb{S}^{d-1}_{\mathbb{F}}), I_G(\mu) = I_G(\nu)$ whenever $I_G(\mu_{U(\mathbb{F})}) = I_G(\nu_{U(\mathbb{F})}).$

Now, let $p_{\mathbb{F}}: \mathbb{S}^{d-1}_{\mathbb{F}} \to \mathbb{FP}^{d-1}$ be the projection of the sphere onto the projective space,

i.e.

$$p_{\mathbb{F}}(x) = x\mathbb{F}$$

as defined by (2.16). Defining $F : [-1,1] \to \mathbb{R}$ by

$$F(\tau(p_{\mathbb{F}}(x), p_{\mathbb{F}}(y))) = F(\tau(x\mathbb{F}, y\mathbb{F})) = F(2|\langle x, y \rangle|^2 - 1) = G(|\langle x, y \rangle|) = G(\langle x, y \rangle),$$

for $x, y \in \mathbb{S}_{\mathbb{F}}^{d-1}$ we then see that for any $\mu \in \mathscr{B}(\mathbb{S}_{\mathbb{F}}^{d-1})$,

$$\begin{split} \int_{\mathbb{FP}^{d-1}} \int_{\mathbb{FP}^{d-1}} F(\tau(x,y)) d((p_{\mathbb{F}})_* \mu)(x) d((p_{\mathbb{F}})_* \mu)(y) &= \int_{\mathbb{S}^{d-1}_{\mathbb{F}}} \int_{\mathbb{S}^{d-1}_{\mathbb{F}}} F(\tau(p_{\mathbb{F}}(x), p_{\mathbb{F}}(y))) d\mu(x) d\mu(y) \\ &= \int_{\mathbb{S}^{d-1}_{\mathbb{F}}} \int_{\mathbb{S}^{d-1}_{\mathbb{F}}} G(\langle x, y \rangle) d\mu(x) d\mu(y), \end{split}$$

where $(p_{\mathbb{F}})_*\mu$ is the push-forward measure defined by $(p_{\mathbb{F}})_*\mu(B) = \mu(p_{\mathbb{F}}^{-1}(B))$ for all Borel subsets of \mathbb{FP}^{d-1} .

This discussion shows that for *G* as above, a minimizing measure on the sphere for I_G , can be taken to be symmetric, and that the problem of minimizing over symmetric measures on spheres is equivalent to minimizing energy over measures on the projective spaces. In Chapters 6 and 7 we will use these relations in order to address optimization problems for kernels *G* of this form.

2.5 The Sphere

Of the compact, connected two-point homogeneous spaces, real unit spheres \mathbb{S}^{d-1} are by far the most well studied, and will be the space we work in most often in this paper. When specifically working on the sphere, we will use σ , instead of η , to denote the surface measure, i.e. the normalized (d-1)-dimensional Lebesgue/Hausdorff measure on the sphere.

Like the other compact, connected, two-point homogeneous spaces, the sphere \mathbb{S}^{d-1}

may be represented as a quotient space [Wol11]. For d > 2, SO(d) acts transitively on \mathbb{S}^{d-1} and $SO(d-1) \times \{1\}$ is the stabilizer of any point $x \in \mathbb{S}^{d-1}$, whereas for d = 2, these groups are O(2) and $O(1) \times \{1\}$ so

$$\begin{split} \mathbb{S}^{d-1} &= SO(d) / \Big(SO(d-1) \times \{1\} \Big) \quad d > 2 \\ \mathbb{S}^2 &= O(2) / \Big(O(1) \times \{1\} \Big). \end{split}$$

On the sphere, the chordal metric is simply the Euclidean metric, i.e. $\rho(x, y) = ||x - y||$, and the geodesic distance is $\vartheta(x, y) = \arccos(\langle x, y \rangle)$, meaning that on the sphere

$$\tau(x,y) = \cos(\vartheta(x,y)) = \langle x,y \rangle.$$

Since \mathbb{S}^{d-1} is a two-point homogeneous space, $L^2(\mathbb{S}^{d-1}, \sigma, \mathbb{C})$ has a decomposition of the form (2.18). In this setting the space $V_n^{(\frac{d-3}{2}, \frac{d-3}{2})}$ is exactly the space of *spherical harmonics* of degree *n* (i.e. the restrictions to \mathbb{S}^{d-1} of homogeneous polynomials, of degree *n*, on \mathbb{R}^d that are in the kernel of the Euclidean Laplacian) [Sha01]. Clearly, any complexvalued spherical harmonic of degree *n* must be of the form p(x) + iq(x), where *p* and *q* are real-valued spherical harmonics of degree *n*. Thus, each $V_n^{(\frac{d-3}{2}, \frac{d-3}{2})}$, and thereby $L^2(\mathbb{S}^{d-1}, \sigma, \mathbb{C})$, has an orthonormal basis of real-valued spherical harmonics. Denoting the space of real-valued spherical harmonics of degree *n* on \mathbb{S}^{d-1} by \mathcal{H}_n^d , we see that this means that

$$L^2(\mathbb{S}^{d-1},\sigma) = \bigoplus_{n>0} \mathscr{H}_n^d$$

and that for all $n \in \mathbb{N}_0$ and $d \ge 2$,

$$\dim(V_n^{(\frac{d-3}{2},\frac{d-3}{2})}) = \dim(\mathscr{H}_n^d) = \binom{n+d-1}{n} - \binom{n+d-3}{n-2}$$

Our addition formula (2.19) now tells us that for all $n \in \mathbb{N}_0$, if $Y_{n,1}, \dots, Y_{n,\dim(\mathscr{H}_n^d)}$ is an

orthonormal basis of \mathscr{H}_n^d , then

$$C_n^{\left(\frac{d-3}{2},\frac{d-3}{2}\right)}(\langle x,y\rangle) = \frac{1}{\dim(\mathscr{H}_n^d)} \sum_{j=1}^{\dim(\mathscr{H}_n^d)} Y_{n,j}(x)Y_{n,j}(y).$$

In certain instances, it will be more convenient to use the standard Gegenbauer/ultraspherical polynomials instead of the normalized ones.

Gegenbauer Polynomials

For a parameter $\lambda > 0$, consider the measure

$$dw_{\lambda}(t) = d\mathbf{v}^{(\lambda - \frac{1}{2}, \lambda - \frac{1}{2})}(t) = \frac{\Gamma(\lambda + 1)}{\sqrt{\pi}\Gamma(\lambda + \frac{1}{2})}(1 - t^2)^{\lambda - \frac{1}{2}}dt$$

on the interval [-1,1]. For the rest of this text, we will assume that $\lambda = \frac{d-2}{2}$, so that

$$dw_{\lambda}(t) = \frac{A_{d-2}}{A_{d-1}} (1-t^2)^{\frac{d-3}{2}} dt,$$

where $A_{d-1} = \frac{2\pi^{d/2}}{\Gamma(d/2)}$ is the (d-1)-dimensional Hausdorff surface measure of \mathbb{S}^{d-1} . The weighted measure w_{λ} is related to integration on the sphere \mathbb{S}^{d-1} in the following way: for a unit vector $p \in \mathbb{S}^{d-1}$,

$$\int_{\mathbb{S}^{d-1}} F(\langle x, p \rangle) d\sigma(x) = I_F(\sigma) = \int_{-1}^{1} F(t) dw_{\lambda}(t)$$
(2.24)

where, as before, σ is the normalized surface measure on the sphere \mathbb{S}^{d-1} .

The Gegenbauer polynomials C_n^{λ} , $n \ge 0$, as a special case of the Jacobi polynomials, clearly form an orthogonal basis of $L^2([-1,1], v^{(\lambda-\frac{1}{2},\lambda-\frac{1}{2})}) = L^2([-1,1], w_{\lambda})$. However, in this instance, we are not normalizing these polynomials, which means that we must properly adjust our formulae from Section 2.4. Every function $F \in L^1([-1,1], w_{\lambda})$ has a Gegenbauer (ultraspherical) expansion

$$F(t) \sim \sum_{n=0}^{\infty} \widehat{F}(n,\lambda) \frac{n+\lambda}{\lambda} C_n^{\lambda}(t) \quad t \in [-1,1]$$
(2.25)

where

$$\widehat{F}(n,\lambda) := \frac{1}{C_n^{\lambda}(1)} \int_{-1}^{1} F(t) C_n^{\lambda}(t) dw_{\lambda}(t).$$
(2.26)

For $F \in L^2([-1,1], w_{\lambda})$ this expansion converges to F in the L^2 sense.

In the case of \mathbb{S}^1 , when $\lambda = 0$, the relevant polynomials are the Chebyshev polynomials of the first kind

$$T_n(t) = \cos\left(n \arccos(t)\right) = \frac{1}{2} \lim_{\lambda \to 0} \frac{n+\lambda}{\lambda} C_n^{\lambda}(t), \qquad (2.27)$$

and for \mathbb{S}^2 , the polynomials are appropriately scaled Legendre polynomials [Sze75].

For any $n \in \mathbb{N}_0$ and $d \ge 2$, let $\{Y_{n,j}\}_{j=1}^{\dim(\mathscr{H}_n^d)}$ be any orthonormal basis in \mathscr{H}_n^d . The Gegenbauer polynomials are related to the spherical harmonics by the following *addition formula*

$$\sum_{j=1}^{\dim(\mathscr{H}_n^d)} Y_{n,j}(x) Y_{n,j}(y) = \frac{n+\lambda}{\lambda} C_n^{\lambda}(\langle x, y \rangle) \quad \text{for all } x, y \in \mathbb{S}^{d-1}.$$
(2.28)

Comparing this to the normalized Gegenbauer polynomials $C_n^{(\frac{d-3}{2},\frac{d-3}{2})}$, we see that for all $n \in \mathbb{N}_0$ and $d \ge 2$,

$$\widehat{F}_n C_n^{(\frac{d-3}{2},\frac{d-3}{2})}(t) = \widehat{F}(n,\lambda) \frac{n+\lambda}{\lambda} C_n^{\lambda}(t) \quad t \in [-1,1],$$

as one would expect. Thus, Lemma 2.4.1 and Proposition 2.4.2 hold with the Gegenbauer expansion (2.25). Likewise, all results in this text that are given for general compact, connected, two-point homogeneous spaces using Jacobi expansions (with normalized Jacobi polynomials) hold on the sphere with the standard Gegenbauer polynomials.

Expanding in terms of standard Gegenbauer polynomials will be more convenient for formulating various results in this text when working on the sphere, such as the Funk-Hecke

formula, which clearly follows from orthogonality, (2.25), and the addition formula (2.28).

Theorem 2.5.1. Let $F \in L^2([-1,1], w_{\lambda})$. Then for every $Y_n \in \mathscr{H}_n^d$,

$$\int_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) Y_n(y) d\boldsymbol{\sigma}(y) = \widehat{F}(n, \lambda) Y_n(x).$$
(2.29)

We complete this section with the following corollary of Proposition 2.3.6, where our decision to use standard Gegenbauer polynomials simplifies the result.

Corollary 2.5.2. Let $F \in C([-1,1])$. Then F is positive definite on \mathbb{S}^{d-1} if and only if there exists some $f \in L^2([-1,1], w_\lambda)$ such that

$$F(\langle x, y \rangle) = \int_{\mathbb{S}^{d-1}} f(\langle x, z \rangle) f(\langle z, y \rangle) d\boldsymbol{\sigma}(z).$$
(2.30)

If such a function f exists, then for all $n \in \mathbb{N}_0$, $\widehat{F}(n, \lambda) = \left(\widehat{f}(n, \lambda)\right)^2$.

Proof. For each $n \in \mathbb{N}_0$, let $Y_{n,1}, ..., Y_{n,\dim(\mathscr{H}_n^d)}$ be an orthonormal basis of \mathscr{H}_n^d . Then we know that $\{Y_{n,k} : n \in \mathbb{N}_0, k \in \{1, ..., \dim(\mathscr{H}_n^d)\}\}$ forms an orthonormal basis of $L^2(\mathbb{S}^{d-1}, \sigma)$ which consists of continuous eigenfunctions of the linear integral operator $T_{K,\sigma}$, with $K(x,y) = F(\langle x, y \rangle)$ for all $x, y \in \mathbb{S}^{d-1}$. From (2.25) and (2.28), we know that

$$F(\langle x, y \rangle) = \sum_{n=0}^{\infty} \widehat{F}(n, \lambda) \sum_{k=1}^{\dim(\mathscr{H}_n^d)} Y_{n,k}(x) Y_{n,k}(y)$$

Proposition 2.3.6 then tells us that *F* is positive definite if and only if there exists some $k \in L^2(\mathbb{S}^{d-1} \times \mathbb{S}^{d-1}, \sigma \times \sigma)$ such that

$$F(\langle x, y \rangle) = \int_{\mathbb{S}^{d-1}} k(x, z) k(z, y) d\sigma(z), \qquad (2.31)$$

and that if such a k exists, then

$$k(x,y) = \sum_{n=0}^{\infty} \sum_{k=1}^{\dim(\mathscr{H}_n^d)} \kappa_{n,k} Y_{n,k}(x) Y_{n,k}(y),$$

with $\kappa_{n,k}^2 = \widehat{F}(n,\lambda)$. Moreover, if such a *k* exists, then there are infinitely many choices of *k* satisfying (2.31), as we may change the sign of each $\kappa_{n,k}$. Thus, if there exists some *k* satisfying (2.31), then we may choose its coefficients such that

$$k(x,y) = \sum_{n=0}^{\infty} \sqrt{\widehat{F}(n,\lambda)} \frac{n+\lambda}{\lambda} C_n^{\lambda}(\langle x,y \rangle)$$

meaning that k only depends on the distance between x and y, and so can be expressed by a function $f \in L^2([-1,1], w_\lambda)$, i.e. $f(t) = \sum_{n=0}^{\infty} \sqrt{\widehat{F}(n,\lambda)} \frac{n+\lambda}{\lambda} C_n^{\lambda}(t)$. Our claim now follows.

For more background information on spherical harmonics, Gegenbauer polynomials, and harmonic analysis on the sphere, we refer the reader to [DX13,Mül66,Gro96].

2.6 Designs

We now treat the topic of designs in the compact, connected, two-point homogeneous spaces Φ . Any finite (non-empty) set, also known as a *code*, $\mathscr{C} \subset \Omega$ is characterized by its distance set

$$\mathscr{A}(\mathscr{C}) = \{\tau(x, y) : x, y \in \mathscr{C}\}.$$
(2.32)

We note that \mathscr{A} is the set of cosines of geodesic distances between points, rather than the distances themselves, which is more convenient for our purposes. The *degree* of \mathscr{C} is the number of distinct distances that occur between distinct elements of \mathscr{C} , i.e. $|\mathscr{A}(\mathscr{C}) \setminus \{1\}|$. Codes in two-point homogeneous spaces, particularly the sphere, have a rich history, and have been the basis of several famous optimization problems. For instance, for any $A \subseteq$

[-1,1), we call $\mathscr{C} \subset \Phi$ an *A*-code if $\mathscr{A}(\mathscr{C}) \subseteq A \cup \{1\}$. Often, one is interested in finding upper bounds of the cardinality $|\mathscr{C}|$ of *A*-codes, and determining extremal configurations with respect to such bounds. On \mathbb{S}^{d-1} , for A = [-1,b], for some b < 1, this is equivalent to the classical problem of placing non-overlapping spherical caps of angular radius $\frac{1}{2} \arccos(b)$ on the sphere [DGS77, Ran55]. In particular, when $A = [-1, \pi/3]$, this becomes a problem of finding the *kissing number* of the sphere, i.e. the maximum number of non-overlapping balls in \mathbb{R}^d of radius *r* that can all touch a single ball of radius *r*, which is only known for $d \in \{1, 2, 3, 4, 8, 24\}$ [SvdW53, Lev79, OS79, Mus08]. Another famous optimization problem is the classification of tight designs.

A code $\mathscr{C} \subset \Phi$ is called a *t*-design if

$$\frac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} p(x) = \int_{\Omega} p(x) d\eta(x) = 0 \quad \forall p \in \bigcup_{n=1}^{t} V_n.$$
(2.33)

A relaxation of the above identity allows the configuration to be weighted, so that the equality

$$\sum_{x \in \mathscr{C}} w(x)p(x) = \int_{\Omega} p(x) d\eta(x) = 0 \quad \forall p \in \bigcup_{n=1}^{r} V_n$$
(2.34)

holds for some weights $\{w(x)\}_{x\in\mathscr{C}} \subset \mathbb{R}_{\geq 0}$, satisfying $\sum_{x\in\mathscr{C}} w(x) = 1$. Such weighted formulas are called *cubature formulas* or *weighted t-designs*. We will make a slight abuse of terminology, and refer to the measure

$$\mu_{\mathscr{C},w} = \sum_{x \in \mathscr{C}} w(x) \delta_x \tag{2.35}$$

as a weighted *t*-design if \mathscr{C} , with weight function *w*, is a weighted *t*-design. If $w(x) = \frac{1}{|\mathscr{C}|}$ for all $x \in \mathscr{C}$, i.e. \mathscr{C} is a *t*-design, we shall denote our measure $\mu_{\mathscr{C}} := \mu_{\mathscr{C},w}$, and refer to it as a *t*-design. We note that our definition implies that if \mathscr{C} is a weighted *t*-design with

weight function w, then

$$\sum_{x,y\in\mathscr{C}} w(x)w(y)C_n(\tau(x,y)) = 0 \quad \text{ for } 1 \le n \le t.$$

The *strength* of a (weighted) design is the maximum value of t for which identity (2.33) (or (2.34), accordingly) holds.

The concept of spherical designs was originally introduced in 1977 by Delsarte, Goethal, and Seidel in [DGS77], in which they determined a lower bound on $N(\mathbb{S}^{d-1},t)$, the minimum number of points necessary to construct a spherical *t*-design, and provided several important examples of designs. The existence of spherical *t*-designs for any $d \ge 2$ and $t \in \mathbb{N}$ was proved in 1984 by Seymour and Zaslavsky in [SZ84]. However, this proof is nonconstructive, and gives no indication of the size of $N(\mathbb{S}^{d-1},t)$. Various upper bounds of $N(\mathbb{S}^{d-1},t)$ were given in [WV91, Baj92, KM93, BV10], culminating in the results of Bondarenko, Radchenko, and Viazovska in [BRV13, BRV15]. In their works, they showed that not only is $N(\mathbb{S}^{d-1},t)$ on the order of t^{d-1} (the lower bound given in Lemma 2.6.4), but for large enough *N*, relative to *t* and *d*, well separated spherical *t*-designs of size *N* always exist. This solved a long-standing conjecture of Korevaar and Meyers [KM93].

Theorem 2.6.1 ([BRV13, BRV15]). For all $d \ge 2$ there exists some $C_d > 0$ and $b_d > 0$ such that for all $N \ge C_d t^{d-1}$, there exists a spherical t-design $\mathscr{C} = \{z_1, ..., z_N\} \subset \mathbb{S}^{d-1}$ such that $||z_i - z_j|| > b_d N^{-1/(d-1)}$ for $i \ne j$.

The concept of codes and designs was generalized to projective spaces (as well as other spaces) in the 1980's by Neumaier, Bannai, and Hoggar [BH85, BH89, Hog82a, Hog82b, Neu81], among others, though they were studied previously (see, e.g., [DGS75]). In the following decades, a general theory was developed, a good overview of which, as well as additional background, can be found in, for instance [BB09, Lev92, Lev98, Lev98]. Concerning the relationship between *t*-designs and classical designs in ranked partially ordered sets, see [Del76, Sta81, Sta86].

The equalities (2.33) and (2.34) show that (weighted) designs can provide a good approximation to the whole space Φ , and the strength of the design characterizes the degree of such an approximation. This makes the problem of finding the minimum cardinality of a (weighted) design of a certain strength of significant interest in factorial experiments, cryptography, complexity theory, and calculation theory [AGHP92, CS99, GMS74, Rao47, Sti93].

Linear programming bounds imply exact constraints on the size of designs, in particular giving the cardinality of *tight t-designs*, which have the smallest possible number of pairwise distances between their elements, for a design of strength *t*. The exact definition may be given as follows.

Definition 2.6.2. A code $\mathscr{C} \subset \Omega$ is called a **tight** *t*-**design** if one of the following conditions is satisfied.

- (i) \mathscr{C} is a design of strength t = 2m 1 and degree m, and $-1 \in \mathscr{A}(\mathscr{C})$, i.e. there is at least one pair of points diameter apart;
- (ii) \mathscr{C} is a design of strength t = 2m and degree m.

Often, tight designs are defined by their cardinality rather than their degree. Though the latter is of greater use for our needs in this paper, it is worth connecting the two definitions and showing the bounds on cardinality. Let $\mathscr{C} \subset \Phi$ be code of degree *m*. Set $e = |\mathscr{A}(\mathscr{C}) \setminus \{-1\}| - 1$. The annihilating polynomial of a configuration is defined by $Ann(\mathscr{C}) := \prod_{a \in \mathscr{A}(\mathscr{C}) \setminus \{1\}} (x - a)$. For a positive number *t*, let $(t)^{\overline{k}} = t(t + 1) \dots (t + k - 1)$ be the rising factorial and $(t)^{\underline{k}} = t(t - 1) \dots (t - k + 1)$ be the falling factorial. We define, for $k \in \mathbb{N}$

$$R_{k,\lambda}(x) := \sum_{j=0}^{k} \frac{j+\lambda}{\lambda} C_j^{\lambda}(x), \qquad (2.36)$$

$$Q_{k,\lambda}(x) := \sum_{j=0}^{\lfloor k/2 \rfloor} \frac{k - 2j + \lambda}{\lambda} C_{k-2j}^{\lambda}(x), \qquad (2.37)$$

$$T_{m,e}^{(\alpha,\beta)} := \frac{(\alpha+\beta+2)^{\overline{m+e}}}{(\beta+1)^{\overline{m}}e!} \sum_{j=0}^{e} (-1)^{j} \frac{(m+\beta)^{j}}{(m+e+\alpha+\beta+1)^{j}} \left(\frac{x+1}{2}\right)^{e-j}.$$
 (2.38)

The following lemmas from [DGS77] now let us relate the degree, strength, and cardinality of spherical designs, and provide us with information about the distance set $\mathscr{A}(\mathscr{C})$ of tight designs.

Lemma 2.6.3. Let $\mathscr{C} \subset \mathbb{S}^{d-1}$ be code of degree *m* such that $|\mathscr{C}| = N$.

- (i) If \mathscr{C} is a design of strength t, then $t \leq 2m$ and $N \leq R_{m,\lambda}(1)$. Equality holds in either of these inequalities if and only if \mathscr{C} is a tight 2m-design
- (ii) If C is an antipodal design of strength t, then t ≤ 2m − 1, and N ≤ 2Q_{m−1,λ}(1).
 Equality holds in either of these inequalities if and only if C is a tight (2m − 1)-design.

Lemma 2.6.4. Let $\mathscr{C} \subset \mathbb{S}^{d-1}$ be a *t*-design such that $|\mathscr{C}| = N$.

(i) If t = 2k, then

$$N \ge R_{k,\lambda}(1) = \binom{d-k-1}{d-1} + \binom{d-k-2}{d-1}.$$
(2.39)

Equality holds (i.e. \mathscr{C} is a tight spherical (2k)-design) if and only if $\mathscr{A}(\mathscr{C})$ is exactly the set of roots of $(1-x)R_{k,\lambda}(x)$.

(*ii*) If t = 2k - 1, then

$$N \ge 2Q_{k-1,\lambda}(1) = 2\binom{d-k-2}{d-1}.$$
(2.40)

Equality holds (i.e. \mathscr{C} is a tight spherical (2k-1)-design) if and only if $\mathscr{A}(\mathscr{C})$ is exactly the set of roots of $(1-x^2)C_{k-1,\lambda}(x)$.

Similar results exist for the projective settings [Hog82a].

Lemma 2.6.5. Let $\mathscr{C} \subset \mathbb{FP}^{d-1}$ be a code of degree *m* such that $|\mathscr{C}| = N$.

42

and

(i) We have

$$N \le T_{m,e}^{(\alpha,\beta)}(1) = \frac{(\alpha+\beta+2)^{\overline{m}}(\alpha+2)^{\overline{e}}}{(\beta+1)^{\overline{m}}e!}$$
(2.41)

with equality if and only if $Ann(\mathscr{C})(x) = \left(\frac{x+1}{2}\right)^{m-e} T_{m,e}^{(\alpha,\beta)}(x).$

(ii) If \mathscr{C} is a design of strength t, then $t \leq m + e$, and equality holds (i.e. \mathscr{C} is a tight *t*-design) if and only if equality holds in (2.41).

The bounds on cardinality one achieves from these results were generalized to weighted designs by Levenshtein in [Lev98].

Theorem 2.6.6. Let (\mathcal{C}, w) be a weighted t-design in Φ . Then with $k = \lfloor \frac{t+1}{2} \rfloor$ and $\varepsilon = 2k - t$, we have the following:

1. If $\Phi = \mathbb{S}^{d-1}$, then

$$|\mathscr{C}| \ge \binom{d+k-1-\varepsilon}{d-1} + \binom{d+k-2}{d-1}.$$
(2.42)

2. If
$$\Phi = \mathbb{FP}^{d-1}$$
, then
 $|\mathscr{C}| \ge \frac{(\alpha + \beta + 2)^{\overline{k}} (\alpha + 2)^{\overline{k} - \varepsilon}}{(\beta + 1)^{\overline{k}} (k - \varepsilon)!}.$
(2.43)

In both cases, equality holds if and only if $w(x) = \frac{1}{|\mathscr{C}|}$ for all $x \in \mathscr{C}$ and \mathscr{C} is a tight t-design.

Note that this means that any weighted tight design is a design (known also as a "simple" design), a result that was shown for the sphere in [Tay95]. In addition, observe that from our above lemmas, it is clear that if C_1 and C_2 are tight *t*-designs on Φ , then $\mathscr{A}(C_1) = \mathscr{A}(C_2)$ and $|C_1| = |C_2|$. However, tight designs, generally, are not unique (not even up to unitary equivalence). This is known to be true in particular for the tight projective 2-designs on \mathbb{CP}^2 , through the characterization of all such designs in [Szö].

Table 2.1 provides a list of known tight spherical designs, as well as the 600-cell, which is not a tight design, but will be of interest in Section 6.3. Tight spherical 1-designs are

clearly just pairs of antipodal points, and tight 2-designs and 3-designs are precisely the vertices of regular simplices and cross polytopes, respectively. On the circle \mathbb{S}^1 , tight tdesigns exist for all $t \in \mathbb{N}$: they are simply the vertices of the regular (t+1)-gon. For $d \ge 3$ and $t \ge 4$ there are eight tight spherical designs known. Aside from the icosahedron, these designs do not come from regular polytopes. Instead, several are derived from the E_8 root lattice in \mathbb{R}^8 and the Leech Lattice Λ_{24} in \mathbb{R}^{24} , objects that provide optimal spherical packings in their respective dimensions [Via17, CKM⁺17]. The 240-point and 196560-point configurations are the minimal (nonzero) vectors in those lattices. Each arrangement labeled "kissing" is the kissing configuration of the set below it: by centering non-overlapping congruent spherical caps of maximal height at each of the points in a given configuration, one arrives at a "sphere packing" on the sphere, and the resulting points of tangency on a given cap form a spherical code in a lower dimensional space. Some of the kissing configurations are of independent interest. The configuration of 240 points produces a set of 120 equiangular lines passing though the origin in \mathbb{R}^7 , and the configuration of 27 points forms the Schläfli arrangement. The remaining two designs are the configuration of 552 points in \mathbb{R}^{23} which comes from an equiangluar arrangement of 276 lines described by the unique regular two-graph on 276 vertices, and the resulting kissing configuration, which is a 275 point arrangement associated with the McLaughlin group in S^{21} .

Tight spherical designs with $d \ge 3$ and $t \ge 4$ may only exist for t = 4,5, and 7 with the one exception of the spherical 11-design formed by the Leech lattice minimal vectors [BD79, BD80]. The problem of finding tight spherical 5-designs is the same as that of finding maximal equiangular tight frames, and it is known that existence of a tight spherical 5-design in S^{d-1} is possible only for d = 1, 2, 3 and for dimensions of the form $d = (2k + 1)^2 - 2$, where $k \ge 1$; see [BD79, BD80, DGS77, LS73] for details on how these conditions arise. A direct correspondence with such spherical designs and regular graphs has long been recognized [Sei73], and, in connection, it is known that for infinitely many values of k, a tight spherical 5-design cannot exist in dimension $d = (2k+1)^2 - 2$ [BMV04, Mak02].

d	N	М	Inner products	Name
d	2	1	±1	Pair of antipodal points
d	d+1	2	-1/d, 1	Regular simplex
d	2d	3	$0,\pm 1$	Cross-polytope
2	N	N-1	$\cos(2j\pi/N), 0 \le j \le N/2$	Regular N-gon
3	12	5	$\pm 1/\sqrt{5},\pm 1$	Icosahedron
4	120	11	$0, (\pm 1 \pm \sqrt{5})/4, \pm 1/2, \pm 1$	600-cell
6	27	4	-1/2, 1/4, 1	Kissing/Schläfli
7	56	5	$\pm 1/3, \pm 1$	Kissing/Equiangular lines
8	240	7	$0,\pm 1/2,\pm 1$	E_8 root system
22	275	4	-1/4, 1/6, 1	Kissing/McLaughlin
23	552	5	$\pm 1/5, \pm 1$	Equiangular lines
23	4600	7	$0,\pm 1/3,\pm 1$	Kissing
24	196560	11	$0,\pm rac{1}{4},\pm rac{1}{2},\pm 1$	Leech Lattice minimal vectors

Table 2.1: A list of known tight spherical designs (with the 600-cell). Here *M* denotes the strength of the design, *d* the dimension of the ambient space \mathbb{R}^d , and *N* is the size of the design.

Table 2.2 lists all known tight projective designs, except those for the spaces \mathbb{FP}^1 , as those are congruent to real spheres. Identifying tight projective designs is simple in the real setting. Tight spherical designs of odd strength must be centrally symmetric [DGS77], and by choosing points from each antipode in an odd tight design, one arrives at a real projective tight design. Thus, all tight designs of odd strength in Table 2.1 correspond to entries in Table 2.2.

For the other projective spaces, the vertices of a cross-polytope (i.e. an orthonormal basis in the projective space) always provide a tight 1-design, as they did in \mathbb{RP}^{d-1} . However, unlike the real case, it is known that no tight *t*-designs exist in the complex or quaternionic setting whenever $t \ge 4$ and $d \ge 3$ [BH89, Hog89, Lyu09]. In the complex setting, tight 2-designs, also known as *symmetric, informationally complete, positive operator-valued measures (SIC-POVMs)*, are known to exist for $d \le 16$, d = 19, 24, 28, 35, 48, and numerical experiments suggest that they may exist in every dimension [ABB+14, RBKSC04, SG10, Zau11]. With exception of the (3, 15) quaternionic and (3, 27) octonionic designs

from [CKM16], explicit constructions are readily found for the other designs mentioned in Table 2.2 [Hog82a].

Table 2.2: A list of parameters for the known to exist projective tight designs (besides designs in \mathbb{FP}^1 for $\mathbb{F} \neq \mathbb{R}$). Here *M* denotes the strength of the design, *d* the dimension of the ambient space \mathbb{F}^d , and *N* is the size of the design. For SIC-POVMs, the (*) indicates that these exist for certain values of *d*, and may or may not exist for all values.

d	N	М	$ \langle x,y\rangle ^2$	F	Name
d	d+1	1	0,1	\mathbb{R}	Cross-polytope/ONB
2	N	N-1	$\cos^2(\pi j/N), 1 \le j \le N$	$\mathbb R$	Regular 2 <i>N</i> -gon
3	6	2	1/5, 1	$\mathbb R$	Icosahedron
7	28	2	1/9, 1	$\mathbb R$	Kissing configuration for E_8
8	120	3	0, 1/4, 1	$\mathbb R$	Roots of E_8 lattice
23	276	2	1/25, 1	$\mathbb R$	Tight simplex
23	2300	3	0, 1/9, 1	$\mathbb R$	Kissing configuration for Λ_{24}
24	98280	5	0, 1/16, 1/4, 1	\mathbb{R}	Minimal vectors of Λ_{24}
d	d+1	1	0, 1	\mathbb{C}	Cross-polytope/ONB
d(*)	d^2	2	1/(d+1), 1	\mathbb{C}	SIC-POVM
4	40	3	0, 1/3, 1	\mathbb{C}	Eisenstein structure on E_8
6	126	3	0, 1/4, 1	\mathbb{C}	Eisenstein structure on K_{12}
d	d+1	1	0,1	\mathbb{H}	Cross-polytope/ONB
3	15	2	2/7, 1	\mathbb{H}	Tight simplex
5	165	3	0, 1/4, 1	\mathbb{H}	Quaternionic reflection group
3	d+1	1	0,1	\mathbb{O}	Cross-polytope/ONB
3	27	2	2/13, 1	\mathbb{O}	Tight simplex
3	819	5	0,1/4,1/2,1	\mathbb{O}	Generalized hexagon of order $(2,8)$

A weaker property of a design is sharpness, which will not play a role here. The paper [CK07] proves that sharp designs, and tight designs in particular, are minimizers for discrete energy minimization problems with absolutely monotone kernels.

Chapter 3

Elementary Aspects of Energy Optimization

We now provide several results which connect properties of the kernel K, the functional I_K , and the minimizers of this continuous energy. Many seem to be new, and those which are known do not appear to have been previously collected in a single text, though [BHS19, Chp. 4] provides a similar exposition for lower semi-continuous kernels on compact sets.

In Sections 3.1, 3.2, and 3.3, we present these results for general compact metric spaces. Section 3.1 discusses the equivalences and implications which hold without requiring that the potential U_K^{μ} is constant or the supp $(\mu) = \Omega$ for some probability measure μ . Bounds on the mutual energy, the convexity of I_K , and a condition for the uniqueness of the equilibrium measure are addressed in this section, as well as two results that will be particularly useful in later chapters: If μ is a local minimizer of I_K , then on supp (μ) , U_K^{μ} is constant and Kis positive definite modulo a constant. In Section 3.2, we assume that μ is K-invariant, i.e. that the potential U_K^{μ} is constant on all of Ω . With this assumption, we find that I_K , which is a quadratic functional, behaves linearly about μ , that the conditionally positive definiteness of K is equivalent positive definiteness modulo a constant, and that if μ is a local minimizer, then it is in fact a global minimizer. In Section 3.3 we make an additional assumption: that μ has full support. This final assumption allows us to equate *K* being conditionally positive definite to μ being a global minimizer of I_K and we collect several conditions on *K*, I_K , or μ that are equivalent to *K* being conditionally positive definite, positive definite, or conditionally strictly positive definite, Theorems 3.3.1, 3.3.2, and 3.3.3, respectively.

In Sections 3.4 and 3.5 we discuss energy on two-point homogeneous spaces and, more specifically, the sphere. For these spaces, for any isometry invariant kernel F, one is often interested in determining if the surface measure η is a minimizer of I_F . Since η naturally has full support and satisfies $U_F^{\eta}(x) = I_F(\eta)$ for all $x \in \Phi$, our results from the previous sections all apply. In these sections, we discuss how these results relate to the Jacobi or Gegenbauer expansions of F, and provide the results that we will make greatest use of later in the text.

3.1 Positive Definite Functions

We start by discussing some properties and characterizations of (conditional) positive definiteness related to the behavior of the energy functionals and their minimizers.

Positive Definiteness and Inequalities for Mixed Energies

We first make the observation that the (conditional) positive definiteness of the kernel can be characterized by the inequalities for mixed energies in terms of arithmetic or geometric means. While the validity of such inequalities for positive definite kernels is well known, see e.g. Chapter 4 of [BHS19], their sufficiency doesn't seem to have appeared in the literature. We summarize these facts in the following two lemmas. The first one connects conditional positive definiteness to the arithmetic mean inequality.

Lemma 3.1.1. Suppose K is a kernel on Ω^2 . Then the following conditions are equivalent:

1. K is conditionally positive definite.

2. For every pair of Borel probability measures μ_1 and μ_2 on Ω , the mutual energy $I_K(\mu_1, \mu_2)$ satisfies

$$I_{K}(\mu_{1},\mu_{2}) \leq \frac{1}{2} (I_{K}(\mu_{1}) + I_{K}(\mu_{2})).$$
(3.1)

3. Inequality (3.1) is satisfied for any pair of signed Borel measures of total mass one.

If K is conditionally strictly positive definite, then equality in 3.1 holds if and only if $\mu_1 = \mu_2$ on Borel subsets of Ω .

Proof. Suppose that *K* is conditionally positive definite. Then for any $\mu_1, \mu_2 \in \widetilde{\mathbb{P}}(\Omega), \mu_1 - \mu_2 \in \mathscr{Z}(\Omega)$, so

$$0 \leq I_K(\mu_1 - \mu_2) = I_K(\mu_1) - 2I_K(\mu_1, \mu_2) + I_K(\mu_2),$$

which proves (3.1). Thus, (1) implies (3), which in its turn obviously implies (2).

Now assume condition (2), i.e. that (3.1) holds for all $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$. For any $\mu \in \mathscr{Z}(\Omega)$, there exists $c \ge 0$ and probability measures $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$ such that $v = c(\mu_1 - \mu_2)$. We then have that

$$I_K(\mu) = c^2 \Big(I_K(\mu_1) - 2I_K(\mu_1, \mu_2) + I_K(\mu_2) \Big) \ge 0,$$

so *K* must be conditionally positive definite.

Let us now assume that *K* is conditionally strictly positive definite. Equality clearly holds in (3.1) if $\mu_1 = \mu_2$ on all Borel sets of Ω . If, however, the restriction of μ_1 to Borel subsets of Ω does not coincide with that of μ_2 , the measure $\mu_1 - \mu_2$ is nonzero on some Borel subset of Ω , and the conditionally strict positive definiteness of *K* gives us that $I_K(\mu_1 - \mu_2) > 0$, making (3.1) a strict inequality.

The second lemma is very similar: it shows that positive definiteness is equivalent to the geometric mean inequality for the mixed energy.

Lemma 3.1.2. Suppose K is a kernel on Ω^2 . Then K is positive definite if and only if for all $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$, the mutual energy $I_K(\mu_1, \mu_2)$ satisfies

$$I_K(\mu_1,\mu_2) \le \sqrt{I_K(\mu_1)I_K(\mu_2)},$$
(3.2)

and $I_K(\mu_1) \ge 0$. If K is strictly positive definite, then equality in (3.2) holds if and only if $\mu_1 = \mu_2$ on Borel subsets of Ω .

Observe that, just like in Lemma 3.1.1, we could replace $\mathbb{P}(\Omega)$ with $\widetilde{\mathbb{P}}(\Omega)$.

Proof. Suppose that *K* is positive definite. Then for any $t \in \mathbb{R}$ and $\mu_1, \mu_2 \in \mathbb{P}(\Omega), t\mu_1 - \mu_2 \in \mathcal{M}(\Omega)$ so

$$g(t) := t^2 I_K(\mu_1) - 2t I_K(\mu_1, \mu_2) + I_K(\mu_2) = I_K(t\mu_1 - \mu_2) \ge 0.$$

Thus the discriminant $4I_K(\mu_1,\mu_2)^2 - 4I_K(\mu_1)I_K(\mu_2)$ of the quadratic polynomial g(t) is nonpositive, which yields (3.2).

Suppose instead that (3.2) holds for all probability measures. For any $\mu \in \mathscr{M}(\Omega)$, there exists $a, b \ge 0$ and $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$ such that $\mu = a\mu_1 - b\mu_2$. We then have that

$$I_{K}(\mu) = a^{2}I_{K}(\mu_{1}) - 2abI_{K}(\mu_{1},\mu_{2}) + b^{2}I_{K}(\mu_{2}) \ge (a\sqrt{I_{K}(\mu_{1})} - b\sqrt{I_{K}(\mu_{2})})^{2} \ge 0,$$

implying that *K* is positive definite.

If *K* is strictly positive definite and equality holds in (3.2), then g(t) has a unique root t'. Since *K* is strictly positive definite, we have $\mu_1 = t'\mu_2$ on Borel subsets of Ω and since $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$, we see that t' = 1 and $\mu_1 = \mu_2$.

Finally, we use the arithmetic mean inequality of Lemma 3.1.1 to show that minimizers are unique for strictly conditionally positive definite kernels.

Theorem 3.1.3. Suppose that K is conditionally strictly positive definite. Then I_K has a unique minimizer (either in $\mathbb{P}(\Omega)$ or in $\widetilde{\mathbb{P}}(\Omega)$).

Proof. Suppose that $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$ (the argument for $\widetilde{\mathbb{P}}(\Omega)$ is identical) are equilibrium measures of I_K , i.e. energy minimizers. Due to Lemma 3.1.1 the probability measure $\mu = \frac{1}{2}(\mu_1 + \mu_2)$ satisfies

$$I_{K}(\mu) = \frac{1}{4}I_{K}(\mu_{1}) + \frac{1}{2}I_{K}(\mu_{1},\mu_{2}) + \frac{1}{4}I_{K}(\mu_{2}) \le \frac{1}{2}I_{K}(\mu_{1}) + \frac{1}{2}I_{K}(\mu_{1}) = \mathscr{I}_{K}(\Omega).$$
(3.3)

Thus, μ must also be a minimizer of I_K , which means that $2I_K(\mu_1, \mu_2) = I_K(\mu_1) + I_K(\mu_2)$. By Lemma 3.1.1, $\mu_1 = \mu_2$ on Borel subset of Ω , which proves the claim.

Positive Definiteness and Convexity of the Energy Functional

Convexity often plays an important role in optimization problems, so it is natural to suspect that the convexity of energy functionals relates to energy minimization, and it is well known that conditional positive definiteness does as well. Therefore, it is not surprising that the two notions are related. We shall demonstrate that they are, in fact, equivalent. This equivalence seems to have been overlooked in most of the literature. The proof presented below has appeared in the author and coauthors' recent paper [BFG⁺a].

Definition 3.1.4. Let $K : \Omega \times \Omega \to \mathbb{R}$ be a kernel. We say that I_K is **convex at** $\mu \in \mathbb{P}(\Omega)$ if for every $v \in \mathbb{P}(\Omega)$ there exists some $t_v \in (0,1]$, such that for all $t \in [0,t_v)$

$$I_K((1-t)\mu + t\nu) \le (1-t)I_K(\mu) + tI_K(\nu).$$
(3.4)

We say I_K is convex on $\mathbb{P}(\Omega)$ if inequality (3.4) holds for every μ , $\nu \in \mathbb{P}(\Omega)$ and all $t \in [0, 1]$.

We observe that convexity of I_K on $\mathbb{P}(\Omega)$ is equivalent to the fact that I_K is convex at all $\mu \in \mathbb{P}(\Omega)$. Indeed, if (3.4) fails for some μ , $\nu \in \mathbb{P}(\Omega)$, then the polynomial $f(t) = I_K((1-t)\mu + t\nu)$ is not convex on the interval [0,1], i.e. f''(t) < 0 on some

subinterval $[a,b] \subset [0,1]$. But in this case, one can easily see that I_K fails to be convex at $\mu_a = (1-a)\mu + a\nu$.

We first show that convexity is equivalent to the arithmetic mean inequality (3.1) for mixed energies.

Lemma 3.1.5. The energy functional I_K is convex at $\mu \in \mathbb{P}(\Omega)$ if and only if for all $\nu \in \mathbb{P}(\Omega)$,

$$I_{K}(\mu, \nu) \leq \frac{1}{2} (I_{K}(\mu) + I_{K}(\nu)).$$
(3.5)

Consequently, I_K *is convex on* $\mathbb{P}(\Omega)$ *if and only if inequality* (3.5) *holds for all* $\mu, \nu \in \mathbb{P}(\Omega)$ *.*

Proof. Let $v \in \mathbb{P}(\Omega)$ and assume that the arithmetic mean inequality (3.5) holds. Then for all $t \in [0, 1]$,

$$I_{K}((1-t)\mu+t\nu) = (1-t)^{2}I_{K}(\mu) + 2(1-t)tI_{K}(\mu,\nu) + t^{2}I_{K}(\nu) \le (1-t)I_{K}(\mu) + tI_{K}(\nu).$$
(3.6)

So I_K is indeed convex at μ .

For the converse direction, assume that I_K is convex at μ . Then for any $v \in \mathbb{P}(\Omega)$ and t > 0 sufficiently small, i.e. $t \in (0, t_v)$, inequality (3.6) holds, so

$$2(1-t)tI_{K}(\mu,\nu) \leq t(1-t)(I_{K}(\mu)+I_{K}(\nu)).$$

Dividing by t(1-t), we obtain the arithmetic mean inequality (3.5).

Lemmas 3.1.1 and 3.1.5 together clearly imply the desired equivalence.

Proposition 3.1.6. Let $K : \Omega \times \Omega \to \mathbb{R}$ be a kernel. Then K is conditionally positive definite *if and only if* I_K *is convex on* $\mathbb{P}(\Omega)$.

Minimizing Measures: Basic Potential Theory

It is well known that the behavior of the minimizing measures is closely connected to the behavior of the potential of the kernel with respect to the minimizing measure. For a detailed account of the topic, we refer the reader to [BHS19] book. The following simple statement is classical, see e.g. [Bjö56]. We provide its proof for completeness.

Theorem 3.1.7. Suppose that μ is a minimizer of I_K over $\mathbb{P}(\Omega)$. Then $U_K^{\mu}(x) = I_K(\mu)$ on $\operatorname{supp}(\mu)$ and $U_K^{\mu}(x) \ge I_K(\mu)$ on Ω .

Proof. Let $v \in \mathscr{Z}(\Omega)$ such that $\mu(A) + \varepsilon v(A) \ge 0$ for all Borel subsets $A \subseteq \Omega$ and $0 \le \varepsilon \le 1$. This clearly means that $\mu + \varepsilon v \in \mathbb{P}(\Omega)$, so

$$I_K(\mu) \leq I_K(\mu + \varepsilon \nu) = I_K(\mu) + 2\varepsilon I_K(\mu, \nu) + \varepsilon^2 I_K(\nu).$$

Thus, for $0 \le \varepsilon \le 1$,

$$0 \leq \varepsilon \left(2I_K(\mu, \nu) + \varepsilon I_K(\nu) \right).$$

This means that $I_K(\mu, \nu) \ge 0$.

Suppose, indirectly, that there exist $a, b \in \mathbb{R}$, $z \in \text{supp}(\mu)$, and $y \in \Omega$ such that

$$a = U_K^\mu(z) > U_K^\mu(y) = b$$

Let *B* be a ball centered at *z*, small enough so that $y \notin B$ and the oscillation of $U_K^{\mu}(x)$ (i.e. $\max_{x \in B} U_K^{\mu}(x) - \min_{y \in B} U_K^{\mu}(y)$) is at most $\frac{a-b}{2}$, and let $m = \mu(B)$. Let *v* be defined by

$$\mathbf{v}(A) = m\delta_{\mathbf{v}}(A) - \boldsymbol{\mu}(A \cap B). \tag{3.7}$$

Then

$$I_K(\mu,\nu) = U_K^{\mu}(y) \cdot m - \int_B U_K^{\mu}(x) d\mu(x) \le bm - \left(a - \frac{a-b}{2}\right)m < 0,$$

which is a contradiction. Thus, if $U_K^{\mu}(z) = a$ for some $z \in \text{supp}(\mu)$, then $U_K^{\mu}(x) \ge a$ for all $x \in \Omega$. Our claim then follows.

Definition 3.1.8. We shall say that μ is a **local minimizer** of I_K if it is a local minimizer in every direction, in other words, if for each $v \in \mathbb{P}(\Omega)$, there exists $\tau_v \in (0,1]$ such that for all $t \in [0, \tau_v]$ we have

$$I_K((1-t)\mu+t\nu)\geq I_K(\mu).$$

Observe that this definition differs from the definition of local minimizers with respect to some metric, such as the Wasserstein d_{∞} metric or the total variation norm (the difference is similar to that between the Gateaux and Fréchet derivatives). In particular, local minimizers with respect to total variation norm are also local minimizers in the above sense, but not vice versa. In the present work, we shall always use the words *local minimizers* in the sense of Definition 3.1.8. The following proposition provides a relation between the local and global minimizers.

Analyzing the proof of Theorem 3.1.7, we find that for v defined in (3.7), we can write $\mu + \varepsilon v = (1 - \varepsilon)\mu + \varepsilon \tilde{v}$ with $\tilde{v} = \mu + v \in \mathbb{P}(\Omega)$. Hence, one arrives at a contradiction even if μ is just a local minimizer.

Corollary 3.1.9. The statement of Theorem 3.1.7 remains true if we only assume that μ is a local (not global) minimizer of I_K .

As we shall see in Theorem 3.2.10, under some additional conditions, in particular, if *K* is conditionally positive definite, the statement of Theorem 3.1.7 can be reversed.

Energy Minimizers and Hilbert–Schmidt Operators

There is a close relation between energy minimizers and the properties of the associated Hilbert–Schmidt operator $T_{K,\mu}$ in $L^2(\Omega,\mu)$. We have the following statement.

Lemma 3.1.10. Let K be a kernel on $\Omega \times \Omega$ and assume that $\mu \in \mathbb{P}(\Omega)$ is a global or local minimizer of I_K with $I_K(\mu) \ge 0$. Then the Hilbert–Schmidt operator $T_{K,\mu}$ is positive.
Proof. We start by observing that if μ is a (global or local) minimizer of I_K , then the constant function $\mathbb{1}_{\Omega}$ is an eigenfunction of $T_{K,\mu}$ in $L^2(\Omega,\mu)$. Indeed, according to Theorem 3.1.7 or Corollary 3.1.9, for each $x \in \text{supp}(\mu)$,

$$T_{K,\mu} \mathbb{1}_{\Omega}(x) = \int_{\Omega} K(x,y) d\mu(y) = U_K^{\mu}(x) = I_K(\mu) \mathbb{1}_{\Omega}(x).$$
(3.8)

Assume, indirectly, that $T_{K,\mu}$ is not positive. By Lemma 2.3.1, $T_{K,\mu}$ is compact and self-adjoint, so there exists an eigenfunction ϕ such that $T_{K,\mu}\phi = \lambda\phi$ with $\lambda < 0$. Since ϕ is continuous, and therefore bounded, on $\operatorname{supp}(\mu)$, we have that for sufficiently small |t|, the measure

$$\boldsymbol{\mu}_t = (1 + t\boldsymbol{\phi})\boldsymbol{\mu}$$

is positive. As we noted above, $\mathbb{1}_{\Omega}$ is an eigenfunction of $T_{K,\mu}$ corresponding to the eigenvalue $I_K(\mu) \ge 0$. Clearly, then, $\mathbb{1}_{\Omega}$ and ϕ are orthogonal, so

$$\mu_t(\Omega) = \int_{\Omega} (1 + t\phi(x)) d\mu(x) = \mu(\Omega) = 1$$

and for s > 0

$$I_K((1-s)\mu + s\mu_t) = I_K(\mu_{st})$$

= $\int_{\Omega} \int_{\Omega} K(x,y)(1 + st\phi(x))(1 + st\phi(y)d\mu(x)d\mu(y))$
= $I_K(\mu) + \lambda t^2 \int_{\Omega} |\phi(x)|^2 d\mu(x) < I_K(\mu)$

which contradicts the (local) minimality of μ over probability measures.

Recall that according to Lemma 2.3.1, the operator $T_{K,\mu}$ is positive if and only if K is positive definite on the support of μ . If the condition $I_K(\mu) \ge 0$ in Lemma 3.1.10 is not satisfied, we can replace K by $K'(x,y) = K(x,y) - I_K(\mu)$, which does not affect energy min-

imizers. Then $I_{K'}(\mu) = 0$, and hence, Lemma 3.1.10 applies. Therefore $T_{K',\mu}$ is positive, i.e. K' is positive definite on $\widetilde{\Omega} = \operatorname{supp}(\mu)$. In other words, K is positive definite up to an additive constant, as a kernel on $\widetilde{\Omega} \times \widetilde{\Omega}$. We arrive at the following important fact.

Lemma 3.1.11. Let K be a kernel on $\Omega \times \Omega$. Suppose that μ is a minimizer of I_K over $\mathbb{P}(\Omega)$. Then the kernel K must be positive definite modulo a constant on $\operatorname{supp}(\mu)$, i.e. as a kernel on $\operatorname{supp}(\mu) \times \operatorname{supp}(\mu)$. If $I_K(\mu) \ge 0$, then K is positive definite on $\operatorname{supp}(\mu)$.

Various statements of this type are known in the literature [CFP17, FS13].

3.2 Invariant Measures

As explained in the previous section, measures with constant potentials are particularly interesting from the point of view of energy minimization. They also naturally arise in metric geometry, in connection with the so called "rendezvous numbers" [CMY86], and we draw the term "invariant" from this literature. These applications and various interesting properties warrant a separate discussion of such measures.

Definition, Examples, and Comments

We start with the following definition.

Definition 3.2.1. Let K be a kernel on $\Omega \times \Omega$. We say that a measure $\mu \in \mathbb{P}(\Omega)$ is Kinvariant if the potential of K with respect to this measure is constant on Ω , i.e.

$$U_K^{\mu}(x) = I_K(\mu) \quad \text{for every } x \in \Omega.$$
(3.9)

We shall see shortly that these measures have an array of remarkable properties. Notice that the definition does not require that μ has full support: while some of the statements in this chapter will require this additional assumption, i.e. $\operatorname{supp}(\mu) = \Omega$, the majority of them hold even in its absence. Before proceeding to these properties we shall provide some examples, showing that *K*-invariant measures is a rather rich notion. Observe that most of the examples below have full support. Numerous statements that we shall prove for *K*-invariant measures will apply to all these natural examples.

- If μ (locally) minimizes I_K and has full support, then according to Theorem 3.1.7 and Corollary 3.1.9, the measure μ is *K*-invariant.
- Let $\Omega = \mathbb{S}^{d-1}$ and assume that *K* is rotationally invariant, i.e. $K(x,y) = F(\langle x,y \rangle)$. Then the normalized uniform surface measure is *K*-invariant, since the potential $U_F^{\sigma}(x) = \int_{\mathbb{S}^{d-1}} F(\langle x,y \rangle) d\sigma(y)$ is obviously independent of $x \in \mathbb{S}^{d-1}$.
- If, moreover, the F from the previous example is a polynomial of degree M, and μ is a spherical M-design, then μ must also be K-invariant.
- Similarly, assume that Ω is a compact two-point homogeneous space, K is invariant with respect to the group of isometries, and η is the normalized uniform measure on Ω. Then η is K-invariant. This equally applies to connected (e.g., projective spaces) and discrete (e.g., Hamming cube) two-point homogeneous spaces.
- If Ω is a compact topological group, µ is its normalized Haar measure, and K is invariant with respect to the group operation, i.e. K(x,y) = F(x-y), then µ is K-invariant.
- Let $\Omega = [0, 1]$ and K(x, y) = |x y|. Then the measure $\mu = \frac{1}{2}(\delta_0 + \delta_1)$ is *K*-invariant. Notice that this example provides an invariant measure which does not have full support.

Despite an abundance of examples, the existence of a *K*-invariant measure is possible only under significant restrictions on the geometry of the domain Ω and the structure of the kernel *K*. For example, the following statement is true [CMY86].

Lemma 3.2.2. Assume that U is a finite-dimensional vector space endowed with a strictly convex norm, i.e. $||x + y||_U < ||x||_U + ||y||_U$ unless x and y are linearly dependent. Let $\Omega \subset U$ be compact and set $K(x,y) = ||x - y||_U$. If there exists a K-invariant measure on Ω , then either Ω is a line segment or no three points of Ω are colinear.

For the case when K(x,y) = ||x - y|| is the Euclidean distance, this lemma shows, for example, that an invariant measure doesn't exist for the unit ball, while, as we know, it does exist for the sphere.

Finally, we make the remark that if a measure μ is *K*-invariant, it implies that a constant function is an eigenfunction of the Hilbert–Schmidt operator $T_{K,\mu}$ in $L^2(\Omega,\mu)$ with eigenvalue $\lambda = I_K(\mu)$, which is implied by (3.8).

A Crucial Identity

The following simple relation provides a direct link between energy minimization and (conditional) positive definiteness and will play a decisive role in many results of this section. It is also an important first step in the proof of the Generalized Stolarsky principle (Theorem 5.3.1). In a nutshell, this lemma states that, while I_K is a quadratic functional, it behaves linearly around a *K*-invariant measure.

Lemma 3.2.3. Let K be a kernel on $\Omega \times \Omega$ and let μ be a K-invariant measure, i.e. $U_K^{\mu}(x) = I_K(\mu)$ for all $x \in \Omega$. Then for any $v \in \widetilde{\mathbb{P}}(\Omega)$,

$$I_{K}(\nu - \mu) = I_{K}(\nu) - I_{K}(\mu).$$
(3.10)

More generally, if $\mu \in \mathbb{P}(\Omega)$ such that $U_K^{\mu}(x) \ge I_K(\mu)$, with equality on $\operatorname{supp}(\mu)$, then for any $\nu \in \mathbb{P}(\Omega)$

$$I_K(v-\mu) \le I_K(v) - I_K(\mu),$$
 (3.11)

and equality (3.10) holds for any measure $v \in \widetilde{\mathbb{P}}(\Omega)$ with $\operatorname{supp}(v) \subseteq \operatorname{supp}(\mu)$.

Proof. If μ is *K*-invariant, then for any $\nu \in \widetilde{\mathbb{P}}(\Omega)$,

$$I_K(\mu, \mathbf{v}) = \int_{\Omega} U_K^{\mu}(x) d\mathbf{v}(x) = \int_{\Omega} I_K(\mu) d\mathbf{v}(x) = I_K(\mu).$$
(3.12)

Therefore

$$I_K(\mathbf{v}-\boldsymbol{\mu}) = I_K(\mathbf{v}) - 2I_K(\boldsymbol{\mu},\mathbf{v}) + I_K(\boldsymbol{\mu}).$$

For the second part of our claim, observe that for any $v \in \mathbb{P}(\Omega)$, instead of equality (3.12), one has the inequality $I_K(\mu, v) \ge I_K(\mu)$, and thus,

$$I_K(v - \mu) = I_K(v) - 2I_K(\mu, v) + I_K(\mu) \le I_K(v) - I_K(\mu).$$

Finally, the last statement follows from the first by replacing Ω with supp(μ).

Theorem 3.1.7 and Corollary 3.1.9 show that if μ is a global (or at least local) minimizer of I_K , it satisfies the the conditions of the second statement in Lemma 3.2.3, and if in addition μ has full support, it also satisfies the first condition, i.e. μ is *K*-invariant. Thus Lemma 3.2.3 applies to (local) energy minimizers, which results in the following corollary:

Corollary 3.2.4. Let *K* be a kernel on $\Omega \times \Omega$ and μ be a (local) minimizer of I_K . Then for any $\nu \in \mathbb{P}(\Omega)$

$$I_K(v - \mu) \le I_K(v) - I_K(\mu).$$
 (3.13)

For any $v \in \widetilde{\mathbb{P}}(\Omega)$ *such that* $\operatorname{supp}(v) \subseteq \operatorname{supp}(\mu)$ *, then*

$$I_K(\nu - \mu) = I_K(\nu) - I_K(\mu).$$
(3.14)

Conditional Positive Definiteness and Energy Minimization

Identity (3.10) of Lemma 3.2.3 provides a clear link between energy minimization and conditional positive definiteness. We would like to emphasize that relation (3.10) holds not just for probability measures v, but for arbitrary signed measures of total mass one, i.e. $v \in \widetilde{\mathbb{P}}(\Omega)$. Therefore, one can immediately deduce the following equivalence.

Theorem 3.2.5. Let K be a kernel on $\Omega \times \Omega$ and assume that μ is K-invariant. Then μ minimizes I_K over the set $\widetilde{\mathbb{P}}(\Omega)$ of normalized signed Borel measures if and only if K is conditionally positive definite.

Moreover, μ uniquely minimizes I_K over $\widetilde{\mathbb{P}}(\Omega)$ if and only if K is conditionally strictly positive definite.

Proof. Suppose that *K* is conditionally positive definite. Then for any $v \in \widetilde{\mathbb{P}}(\Omega)$, equality (3.10) holds and, since $(v - \mu)(\Omega) = 0$, we have

$$I_K(\mathbf{v})-I_K(\boldsymbol{\mu})=I_K(\mathbf{v}-\boldsymbol{\mu})\geq 0,$$

which shows that μ minimizes I_K over $\widetilde{\mathbb{P}}(\Omega)$. If *K* is conditionally strictly positive definite, then $I_K(v) = I_K(\mu)$ only if $v - \mu = 0$ on all Borel sets of Ω , i.e. μ is the unique minimizer.

Assume conversely that $I_K(\mu) \leq I_K(\nu)$ for each $\nu \in \widetilde{\mathbb{P}}(\Omega)$. Consider an arbitrary signed measure $\gamma \in \mathscr{Z}(\Omega)$. Define $\nu = \mu + \gamma$, then $\nu(\Omega) = 1$, i.e. $\nu \in \widetilde{\mathbb{P}}(\Omega)$. Thus, applying (3.10) once again, we find that

$$I_K(\gamma) = I_K(\nu - \mu) = I_K(\nu) - I_K(\mu) \ge 0,$$

hence *K* is conditionally positive definite. If μ is the unique minimizer, then the expression above equals zero only for $\gamma = 0$, i.e. *K* is conditionally strictly positive definite.

Obviously, one of the implications holds for minimizers over probability measures $\mathbb{P}(\Omega).$

Corollary 3.2.6. Let K be a kernel on $\Omega \times \Omega$ and assume that μ is K-invariant. If K is conditionally (strictly) positive definite, then μ (uniquely) minimizes I_K over the set $\mathbb{P}(\Omega)$ of Borel measures probability measures.

Some remarks are in order. We would like to remind the reader that in some specific cases, such as the sphere with the uniform surface measure (or more generally, two-point homogeneous spaces with the corresponding uniform measures), the relation between energy minimization and some form of positive definiteness of the kernel is well known [Sch42, BDM18]. However, it is usually demonstrated using the representation theory of the underlying space and the associated orthogonal polynomial (Gegenbauer, Jacobi, Krawtchouk) expansions. Theorem 3.2.5 above is a blanket statement that covers all of these examples and beyond. Moreover, it relies only on the completely elementary identity (3.10), thus simplifying the known proofs in all of the specific cases. In the spherical case, a similar approach has been recently employed in [BDM18].

Conditional Positive Definiteness vs. Positive Definiteness up to an Additive Constant

As we have observed in the previous discussions, two properties, which are somewhat weaker than positive definiteness, play an important role in energy minimization: namely, conditional positive definiteness and positive definiteness up to an additive constant. We have already demonstrated in Lemma 2.2.4 that the latter always implies the former, and the converse implication is not true in general. We shall now show that the converse implication also holds, i.e. conditional positive definiteness implies positive definiteness up to an additive constant, if we additionally assume the existence of a *K*-invariant measure. Moreover, the statement also holds for the "strict" version of these properties.

Lemma 3.2.7. Let K be a kernel on $\Omega \times \Omega$ and assume that K is conditionally (strictly) positive definite. Suppose also that there exists a K-invariant measure $\mu \in \mathbb{P}(\Omega)$. Then K

is (strictly) positive definite up to an additive constant.

Proof. Let *K* be conditionally positive definite. Set $C = -I_K(\mu) + 1$. Then K + C is still conditionally (strictly) positive definite, μ is (K+C)-invariant, and $I_{K+C}(\mu) = (\mu(\Omega))^2 = 1$. For any signed measure ν with $\nu(\Omega) = 1$, identity (3.10) implies that

$$I_{K+C}(v) - I_{K+C}(\mu) = I_{K+C}(v-\mu) = I_K(v-\mu) \ge 0.$$

Therefore, $I_K(v) \ge 1 > 0$.

Now consider an arbitrary measure $\gamma \in \mathscr{M}(\Omega)$. If $\gamma(\Omega) = 0$, then $I_{K+C}(\gamma) \ge 0$ by conditional positive definiteness (and $I_{K+C}(\gamma) > 0$ for $\gamma \ne 0$ for the "strict" version if $\gamma \ne 0$).

If $\gamma(\Omega) = c \neq 0$, we can write $\gamma = cv$ for some $v \in \widetilde{\mathbb{P}}(\Omega)$. Therefore, $I_{K+C}(\gamma) = c^2 I_{K+C}(v) \ge c^2 > 0$. Hence K + C is (strictly) positive definite. \Box

Lemmas 2.2.4 and 3.2.7 together show that in the presence of a *K*-invariant measure, conditional positive definiteness and positive definiteness modulo an additive constant are equivalent notions. This is the case, for example, for rotationally invariant kernels *K* on the sphere, since the uniform surface measure σ is *K*-invariant for all such kernels.

Local and Global Minimizers

Under some additional assumptions local minimizers of I_K are necessarily global minimizers. Some facts of this type have been observed in the papers of the author with various coauthors [BFG⁺a, BGM⁺a]. Here we prove a variety of more general statements with the same flavor.

Proposition 3.2.8. Suppose that μ is a local minimizer of I_K and any of the following two conditions holds:

- 1. the measure μ is K-invariant, i.e. $U_K^{\mu}(x) = I_K(\mu)$ for all $x \in \Omega$;
- 2. *K* is conditionally positive definite.

Then μ is a global minimizer of I_K .

Observe that, according to Corollary 3.1.9, condition (1) is automatically satisfied if μ has full support.

Proof. Assume that (1) holds. Then for any $v \in \mathbb{P}(\Omega)$, we have $I_K(\mu, v) = I_K(\mu)$ and therefore, since μ is a local minimizer, for small t > 0, we have

$$I_{K}(\mu) \leq I_{K}((1-t)\mu + t\nu) = (1-t)^{2}I_{K}(\mu) + 2t(1-t)I_{K}(\mu,\nu) + t^{2}I_{K}(\nu)$$
$$= (1-t)^{2}I_{K}(\mu) + 2t(1-t)I_{K}(\mu) + t^{2}I_{K}(\nu) = (1-t^{2})I_{K}(\mu) + t^{2}I_{K}(\nu).$$

Thus $I_K(\mu) \leq I_K(\nu)$, i.e. μ is a global minimizer of I_K in $\mathbb{P}(\Omega)$.

Now assume that *K* is conditionally positive definite. Then for each $v \in \mathbb{P}(\Omega)$, the measure $v - \mu$ has total mass zero. Therefore, according to conditional positive definiteness of *K* and inequality (3.13) of Corollary 3.2.4,

$$I_K(\mathbf{v}) - I_K(\mu) \ge I_K(\mathbf{v} - \mu) \ge 0.$$
 (3.15)

Therefore, in this case, μ also minimizes I_K .

If both conditions (1) and (2) hold simultaneously, an even stronger conclusion can be drawn.

Proposition 3.2.9. Let μ be a local minimizer of I_K . Assume in addition that μ is *K*-invariant and *K* is conditionally positive definite. Then μ is a global minimizer of I_K over $\widetilde{\mathbb{P}}(\Omega)$, the set of all signed Borel measures with total mass one.

Proof. Let $v \in \widetilde{\mathbb{P}}(\Omega)$. Since μ is a *K*-invariant local minimizer, we can apply conditional positive definiteness of *K* in conjunction with equality (3.10) of Lemma 3.2.3 to obtain

$$I_K(\mathbf{v}) - I_K(\boldsymbol{\mu}) = I_K(\mathbf{v} - \boldsymbol{\mu}) \ge 0,$$

giving us our claim.

Finally, another version of the local-to-global minimization principle may be proved under the assumption that I_K is convex at μ , which according to Lemma 3.1.5, is equivalent to the fact that the arithmetic mean inequality (3.5) holds for any measure $v \in \mathbb{P}(\Omega)$. We have the following statement.

Theorem 3.2.10. Suppose that K is a kernel on $\Omega \times \Omega$ and that for some $\mu \in \mathbb{P}(\Omega)$ there exists a constant $M \in \mathbb{R}$ such that $U_K^{\mu}(x) \ge M$, with equality on $\operatorname{supp}(\mu)$. If I_K is convex at μ , then μ is a global minimizer of I_K over $\mathbb{P}(\Omega)$.

Before proving this statement we observe that its first assumption is satisfied in any of the following two cases: (a) if μ is *K*-invariant; (b) if μ is a local minimizer, according to Corollary 3.1.9. In addition, if convexity at μ were replaced with convexity of I_K on $\mathbb{P}(\Omega)$, then in view of Proposition 3.1.6, this would be equivalent to the conditional positive definiteness of *K*. Thus, this theorem recovers and strengthens part (2) of Proposition 3.2.8. Moreover, we have the following immediate corollary, which will be used later.

Corollary 3.2.11. Let K be a kernel on $\Omega \times \Omega$ and let $\mu \in \mathbb{P}(\Omega)$ be a K-invariant measure. If I_K is convex at μ , then μ is a global minimizer of I_K over $\mathbb{P}(\Omega)$.

Proof of Theorem 3.2.10. Observe first that the constant *M* is necessarily equal to $I_K(\mu)$:

$$I_K(\mu) = \int_{\Omega} U_K^{\mu}(x) d\mu(x) = \int_{\operatorname{supp}(\mu)} M d\mu(x) = M.$$

For any $v \in \mathbb{P}(\Omega)$, we have that

$$I_K(\mu, \nu) = \int_{\Omega} U_K^{\mu}(x) d\nu(x) \ge I_K(\mu).$$

Convexity of I_K at μ , according to Lemma 3.1.5, is equivalent to the arithmetic mean

inequality (3.5). Thus,

$$I_K(\mu) \leq I_K(\mu, \mathbf{v}) \leq rac{1}{2}I_K(\mu) + rac{1}{2}I_K(\mathbf{v}),$$

so $I_K(\mathbf{v}) \ge M = I_K(\boldsymbol{\mu})$.

3.3 Invariant Measures and Minimizers with Full Support

It is now time to summarize the results of the previous sections. It may not be yet obvious, but we have proven (sometimes quite surprising) equivalences between many different notions. We shall restrict our attention to the case when the measure μ is *K*-invariant (i.e. has constant potential) and has full support. As we have discussed before, these conditions are satisfied by many natural candidates (the uniform measure on the sphere or other two-point homogeneous spaces, the Haar measure on a compact topological group, etc). Though a majority of the implications are valid even just for *K*-invariant measures without the full support assumption, assuming that μ has full support truly ties the picture together. We shall carefully trace which of the conclusions require this condition.

We start with the following long list of equivalences.

Theorem 3.3.1. Let K be a kernel on $\Omega \times \Omega$. Assume that there exists a measure $\mu \in \mathbb{P}(\Omega)$, which is K-invariant and has full support, i.e. $U_K^{\mu}(x) = I_K(\mu)$ for all $x \in \Omega$ and $\operatorname{supp}(\mu) = \Omega$.

Then the following conditions are equivalent:

1. K positive definite modulo a constant.

2. K is conditionally positive definite.

3. μ is a local minimizer of I_K .

- 4. μ is a global minimizer of I_K over $\mathbb{P}(\Omega)$.
- 5. μ is a global minimizer of I_K over $\widetilde{\mathbb{P}}(\Omega)$.
- 6. I_K is convex.
- 7. I_K is convex at μ .
- 8. The arithmetic mean inequality (3.1) holds for all $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$ (or, equivalently for all $\mu_1, \mu_2 \in \widetilde{\mathbb{P}}(\Omega)$).
- 9. The arithmetic mean inequality (3.1) holds when $\mu_1 = \mu$.
- 10. The kernel K can be represented as

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x) \phi_j(y)$$

where the series converges uniformly and absolutely, the function ϕ_1 is constant, and $\lambda_j \ge 0$ for $j \ge 2$.

Proof. For the reader's convenience the implications proving this theorem are summarized in Figure 3.1.

We open with the list the implications that do not require any assumptions on μ . It is obvious that (5) implies (4), which in turn implies (3). Also, (6) implies (7), and similarly, (8) implies (9).

The equivalence between (2), (6), and (8) is proved in Lemmas 3.1.1 and 3.1.5 together with Proposition 3.1.6. Lemma 3.1.5 also establishes the equivalence between (7) and (9). Lemma 2.2.4 shows that (1) implies (2).

The following implications rely on the fact that μ is *K*-invariant, but do not require μ to have full support. Lemma 3.2.7 demonstrates that (2) implies (1). Theorem 3.2.5 yields the equivalence of (2) and (5). Corollary 3.2.11 to Theorem 3.2.10 shows that (4) follows



Figure 3.1: Equivalences in Theorem 3.3.1: double line (cyan) arrows are implications that hold without additional assumptions; single line (blue) ones require *K*-invariance, but not full support; the dashed (black) arrows represent the implications which do require the assumption of full support.

from (7). Finally, part (1) of Proposition 3.2.8 guarantees that (3) implies (4).

The equivalence between (1) and (10) is discussed in Lemma 2.2.2 and Mercer's Theorem (see Theorem 2.3.3), or more specifically Corollary 2.3.4. The fact that (1) implies (10) relies on the fact that for a *K*-invariant measure μ , the constant function $\mathbb{1}_{\Omega}$ is an eigenfunction of the Hilbert–Schmidt operator $T_{K,\mu}$ in $L^2(\Omega,\mu)$, and the condition $\operatorname{supp}(\mu) = \Omega$ implies that the expansion in Mercer's Theorem is valid on all of Ω . The implication (10) \Rightarrow (1) holds without any additional assumptions, according to Lemma 2.2.2.

In conclusion, we observe that Lemma 3.1.11 demonstrates that (4) implies (1), which closes the loop of implications – and, except for the standalone equivalence between (1) and (10), this is the only implication in our proof where the fact that $\operatorname{supp}(\mu) = \Omega$ is used. Indeed, Lemma 3.1.11 only guarantees that the kernel *K* is positive definite (up to constant) on the support of the minimizer. Observe also that due to Theorem 3.1.7, if (4) holds and μ has full support.

To reiterate, this theorem reveals several interesting novel effects that happen to a *K*-invariant measure (with full support):

Equivalence between minimization over the set P(Ω) of probability measures and the set P(Ω) of all signed measures of mass one. This effect has been observed for rotationally invariant kernels on the sphere and the surface measure σ by the author and collaborators [BDM18]. This is not necessarily the case in other settings. In particular, for the integral over the unit ball B^d

$$\int_{\mathbb{B}^d}\int_{\mathbb{B}^d} \|x-y\|d\mu(x)d\mu(y),$$

according to [Bjö56], the unique maximizer over both $\mathbb{P}(\mathbb{B}^d)$ and $\mathbb{P}(\mathbb{S}^{d-1})$ is σ . According to the aforementioned equivalence, σ is also a maximizer over $\widetilde{\mathbb{P}}(\mathbb{S}^{d-1})$,

while in the case of signed measures on the ball, the maximizer does not exist [HNW11].

- Equivalence between being a local and global energy minimizer. This effect, in a slightly less general form, has been observed by the author and collaborators in [BFG⁺a, BGM⁺a].
- Equivalence between other local and global properties: e.g., the energy functional I_K is convex on the whole set of probability measures if and only if it is convex just at the special measure μ .
- Equivalence between conditional positive definiteness of the kernel and positive definiteness up to constant, which is not true in general.

We now formulate similar theorems for kernels which have the "strict" version of the properties and for positive definite kernels. We start with the latter.

Theorem 3.3.2. Suppose that K is a kernel on $\Omega \times \Omega$ and that there exists a measure $\mu \in \mathbb{P}(\Omega)$, which is K-invariant and has full support, i.e. $U_K^{\mu}(x) = I_K(\mu)$ for all $x \in \Omega$ and $\operatorname{supp}(\mu) = \Omega$. Then the following conditions are equivalent:

- 1. The kernel K is positive definite.
- 2. The geometric mean inequality (3.2) and $I_K(\mu_1) \ge 0$ hold for all $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$.
- 3. The measure μ is a global minimizer of I_K and satisfies $I_K(\mu) \ge 0$.
- 4. The kernel K can be represented as

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x) \phi_j(y)$$

where the series converges uniformly and absolutely, and $\lambda_j \ge 0$ for $j \ge 1$.

5. There exists some symmetric $k \in L^2(\Omega \times \Omega, \mu \times \mu)$ such that for all $x, y \in \Omega$,

$$K(x,y) = \int_{\Omega} k(x,z)k(z,y)d\mu(z).$$

Proof. Lemma 3.1.2, Corollary 2.3.4, and Proposition 2.3.6 show the equivalence of (1), (2), (4), and (5). Positive definiteness, i.e. condition (1), guarantees that $I_K(\mu) \ge 0$, and that μ is a minimizer, due to Theorem 3.3.1. Conversely, Lemma 3.1.11 shows that (3) implies (1), finishing our proof. Observe also that according to Theorem 3.3.1 it does not matter whether we mean global minimization over $\mathbb{P}(\Omega)$ or $\widetilde{\mathbb{P}}(\Omega)$ in condition (3).

Theorem 3.3.3. Suppose that K is a kernel on Ω^2 and that that there exists a measure $\mu \in \mathbb{P}(\Omega)$ which is K-invariant, i.e. $U_K^{\mu}(x) = I_K(\mu)$ for all $x \in \Omega$. Then the following conditions are equivalent:

- 1. *K* is conditionally strictly positive definite.
- 2. *K* is strictly positive definite modulo a constant.
- 3. μ is the unique minimizer of I_K over $\widetilde{\mathbb{P}}(\Omega)$.

If in addition supp $(\mu) = \Omega$, i.e. μ has full support, then each of the conditions (1)–(3) implies the following

4. The kernel K can be represented as

$$K(x,y) = \sum_{j=1}^{\dim(L^2(\Omega,\mu))} \lambda_j \phi_j(x) \phi_j(y)$$

where $\{\phi_j\}$ is the orthonormal basis consisting of eigenfunctions of the Hilbert– Schmidt operator $T_{K,\mu}$ in $L^2(\Omega,\mu)$, the function ϕ_1 is a constant, the series converges uniformly and absolutely, and $\lambda_j > 0$ for $j \ge 2$.

Moreover, if the span of $\{\phi_j\}$ is dense in $C(\Omega)$, then (4) also implies conditions (1)–(3).

Proof. Lemma 2.2.4 shows that (2) implies (1), while Lemma 3.2.7 provides the converse implication. The equivalence of (3) and (1) is proved in Theorem 3.2.5.

Before we turn to dealing with condition (4), recall that *K*-invariance of μ implies that a constant is an eigenfunction of the operator $T_{K,\mu}$, so we shall assume that $\phi_1 = \mathbb{1}_{\Omega}$.

Now we show that (3) implies (4). Since μ minimizes I_K and has full support, by Theorem 3.3.1, K is positive definite up to a constant, and since $\phi_1 = 1$, by Mercer's Theorem, the expansion in part (4) holds with some $\lambda_j \in \mathbb{R}$ and with the series converging uniformly and absolutely. Suppose, indirectly, that μ is the unique minimizer of I_K over $\widetilde{\mathbb{P}}(\Omega)$, but there exists some $l \ge 2$ such that $\lambda_l \le 0$. Let $dv(x) = (1 + \phi_l(x))d\mu(x)$. Orthogonality implies that $\int_{\Omega} \phi_l(x)d\mu(x) = 0$, therefore $v \in \widetilde{\mathbb{P}}(\Omega)$. Then we obtain

$$\begin{split} I_{K}(\mathbf{v}) &= \int_{\Omega} \int_{\Omega} K(x, y) (1 + \phi_{l}(x)) (1 + \phi_{l}(y)) d\mu(x) d\mu(y) \\ &= I_{K}(\mu) + 2 \langle T_{K,\mu} \phi_{l}, \mathbb{1}_{\Omega} \rangle_{L^{2}(\Omega,\mu)} + \langle T_{K,\mu} \phi_{l}, \phi_{l} \rangle_{L^{2}(\Omega,\mu)} \\ &= I_{K}(\mu) + 2 \lambda_{l} \langle \phi_{l}, \mathbb{1}_{\Omega} \rangle_{L^{2}(\Omega,\mu)} + \lambda_{l} \| \phi_{l} \|_{L^{2}(\Omega,\mu)}^{2} \\ &= I_{K}(\mu) + \lambda_{l} \leq I_{K}(\mu), \end{split}$$

which contradicts the fact that μ is the unique minimizer over $\widetilde{\mathbb{P}}(\Omega)$.

Finally, we show that (4) implies (2) under the aforementioned additional assumption. Let $K'(x,y) = K(x,y) - \lambda_1 + 1$ and $v \in \mathcal{M}(\Omega)$. Then

$$\begin{split} I_{K'}(\mathbf{v}) &= \int_{\Omega} \int_{\Omega} K'(x,y) d\mathbf{v}(x) d\mathbf{v}(y) \\ &= (\mathbf{v}(\Omega))^2 + \sum_{j=2}^{\dim(L^2(\Omega,\mu))} \int_{\Omega} \int_{\Omega} \lambda_j \phi_j(x) \phi_j(y) d\mathbf{v}(x) d\mathbf{v}(y) \\ &= (\mathbf{v}(\Omega))^2 + \sum_{j=2}^{\dim(L^2(\Omega,\mu))} \lambda_j \Big(\int_{\Omega} \phi_j(x) d\mathbf{v}(x)\Big)^2 \ge 0. \end{split}$$

Clearly, the only way that $I_{K'}(v) = 0$ is if $\int_{\Omega} \phi_j(x) dv(x) = 0$ for all $j \ge 1$. By the density of span $(\{\phi_j\}_{j=1}^{\dim(L^2(\Omega,\mu))})$ in $C(\Omega)$, we can conclude that this implies dv = 0, so K' must be strictly positive definite, which completes the proof.

We conclude with the remark that the additional condition imposed for the sufficiency of condition (4) is not very restrictive in practice. For example in the case of rotationally invariant kernels on the sphere and $\mu = \sigma$, the eigenfunction ϕ_j are simply spherical harmonics, which span all polynomials on the sphere and thus their span is dense in the space continuous functions.

3.4 Energies on Two-point Homogeneous Spaces

On a compact, connected, two-point homogeneous set Φ , it is natural to ask, for an isometry invariant kernel *F*, whether the uniform surface measure η is a minimizer of the energy I_F . Since *F* is isometry invariant, we have that for every isometry θ on Φ ,

$$U_F^{\eta}(\theta x) = \int_{\Phi} F(\tau(\theta x, y)) d\eta(y) = \int_{\Phi} F(\tau(x, \theta^{-1}y)) d\eta(\theta^{-1}y)$$
$$= \int_{\Phi} F(\tau(x, y)) d\eta(y) = U_F^{\eta}(x),$$

so the potential U_F^{η} is constant on Φ . This means that η is an *F*-invariant measure and hence Theorems 3.3.1, 3.3.2, and 3.3.3 hold. In particular, η minimizes I_F if and only if *F* is positive definite modulo a constant. Due to (2.35), this means that if *F* is a positive definite (modulo a constant) polynomial of degree *M*, then any weighted *M*-design on Φ is a minimizer of I_F .

We now show that having nonnegative coefficients in the Jacobi expansion of F is equivalent to the fact that η is a minimizer of I_F over $\mathbb{P}(\Phi)$, which has been shown in a number of papers, see for instance [DG03, BD19, BGM⁺a]. It immediately follows from the equivalence of η being a minimizer of I_F and positive definiteness (modulo a constant) of *F*, from Theorem 3.3.1, and Proposition 2.4.2, however, we will provide a direct proof. Let, as before $C_n = C_n^{(\alpha,\beta)}$ denote the normalized Jacobi polynomials associated to Φ .

Proposition 3.4.1. Let $F \in C([-1,1])$, $F(t) = \sum_{n=0}^{\infty} \widehat{F}_n C_n(t)$, and $\mu \in \mathbb{P}(\Phi)$. Then, the following are equivalent:

- 1. $\widehat{F}_n \ge 0$ for all $n \ge 1$,
- 2. the surface measure η is a minimizer of I_F .

Moreover, η is the unique minimizer of I_F if and only if $\widehat{F}_n > 0$ for all $n \ge 1$.

Proof. We first show that η is a minimizer of I_F . Assume that $\widehat{F}_n \ge 0$ for all $n \ge 1$. Then by Lemma 2.4.1, the Fubini theorem, and the addition formula (2.19), we have, for any $\mu \in \mathbb{P}(\Phi)$,

$$\begin{split} I_F(\mu) &= \sum_{n=0}^{\infty} \widehat{F}_n \int_{\Phi} \int_{\Phi} C_n(\tau(x,y)) d\mu(x) d\mu(y) \\ &= \sum_{n=0}^{\infty} \frac{\widehat{F}_n}{\dim V_n} \sum_{n=1}^{\dim V_n} \int_{\Omega} \int_{\Omega} Y_{n,k}(x) \overline{Y_{n,k}(y)} d\mu(x) d\mu(y) \\ &= \widehat{F}_0 + \sum_{n=1}^{\infty} \frac{\widehat{F}_n}{\dim V_n} \cdot \sum_{k=1}^{\dim V_n} \left| \int_{\Omega} Y_{n,k}(x) d\mu(x) \right|^2, \\ &\geq \widehat{F}_0 = I_F(\eta). \end{split}$$

If $\widehat{F}_n > 0$ for all $n \ge 1$, then equality can be achieved above only if μ is orthogonal to all spaces V_n , which directly implies that $\mu = \eta$.

If $\widehat{F}_n < 0$ for some $n \ge 1$, then set, for some $p \in \Phi$, $Y_n(x) = C_n(\tau(x,p)) \in V_n$ and $d\mu(x) = (1 + \varepsilon Y_n(x))d\eta(x)$, where $\varepsilon > 0$ is sufficiently small so that $(1 + \varepsilon Y_n(x)) \ge 0$ on Ω . Orthogonality and the fact that dim $(V_n)C_n(\tau(x,y))$ is the reproducing kernel of V_n show

us that

$$\begin{split} I_F(\mu) &= \int\limits_{\Phi} \int\limits_{\Phi} F(\tau(x,y))(1 + \varepsilon Y_n(x))(1 + \varepsilon Y_n(y))d\eta(x)d\eta(y) \\ &= I_F(\eta) + \varepsilon^2 \widehat{F}_n \int\limits_{\Phi} Y_n(x)^2 d\eta(x) < I_F(\eta), \end{split}$$

implying that η is not a minimizer for I_F . If $\widehat{F}_n = 0$ for some $n \ge 1$, the same argument shows that $I_F(\mu) = I_F(\eta)$, i.e. η is not the unique minimizer.

Observe that by combining the equivalence of positive definiteness (modulo a constant) on Φ and η being a minimizer with Lemma 3.1.11, we immediately achieve the following:

Corollary 3.4.2. Let $F \in C([-1,1])$. Then either η is a minimizer of I_F , or every minimizer of I_F is supported on a proper subset of Φ .

3.5 Energy on the Sphere

Clearly, everything that was said about the invariant measures in general and the case of compact connected two-point homogeneous spaces applies to the case of the sphere \mathbb{S}^{d-1} and the uniform surface measure σ . In particular, Proposition 3.4.1 holds on \mathbb{S}^{d-1} if we replace the normalized Jacobi expansion with a Gegenbauer expansion. We now collect some of our results for the sphere, for convenience.

Proposition 3.5.1. For a function $F \in C([-1,1])$ and $\lambda = \frac{d-2}{2}$, the following conditions are equivalent:

- 1. *F* is positive definite on \mathbb{S}^{d-1} .
- 2. All Gegenbauer coefficients of F are non-negative, i.e.

$$\widehat{F}(n,\lambda) \ge 0$$
 for all $n \ge 0$.

3. There exists a function $f \in L^2([-1,1], w_{\lambda})$ such that

$$F(\langle x, y \rangle) = \int_{\mathbb{S}^{d-1}} f(\langle x, z \rangle) f(z \cdot y) d\boldsymbol{\sigma}(z), \ x, y \in \mathbb{S}^{d-1},$$
(3.16)

- *i.e. F* is the spherical convolution of *f* with itself.
- 4. The normalized Lebesgue measure σ is a minimizer of I_F over $\mathbb{P}(\mathbb{S}^{d-1})$ and $I_F(\sigma) \ge 0$.

While this theorem clearly follows from Corollary 2.5.2, Theorem 3.3.2 and Proposition 3.4.1, we also note that the equivalence of (1) and (2) is a celebrated theorem of Schoenberg [Sch42].

In some situations, Gegenbauer coefficients can give some information about the minimizers, even when σ does not minimize the energy. In particular, the facts that Gegenbauer polynomials C_n^{λ} achieve their maximum at t = 1, are even if *n* is even, and are odd if *n* is odd, provide certain conditions, other than being a positive definite polynomial, under which *there exist* discrete minimizers or *all* minimizers are discrete. These results can be found in [BD19].

Proposition 3.5.2. *Let* $F \in C([-1,1])$ *. Then the following hold:*

- 1. If $\widehat{F}(n,\lambda) \leq 0$ for all $n \geq 1$, then a Dirac delta mass $\mu = \delta_z$, for any $z \in \mathbb{S}^{d-1}$, is a minimizer of I_F . If F has a strict absolute minimum at t = 1 (in particular, if $\widehat{F}(n,\lambda) < 0$ for all $n \geq 1$), then every minimizer is a Dirac mass. Observe that this case resonates with Theorem 4.1.3.
- 2. If $(-1)^{n+1}\widehat{F}(n,\lambda) \ge 0$ for all $n \ge 1$, and the Gegenbauer expansion of F converges uniformly to F, then a measure of the form $\mu = \frac{1}{2}(\delta_z + \delta_{-z})$, for any $z \in \mathbb{S}^{d-1}$, is a minimizer of I_F . Moreover, all minimizers are of this form if the strict inequality $(-1)^{n+1}\widehat{F}(n,\lambda) > 0$ holds.

3. If $\widehat{F}(2n,\lambda) = 0$ and $\widehat{F}(2n-1,\lambda) \ge 0$ for all $n \ge 1$, then every centrally symmetric measure minimizes I_F . In particular, there exist discrete minimizers. Moreover, if $\widehat{F}(2n,\lambda) = 0$ and $\widehat{F}(2n-1,\lambda) > 0$ for all $n \ge 1$, then all minimizers of I_F are centrally symmetric.

Finally, we would like to remind the reader that Jacobi and Gegenbauer expansions are closely connected to Mercer's theorem (Theorem 2.3.3), and one could also take this approach in developing Schoenberg's theory. Indeed, the Funk–Hecke formula (2.29) shows that spherical harmonics of order *n* are eigenfunctions of the Hilbert–Schmidt operator $T_{K,\sigma}$ with the kernel $K(x,y) = F(\langle x,y \rangle)$, corresponding to the eigenvalue $\hat{F}(n,\lambda)$. Thus for positive definite kernels *F*, all Gegenbauer coefficients are non-negative, which recovers Schoenberg's theorem [Sch42]. Moreover, using Mercer's theorem and the addition formula, we find that

$$F(\langle x, y \rangle) = K(x, y) = \sum_{n=0}^{\infty} \widehat{F}_n \sum_{k=1}^{N(n,d)} Y_{n,k}(x) Y_{n,k}(y) = \sum_{n=0}^{\infty} \widehat{F}(n,\lambda) \frac{n+\lambda}{\lambda} C_n^{\lambda}(\langle x, y \rangle),$$

and hence the Gegenbauer expansion of a function $F \in C[-1, 1]$, which is positive definite on \mathbb{S}^{d-1} converges absolutely and uniformly.

Chapter 4

Support of Minimizers

Lemma 3.1.11 implies that for kernels K which are not positive definite (modulo a constant), the support of any minimizing probability measure $\mu \in \mathbb{P}(\Omega)$ cannot be the entire set Ω . Moreover, parts 1 and 2 of Proposition 3.5.2 show us that for kernels of a certain type on the sphere, not only are the minimizers of the energy necessarily discrete, but we can in fact determine their cardinality. Furthermore, we can exactly characterize what those minimizers are. This led us to several interesting questions. In particular, under certain conditions on K (and Ω), are all minimizers of I_K necessarily discrete, or otherwise concentrated? Such a phenomenon has been repeatedly observed for repulsive-attractive potentials (i.e. potentials that depend only on distance and under which particles repel each other at close range, but attract each other at distant ranges) in the Euclidean space \mathbb{R}^d .

Definition 4.0.1. Let (Ω, ρ) be a (not necessarily compact) metric space and suppose that $W : [0, diam(\Omega)) \to \mathbb{R}$ is continuous (with W also being defined and continuous at $diam(\Omega)$ if Ω is compact). We call a potential $K(x,y) = W(\rho(x,y))$ on Ω repulsive-attractive if there exist some $R_0 \in (0, diam(\Omega))$ such that W is strictly decreasing (repelling) on $(0, R_0)$ and strictly increasing (attracting) on $(R_0, diam(\Omega))$.

As mentioned in Section 2.1, energy optimization in non-compact spaces is possible, so long as some constraint on the problem forces the support of minimizers to be bounded.

Repulsive-attractive potentials provide such a restriction, as shown by Carillo, Figalli, and Patacchini in the Euclidean setting.

Lemma 4.0.2. [CFP17, Lemma 2.6] Let $W : [0, \infty) \to \mathbb{R}$ be continuous and repulsiveattractive. Suppose that there exist some R' > 0 such that for r > R', W(r) > W(0). Then the minimizers of the energy

$$I_W(\mu) := \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} W(||x-y||) d\mu(x) d\mu(y)$$
(4.1)

over $\mathbb{P}(\mathbb{R}^d)$ have compact support.

In many energy minimization problems in the compact setting, one deals with purely repulsive potentials, such as the Riesz potentials. However, there are many phenomena that can be modeled by repulsive-attractive potentials. For instance, interactions between neutral molecules, which involve repulsion due to overlapping electron orbitals and attraction due to London dispersion forces, are often simulated by the Lennard-Jones potential [ZD83] The minimization of repulsive-attractive energies has been observed to result in a clustering phenomenon in various models in computational chemistry, mathematical biology, social sciences, and physics [BLT06, BC14, CMV03, FS13, HP06, MEK99, VBUKB12, WS15]. Carrillo, Figalli, and Patacchini provided the first theoretical explanation for this discreteness of minimizers for *mildly repulsive* repulsive-attractive energies in Euclidean space.

Theorem 4.0.3 ([CFP17]). Let W satisfy the conditions of Lemma 4.0.2. Suppose that W(0) = 0 and for some $\gamma > 2$ and C > 0,

$$\lim_{r\to 0^+} W(r)r^{-\gamma}\to -C,$$

i.e. W is mildly repulsive. Then any minimizer $\mu \in \mathbb{P}(\mathbb{R}^d)$ of I_W has finite support.

An analogue of this result for certain compact sets has recently been discovered by Vlasiuk [Vla].

Though little seems to be known about the minimizers of such energies when $\gamma = 2$, this value acts as a "breaking point": for $\gamma < 2$, i.e. when *W* is "strongly repulsive", minimizers are no longer discrete.

Theorem 4.0.4 ([BCLR13]). *Let* W be a repulsive-attractive potential such that W(0) = 0and for some min $\{0, 2 - d\} < \gamma < 2$ and C > 0,

$$\lim_{r\to 0^+} W(r)r^{-\gamma}\to -C.$$

Then the support of any minimizer $\mu \in \mathbb{P}(\mathbb{R}^d)$ of I_W has Hausdorff dimension greater than or equal to $2 - \gamma$.

As we will soon discuss, this result is specific to the Euclidean setting, and does not hold in the compact setting.

We note that Theorem 4.0.3 is not quantitative, and except for the one dimensional case \mathbb{R} [CFP17], there are few estimates on the cardinality of the support of these discrete minimizers. The recent paper of Lim and McCann [LMa] addresses this question for certain repulsive-attractive potentials with $\gamma \ge 2$, showing that the corresponding energies are uniquely minimized by discrete measures on regular simplices, and Kang, Kim, Lim, and Seo have classified certain repulsive-attractive potentials for which two-point measures appear as minimizers in [KKLS21]. However, the problem of generally determining cardinalities of minimizers in the Euclidean setting remains largely unsolved. The problem of understanding minimizers when $\gamma = 2$ is open as well, though some known energies in both the Euclidean case [LMa] and the compact case (see, e.g., Chapter 6) have only discrete minimizers.

We now turn our attention to the compact setting, where we do not require our potentials to be eventually attractive for global minimizers to be well-defined. Comparing Theorems 4.0.3 and 4.0.4 to Theorem 2.1.2 results in a few important observations. First, weak-repulsiveness might be a sufficient condition in the compact setting to guarantee only discrete minimizers. Second, the geometry of a compact domain Ω affects the support of the minimizers to the point where even strongly-repulsive energies may have discrete minimizers, unlike the Euclidean case. Moreover, as we will show in Chapter 5, there exist strongly-repulsive kernels on the sphere (of the same strength of repulsion, in fact) for which minimizers can be discrete and for which only σ , the normalized Lebesgue measure on the sphere, is a minimizer. This clearly means that the geometry of Ω and the strength of repulsion, γ , may not be sufficient to determine discreteness or dimension of minimizers, as they are in the Euclidean case. Finally, much like the Euclidean case, there appears to be a breaking point at $\gamma = 2$ that may change the behavior of minimizers, and little is known for what generally happens in this instance. As many of the kernels we are most interested in, e.g. (6.1), (5.15), and (7.1), are not weakly-repulsive (or not weakly-attractive for the latter two), in this chapter we develop methods to determine properties of minimizers that ignore our potentials' strength of repulsion (attraction).

In Section 4.1, we prove a fairly general condition on a kernel F on the sphere guaranteeing the existence of a discrete minimizer and obtain bounds on the cardinality of the support of such minimizers. The proof relies on the analysis of the structure of extreme points of the set of moment-constrained measures.

In Section 4.2, we show that for any kernel F on the sphere that is real-analytic but not positive definite (modulo a constant), the support of any minimizer of I_F must have empty interior. Moreover, on the circle \mathbb{S}^1 , they are discrete. This generates a certain dichotomy: for an analytic function F, either the energy I_F is minimized by the normalized Lebesgue measure σ , or *all minimizers* have support with empty interior.

In Section 4.3, we expand our focus to all connected, compact, two-point homogeneous spaces Φ , and show that for a certain class of functions, not only can we determine the existence of discrete minimizers and bounds on the cardinality of their support, we can

in fact provide an explicit minimizer. Using arguments based on the linear programming method which goes back to Delsarte and Yudin [Del73, Yud92] and which are reminiscent of those appearing in [CK07], we demonstrate that tight *M*-designs minimize the energy I_F for all absolutely monotonic functions of degree *M* with $F^{(M+1)} \leq 0$. Moreover, if *F* is strictly absolutely monotonic of degree *M*, tight *M*-designs characterize the minimizers.

4.1 Extreme Points for Sets of Moment-constrained Mea-

sures

In the present section, we exhibit a large class of kernels K for which there exist discrete minimizers of the energies I_K . The methods that we employ are closely related to *moment problems*.

Let Ω be a compact metric space and let $\mathscr{B}(\Omega)$ denote the set of positive Borel measures on Ω . Given continuous functions $\phi_0, ..., \phi_n$ on Ω and non-negative constants c_i , we consider the set

$$H = \left\{ \mu \in \mathscr{B}(\Omega) : \int_{\Omega} \phi_i(x) d\mu(x) = c_i, i = 0, 1, \dots, n \right\},$$
(4.2)

which consists of Borel measures whose moments with respect to $\phi_i \in C(\Omega)$ are fixed. We always set $\phi_0 \equiv 1$ and $c_0 = 1$, so that $\mu \in H$ guarantees that μ is a probability measure.

It is easy to see that *H* is convex, bounded, and weak^{*} closed, and therefore is weak^{*} compact. By the Krein–Milman Theorem, *H* is the weak^{*} closure of ext(H) — the set of extreme points of *H* (i.e. the measures $\mu \in H$ such that μ cannot be written as a convex combination of other measures in *H*). The results presented below describe the structure of ext(H) – in particular, the discreteness of its elements.

We start with a theorem that gives a necessary condition for μ to be an extreme point of *H*.

Theorem 4.1.1 ([Dou64]). *Assume that* $\mu \in ext(H)$. *Then*

$$L^{1}(\mu) = \operatorname{span}\{\phi_{0} = 1, \phi_{1}, \dots, \phi_{n}\}.$$
(4.3)

Proof. Assume that $g \in L^{\infty}(\mu)$ satisfies

$$\int_{\Omega} \phi_i g d\mu = 0, \qquad i = 0, 1, \dots, n.$$

Multiplying g by a constant, we may assume that $||g||_{L^{\infty}(\mu)} < 1$. Then the measures μ_{\pm} , defined by $d\mu_{\pm} = (1 \pm g)d\mu$, belong to *H*, since $\mu_{\pm} \in \mathscr{B}(\Omega)$ and

$$\int_{\Omega} \phi_i d\mu_{\pm} = \int_{\Omega} \phi_i (1 \pm g) d\mu = \int_{\Omega} \phi_i d\mu = c_i.$$

At the same time, $\mu = \frac{1}{2}(\mu_{-} + \mu_{+})$. Since $\mu \in ext(H)$, this implies that $\mu_{\pm} = \mu$ and hence $g = 0 \mu$ -a.e. Therefore, the functions ϕ_i span $L^1(\mu)$.

Remark: An infinite version of this statement also holds. In this case, for $\mu \in \text{ext}(H)$ the span of $\{\phi_i\}_{i=0}^{\infty}$ is dense in the space $L^1(d\mu)$.

We now state and prove a result, which demonstrates the discreteness of the elements of ext(H). This result has a number of precursors and extensions; see [Ric57, Rog58, Rog62, Ros51, Win98, Zuh62].

Theorem 4.1.2 ([Kar83]). Let $\mu \in H$. Then the following statements are equivalent:

- 1. $\mu \in \text{ext}(H)$.
- 2. The cardinality of supp(μ) is at most n + 1. Moreover, if we denote supp(μ) = $\{x_1, \ldots, x_k\}$, then the vectors $v_j = (1, \phi_1(x_j), \ldots, \phi_n(x_j)), j = 1, 2, \ldots, k$, are linearly independent.

Proof. (1) \Rightarrow (2). Assume that there exist points $\{x_1, \ldots, x_{n+2}\} \subseteq \text{supp}(\mu)$. Then one can find a vector $y \in \mathbb{R}^{n+2}$, which is not in the span of the vectors $(\phi_i(x_1), \phi_i(x_2), \ldots, \phi_i(x_{n+2}))$,

i = 0, 1, ..., n, since the latter subspace is at most n + 1 dimensional. Appealing to Urysohn's Lemma, one can construct a function $g \in C(\Omega) \subseteq L^1(\mu)$ such that $g(x_i) = \langle y, e_i \rangle$ (where $\{e_1, ..., e_{n+2}\}$ is an orthonormal basis of \mathbb{R}^{n+2}) for i = 1, 2, ..., n+2. But then $g \notin \text{span}\{\phi_i\}$, which contradicts Theorem 4.1.1, so $|\text{supp}(\mu)| \leq n+1$.

Now that it is known that $\mu = \sum_{i=1}^{k} t_i \delta_{x_i}$ with $k \le n+1$, $t_i > 0$, $\sum t_i = 1$, consider the linear system

$$\begin{pmatrix} 1 & \dots & 1 \\ f_1(x_1) & \dots & f_1(x_k) \\ \vdots & \ddots & \vdots \\ f_n(x_1) & \dots & f_n(x_k) \end{pmatrix} \begin{pmatrix} a_1 \\ a_2 \\ \vdots \\ a_k \end{pmatrix} = \begin{pmatrix} 1 \\ c_1 \\ \vdots \\ c_n \end{pmatrix}.$$
(4.4)

This system has a unique solution $a_i = t_i$, since if the solution is not unique, then there is a whole affine subspace of solutions and one could perturb the values of t_i in opposite directions. In other words, on could find two solutions of the form $\{t_i \pm \tau_i\}$ and construct two measures $\mu_{\pm} = \sum_{i=1}^{k} (t_i \pm \tau_i) \delta_{x_i}$ so that $\mu_{\pm} \ge 0$ and $\int \phi_i d\mu_{\pm} = \int \phi_i d\mu$. This would mean $\mu_{\pm} \in K$, and since $\mu = \frac{1}{2}(\mu_+ + \mu_-)$, this would contradict the fact that $\mu \in \text{ext}(H)$. This proves the linear independence of the columns of the matrix above.

(2) \Rightarrow (1). Assume that (2) holds. Then the system (4.4) has a unique solution, i.e. μ is uniquely determined by the condition supp $(\mu) \subseteq \{x_1, \dots, x_k\}$. If $\mu = t\mu_1 + (1-t)\mu_2$ for some $t \in (0,1)$, then supp $(\mu) \subseteq$ supp $(\mu_1) \cup$ supp (μ_2) , and thus supp $(\mu_j) \subseteq \{x_1, \dots, x_k\}$ for j = 1, 2. Therefore $\mu_1 = \mu_2 = \mu$, i.e. $\mu \in \text{ext}(H)$.

We remark that convex geometry plays heavily into similar characterizations of solutions to infinite dimensional optimization problems in the recent papers [BCDC⁺19, CRPW12, UFW17].

Applications of Karr's Theorem: Existence of Discrete Minimizers

We now apply the results on moment-constrained measures to show that for kernels K with certain expansions, there exist discrete minimizers of I_K . We specifically prove it for

rotationally invariant kernels F on the sphere, but the generalization to other spaces is clear.

Let $\widehat{F}(n,\lambda)$ denote the coefficients in the Gegenbauer expansion (2.25) of the function $F \in C([-1,1])$. Consider the sets $N_+(F) = \{n \ge 0 : \widehat{F}(n,\lambda) > 0\}$ and $N_-(F) = \{n \ge 0 : \widehat{F}(n,\lambda) < 0\}$. We shall assume that

$$\left|N_{+}(F)\right| < \infty, \tag{4.5}$$

i.e. there are only finitely many terms of (2.25) with $\widehat{F}(n,\lambda) > 0$. In this case, Lemma 2.4.1 implies that the Gegenbauer expansion of *F* converges uniformly and absolutely.

Recall that \mathscr{H}_n^d denotes the space of spherical harmonics of degree *n* on \mathbb{S}^{d-1} . We are now ready to state the main theorem of the section.

Theorem 4.1.3. Assume that the Gegenbauer expansion (2.25) of the function $F \in C([-1, 1])$ satisfies

$$\left|N_{+}(F)\right| = \left|\{n \ge 0 : \widehat{F}(n,\lambda) > 0\}\right| < \infty$$

i.e. the Gegenbauer expansion has only finitely many positive terms. Then there exists a discrete measure $\mu^* \in \mathbb{P}(\mathbb{S}^{d-1})$ such that

$$\left|\operatorname{supp}(\boldsymbol{\mu}^*)\right| \le \sum_{n \in N_+(F) \cup \{0\}} \dim(\mathscr{H}_n^d), \tag{4.6}$$

and μ^* minimizes the energy I_F over $\mathbb{P}(\mathbb{S}^{d-1})$, i.e.

$$I_F(\boldsymbol{\mu}^*) = \mathscr{I}_F(\mathbb{S}^{d-1}). \tag{4.7}$$

Proof. Let $v \in \mathbb{P}(\mathbb{S}^{d-1})$ be any minimizer of I_F . We shall use the addition formula for spherical harmonics (2.28), as well as the absolute convergence of (2.25), to re-write $I_F(\mu)$,

for any measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$, as

$$\begin{split} I_F(\mu) &= \sum_{n=0}^{\infty} \widehat{F}(n,\lambda) \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \frac{n+\lambda}{\lambda} C_n^{\lambda}(\langle x,y \rangle) d\mu(x) d\mu(y) \\ &= \sum_{n=0}^{\infty} \widehat{F}(n,\lambda) \left[\sum_{j=1}^{\dim(\mathscr{H}_n^d)} \left(\int_{\mathbb{S}^{d-1}} Y_{n,j}(x) d\mu(x) \right)^2 \right] \\ &= \sum_{n \in N_+(F)} \widehat{F}(n,\lambda) \left[\sum_{j=1}^{\dim(\mathscr{H}_n^d)} \left(\int_{\mathbb{S}^{d-1}} Y_{n,j}(x) d\mu(x) \right)^2 \right] \\ &- \sum_{n \in N_-(f)} \left(-\widehat{F}(n,\lambda) \right) \left[\sum_{j=1}^{\dim(\mathscr{H}_n^d)} \left(\int_{\mathbb{S}^{d-1}} Y_{n,j}(x) d\mu(x) \right)^2 \right], \end{split}$$

the last of which we define as the difference of functionals $\mathscr{F}(\mu) - \mathscr{G}(\mu)$. It is easy to see that \mathscr{G} is convex with respect to μ since it is a positive linear combination of squares of linear functionals of μ . Let us set

$$H_F = \left\{ \mu \in \mathscr{B}(\mathbb{S}^{d-1}) : \int_{\mathbb{S}^{d-1}} Y_{n,j}(x) d\mu(x) = \int_{\mathbb{S}^{d-1}} Y_{n,j}(x) d\nu(x), n \in N_+(F), j = 1, \dots, \dim(\mathscr{H}_n^d) \right\},$$

so that $v \in H_F$ and $\mathscr{F}(\mu) = \mathscr{F}(v)$ for $\mu \in H_F$. Without loss of generality, we shall assume that $0 \in N_+(F)$. This guarantees that $\mu \in H_F$ is a probability measure (similarly to setting $c_0 = 1$ and $\phi_0 \equiv 1$ earlier). Since $|N_+(F)| < \infty$, the set H_F has finitely many moment constraints and Theorem 4.1.2 is applicable. In fact, the number of constraints is exactly the right-hand side of (4.6).

Given that \mathscr{G} is convex in μ and H_F is a convex weak^{*} compact subset of $\mathscr{B}(\mathbb{S}^{d-1})$, we conclude that $\mathscr{G}(\mu)$ achieves its maximum on H_F at a point of $ext(H_F)$. Hence there exists

a measure $\mu^* \in \text{ext}(H_F)$ such that $\mathscr{G}(\mu^*) = \sup_{\mu \in H_F} \mathscr{G}(\mu)$. We then find that

$$\begin{split} \mathscr{I}_F(\mathbb{S}^{d-1}) &= \inf_{\mu \in \mathbb{P}(\mathbb{S}^{d-1})} I_F(\mu) = I_F(\nu) = \mathscr{F}(\nu) - \mathscr{G}(\nu) = \mathscr{F}(\mu^*) - \mathscr{G}(\nu) \\ &\geq \mathscr{F}(\mu^*) - \mathscr{G}(\mu^*) = I_F(\mu^*) \geq \mathscr{I}_F(\mathbb{S}^{d-1}), \end{split}$$

i.e. $I_F(\mu^*) = M$ and μ^* is also a minimizer of I_F .

Since $\mu^* \in ext(H_F)$, we can apply Karr's theorem (Theorem 4.1.2) to finish our proof.

4.2 Minimizers of Energies with Analytic Kernels

It follows from Lemma 3.1.11 that if *K* is not positive definite (modulo a constant) on Ω , then the support of a minimizer μ of I_K must be a proper subset of Ω . When Ω is a real-analytic manifold and *K* is real-analytic, we can make a stronger claim: the support of any minimizer must have empty interior. For simplicity, we will prove this for rotationally invariant kernels *F* on the sphere \mathbb{S}^{d-1} , though the generalization is clear.

Theorem 4.2.1. Assume that F is a real-analytic function on [-1,1], such that σ is not a minimizer of I_F , i.e. F is not (up to an additive constant) positive definite on \mathbb{S}^{d-1} . Let μ be a minimizer of I_F , then $(\operatorname{supp}(\mu))^\circ = \emptyset$. Moreover, if d = 2, then $\operatorname{supp}(\mu)$ must be discrete.

In order to prove this theorem, we need the following lemma (see, e.g. [KP02, MT]):

Lemma 4.2.2 (Principle of Analytic Continuation). Let M be a real-analytic manifold, and $f: M \to \mathbb{R}$ be real-analytic. If f is constant on an open set of M, then f is constant on all of M. Moreover, if the manifold is one-dimensional, then it suffices that f is constant on some $U \subseteq M$ that has an accumulation point in M.

Proof of Theorem 4.2.1. Suppose, indirectly, that $(\operatorname{supp}(\mu))^{\circ} \neq \emptyset$. By Theorem 3.1.7, we

know that the potential

$$U_F^{\mu}(x) = \int_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) d\mu(y)$$

is constant on supp(μ). Since $F(\langle x, y \rangle)$ is real-analytic on $\mathbb{S}^{d-1} \times \mathbb{S}^{d-1}$, $U_F^{\mu}(x)$ is realanalytic on \mathbb{S}^{d-1} . Since U_F^{μ} is real-analytic and constant on an open set in \mathbb{S}^{d-1} , it is constant on all of \mathbb{S}^{d-1} , by Lemma 4.2.2. In addition, $U_F^{\sigma}(x) = I_F(\sigma)$ is constant on \mathbb{S}^{d-1} due to rotational invariance. We then obtain

$$\begin{split} I_F(\mu) &= \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) d\mu(y) d\mu(x) = \int\limits_{\mathbb{S}^{d-1}} U_F^{\mu}(x) d\mu(x) = \int\limits_{\mathbb{S}^{d-1}} U_F^{\mu}(x) d\sigma(x) \\ &= \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) d\mu(y) d\sigma(x) = \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) d\sigma(x) d\mu(y) \\ &= \int\limits_{\mathbb{S}^{d-1}} U_F^{\sigma}(x) d\mu(y) = I_F(\sigma). \end{split}$$

This is clearly a contradiction, as by the assumption, I_F is not minimized by σ . Our first claim then follows.

For \mathbb{S}^1 , Lemma 4.2.2 tells us that if U_F^{μ} is constant on a set $\{z_1, z_2, ...\} \subset \mathbb{S}^1$ with an accumulation point, U_F^{μ} is constant on \mathbb{S}^1 . The proof of our second claim then follows as above.

If \mathbb{S}^{d-1} is replaced with one of the projective spaces \mathbb{FP}^{d-1} , we can derive a similar result.

In the spirit of Theorem 4.2.1, as well as Corollary 3.4.2, it may be tempting to conjecture that if *F* (not necessarily analytic) is not positive definite on \mathbb{S}^{d-1} (up to constant), i.e $I_F(\mu)$ is *not* minimized by σ , then the support of any minimizer of I_F must have empty interior. However, this is not true, as the following simple example shows.

Example 4.2.3. Assume that $F \in C([-1,1])$ is constant near t = 1 and strictly decreasing

otherwise, i.e. it satisfies for some fixed $\gamma \in (0, 1)$,

$$F(1) = F(t') = \min_{t \in [-1,1]} F(t)$$
 for any $t' \in [1 - \gamma, 1]$,

and F(t') > F(1) for all $t' \in [-1, 1 - \gamma)$. It is then evident that for any $z \in \mathbb{S}^{d-1}$

$$\min_{\boldsymbol{\mu}\in\mathbb{P}(\mathbb{S}^{d-1})}I_F(\boldsymbol{\mu})=I_F(\boldsymbol{\delta}_z)=F(1),$$

and $I_F(\sigma) > I_F(\delta_z)$, *i.e.* σ *is not a minimizer of* I_F . Let $C(z,h) = \{x \in \mathbb{S}^{d-1} : \langle x, z \rangle > h\}$ denote the spherical cap of "height" h centered at $z \in S^{d-1}$. Let v be the normalized uniform measure on C(z,h), *i.e.*

$$d\mathbf{v}(x) = \frac{\mathbb{1}_{C(z,h)}(x)}{\boldsymbol{\sigma}(C(z,h))} d\boldsymbol{\sigma}(x),$$

with $h = 1 - \frac{\gamma}{4}$. Then for each $x, y \in C(z, h)$, we have $\langle x, y \rangle > 1 - \gamma$, and hence

$$I_F(\mathbf{v}) = I_F(\boldsymbol{\delta}_z) = F(1),$$

i.e. v is also a minimizer of I_F , but its support has non-empty interior.

Applications to Polynomial Energies

We observe that the results of Sections 4.1 and 4.2 apply if F is a polynomial. Indeed, Theorem 4.2.1 is applicable since polynomials are analytic, while the conditions of Theorem 4.1.3 hold because the Gegenbauer expansion has only finitely many terms. We summarize these statements in the following corollary. As in the preceding sections, this can be generalized to the other two-point homogeneous spaces. **Corollary 4.2.4.** Assume that F is a polynomial whose Gegenbauer expansion is

$$F(t) = \sum_{n=0}^{m} a_n C_n^{\lambda}(t).$$

1. There exists a discrete minimizer $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ with

$$|\operatorname{supp}(\mu)| \le 1 + \sum_{\{n: a_n > 0, 1 \le n \le m\}} \dim(\mathscr{H}_n^d).$$

2. If, moreover, σ is not a minimizer of I_F over $\mathbb{P}(\mathbb{S}^{d-1})$, i.e. there exists $n \ge 1$ such that $a_n < 0$, then the support of any minimizer of I_F has empty interior. For \mathbb{S}^1 , the support is finite.

We observe that when $a_n \ge 0$ for n = 1, ..., m, i.e. F is a polynomial that is positive definite on \mathbb{S}^{d-1} (up to constant), the statement of Theorem 4.1.3 (and hence also part (1) of the above corollary) is known. In this case, the discrete minimizers $\mu = \sum w_{z_i} \delta_{z_i}$ are exactly *weighted spherical m-designs*, see Section 2.6.

Existence of weighted *m*-designs of cardinality $\sum_{n=0}^{m} \dim(\mathscr{H}_{n}^{d})$ has been shown in [Tch57, Rog62]. A certain well-known generalization of this fact can also be easily deduced from part (1) of Corollary 4.2.4. Let $\mathscr{N} \subset \mathbb{N}_{0}$ with $0 \in \mathscr{N}$. Call a set $\{z_{i}\}_{i=1}^{N} \subset \mathbb{S}^{d-1}$ with positive weights $\omega_{z_{i}}$ a weighted \mathscr{N} -design if for every $m \in \mathscr{N}$ and every spherical harmonic $Y \in \mathscr{H}_{m}^{d}$ one has

$$\sum_{i=1}^{N} \omega_{z_i} Y(z_i) = \int_{\mathbb{S}^{d-1}} Y(x) d\sigma(x).$$

When $\mathscr{N} = \{0, 1, ..., m\}$, this definition coincides with the definition of a weighted *m*design. Such objects arise naturally for some configurations. For example, the 600-cell, one of the six 4-dimensional convex regular polytopes with vertices which form a 120point subset of \mathbb{S}^3 , yields an exact cubature formula for spherical harmonics of degrees up to 19, *excluding* degree 12 (see, e.g., [CK07, Section 7]). In other words, it is an \mathscr{N} -design for $\mathscr{N} = \{0, 1, ..., 11\} \cup \{13, ..., 19\}$. By taking $a_n > 0$ only for $n \in \mathscr{N}$ and applying part (1) of Corollary 4.2.4, one easily concludes existence of weighted \mathscr{N} -designs on the sphere \mathbb{S}^{d-1} of cardinality at most $\sum_{n \in \mathscr{N}} \dim(\mathscr{H}_n^d)$. This statement is encompassed by more general results [Tch57, Rog62].

Theorem 4.1.3 and part (1) of Corollary 4.2.4 vastly generalize these well-known statements, essentially showing that the addition of any number of negative definite terms does not destroy the statement: discrete minimizers with the same cardinality still exist.

4.3 Linear Programming and Optimality of Tight Designs

Though results such as Theorem 4.0.3 are useful in showing that minimizers of some energy must be discrete, and results along the lines of Theorem 4.1.3 tell us that discrete minimizers of a certain cardinality must exist, it would be ideal to determine what those minimizers are. The main goal of this section is to show that for those dimensions d and values of M for which tight designs exist, these measures minimize the energy I_F for a large class of potentials F. We will use linear programming bounds, discussed below, to this end.

The main result of this section is the following:

Theorem 4.3.1. Let F be absolutely monotonic of degree M, with $F^{(M+1)}(t) \le 0$ for $t \in (-1,1)$. Then for a tight M-design C, the measure

$$\mu_{\mathscr{C}} := rac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} \delta_x$$

is a minimizer of

$$I_F(\mu) = \int_{\Phi} \int_{\Phi} F(\tau(x,y)) d\mu(x) d\mu(y)$$

over $\mathbb{P}(\Phi)$.
Linear Programming

In the 1970s, Delsarte developed a method to bound codes on finite fields that yielded an upper bound on the kissing numbers and upper bounds for the size of spherical codes given minimal distance as solutions to linear programs [Del73]. This linear programming method provides bounds for optima in various optimization problems, and its use is often aided by computational tools, where a problem is approximated by a finite-dimensional or discretized counterpart, then solved with a computer. Despite being a relatively simple method, it often provides optimal bounds, such as those in Theorem 2.6.6. In particular, the linear programming method has been used to find the kissing number in dimensions 8 and 24 by Odlyzko and Sloane [OS79], and independently by Levenshtein [Lev79], as well as in dimension 4 by Musin [Mus08]. This technique applies to all the compact twopoint homogeneous spaces Φ described in Section 2.4. Our application of the method can be summed up in the following lemma, which is a measure-theoretic counterpart of the linear programming bound of Yudin [Yud92]. In what follows, we will use the notion from Section 2.4, i.e. C_n is the appropriate normalized Jacobi polynomial of degree *n*, and \hat{h}_n is the corresponding coefficient in the Jacobi expansion of *h*.

Lemma 4.3.2. Let $h \in C([-1,1])$ be positive-definite modulo a constant, i.e.

$$h(t) = \sum_{n=0}^{\infty} \widehat{h}_n C_n(t)$$

with $\hat{h}_n \ge 0$ for all $n \ge 1$.

1. Assume that $h(t) \leq F(t)$ for all $t \in [-1, 1]$. Then for any $\mu \in \mathbb{P}(\Phi)$,

$$I_F(\mu) \geq \widehat{h}_0 = I_h(\eta).$$

2. Assume further that h is a polynomial of degree k and that there exists a k-design

 $\mathscr{C} \subset \Phi$ such that h(t) = F(t) for each $t \in \mathscr{A}(\mathscr{C})$. Then for any $\mu \in \mathbb{P}(\Phi)$,

$$I_F(\mu) \ge I_F\Big(rac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} \delta_x\Big),$$

i.e. I_F is minimized by the uniform distribution on \mathcal{C} .

Proof. For the first part observe that

$$I_F(\mu) \geq I_h(\mu) \geq I_h(\eta) = h_0,$$

where the first inequality follows from the fact that $F \ge h$, while the second one is due to Proposition 3.4.1, since *h* is positive definite modulo a constant.

For the second part, one can continue as follows

$$I_h(\boldsymbol{\eta}) = I_h\Big(\frac{1}{|\mathscr{C}|}\sum_{x\in\mathscr{C}}\boldsymbol{\delta}_x\Big) = I_F\Big(\frac{1}{|\mathscr{C}|}\sum_{x\in\mathscr{C}}\boldsymbol{\delta}_x\Big).$$

The first equality follows from the fact that \mathscr{C} is a *k*-design, and the second from the fact that *F* and *h* coincide on the set $\mathscr{A}(\mathscr{C})$. Together with part (1) this proves the statement in part (2).

This lemma provides insights in two different ways for how the linear programming method can be applied. If a candidate \mathscr{C} is available, one can apply part (2) of Lemma 4.3.2 by constructing a polynomial $h \leq F$ as a Hermite interpolant of the function F at the points of $\mathscr{A}(\mathscr{C})$. This reasoning, which lies behind the proof of Theorems 4.3.1 and 6.2.3, explains the appearance of tight designs: indeed, the number of elements in the set of interpolation points (i.e. distinct distances between the points of \mathscr{C}) determines the degree of the interpolant h – hence one wants a design of high strength, but low degree.

The same reasoning as above applies to the emergence of sharp designs as universally optimal sets in [CK07], and it also explains why this slightly weaker notion does not suffice for our purposes: since we are working with general measures rather than point sets with

fixed cardinality, we cannot avoid interpolating at the point t = 1, which requires a design of higher strength. The main technical difficulty in this setting is proving positive definiteness of the Hermite interpolating polynomial *h*. We take this approach to Theorem 4.3.1 and carry out the technicalities in the following subsections.

If a suitable candidate is not available, one can still rely on part (1) of Lemma 4.3.2 and attempt to optimize the value of the energy $I_h(\eta)$ over auxiliary polynomials h (i.e. polynomials h satisfying the conditions of Lemma 4.3.2), obtaining a lower bound for the energy over all probability measures. If the degree of an auxiliary function h is bounded by D, we have D+1 non-negative variables \hat{h}_i , $0 \le i \le D$, and infinitely many linear constraints $h(t) \le F(t)$ for all $t \in [-1, 1]$. In order to get the best possible lower bound, we need to maximize \hat{h}_0 given these linear conditions. This particular method will be used in Sections 7.2.

This problem is, generally, intractable as a linear optimization problem. However, when F is a polynomial, the condition $F(t) - h(t) \ge 0$ for all $t \in [-1, 1]$ may be represented as a finite-size positive semi-definite constraint on the coefficients \hat{h}_i . In particular, the polynomial inequality may be rewritten as a sum-of-squares optimization problem (see, for instance, [Nes00]) and thus solved as a semi-definite program.

Properties of Orthogonal Polynomials

Recall that, for fixed Φ , we write simply $C_n(t) = C_n^{(\alpha,\beta)}(t)$ and have $C_n(1) = 1$. In some of the arguments below we will use instead the monic Jacobi polynomials, which we will denote as $Q_n(t) = Q_n^{(\alpha,\beta)}(t)$.

We now collect several results about orthogonal polynomials relevant to the proof of Theorem 4.3.1 and which are presented and covered in greater detail in [CK07]. Fix a space Φ , and let α and β be the corresponding parameters of the associated Jacobi polynomials. According to Proposition 3.4.1, positive definiteness on Φ is equivalent to the positivity of coefficients in the monic Jacobi expansion, i.e. the expansion with respect to $Q_n^{(\alpha,\beta)}$.

It will be useful to consider *adjacent* Jacobi polynomials, defined as one of the three sequences $Q_n^{k,l} = Q_n^{(\alpha+k,\beta+l)}$ with $k,l \in \{0,1\}, k+l > 0$. Specifically, we will need the following corollary which comes out of representing $Q_n^{1,0}$ through $Q_n^{0,0}$ [Lev92, equation (3.4)]:

Proposition 4.3.3. Adjacent Jacobi polynomials $Q_n^{1,0}$ are positive definite on Φ .

On the other hand, adjacent polynomials $Q_n^{1,1}$, defined as orthogonal with respect to the measure $(1-t^2)dv^{(\alpha,\beta)}$, are not positive definite on Φ . The following property, a special case of the strengthened Krein condition [Lev98, Lemma 3.22], can serve as a substitute.

Lemma 4.3.4. $(t+1)Q_n^{1,1}(t)$ are positive definite on Φ for $n \ge 0$.

Proof. For all $n \in \mathbb{N}_0$, $(t+1)Q_n^{1,1}$ is orthogonal to all polynomials of degree less than n with respect to the measure $(1-t)dv^{(\alpha,\beta)} = c_{\alpha,\beta}dv^{(\alpha+1,\beta)}$, so it can be expressed through the orthogonal polynomials corresponding to $dv^{(\alpha+1,\beta)}$ as

$$(t+1)Q_n^{1,1}(t) = Q_{n+1}^{1,0}(t) + bQ_n^{1,0}(t),$$

for some constant *b*. Since all the roots of $Q_n^{1,0}$ lie in (-1,1), $\operatorname{sgn} Q_n^{1,0}(-1) = (-1)^n$. Substituting t = -1 in the last equation gives $Q_{n+1}^{1,0}(-1) + bQ_n^{1,0}(-1) = 0$, and so $b \ge 0$. By Proposition 4.3.3, each $Q_n^{1,0}(t)$ is positive definite, and thus $(t+1)Q_n^{1,1}(t)$ is also positive definite.

Lastly, we will need the strict positive-definiteness of polynomials annihilated by subsets of roots of $Q_n + \gamma Q_{n-1}$, which is provided by the following result.

Proposition 4.3.5 ([CK07, Theorem 3.1]). Let $\mu \in \mathscr{B}(\mathbb{R})$ such that all polynomials are integrable with respect to μ and for all polynomials p, $\int_{\mathbb{R}} (p(t))^2 d\mu(t) > 0$ if p is not identically zero. Set $p_0(t), p_1(t), ...$ to be a sequence of monic orthogonal polynomials for μ such that deg $(p_k) = k$ for all $k \in \mathbb{N}$. If $t_1 < ... < t_n$ are the zeros of $p_n + \gamma p_{n-1}$ for some fixed γ , then the polynomials

$$\prod_{i=1}^k (t-t_i), \qquad 1 \le k < n,$$

can be represented as a linear combination of $p_0(t), p_1(t), \ldots, p_n(t)$ with positive coefficients.

Hermite Interpolation

Let $F \in C^B[a,b]$, for some $B \in \mathbb{N}_0$, and let a collection $t_1 < \ldots < t_m \subset [a,b]$, as well as positive integers k_1, \ldots, k_m be given with

$$\max\{k_1,\ldots,k_m\} \le B+1.$$

There exists a polynomial *p* of degree less than $D = \sum_{i=1}^{m} k_i$, such that for $1 \le i \le m$ and $0 \le k < k_i$,

$$p^{(k)}(t_i) = F^{(k)}(t_i).$$

Such a p is called the *Hermite interpolating polynomial* that agrees with F to order k_i at each t_i (we will simply call it the Hermite interpolating polynomial for short); it always exists and is unique because the linear map that takes a polynomial p of degree less than D to

$$(p(t_1), p'(t_1), \dots, p^{(k_1-1)}(t_1), p(t_2), p'(t_2), \dots, p^{k_m-1}(t_m))$$

is bijective.

It is convenient to organize both the collection $t_1 < ... < t_m$ and the orders of derivatives $k_1, ..., k_m$ into a polynomial g(t). Given such a polynomial

$$g(t) = \prod_{i=1}^{m} (t - t_i)^{k_i}$$
(4.8)

where $D = \deg(g) \ge 1$, we write H[F,g] for the interpolating polynomial of degree less than *D* that agrees with *F* at each t_i to the order k_i .

We can now state an important remainder formula for Hermite interpolation [Dav63, Theorem 3.5.1].

Lemma 4.3.6. Suppose that $F \in C^B[a,b] \cap C^D(a,b)$ and g is as in (4.8). Then for each $t \in [a,b]$, there exists $\xi \in (a,b)$ such that $\min(t,t_1,...,t_m) < \xi < \max(t,t_1,...,t_m)$ and

$$F(t) - H[F,g](t) = \frac{F^{(D)}(\xi)}{D!}g(t).$$
(4.9)

Similarly, we let

$$\mathcal{Q}[F,g](t) = \frac{F(t) - H[F,g](t)}{g(t)},$$

be the *divided difference* associated with the polynomial g. Since, for $1 \le i \le m$,

$$F(t) - H[F,g](t) = O((t - t_i)^{k_i})$$

as $t \to t_i$, it follows that Q[F,g] extends to a continuous function at the roots of g. If we assume that $F \in C^D(a,b)$, then the continuity of $F^{(D)}$ and Lemma 4.3.6 implies that for all $t \in [a,b]$, there there exists some $\xi \in (a,b)$ such that $\min(t,t_1,...,t_m) < \xi < \max(t,t_1,...,t_m)$ and

$$Q[F,g](t) = \frac{F^{(D)}(\xi)}{D!}.$$
(4.10)

We can enumerate the roots of g with multiplicities in increasing order, and denote these by $s_j, 1 \le j \le D$, where $s_j \le s_{j+1}$. Let g_n^* be the polynomial annihilated on the first nelements of the sequence s_1, \ldots, s_D :

$$g_n^*(t) = \prod_{j=1}^n (t - s_j), \qquad 1 \le n \le D.$$

The usual assignment of the empty product applies here: $g_0^*(t) = 1$.

By the Newton's formula [DL93, Chapter 4.6–7], the Hermite interpolating polynomial H[F,g] can be represented as

$$H[F,g](t) = F(s_1) + \sum_{j=1}^{D-1} g_j^*(t) Q[F,g_j^*](s_{j+1}).$$
(4.11)

It will be convenient to introduce notation for this number of nonnegative derivatives of a function.

Definition 4.3.7. Let $F \in C^{D}(a,b)$. We say that F is absolutely monotonic of degree D if $F^{(k)}(t) \ge 0$ for $0 \le k \le D$ and $t \in (a,b)$. If these derivatives are positive, we say that F is strictly absolutely monotonic of degree D.

The benefit of these nonnegative derivatives lies in the fact that the Hermite interpolant of an absolutely monotonic function, F, of degree D with negative (D+1)st derivative will stay below F, as shown in the following observation.

Lemma 4.3.8. Let $F : [-1,1] \to \mathbb{R}$ be absolutely monotonic of degree D, and $F^{(D+1)}(t) \le 0$ for all $t \in (-1,1)$. If the roots of a polynomial g of the form (4.8) (and thus of degree D) are contained in [-1,1], and, in addition, $g(t) \le 0$ for $t \in [-1,1]$, then,

$$F(t) \ge H[F,g](t), \quad t \in [-1,1].$$

Proof. According to Lemma 4.3.6, for all $t \in [-1,1]$ there exists $\xi \in (-1,1)$ such that $\min(t,t_1) < \xi < \max(t,t_m)$, where the roots of *g* are $t_1 < \ldots < t_m$, and

$$F(t) - H[F,g](t) = \frac{F^{(D+1)}(\xi)}{(D+1)!}g(t).$$

The expression on the right is nonnegative, so the conclusion of the lemma follows. \Box

Optimality of Tight Designs

As above, Φ is a compact, connected two-point homogeneous space and $Q_0, Q_1, Q_2, ...$ are the corresponding monic orthogonal polynomials. Recall that Q_n are orthogonal with respect to the measure $dv^{(\alpha,\beta)} = \frac{1}{\gamma_{\alpha,\beta}}(1-t)^{\alpha}(1+t)^{\beta}dt$, where the parameters α , β are chosen as in Section 2.4.

In what follows, we give a proof of the Theorem 4.3.1, splitting it into two separate cases, depending on whether the code \mathscr{C} contains two points separated by the diameter of Φ , or, equivalently, depending on the parity of the strength *M* of \mathscr{C} .

Proposition 4.3.9. Theorem 4.3.1 holds when M = 2m, $m \ge 1$.

Proof. Let $-1 \le t_1 < \ldots < t_m < t_{m+1} = 1$ be the values within $\mathscr{A}(\mathscr{C})$ and define

$$g_k(t) := \prod_{i=1}^k (t - t_i), \qquad 1 \le k \le m + 1.$$

and

$$g(t) := g_m(t) g_{m+1}(t) = (t-1)g_m^2(t)$$
(4.12)

To prove the statement of the proposition, we verify the following chain of inequalities, satisfied for arbitrary $\mu \in \mathbb{P}(\Phi)$, similar to the proof of Lemma 4.3.2,

$$I_{F}(\mu) \ge I_{H[F,g]}(\mu) \ge I_{H[F,g]}(\eta) = I_{H[F,g]}(\mu_{\mathscr{C}}) = I_{F}(\mu_{\mathscr{C}}).$$
(4.13)

The equality $I_{H[F,g]}(\eta) = I_{H[F,g]}(\mu_{\mathscr{C}})$ follows since \mathscr{C} is a design of strength $2m \ge \deg H[F,g]$. The last equality holds since H[F,g] agrees with F at the cosines of distances occurring in \mathscr{C} . Since $g(t) \le 0$ for $t \in [-1,1]$, Lemma 4.3.8 implies that $F(t) \ge H[F,g](t)$ for $t \in [-1,1]$, which gives the first inequality. It remains to show the second inequality: it will follow from the positive definiteness of H[F,g], which we will now demonstrate.

For any n < m, the degree of $g_{m+1}(t)Q_n(t)$ is at most 2m. As \mathscr{C} is a 2m-design, for every fixed $y \in \mathscr{C}$ we have that

$$\int_{-1}^{1} g_{m+1}(t)Q_n(t)d\mathbf{v}^{(\alpha,\beta)}(t) = \int_{\Phi} g_{m+1}(\tau(x,y))Q_n(\tau(x,y))d\eta(x)$$
$$= \frac{1}{|\mathscr{C}|}\sum_{x\in\mathscr{C}} g_{m+1}(\tau(x,y))Q_n(\tau(x,y))$$
$$= \frac{1}{|\mathscr{C}|}\sum_{i=1}^{m+1} c_i g_{m+1}(t_i)Q_n(t_i) = 0$$

since, by construction, g_{m+1} is annihilated on all the t_i . The constants c_i are given by

$$c_i = |\{x \in \mathscr{C} \mid \tau(x, y) = t_i\}|.$$

Since both g_{m+1} and Q_{m+1} are monic and g_{m+1} is orthogonal to each Q_n for n < m, we conclude that

$$g_{m+1}(t) = Q_{m+1}(t) + \gamma Q_m(t),$$

for some $\gamma \in \mathbb{R}$. By Proposition 4.3.5, subproducts of factors of g_{m+1} , which we denote by g_k , $1 \le k \le m$, can be expressed as linear combinations of Q_n with positive coefficients, and therefore are positive definite.

According to Newton's formula (4.11), the Hermite interpolant of F can be expressed as the sum of partial products of factors of g multiplied by the appropriate divided difference. We will use this formula to show that H[F,g] is positive definite. Indeed, (4.11) gives

$$H[F,g](t) = F(t_1) + \sum_{k=1}^{m} \left(g_k(t)g_{k-1}(t)Q[F,g_kg_{k-1}](t_k) + g_k^2(t)Q[F,g_k^2](t_{k+1}) \right), \quad (4.14)$$

where, as usual, $g_0 = 1$. Observe that the divided differences in the last equation are nonnegative due to (4.10), as the function F is absolutely monotonic of degree 2m. Since we have shown that each g_k is positive definite, Schur's theorem implies that so are g_k^2 and $g_k g_{k+1}$, and it follows that H[F,g] is positive definite as well.

Before turning to the proof of Theorem 4.3.1 for tight designs of odd strength, recall

the definition of the adjacent polynomials $Q_n^{1,1} = Q_n^{(\alpha+1,\beta+1)}$ for $n \ge 0$. They are monic and orthogonal with respect to the measure

$$d\mathbf{v}^{(\alpha+1,\beta+1)}(t) = \frac{1}{\gamma_{\alpha+1,\beta+1}}(1-t)^{\alpha+1}(1+t)^{\beta+1}dt = \frac{\gamma_{\alpha,\beta}}{\gamma_{\alpha+1,\beta+1}}(1-t^2)d\mathbf{v}^{(\alpha,\beta)}(t).$$

Proposition 4.3.10. *Theorem 4.3.1 holds when* M = 2m - 1, $m \ge 1$.

Proof. Suppose that $\mathscr{C} \subset \Phi$ is a tight (2m-1)-design. As discussed in Section 2.6 tight designs of odd strength necessarily contain antipodal points, i.e. there exist $x, y \in \mathscr{C}$ such that $\vartheta(x, y) = \pi$ and thus $-1 \in \mathscr{A}(\mathscr{C})$. Let $-1 = t_1 < \ldots < t_m < t_{m+1} = 1$ be the values of $\mathscr{A}(\mathscr{C})$, and set

$$w(t) = \prod_{j=2}^{m} (t - t_j)$$

and

$$g(t) = w^2(t)(t^2 - 1).$$
 (4.15)

As in the proof of Proposition 4.3.9, we need to verify the inequalities (4.13). Applying Lemma 4.3.8 to H[F,g] gives the first inequality; it remains to show positive-definiteness of H[F,g].

For n < m-1, the degree of $(1-t^2)w(t)Q_n^{1,1}(t)$ is at most 2m-1, so for any $y \in \mathscr{C}$ there holds

$$\begin{split} \frac{\gamma_{\alpha+1,\beta+1}}{\gamma_{\alpha,\beta}} &\int_{-1}^{1} w(t) Q_n^{1,1}(t) d\nu^{(\alpha+1,\beta+1)} = \int_{\Phi} (1 - \tau^2(x,y)) w(\tau(x,y)) Q_n^{1,1}(\tau(x,y)) d\eta(x) \\ &= \frac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} (1 - \tau^2(x,y)) w(\tau(x,y)) Q_n^{1,1}(\tau(x,y)) \\ &= \frac{1}{|\mathscr{C}|} \sum_{j=1}^{m+1} c_j (1 - t_j^2) w(t_j) Q_n^{1,1}(t_j) = 0 \end{split}$$

as $(1-t^2)w(t)$ is annihilated on the cosines of distances from \mathcal{C} . Because w(t) is a degree

(m-1) monic polynomial, the above implies $w(t) = Q_{m-1}^{1,1}(t)$. By Proposition 4.3.5, this also means that for $2 \le k \le m-1$, polynomials $\prod_{j=2}^{k}(t-t_j)$ are linear combinations of $Q_n^{1,1}$ with positive coefficients. Since the cone of functions with nonnegative Jacobi coefficients with respect to $Q_n^{1,1}$ is closed under multiplication, for $2 \le k \le m$, polynomials $\prod_{j=2}^{k}(t-t_j)^2$ and $(t-t_k)\prod_{j=2}^{k-1}(t-t_j)^2$ also have nonnegative Jacobi coefficients in $Q_n^{1,1}$. Due to Lemma 4.3.4, since $t - t_1 = t + 1$, we obtain that

$$a_k(t) := (t - t_1)(t - t_k) \prod_{j=2}^{k-1} (t - t_j)^2$$
 and $b_k(t) := (t - t_1) \prod_{j=2}^k (t - t_j)^2$, (4.16)

are linear combinations of $Q_n^{(\alpha,\beta)}$ with positive coefficients; that is, they are positive definite on Φ for $1 \le k \le m$.

We conclude by the same observations as in the proof of Proposition 4.3.9; in particular, the positive definiteness of the Hermite interpolant H[F,g] follows from the representation

$$H[F,g](t) = F(t_1) + b_1(t)Q[F,b_1](t_2) + \sum_{k=2}^{m} \left(a_k(t)Q[F,a_k](t_k) + b_k(t)Q[F,b_k](t_{k+1}) \right),$$
(4.17)

combined with the absolute monotonicity of *F* to degree 2m - 1, which implies positivity of the divided differences *Q*.

Example 4.3.11. As an example of another application of Theorem 4.3.1, consider the case that $F(t) = a + bt + ct^2 + dt^3$ is given as the potential function. In this case, some elementary considerations show that if

- (*i*) $d \le 0$,
- (*ii*) $c \geq -3d$,
- $(iii) \ c^2 3bd \ge 0,$
- (*iv*) $-c \sqrt{c^2 3bd} \le 3d$, and,

$$(v) -c + \sqrt{c^2 - 3bd} \ge 3d,$$

then F is absolutely monotonic of degree 2 up to a constant. Hence, the energy of any potential function of the above form on a two-point homogeneous space Φ is minimized by a tight 2-design. Note that the constant term can be ignored, so it suffices to only consider the sign of derivatives. In particular, if b > 0 and d becomes sufficiently small in magnitude, the above inequalities will hold.

As discussed earlier this chapter, we would like to determine conditions for which all minimizers of an energy are discrete, as well as characterize the minimizers whenever possible. This is possible when F is strictly absolutely monotonic of degree M and tight M-designs exist, as then the tight designs are exactly the minimizers of I_F .

Theorem 4.3.12. Suppose that a tight *M*-design \mathcal{C} minimizes the *F*-energy integral, for *F* strictly absolutely monotonic of degree *M* and such that $F^{(M+1)}(t) < 0$ for $t \in (-1,1)$. Then any minimizer of I_F must be a tight *M*-design.

Proof. The argument developed to prove Theorem 4.3.1 may be described concisely through the following string of inequalities

$$I_F(\mu) \ge I_{H[F,g]}(\mu) \ge I_{H[F,g]}(\eta) = I_{H[F,g]}(\mu_{\mathscr{C}}) = I_F(\mu_{\mathscr{C}}),$$

where g is of the form (4.12) or (4.15), as is appropriate. In order for $I_F(\mu) = I_F(\mu_{\mathscr{C}})$ to hold, the inequalities must be equalities. The first inequality can only be sharp in the case that $\mathscr{A}(\operatorname{supp}(\mu)) \subseteq \mathscr{A}(\mathscr{C})$. This follows from the fact that H[F,g](t) < F(t) for all $t \notin \mathscr{A}(\mathscr{C})$ by the remainder formula from Lemma 4.3.8. In particular, this shows that $|\operatorname{supp}(\mu)|$ is finite.

We now wish to show that the second inequality is sharp only when μ is a weighted design of at least the strength of the minimizing tight design. We first note that since *F* is strictly absolutely monotonic, the divided differences appearing in (4.14) or (4.17) are

all positive due to (4.10). Thus, H[F,g] (possibly modulo a constant) is a linear combination, with positive coefficients, of positive definite polynomials of degrees 0, 1, ..., M, so $H[F,g] = a_0 + \sum_{j=1}^{M} a_j C_j$, where $a_j > 0$ for j > 0. We see that μ must indeed be a weighted *M*-design, and due to Theorem 2.6.6 and the fact that \mathscr{C} is a tight *M*-design, $|\operatorname{supp}(\mu)| \ge |\mathscr{C}|$. Lemmas 2.6.3 and 2.6.5 then tell us that this is only possible if

$$|\mathscr{A}(\operatorname{supp}(\boldsymbol{\mu}))| \geq |\mathscr{A}(\mathscr{C})|.$$

Thus, the distance sets must be the same, and therefore so are the cardinalities of the sets, making $supp(\mu)$ a tight *M*-design. Since μ is a weighted *M*-design, it must be a tight *M*-design.

Causal Variational Principle

We now turn to another application of the linear programming method. Define the kernel

$$F(t) = F_{\tau}(t) := \max\{0, 2\tau^2 (1+t) (2-\tau^2 (1-t))\}.$$
(4.18)

for $\tau > 0$. The minimization problem for the energy

$$I_F(\mu) = \int_{\mathbb{S}^2} \int_{\mathbb{S}^2} F(\langle x, y \rangle) d\mu(x) d\mu(y)$$
(4.19)

is known as the *causal variational principle* on the sphere and is connected to relativistic quantum field theory. It is conjectured in [FS13] that there exist discrete minimizers for $\tau \ge 1$ and, moreover, that all the minimizers of (4.19) are discrete whenever $\tau > \sqrt{2}$. The background on this problem can be found in [FS13, BFSvdM19].

Here we confirm this conjecture for two values of $\tau > 0$, for which we can show that the cross-polytope (or orthoplex) and the icosahedron indeed minimize the energy, which was suggested by numerical experiments in [FS13]. The proofs are another application of the linear programming framework. In this instance, Hermite interpolation is unavailable to us as F is not differentiable on (-1, 1). However, as we are dealing with a single kernel, instead of a class of them as in the previous section, we need only construct the correct auxiliary function.

We address the cross-polytope first. When $\tau = \sqrt{2}$, we have

$$F_{\tau}(t) = \max\{0, 8t^2 + 8t\},\$$

and thus $F_{\tau}(0) = 0$. Setting the measure

$$\mathbf{v} = rac{1}{6} \sum_{i=1}^{3} \left(\delta_{e_i} + \delta_{-e_i}
ight),$$

where $\{e_1, e_2, e_3\}$ is an orthonormal basis of \mathbb{R}^3 , i.e. *v* is a measure whose mass is equally concentrated in the vertices of a cross-polytope, we have the following:

Proposition 4.3.13. The measure v is a minimizer for the energy $I_{F_{\sqrt{2}}}$ over $\mathbb{P}(\mathbb{S}^2)$.

Proof. The function

$$h(t) = 8t^2 + 8t.$$

is positive definite on \mathbb{S}^2 and clearly satisfies

$$h(t) \le F_{\sqrt{2}}(t)$$
 for all $t \in [-1, 1]$,

and

$$h(-1) = F_{\sqrt{2}}(-1) = 0, \quad h(0) = F_{\sqrt{2}}(0) = 0, \quad h(1) = F_{\sqrt{2}}(1) = 16,$$

so that *h* coincides with $F_{\sqrt{2}}$ on the set $\mathscr{A}(\operatorname{supp}(v))$.

We obtain that for any measure $\mu \in \mathbb{P}(\mathbb{S}^2)$,

$$I_{F_{\sqrt{2}}}(\mu) \geq I_h(\mu) \geq I_h(\sigma) = I_h(\nu) = I_{F_{\sqrt{2}}}(\nu),$$

where we have used the fact that $h(t) \leq F_{\sqrt{2}}(t)$ for $t \in [-1,1]$, so $I_{F_{\sqrt{2}}}(\mu) \geq I_h(\mu)$. Since h is positive definite, according to Proposition 3.4.1, we have that σ minimizes I_h , i.e. $I_h(\mu) \geq I_h(\sigma)$. We have also used that the cross-polytope is a 3-design and h is a quadratic polynomial, hence $I_h(\sigma) = I_h(\nu)$. Finally, $h(t) = F_{\sqrt{2}}(t)$ for $t \in \mathscr{A}(\operatorname{supp}(\nu)) = \{0, \pm 1\}$, hence $I_h(\nu) = I_{F_{\sqrt{2}}}(\nu)$.

We now focus on the case of the icosahedron. Here we set $\tau = \sqrt{\frac{2\sqrt{5}}{\sqrt{5}-1}}$ so that $F_{\tau}(\frac{1}{\sqrt{5}}) = 0$. Let $\mathscr{C} \subset \mathbb{S}^2$ be the vertices of a regular icosahedron and let

$$\mathbf{v} = \frac{1}{12} \sum_{x \in \mathscr{C}} \delta_x$$

be the uniform measure on the vertices of the icosahedron.

Proposition 4.3.14. The measure v is a minimizer for the energy, $I_{F_{\tau}}$, over $\mathbb{P}(\mathbb{S}^2)$ for $\tau = \sqrt{\frac{2\sqrt{5}}{\sqrt{5}-1}}$.

Proof. We shall need two facts about the icosahedron: namely that the set of inner products between elements of \mathscr{C} is $\mathscr{A}(\mathscr{C}) = \left\{ \pm 1, \pm 1/\sqrt{5} \right\}$, and \mathscr{C} is a 5-design. For simplicity let us consider the function $F(t) = \frac{F_{\tau}(t)}{F_{\tau}(1)}$ so that F(1) = 1 (which does not effect the minimizers).

We construct the following polynomial:

$$h(t) = \frac{5(5-\sqrt{5})}{32}t^4 + \frac{5}{8}t^3 + \frac{3\sqrt{5}-5}{16}t^2 - \frac{1}{8}t + \frac{1-\sqrt{5}}{32}$$
$$= \frac{5-\sqrt{5}}{28}C_4(t) + \frac{1}{4}C_3(t) + \frac{20+3\sqrt{5}}{84}C_2(t) + \frac{1}{4}C_1(t) + \frac{1}{12}C_0(t),$$

where C_k are the standard Legendre polynomials (i.e. the Gegenbauer polynomials $C_k^{\frac{1}{2}}$). We observe then that *h* is positive definite, and that $h(t) \le F(t)$ for $-1 \le t \le 1$, which follows from the factored form

$$h(t) = \frac{5}{32}(5 - \sqrt{5})(t+1)(t - \frac{1}{\sqrt{5}})(t + \frac{1}{\sqrt{5}}).$$

A glance at this formula gives $h \le F$ for $t \in [-1, \frac{1}{\sqrt{5}}]$, and the fact that F - h is a polynomial with roots

$$t = -1, \frac{1}{\sqrt{5}}, \frac{-1 \pm 4\sqrt{10 + 4\sqrt{5}}}{\sqrt{5}},$$

gives $h \le F$ for $t \in [\frac{1}{\sqrt{5}}, 1]$ which is a subset of the interval $[\frac{1}{\sqrt{5}}, \frac{-1+4\sqrt{10+4\sqrt{5}}}{\sqrt{5}}]$.

By construction, the function *h* (which was obtained by solving the linear equations h(t) = F(t) for $t = \pm 1, \pm 1/\sqrt{5}$, as well as $h'(-1/\sqrt{5}) = 0$) has a local maximum at $-\frac{1}{\sqrt{5}}$ and coincides with *F* on the set $\mathscr{A}(\mathscr{C}) = \{\pm 1, \pm 1\sqrt{5}\}$. The same argument as in the proof of Proposition 4.3.13 finally shows

$$I_F(\mathbf{v}) = \inf_{\boldsymbol{\mu} \in \mathbb{P}(\mathbb{S}^2)} I_F(\boldsymbol{\mu}),$$

i.e. the icosahedron minimizes the energy $I_{F_{\tau}}$ for $\tau^2 = \frac{2\sqrt{5}}{\sqrt{5}-1}$.

Chapter 5

Stolarsky-type Principles

In numerous areas of mathematics and other sciences, one is faced with the problem of distributing large finite sets of points in a set Ω as uniformly as possible. There exist various quantitative measures of uniformity of point distributions. It should be evident from our discussion of Riesz energies in Section 2.1 that energy is one. Another popular candidate is discrepancy. We will focus on the sphere in this chapter, but a more complete exposition on general Discrepancy Theory can be found in [DT97, Mat99].

Let $\omega_N = \{z_1, ..., z_N\} \subset \mathbb{S}^{d-1}$. For a given subset of the sphere, $A \subset \mathbb{S}^{d-1}$, the discrepancy of ω_N with respect to A is defined as

$$D(\boldsymbol{\omega}_N, A) = \frac{1}{N} \sum_{k=1}^N \mathbb{1}_A(z_k) - \boldsymbol{\sigma}(A), \qquad (5.1)$$

in other words, $D(\omega_N, A)$ indicates how well the Lebesgue measure of A is approximated by the counting measure $\frac{1}{N} \sum_{k=1}^{N} \delta_{z_k}$. To obtain good finite distributions ω_N , one usually evaluates and strives to minimize the supremum (extremal discrepancy) or average (e.g., L^2 discrepancy) of $|D(\omega_N, A)|$ over some rich and well-structured collection of sets A. Typical examples of such collections include spherical caps, slices, convex sets, etc. – the specific choice depends on the problem at hand.

It is known that in some cases these two ways of quantifying equidistribution are closely

connected. One of the first instances of such a connection was obtained in 1973 by Stolarsky [Sto73], who proved that minimizing the L^2 discrepancy with respect to spherical caps is equivalent to maximizing the pairwise sum of Euclidean distances, i.e. $E_{\mathscr{D}}(\omega_N)$. More precisely, he established the identity

$$c_d[D_{L^2,\text{cap}}(\omega_N)]^2 = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} ||x-y|| \, d\sigma(x) \, d\sigma(y) \, - \, \frac{1}{N^2} \sum_{i,j=1}^N ||z_i - z_j||,$$

which came to be known as the *Stolarsky Invariance Principle* (see section 5.1 for more details).

Recently, there has been a spike in interest in applications, analogues, and generalizations of the Stolarsky Invariance Principle in different settings: Brauchart and Dick studied it from the point of view of numerical integration on the sphere [BD13], Bilyk and Lacey connected it to tessellations of the sphere in [BL17], Basu, He, Owen, and Zhao used it in applications to genomics [HBZO19], Skriganov explored Stolarsky principles in general compact metrics space, proving an analogue of the result in projective spaces in [Skr17, Skr20], and Barg and Skriganov further explore this on the Hamming Cube in [Bar20, BS21]. In [BD19, BDM18, BMV], Bilyk, Dai, Vlasiuk, and the author explore a generalization of the Stolarsky Invariance Principle from the context of energy optimization. The results of these papers make up the bulk of this chapter.

We note that the basic strategy behind most versions of Stolarsky principle, at a very low level, is straightforward. Computing the L^2 discrepancy, one squares out the expression in (5.1), thus pairwise interactions between points of ω_N arise from cross terms of the form $\mathbb{1}_A(z_i) \cdot \mathbb{1}_A(z_j)$. When integrated over the test sets A in a given class, this yields the interaction potential $K(z_i, z_j)$, which is often represented as the volume of intersection of test sets "centered" at z_i and z_j , see e.g. (5.8), however, the details in some settings get rather technical. This approach is employed in Sections 5.1 and 5.2. A similar idea has been used by Torquato [Tor10] for "number variance", a quantity very similar to L^2 discrepancy. In Section 5.3 we go in the opposite direction and show that for any positive definite interaction potential one can construct an appropriate notion of discrepancy, so that a version of the Stolarsky principle holds.

In Section 5.1, we revisit the classical Stolarsky Invariance Principle. In Section 5.2, we observe that by replacing *all* spherical caps with *hemispheres* one obtains a variant of the Stolarsky Invariance Principle involving the *Geodesic Riesz* (-1)-*energy*. This allows one to easily characterize the finite point sets on \mathbb{S}^{d-1} which maximize the sum of geodesic distances in all dimensions $d \ge 2$. We then take this idea a step further and show that our analogue of the Stolarsky Invariance Principle holds for general probability measures, which points to a drastic difference between the geodesic and Euclidean Riesz (-1)-energies. We finish the section with a discussion of the Geodesic Riesz *s*-energies for s < 0.

In Section 5.3, we explore the connections between energy optimization and discrepancy on a more general level, showing that for any positive definite function F, one can define a natural notion of discrepancy so that an analogue of the Stolarsky Invariance Principle holds for general probability measures. In Section 5.4, we take this a step further, showing that a similar result holds on all compact metric spaces.

5.1 Stolarsky Invariance Principle

We consider "spherical caps" C(x,h) with center $x \in \mathbb{S}^{d-1}$ and "height" $h \in [-1,1]$, i.e.

$$C(x,h) = \{ z \in \mathbb{S}^{d-1} : \langle z, x \rangle > h \}.$$

We define the L^2 discrepancy of $\omega_N = \{z_1, ..., z_N\}$ with respect to spherical caps by

$$[D_{L^{2}, cap}(\omega_{N})]^{2} = \int_{-1}^{1} \int_{\mathbb{S}^{d-1}} \left| \frac{1}{N} \sum_{j=1}^{N} \mathbb{1}_{C(x,h)}(z_{j}) - \sigma(C(x,h)) \right|^{2} d\sigma(x) dh.$$
(5.2)

The following result was proved by Stolarsky in 1973 [Sto73]:

Theorem 5.1.1 (Stolarsky Invariance Principle). Let $\omega_N = \{z_1, z_2, ..., z_N\} \subset \mathbb{S}^{d-1}$. Then *the following relation holds:*

$$[D_{L^{2},\mathrm{cap}}(\omega_{N})]^{2} = C_{d-1} \left(\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \|x - y\| d\sigma(x) d\sigma(y) - \frac{1}{N^{2}} \sum_{i,j=1}^{N} \|z_{i} - z_{j}\| \right).$$
(5.3)

The constant C_{d-1} satisfies

$$C_{d-1} = \frac{1}{2} \int_{\mathbb{S}^{d-1}} |\langle p, z \rangle| d\sigma(z) = \frac{1}{d-1} \frac{A_{d-2}}{A_{d-1}} = \frac{Vol_{d-1}}{A_{d-1}}$$

$$= \frac{1}{d-1} \frac{\Gamma(d/2)}{\sqrt{\pi} \Gamma((d-1)/2)} \sim \frac{1}{\sqrt{2\pi(d-1)}} \text{ as } d \to \infty,$$
 (5.4)

where A_{d-1} is the surface area of \mathbb{S}^{d-1} , Vol_d is the volume of the unit ball in \mathbb{R}^d , and p is an arbitrary point on the sphere \mathbb{S}^{d-1} .

This theorem states that

- minimizing the L^2 spherical cap discrepancy of ω_N is equivalent to maximizing the sum of Euclidean distances between the points of ω_N .
- the L^2 spherical cap discrepancy can be realized as the difference between the continuous and discrete energies $I_F(\sigma) - E_F(\omega_N)$, with $F(\langle x, y \rangle) = ||x-y|| = \sqrt{2 - 2\langle x, y \rangle}$, or, equivalently, the error of numerical integration of the distance integral

$$I_{\mathscr{D}_1}(\boldsymbol{\sigma}) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \|x - y\| d\boldsymbol{\sigma}(x) d\boldsymbol{\sigma}(y)$$

by the cubature formula with knots at the points of ω_N .

It is well known [Bec84a, Bec84b] that the optimal order of the L^2 spherical cap discrepancy is $N^{-\frac{1}{2}-\frac{1}{2(d-1)}}$, i.e.

$$c_{d-1}N^{-\frac{1}{2}-\frac{1}{2(d-1)}} \le \inf_{\omega_N \subset \mathbb{S}^{d-1}} D_{L^2, \operatorname{cap}}(\omega_N) \le c'_{d-1}N^{-\frac{1}{2}-\frac{1}{2(d-1)}},$$
(5.5)

which in turn bounds the difference of continuous and discrete energies in (5.3).

In addition to the original proof in [Sto73], an alternative proof has been given in [BD13]. Here we present a new short and simple proof of the Stolarsky invariance principle (5.3) [BDM18]. It strongly resonates with the proof in [BD13], but is completely elementary in nature. A similar proof in a probabilistic interpretation has been independently given in [HBZO19] (compare Lemmas 5.1.2 and 5.1.3 below to Proposition 1 of [HBZO19]), and analogous ideas are used in [Skr17] on compact, connected, two-point homogeneous spaces and [Bar20] for the Hamming Cube.

The proof of (5.3) follows the aforementioned strategy: one squares out the integrand, and the discrete part (pairwise interactions) arises naturally from the cross terms. The important ingredient is the following relation between intersections of spherical caps and the Euclidean distance between their centers:

Lemma 5.1.2. For arbitrary $x, y \in \mathbb{S}^{d-1}$ we have

$$\int_{-1}^{1} \int_{S^{d-1}} \mathbb{1}_{C(x,h)}(z) \cdot \mathbb{1}_{C(y,h)}(z) d\sigma(z) dh = \int_{-1}^{1} \sigma \left(C(x,h) \cap C(y,h) \right) dh = 1 - C_{d-1} ||x - y||,$$
(5.6)

where the constant C_{d-1} is given by (5.4).

Proof. We have

$$\int_{-1}^{1} \sigma(C(x,h) \cap C(y,h)) dh = \int_{-1}^{1} \int_{\mathbb{S}^{d-1}} \mathbb{1}_{C(x,h)}(z) \cdot \mathbb{1}_{C(y,h)}(z) d\sigma(z) dh$$
$$= \int_{\mathbb{S}^{d-1}} \int_{-1}^{1} \mathbb{1}_{C(z,h)}(x) \cdot \mathbb{1}_{C(z,h)}(y) dh d\sigma(z)$$
$$= \int_{\mathbb{S}^{d-1}} \int_{-1}^{\min\{\langle x, z \rangle, \langle y, z \rangle\}} dh d\sigma(z)$$
$$= \int_{\mathbb{S}^{d-1}} \left(\min\{\langle x, z \rangle, \langle y, z \rangle\} + 1\right) d\sigma(z).$$

We now write $\min\{\langle x, z \rangle, \langle y, z \rangle\} = \frac{1}{2} (\langle x, z \rangle + \langle y, z \rangle - |\langle (x-y), z \rangle|)$. Clearly, $\int_{\mathbb{S}^{d-1}} \langle x, z \rangle d\sigma(z) = \int_{\mathbb{S}^{d-1}} \langle y, z \rangle d\sigma(z) = 0$ and by rotational invariance, we observe that

$$\int_{\mathbb{S}^{d-1}} \left| \langle (x-y), z \rangle \right| d\sigma(z) = \|x-y\| \cdot \int_{\mathbb{S}^{d-1}} \left| \left\langle \frac{x-y}{\|x-y\|}, z \right\rangle \right| d\sigma(z) = 2C_{d-1} \|x-y\|,$$

and this finishes the proof.

Here we essentially repeated the proof from [BD13], but the proof of the next lemma, which gives the quadratic mean value of the size of the spherical caps, is simpler (does not use reproducing kernels).

Lemma 5.1.3. *For any* $p \in \mathbb{S}^{d-1}$ *we have*

$$\int_{-1}^{1} \left(\sigma(C(p,h)) \right)^2 dh = 1 - C_d \int_{\mathbb{S}^{d-1}} \|x - p\| \, d\sigma(x), \tag{5.7}$$

Proof. It is clear that

$$\int_{\mathbb{S}^{d-1}} \|x-p\| d\sigma(x) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \|x-y\| d\sigma(x) d\sigma(y)$$

We use the result of the previous lemma, i.e. relation (5.6), to compute

$$\begin{split} 1 - C_{d-1} & \int_{\mathbb{S}^{d-1}} \|x - p\| \, d\sigma(x) = \int_{\mathbb{S}^{d-1}} \left(1 - C_{d-1} \|x - p\| \right) d\sigma(x) \\ &= \int_{\mathbb{S}^{d-1}} \int_{-1}^{1} \int_{\mathbb{S}^{d-1}} \mathbb{1}_{C(x,h)}(z) \cdot \mathbb{1}_{C(p,h)}(z) d\sigma(z) \, dh \, d\sigma(x) \\ &= \int_{-1}^{1} \int_{\mathbb{S}^{d-1}} \mathbb{1}_{C(p,h)}(z) \left(\int_{\mathbb{S}^{d-1}} \mathbb{1}_{C(z,h)}(x) \, d\sigma(x) \right) \, d\sigma(z) \, dh \\ &= \int_{-1}^{1} \left(\sigma(C(p,h)) \right)^2 \, dh. \end{split}$$

This is one of numerous examples of a situation in which averaging over scales simplifies computations. For the L^2 discrepancy for spherical caps of fixed height *h*:

$$D_{L^{2}, \operatorname{cap}}^{(h)}(\omega_{N}) := \left(\int_{\mathbb{S}^{d-1}} \left| \frac{1}{N} \sum_{j=1}^{N} \mathbb{1}_{C(x,h)}(z_{j}) - \sigma(C(x,h)) \right|^{2} d\sigma(x) \right)^{1/2},$$
(5.8)

one would have to deal with $\sigma(C(x,h) \cap C(y,h))$, which has complicated structure, and no short relation akin to (5.6) is available, see e.g. [HBZO19]. Hence in this case, there is no formula as succinct and explicit as the Stolarsky principle, however one can still write down a generic relation where the interactions between z_j 's would involve $\sigma(C(z_i,h) \cap C(z_j,h))$.

Proposition 5.1.4. For any $\omega_N = \{z_1, z_2, ..., z_N\} \subset \mathbb{S}^{d-1}$ and a fixed $h \in [-1, 1]$, the following relation holds

$$\left[D_{L^{2},\text{cap}}^{(h)}(\omega_{N})\right]^{2} = \frac{1}{N^{2}} \sum_{i,j=1}^{N} \sigma\left(C(z_{i},h) \cap C(z_{j},h)\right) - \left(\sigma\left(C(p,h)\right)\right)^{2},$$
(5.9)

where $p \in \mathbb{S}^{d-1}$ is arbitrary.

Proof. We note that $\sigma(C(x,h))$ is independent of $x \in \mathbb{S}^{d-1}$, hence

$$\begin{split} \left[D_{L^{2}}^{(h)}(\boldsymbol{\omega}_{N}) \right]^{2} &= \int_{\mathbb{S}^{d-1}} \left| \frac{1}{N} \sum_{j=1}^{N} \mathbb{1}_{C(x,h)}(z_{j}) - \boldsymbol{\sigma} \left(C(x,h) \right) \right|^{2} d\boldsymbol{\sigma}(x) \\ &= \frac{1}{N^{2}} \sum_{i,j=1}^{N} \int_{\mathbb{S}^{d-1}} \mathbb{1}_{C(x,h)}(z_{i}) \cdot \mathbb{1}_{C(x,h)}(z_{j}) d\boldsymbol{\sigma}(x) \\ &\quad - \frac{2}{N} \int_{\mathbb{S}^{d-1}} \sum_{j=1}^{N} \mathbb{1}_{C(z_{j},h)}(x) \cdot \boldsymbol{\sigma} \left(C(x,h) \right) d\boldsymbol{\sigma}(x) + \left(\boldsymbol{\sigma} \left(C(p,h) \right) \right)^{2} \\ &= \frac{1}{N^{2}} \sum_{i,j=1}^{N} \boldsymbol{\sigma} \left(C(z_{i},h) \cap C(z_{j},h) \right) - \left(\boldsymbol{\sigma} \left(C(p,h) \right) \right)^{2}. \end{split}$$
(5.10)

Integrating identity (5.9) with respect to h, as

$$[D_{L^{2},\mathrm{cap}}(\omega_{N})]^{2} = \int_{-1}^{1} [D_{L^{2},\mathrm{cap}}^{(h)}(\omega_{N})]^{2} dh,$$

and applying relations (5.6) and (5.7) we finish the proof of the Stolarsky principle (5.3).

5.2 Hemispheric Stolarsky Principle and Geodesic Riesz s-Energy

In this section, we see that by restricting to only hemispheres, we produce a new invariance principle that involves the discrete geodesic Riesz (-1)-energy in place of the standard Riesz (-1)-energy. This allows one to characterize the finite point sets on \mathbb{S}^{d-1} which maximize the sum of geodesic distances (see Theorem 5.2.10). This can be taken one step further to show that this analogue of the Stolarsky Invariance Principle holds for general probability measures, providing a way to characterize the maximizers of the continuous geodesic Riesz (-1)-energy $I_{\vartheta_{-1}^*}$. This brings up a surprising difference between the geodesic and standard Riesz (-1)-energies. Writing the negative geodesic Riesz (-1)-kernel as a function of the Euclidean distance

$$-\vartheta_{-1}^*(x,y) = -\frac{2}{\pi}\arccos(\langle x,y\rangle) = -\frac{2}{\pi}\arccos\left(1 - \frac{||x-y||^2}{2}\right)$$

we see it has the same order of repulsion as $-\mathcal{D}_{-1}$, i.e.

$$\lim_{r \to 0^+} \frac{-\frac{2}{\pi} \arccos\left(1 - \frac{r^2}{2}\right)}{r} = \frac{4}{\pi}.$$
(5.11)

Thus, both $-\mathcal{D}_{-1}$ and $-\vartheta_{-1}^*$ are *strongly repulsive*, with $\gamma = 1$ in the notation of Theorem 4.0.4. This might lead one to think that the maximizers of $I_{\vartheta_{-1}^*}$ and $I_{\mathcal{D}_{-1}}$ would exhibit similar behavior. However, any symmetric measure is optimal in the former case, which includes both discrete and absolutely continuous measures, while only the uniform measure σ maximizes the latter energy.

Hemispheric Stolarsky Principle

An (open) hemisphere in the direction of $x \in \mathbb{S}^{d-1}$ is simply a spherical cap of height h = 0:

$$H(x) = \{ z \in \mathbb{S}^{d-1} : \langle z, x \rangle > 0 \} = C(x, 0).$$
(5.12)

Since $\sigma(H(x)) = \frac{1}{2}$, the natural L^2 discrepancy for this set system is

$$D_{L^{2},\text{hem}}(\boldsymbol{\omega}_{N}) := \left(\int_{\mathbb{S}^{d-1}} \left| \frac{1}{N} \sum_{j=1}^{N} \mathbb{1}_{H(x)}(z_{j}) - \frac{1}{2} \right|^{2} d\boldsymbol{\sigma}(x) \right)^{1/2} = D_{L^{2},\text{cap}}^{(0)}(\boldsymbol{\omega}_{N}).$$
(5.13)

While, as mentioned above, generally the quantity $\sigma(C(x,h) \cap C(y,h))$ is complicated, in the case h = 0 (hemispheres) it has a very simple representation: for $x, y \in \mathbb{S}^{d-1}$

$$\sigma(H(x) \cap H(y)) = \sigma(C(x,0) \cap C(y,0)) = \frac{1}{2} \cdot (1 - \vartheta^*(x,y)), \qquad (5.14)$$

where

$$\vartheta^*(x,y) := \frac{\vartheta(x,y)}{\pi}$$
(5.15)

is the normalized geodesic distance on the sphere between x and y. This can be very easily seen from Figure 5.1.

Combining (5.14), Proposition 5.1.4 and the fact that $\int_{\mathbb{S}^{d-1}\mathbb{S}^{d-1}} \vartheta^*(x,y) d\sigma(x) d\sigma(y) = \frac{1}{2}$, one arrives at the following result:



Figure 5.1: The size $\sigma(H_x \cap H_y)$ of the intersection of two hemispheres depends linearly on the geodesic distance $\vartheta^*(x, y)$.

Theorem 5.2.1 (Stolarsky Principle for Hemispheres). *For any* $\omega_N \in \mathbb{S}^{d-1}$ *, the following relation holds:*

$$[D_{L^2,\text{hem}}(\omega_N)]^2 = \frac{1}{2} \left(\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \vartheta^*(x,y) \, d\boldsymbol{\sigma}(x) \, d\boldsymbol{\sigma}(y) - \frac{1}{N^2} \sum_{i,j=1}^N \vartheta^*(z_i,z_j) \right).$$
(5.16)

The statement looks strikingly similar to the original Stolarsky principle (5.3). One can say that the Euclidean distance corresponds to the *mean* over $t \in [-1, 1]$, while the geodesic distance corresponds to the *median* (h = 0) of the heights of the spherical caps. Despite the fact that the original Stolarsky principle was proved in 1973, the Hemispheric version has only been discovered very recently, and independently, in [Skr17] and [BDM18]. Relation (5.16) has several interesting features and consequences.

Sum of Geodesic Distances

First of all, the principle of *irregularities of distribution*, discussed in great detail in [BC87], does not hold in this situation, that is, the hemisphere discrepancy can be very small, even zero, for large *N*. Indeed, for any symmetric distribution ω_N , it is easy to see that the L^2

hemisphere discrepancy is equal to zero. Moreover, (5.16) allows us to characterize finite point distributions in \mathbb{S}^{d-1} , which maximize the sum of geodesic distances.

Theorem 5.2.2. *Let* $d \ge 2$ *. Then the following holds:*

1. For any point distribution $\omega_N = \{z_1, \ldots, z_N\} \subset \mathbb{S}^{d-1}$,

$$\frac{1}{N^2} \sum_{i,j=1}^{N} \vartheta^*(z_i, z_j) \le \frac{1}{2}.$$
(5.17)

2. For a given $N \in \mathbb{N}$ the sum above is maximized if and only if the following condition holds: for any $x \in \mathbb{S}^{d-1}$, such that the hyperplane x^{\perp} contains no points of ω_N , the numbers of points of ω_N on either side of x^{\perp} differ by at most one, i.e.

$$\left| \left| \boldsymbol{\omega}_{N} \cap H(x) \right| - \left| \boldsymbol{\omega}_{N} \cap H(-x) \right| \right| \leq 1.$$
(5.18)

3. If N is even,

$$\max_{\omega_N \subset \mathbb{S}^{d-1}} \frac{1}{N^2} \sum_{i,j=1}^N \vartheta^*(z_i, z_j) = \frac{1}{2},$$
(5.19)

and this maximum is achieved if and only if ω_N is a centrally symmetric set.

4. If N is odd,

$$\max_{\omega_N \subseteq \mathbb{S}^{d-1}} \frac{1}{N^2} \sum_{i,j=1}^N \vartheta^*(z_i, z_j) = \frac{1}{2} - \frac{1}{2N^2},$$
(5.20)

and this maximum is achieved if and only if ω_N can be represented as a union $\omega_N = Z_1 \cup Z_2$, where Z_1 is symmetric, while Z_2 lies on a two-dimensional hyperplane (i.e. on a great circle) and satisfies

$$\frac{1}{M^2} \sum_{z_i, z_j \in Z_2} \vartheta^*(z_i, z_j) = \frac{1}{2} - \frac{1}{2M^2},$$

where $M = |Z_2|$. In other words, Z_2 is a maximizer of the sum of geodesic distances on \mathbb{S}^1 .

Before we turn to the proof of the theorem, we briefly discuss the history of these questions. The problem of maximizing the sum of geodesic distances on the sphere was first introduced by Fejes Tóth in [FT59]. In his work, Fejes Tóth proved parts (3) and (4) on \mathbb{S}^1 and conjectured that they held on \mathbb{S}^2 (which he showed for $N \leq 6$). This conjecture was proven for N even (i.e. part (3)) soon after by Sperling [Spe60], and later for odd N (i.e. part (4)) by Nielsen [Nie65]. An earlier proof of part (4) for d = 3 was provided by Larcher in [Lar62], however, there is a mistake in his proof (statement (ii) at the bottom of page 48). Kelly then proved the bounds (5.19) and (5.20) for all N and in all dimensions in [Kel70], though no characterization of maximizers was found. Though parts (1) and (2) can be inferred from parts (3) and (4), they were recently proven directly for d = 2 in [Jia08] in relation to musical rhythms.

Our Stolarsky principle (5.16) allows us to directly prove parts (1) and (2), and makes the proofs of (5.19) and (5.20) simple in all dimensions $d \ge 2$. Moreover, we can now provide a characterization of maximizers, something that was previously known only for $d \le 3$. In the case of odd *N*, we exploit a spherical version of the Sylvester–Gallai theorem (an interesting result from combinatorial geometry), much as Nielsen did in their paper [Nie65], though the main methods are different. We now turn to the proof of Theorem 5.2.2.

Proof. Part (1) of the theorem is now obvious since the left-hand side of (5.16) is nonnegative and $\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \vartheta^*(x,y) d\sigma(x) d\sigma(y) = \frac{1}{2}$.

Part (2) also follows easily from (5.16). Indeed, for every $x \in \mathbb{S}^{d-1}$ such that x^{\perp} does not contain any points of ω_N , the minimal value of the integrand in the left-hand side of (5.16), i.e. the integrand in (5.13), equals 0 for even N (if exactly half the points lie on either side of x^{\perp}), and is $\frac{1}{4N^2}$ for odd N (if the numbers of points on both sides of x^{\perp} differ exactly by 1). Obviously, configurations ω_N for which this is achieved for each such $x \in \mathbb{S}^{d-1}$ are possible: e.g., $\lfloor N/2 \rfloor$ and $\lceil N/2 \rceil$ points in antipodal poles. Moreover, if for some $x \in \mathbb{S}^{d-1}$ with $x^{\perp} \cap \omega_N = \emptyset$ this condition is not satisfied, then it also fails on a small set of positive measure around x, hence $D_{L^2,hem}(\omega_N)$ is not minimal, and therefore $\frac{1}{N^2} \sum \vartheta^*(z_i, z_j)$ is not maximized.

To prove part (3), first observe that symmetric sets ω_N trivially satisfy the condition of part (2). Now assume that for some $x \in \mathbb{S}^{d-1}$ the number of points of ω_N located at xand -x is not the same. Consider a hyperplane passing through x, which contains no other points of ω_N . Perturbing the hyperplane in opposite directions, we find that the difference of number of points on either side changes by at least 2, i.e. cannot stay equal to zero. Thus non-symmetric sets ω_N with an even number of points do not satisfy the condition of part (2), and so cannot maximize the sum of geodesic distances.

We now turn to part (4). We shall rely on the Sylvester–Gallai theorem. In the Euclidean case it states the following: *if a finite set* ω_N *in* \mathbb{R}^d *has the property that for every two points of* ω_N *, the straight line passing through them contains at least one other point of* ω_N *, then all points of* ω_N *lie on the same straight line.* A spherical version of this theorem also holds.

Theorem 5.2.3 (Spherical Sylvester–Gallai Theorem). Assume that a set ω_N of N points on the sphere \mathbb{S}^{d-1} contains no antipodal points and satisfies the following condition: for every two points of ω_N , the great circle passing through them contains at least one more point of ω_N . Then all points of ω_N lie on the same great circle.

For the history and several proofs of these theorems we refer the reader to the book [AZ14, pages 73 and 88]. Normally, these theorems are stated in dimension d = 3, but higher dimensional extensions are simple. Indeed, for $\omega_N \subset \mathbb{S}^{d-1}$, consider a copy of \mathbb{S}^2 which contains $z_1, z_2, z_3 \in \omega_N$. The lower-dimensional version of Theorem 5.2.3 applies, and hence z_1, z_2, z_3 lie on the same great circle. In the same manner, considering a copy of \mathbb{S}^2 containing this great circle and any other point $z_i \in \omega_N$, we find that z_i has to lie on the same great circle.

We are now ready to prove part (4). Assume that N is odd. It follows from (5.16) and the proof of part (2) that the maximal value of $\frac{1}{N^2} \sum \vartheta^*(z_i, z_j)$ is $\frac{1}{2} - \frac{1}{2N^2}$. Observe that adding a pair of antipodal points to ω_N does not change maximality of ω_N , i.e. ω_N is a maximizer if and only if $\omega_N \cup \{p, -p\}$ is a maximizer (with *N* replaced by N+2). Indeed, since $\vartheta^*(x, p) + \vartheta^*(x, -p) = 1$, it is easy to check that

$$\sum_{x,y\in\boldsymbol{\omega}_N\cup\{-p,p\}}\vartheta^*(x,y)=\sum_{x,y\in\boldsymbol{\omega}_N}\vartheta^*(x,y)+2(N+1),$$

thus the second sum equals $\frac{N^2}{2} - \frac{1}{2}$ if and only if the first sum is $\frac{(N+2)^2}{2} - \frac{1}{2}$. This immediately proves sufficiency of the condition in (4). Moreover, it shows that, in order to prove necessity, it is enough to consider maximizers without antipodal points and to prove that they have to be contained in some great circle.

Assume that ω_N maximizes $\frac{1}{N^2} \sum \vartheta^*(z_i, z_j)$ and contains no pair of antipodal points. Consider two arbitrary points $z_1, z_2 \in \omega_N$, and assume that no other point of ω_N lies on the great circle defined by z_1 and z_2 . Since ω_N is finite, there exists a hyperplane containing z_1 and z_2 , which does not contain any other points of ω_N . Since z_1 and z_2 are not antipodal, one can perturb the hyperplane in such a way that it does not touch other points of ω_N and both points z_1 and z_2 end up on the same side of the hyperplane. Perturbing in the opposite direction, we observe that the difference between the numbers of points on opposite sides of the hyperplane changes by 4, and therefore cannot stay equal to ± 1 , so by part (2), ω_N cannot be a maximizer.

We thus conclude that, for any two points of ω_N , at least one other point of ω_N has to lie on the same great circle, i.e. the spherical Sylvester–Gallai theorem, Theorem 5.2.3, applies. Hence ω_N is contained in a great circle.

Remark: Observe that the one-dimensional maximizers of odd cardinality N, which arise in part (4) of Theorem 5.2.2, are characterized by the condition that the sum of any $\lceil N/2 \rceil$ consecutive central angles defined by the points is at least π . In particular, any acute triangle is a maximizer for d = 2 and N = 3.

Theorem 5.2.2 demonstrates that the situation is drastically different from the spherical

cap discrepancy and the sum of Euclidean distances. In the latter case, minimizing the L^2 spherical cap discrepancy (equivalently, maximizing the sum of Euclidean distances) leads to a rather uniform distribution of ω_N [Bec84a, Bec84b, Mat99]. In particular, for d = 2 the sum is maximized by the vertices of a regular *N*-gon [FT56], and in higher dimensions maximizing distributions have to be well-separated [Sto75]. The sum of *geodesic* distances, however, may be maximized by very non-uniform sets, e.g. N/2 points in two antipodal poles.

Geodesic Distance Energy Integral

We now turn our sights to the problem of finding equilibrium distributions of the geodesic distance energy integral. As in the discrete case, we are interested in the maximum value of the energy I_{ϑ^*} , as well as the maximizers of this expression. The former follows from Theorem 5.2.2 and the weak^{*} density of discrete measures in $\mathbb{P}(\mathbb{S}^{d-1})$:

$$\max_{\mu \in \mathbb{P}(\mathbb{S}^{d-1})} I_{\vartheta^*}(\mu) = \lim_{n \to \infty} \sup_{\omega_N \subset \mathbb{S}^{d-1}} E_{\vartheta^*}(\omega_N) = \frac{1}{2}.$$
(5.21)

One possible way to determine maximizing measures would be to show that the Gegenbauer expansion of $F(\langle x, y \rangle) = \frac{1}{2} - \vartheta^*(x, y)$ satisfies the conditions of part (3) of 3.5.2, so symmetric measures are uniquely optimal. However, we shall instead provide a proof making use of our Hemisphere Stolarsky principle (i.e. Theorem 5.2.1), which may be extended to more general measures than the counting measure $\mu = \frac{1}{N} \sum_{i=1}^{N} \delta_{z_i}$.

Theorem 5.2.4 (Hemisphere Stolarsky principle for general measures). Let $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$. *Then the following relation holds*

$$\int_{\mathbb{S}^{d-1}} \left(\mu \left(H(x) \right) - \frac{1}{2} \right)^2 d\boldsymbol{\sigma}(x) = \frac{1}{2} \cdot \left(\frac{1}{2} - I_{\vartheta^*}(\mu) \right).$$
(5.22)

Proof. Notice that

$$\int_{\mathbb{S}^{d-1}H(x)} \int d\mu(y) d\sigma(x) = \int_{\mathbb{S}^{d-1}H(y)} \int d\sigma(x) d\mu(y) = \frac{1}{2} \cdot \int_{\mathbb{S}^{d-1}} d\mu(y) = \frac{1}{2}$$

and, according to (5.14),

$$\int_{\mathbb{S}^{d-1}H(x)H(x)} \int d\mu(y)d\mu(z)d\sigma(x) = \int_{\mathbb{S}^{d-1}\mathbb{S}^{d-1}} \int \sigma(H(y)\cap H(z))d\mu(y)d\mu(z)$$
$$= \int_{\mathbb{S}^{d-1}\mathbb{S}^{d-1}} \int \frac{1}{2} \cdot \left(1 - \vartheta^*(y,z)\right)d\mu(y)d\mu(z) = \frac{1}{2} - \frac{1}{2}I_{\vartheta^*}(\mu).$$

Using the two relations above we obtain

$$\int_{\mathbb{S}^{d-1}} \left(\mu(H(x)) - \frac{1}{2} \right)^2 d\sigma(x) = \int_{\mathbb{S}^{d-1}} \left(\int_{H(x)H(x)} \int_{H(x)} d\mu(y) d\mu(z) - \int_{H(x)} d\mu(y) + \frac{1}{4} \right) d\sigma(x)$$
$$= \frac{1}{2} - \frac{1}{2} I_{\vartheta^*}(\mu) - \frac{1}{2} + \frac{1}{4} = \frac{1}{2} \cdot \left(\frac{1}{2} - I_{\vartheta^*}(\mu) \right),$$

which proves (5.22).

Since the left-hand side of identity (5.22) is non-negative, Theorem 5.2.4 provides another way of showing that $I_{\vartheta^*}(\mu) \leq \frac{1}{2}$ for all probability measures, and an immediate necessary condition to achieve that bound:

Corollary 5.2.5. *Measures* $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ *for which* $I_{\vartheta^*}(\mu) = \frac{1}{2}$ *, are exactly the measures which satisfy the following condition:*

$$\mu(H(x)) = \frac{1}{2} \text{ for } \sigma\text{-a.e. } x \in \mathbb{S}^{d-1}.$$
(5.23)

It is easy to see that if the measure μ is symmetric, it is a maximizer of the energy I_{ϑ^*} . Consider the reflection of a measure μ , which we will call μ^* , i.e. $\mu^*(E) = \mu(-E)$. It is easy to see that

$$\begin{split} I_{\vartheta^*}(\mu,\mu^*) &= \int_{\mathbb{S}^{d-1}} \int_{\vartheta^*} \vartheta^*(x,y) d\mu(x) d\mu^*(y) = \int_{\mathbb{S}^{d-1}} \int_{\vartheta^*} \vartheta^*(x,-y) d\mu(x) d\mu(y) \\ &= \int_{\mathbb{S}^{d-1}} \int_{\vartheta^{d-1}} \left(1 - \vartheta^*(x,y) \right) d\mu(x) d\mu(y) = 1 - I_{\vartheta^*}(\mu). \end{split}$$

If, moreover, μ is symmetric, i.e. $\mu^* = \mu$, then $I_{\vartheta^*}(\mu) = \frac{1}{2}$, as

$$I_{\vartheta^{*}}(\mu) = I_{\vartheta^{*}}(\mu, \mu^{*}) = 1 - I_{\vartheta^{*}}(\mu).$$
(5.24)

Therefore, in particular, every symmetric measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ satisfies (5.23). The converse of this fact is less obvious.

Proposition 5.2.6. Assume that the measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ satisfies the condition

$$\mu(H(x)) = \frac{1}{2} \text{ for } \sigma\text{-a.e. } x \in \mathbb{S}^{d-1}.$$
(5.25)

Then the measure μ is symmetric, i.e. $\mu(E) = \mu(-E)$ for every Borel set $E \subseteq \mathbb{S}^{d-1}$.

The proof for this Proposition is based on spherical harmonics and Gegenbauer polynomials and requires the following auxiliary lemma, which will also be used in Section 5.3.

Lemma 5.2.7. Let γ be a finite signed Borel measure on \mathbb{S}^{d-1} and $F \in L^2([-1,1], w_{\lambda})$ with $\lambda = \frac{d-3}{2}$. Assume that

$$\int_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) \, d\gamma(y) = 0 \text{ for } \sigma \text{-almost every } x \in \mathbb{S}^{d-1}.$$
(5.26)

Assume also that $\widehat{F}(n,\lambda) \neq 0$ for some $n \geq 1$. Then for every spherical harmonic $Y_n \in \mathscr{H}_n^d$,

$$\int_{\mathbb{S}^{d-1}} Y_n(y) d\gamma(y) = 0.$$
(5.27)

Proof. Applying the Funk-Hecke formula (2.29), we find that

$$\int_{\mathbb{S}^{d-1}} Y_n(y) d\gamma(y) = \frac{1}{\widehat{F}(n,\lambda)} \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} F(\langle x,y \rangle) Y_n(x) d\sigma(x) d\gamma(y)$$
$$= \frac{1}{\widehat{F}(n,\lambda)} \int_{\mathbb{S}^{d-1}} \left(\int_{\mathbb{S}^{d-1}} F(\langle x,y \rangle) d\gamma(y) \right) Y_n(x) d\sigma(x) = 0.$$

Proof of Proposition 5.2.6. Let μ^* be the reflection of μ , defined by $\mu^*(E) = \mu(-E)$, and set $\gamma = \mu - \mu^*$. Condition (5.25) then implies that

$$\gamma(H(x)) = \int_{\mathbb{S}^{d-1}} \mathbb{1}_{(0,1]}(\langle x, y \rangle) d\gamma(y) = 0 \text{ for } \sigma\text{-a.e. } x \in \mathbb{S}^{d-1}$$

The Gegenbauer coefficients $\widehat{F}(n,\lambda)$ of the function $F(t) = \mathbb{1}_{(0,1]}(t)$ are non-zero for odd *n* (Lemma 3.4.6 in [Gro96]). Therefore, according to Lemma 5.2.7, relation (5.27) holds for all odd *n*. For even values of *n*, the relation also holds, since $Y_n \in \mathscr{H}_n^d$ is an even function in this case, and γ is antisymmetric. Therefore, $\int_{\mathbb{S}^{d-1}} Y(y) d\gamma(y) = 0$ for every polynomial *Y*, and hence for each $Y \in C(\mathbb{S}^{d-1})$, which implies that $\gamma = 0$. Hence $\mu = \mu^*$, i.e. μ is symmetric.

From the above discussion we obtain the following characterization of the maximizers of $I_{\vartheta^*}(\mu)$:

Theorem 5.2.8. *For a measure* $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ *,*

$$I_{\vartheta^*}(\mu) = \sup_{\gamma \in \mathbb{P}(\mathbb{S}^{d-1})} I_{\vartheta^*}(\gamma) = \frac{1}{2}$$

if and only if μ *is centrally symmetric.*

This behavior of $I_{\vartheta^*}(\mu)$ goes in sharp contrast with the behavior of the seemingly similar energy integral $\int_{\mathbb{S}^{d-1}\mathbb{S}^{d-1}} ||x-y|| d\mu(x)d\mu(y)$. It is known [Bjö56] that the unique maximizer of this energy integral is $\mu = \sigma$, the uniform distribution on \mathbb{S}^{d-1} . In this sense the behavior of $I_{\vartheta^*}(\mu)$ is more similar (albeit still different) to that of $\int_{\mathbb{S}^{d-1}\mathbb{S}^{d-1}} ||x-y||^2 d\mu(x)d\mu(y)$ which is maximized by any measure with center of mass at the origin, which may be easily seen from the relation

$$\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \|x - y\|^2 d\mu(x) d\mu(y) = 2 - 2 \cdot \left\| \int_{\mathbb{S}^{d-1}} x d\mu(x) \right\|^2.$$
(5.28)

It is thus natural to analyze the *Geodesic Riesz s-energy* for general powers s < 0, and contrast it with the standard Riesz *s*-energies, discussed in Section 2.1.

Geodesic Riesz s-Energy

We would like to understand which measures $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ maximize the Geodesic Riesz *s*-energy

$$I_{\vartheta_s^*}(\mu) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \left(\vartheta^*(x, y) \right)^{-s} d\mu(x) d\mu(y),$$
(5.29)

for s < 0, and how maximizers depend on s. The case of 0 < s < d - 1 was studied by Bilyk and Dai in [BD19], where they found that σ was the unique minimizer of $I_{\vartheta_s^*}$.

While the study of the geodesic Riesz *s*-energies is quite recent, the Riesz *s*-energies for Euclidean distance are well investigated, as discussed in Section 2.1. In addition to the properties of maximizers given in Theorem 2.1.2 for general compact spaces, Björck was able to further refine his results in the case $\Omega = \mathbb{S}^{d-1}$.

Theorem 5.2.9 ([Bjö56]). For s < 0, the maximizers of the Euclidean Riesz s-energy

$$I_{\mathscr{D}_s}(\mu) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} ||x-y||^{-s} d\mu(x) d\mu(y)$$

over $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ can be characterized as follows:

- 1. 0 > s > -2: the unique maximizer of $I_{\mathscr{D}_s}(\mu)$ is $\mu = \sigma$.
- 2. s = -2: $I_{\mathscr{D}_s}(\mu)$ is maximized if and only if the center of mass of μ is at the origin.
- 3. s < -2: $I_{\mathscr{D}_s}(\mu)$ is maximized if and only if $\mu = \frac{1}{2}(\delta_p + \delta_{-p})$, i.e. the mass is equally concentrated at two antipodal poles.

The proof of part (1) uses potential analysis, in particular, the semigroup property of the Riesz potentials; part (2) is explained in (5.28); and part (3) follows from part (2) and simple linear programming bounds.

We observe that there is a "breaking point" s = -2 in the behavior of maximizers of the Euclidean Riesz *s*-energy. Surprisingly, for the seemingly similar geodesic Riesz *s*-energy, this critical value is different: s = -1. We have the following theorem:

Theorem 5.2.10. For s < 0, the maximizers of the geodesic Riesz s-energy

$$I_{\vartheta_s^*}(\mu) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \left(\vartheta^*(x, y) \right)^{-s} d\mu(x) d\mu(y)$$

over $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ can be characterized as follows:

- 1. 0 > s > -1: the unique maximizer of $I_{\vartheta_s^*}(\mu)$ is $\mu = \sigma$.
- 2. s = -1: $I_{\vartheta_s^*}(\mu)$ is maximized if and only if μ is centrally symmetric.
- 3. s < -1: $I_{\vartheta_s^*}(\mu)$ is maximized if and only if $\mu = \frac{1}{2}(\delta_p + \delta_{-p})$, i.e. the mass is equally concentrated at two antipodal poles.

Part (1) is proved in [BD19] through extensive analysis of spherical harmonic expansions. Part (2) is the result of Theorem 5.2.8 above, which is a consequence of the hemisphere Stolarsky principle (5.22). The proof of part (3) is quite simple, following from
Lemma 4.3.2: since $\vartheta^*(x, y) \le 1$, we have for s < -1

$$I_{\vartheta_s^*}(\mu) \leq \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} \vartheta^*(x,y) d\mu(x) d\mu(y) \leq \frac{1}{2}.$$

The first inequality turns into an equality when $(\mu \times \mu)\{(x,y) : \vartheta^*(x,y) = 0 \text{ or } 1\} = 1$, while the second bound becomes exact when μ is symmetric, according to part (2). This readily implies that $\mu = \frac{1}{2}(\delta_p + \delta_{-p})$.

This peculiar effect (that geodesic distance energy behaves differently from its Euclidean counterpart) was previously noticed in dimension d = 2, i.e. on the circle, in [BHS12], where the one-dimensional case of parts (1) and (3) of the above theorem have been proved.

5.3 The Generalized Stolarsky Principle for the Sphere

In this section we turn to a generalization of the Stolarsky Invariance Principle for positive definite functions. Assume that $F \in C([-1,1])$ is positive definite, $\lambda = \frac{d-2}{2}$, and the function $f \in L^2([-1,1], w_\lambda)$ is as in Corollary 2.5.2, i.e.

$$F(\langle x, y \rangle) = \int_{\mathbb{S}^{d-1}} f(\langle x, z \rangle) f(\langle z, y \rangle) d\sigma(z)$$

For a Borel probability measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ we define the L^2 discrepancy of μ with respect to f as

$$D_{L^2,f}(\mu) = \left(\int_{\mathbb{S}^{d-1}} \left| \int_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) d\mu(y) - \int_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) d\sigma(y) \right|^2 d\sigma(x) \right)^{\frac{1}{2}}.$$
 (5.30)

The L^2 discrepancy of a finite point-set $\omega_N \subset \mathbb{S}^{d-1}$ is simply

$$D_{L^2,f}(\omega_N) = D_{L^2,f}\left(\frac{1}{N}\sum_{j=1}^N \delta_{z_j}\right) = \left(\int\limits_{\mathbb{S}^{d-1}} \left|\frac{1}{N}\sum_{i=1}^N f(\langle x, z_i \rangle) - \int\limits_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) d\sigma(y)\right|^2 d\sigma(x)\right)^{\frac{1}{2}}.$$

Notice that various choices of f recover different geometric notions of discrepancy (e.g., $\mathbb{1}_{(h,1]}$ for spherical caps of a fixed height, see (5.9)), although this object is more general. We now prove a general version of the Stolarsky principle, which connects the energies with respect to F with the L^2 discrepancy built upon f.

Theorem 5.3.1 (Generalized Stolarsky principle). Let $\mu \in \mathbb{P}^*(\mathbb{S}^{d-1})$, i.e. a signed Borel probability measure, and let *F* be positive definite with *f* as in (3.16). Then

$$I_F(\mu) - I_F(\sigma) = D_{L^2, f}^2(\mu).$$
(5.31)

In particular, in the case of $\mu = \frac{1}{N} \sum_{i=1}^{N} \delta_{z_i}$, this relation becomes

$$\frac{1}{N^2} \sum_{i,j=1}^{N} F(\langle z_i, z_j \rangle) - \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} F(\langle x, y \rangle) d\sigma(x) d\sigma(y) = D_{L^2,f}^2(\omega_N).$$
(5.32)

Proof. According to the definition of $D_{L^2,f}(\mu)$, (3.16), and Lemma 3.2.3, we have

$$D_{L^{2},f}^{2}(\mu) = \int_{\mathbb{S}^{d-1}} \left(\int_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) d(\mu - \sigma)(y) \right)^{2} d\sigma(x)$$

= $\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) f(\langle x, z \rangle) d(\mu - \sigma)(y) d(\mu - \sigma)(z) d\sigma(x)$
= $\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} F(\langle y, z \rangle) d(\mu - \sigma)(y) d(\mu - \sigma)(z) = I_{F}(\mu - \sigma) = I_{F}(\mu) - I_{F}(\sigma).$

This approach brings up several novel points. First of all, in most contexts the Stolarsky identity arises from the notion of the L^2 discrepancy, which in turn dictates the specific

form of the interaction potential *F*. Theorem 5.3.1, on the other hand, allows one to go in the opposite direction: starting with the positive definite potential *F*, one can produce a natural notion of discrepancy, for which the Stolarsky principle holds. The precise form of the function *f*, defined through the identity $(\hat{f}(n,\lambda))^2 = \hat{F}(n,\lambda)$, cannot be made explicit in most cases (in fact, many different choices of *f* corresponding to the same *F* can be constructed by changing the signs of the coefficients $\hat{f}(n,\lambda)$). However, this does not prevent one from being able to obtain estimates for $D_{L^2,f}(\mu)$, as shown by Bilyk and Dai in [BD19] (Theorem 4.2, part (ii)).

Proposition 5.3.2. Suppose that F is positive definite and $f \in L^2_{w_{\lambda}}([-1,1])$ is as in Corollary 2.5.2. Then there exist $c_{d-1}, c'_{d-1} > 0$ such that for all $N \in \mathbb{N}$,

$$c_{d-1}' \min_{1 \le k \le c_{d-1}N^{\frac{1}{d-1}}} \widehat{F}(k,\lambda) \le \inf_{\omega_N \subset \mathbb{S}^{d-1}} D_{L^2,f}^2(\omega_N) \le N^{-1} \max_{0 \le \theta \le c_{d-1}N^{-\frac{1}{d-1}}} F(1) - F(\cos\theta)$$
(5.33)

Hence, e.g., lower bounds can be proved using information about either F or f. In [BD19], the authors use these estimates to give an alternative proof of the spherical cap discrepancy bounds (5.5), and employ (5.32) to obtain sharp asymptotic behavior of the difference between discrete and continuous energies, $E_F(\omega_N) - I_F(\sigma)$, as the number of points $N \to \infty$, both in the case of Euclidean Riesz energies, $F(\langle x, y \rangle) = \mathscr{D}_s(x, y)$ (recovering results of [Wag92, KS98, Bra06]), and the geodesic Riesz energies, $F(\langle x, y \rangle) = \vartheta_s^*(x, y)$, introduced in [BD19, BDM18].

Since $D_{L^2,f}^2(\mu) \ge 0$, the generalized Stolarsky identity (5.31) gives yet another proof that for *F* positive definite, the uniform measure σ is a minimizer of I_F over $\mathbb{P}(\mathbb{S}^{d-1})$ (in fact, over $\widetilde{\mathbb{P}}(\mathbb{S}^{d-1})$). Furthermore, the identity also provides an alternative way of showing that *F* having all positive Gegenbauer coefficients implies that σ is the unique minimizer of I_F . Indeed, without loss of generality, we may assume that $\widehat{F}(0,\lambda) > 0$. Assume also that for each $n \ge 1$, we have $\widehat{f}(n,\lambda) = (\widehat{F}(n,\lambda))^{1/2} \ne 0$. Let μ be a minimizer of $I_F(\mu)$, i.e. $I_F(\mu) = I_F(\sigma)$. Therefore, the Stolarsky principle (5.31) implies that $D^2_{L^2,f}(\mu) = 0$, i.e.

$$\int_{\mathbb{S}^{d-1}} f(\langle x, y \rangle) d(\sigma - \mu)(y) = 0$$

for σ -almost every x. Then by Lemma 5.2.7, relation (5.27), with $\gamma = \sigma - \mu$, holds for every polynomial, and thus for every $Y \in C(\mathbb{S}^{d-1})$. Hence σ is the unique minimizer of $I_F(\mu)$.

5.4 The Generalized Stolarsky Principle on Compact Metric Spaces

The generalized Stolarsky principle on the sphere, Theorem 5.3.1, extends to arbitrary compact domains without significant difficulty.

Let (Ω, ρ) be a compact metric space and let us fix a measure $\mu \in \mathbb{P}(\Omega)$ – this will usually be an energy minimizing (equilibrium) measure or an invariant measure (its role is similar to that of σ in the spherical case). We now define the L^2 discrepancy of an arbitrary probability measure $v \in \mathbb{P}(\Omega)$ (or even a signed measure $v \in \widetilde{\mathbb{P}}(\Omega)$) relative to the equilibrium measure μ with respect to the function $k : \Omega \times \Omega \to \mathbb{R}$ by the identity

$$D_{L^{2},k,\mu}^{2}(\mathbf{v}) = \int_{\Omega} \left| \int_{\Omega} k(x,y) d\mathbf{v}(y) - \int_{\Omega} k(x,y) d\mu(y) \right|^{2} d\mu(x)$$

$$= \int_{\Omega} \left| \int_{\Omega} k(x,y) d(\mathbf{v} - \mu)(y) \right|^{2} d\mu(x).$$
 (5.34)

When *v* is the equal-weight discrete measure associated to the *N*-point set $\omega_N = \{z_1, ..., z_N\} \subset \Omega$, i.e. $v = \frac{1}{N} \sum_{i=1}^N \delta_{z_i}$, this becomes the discrepancy of the set ω_N with respect to *k*:

$$D_{L^{2},k,\mu}^{2}(\omega_{N}) = D_{L^{2},k,\mu}^{2}\left(\frac{1}{N}\sum_{i=1}^{N}\delta_{z_{i}}\right) = \int_{\Omega} \left|\frac{1}{N}\sum_{i=1}^{N}k(x,z_{i}) - \int_{\Omega}k(x,y)d\mu(y)\right|^{2}d\mu(x) \quad (5.35)$$

Observe that changing the function k by an additive constant does not change the value of the discrepancy.

We can now obtain the following general version of the Stolarsky Invariance Principle:

Theorem 5.4.1. Let K be a positive definite (modulo and additive constant C) kernel on $\Omega \times \Omega$. Let us assume that $\mu \in \mathbb{P}(\Omega)$ is a K-invariant measure with full support ($\operatorname{supp}(\mu) = \Omega$). Then for every measure $v \in \widetilde{\mathbb{P}}(\Omega)$, we have the following identity.

$$I_K(\mathbf{v}) - I_K(\mu) = D_{L^2,k,\mu}^2(\mathbf{v}), \tag{5.36}$$

where the function $k \in L^2(\Omega \times \Omega, \mu \times \mu)$ is as in part (5) of Theorem 3.3.2 applied to the positive definite kernel K + C.

In particular, for a discrete set $\omega_N = \{z_1, ..., z_N\} \subset \Omega$,

$$E_{K}(\omega_{N}) - I_{K}(\mu) = D^{2}_{L^{2},k,\mu}(\omega_{N}).$$
(5.37)

This theorem has the following immediate corollary:

Corollary 5.4.2. Let *K* be a kernel on $\Omega \times \Omega$. Assume that $\mu \in \mathbb{P}(\Omega)$ is a global minimizer of the energy functional I_K over $\mathbb{P}(\Omega)$ with $\operatorname{supp}(\mu) = \widetilde{\Omega} \subseteq \Omega$. Then identity (5.36) holds for any signed measure v with total mass one, whose support is contained in the support of μ , i.e. $v \in \widetilde{\mathbb{P}}(\widetilde{\Omega})$. Similarly, relation (5.37) holds for any point set $\omega_N = \{z_1, ..., z_N\} \subset \widetilde{\Omega}$.

Proof. If μ is a global minimizer of I_K , by Theorem 3.1.7, the potential U_K^{μ} is constant on $\widetilde{\Omega}$, i.e. μ is *K*-invariant and has full support if viewed as an element of $\mathbb{P}(\widetilde{\Omega})$. Moreover, according to Lemma 3.1.11, the kernel *K* is positive definite (modulo a constant) on $\widetilde{\Omega}$. Therefore, the statement follows directly from Theorem 5.4.1 applied to $\widetilde{\Omega}$ in place of Ω .

We now turn to the proof of the generalized Stolarsky principle:

Proof of Theorem 5.4.1. Without loss of generality, we can assume that *K* is positive definite, since adding a constant to *K* does affects neither the invariance of μ nor the difference $I_K(\nu) - I_K(\mu)$. We can now use the crucial identity (3.10) of Lemma 3.2.3, as well as part (5) of Theorem 3.3.2, to obtain

$$I_{K}(\mathbf{v}) - I_{K}(\mu) = I_{K}(\mathbf{v} - \mu) = \int_{\Omega} \int_{\Omega} \int_{\Omega} K(x, y) d(\mathbf{v} - \mu)(x) d(\mathbf{v} - \mu)(y)$$
$$= \int_{\Omega} \int_{\Omega} \int_{\Omega} \int_{\Omega} k(x, z) k(z, y) d\mu(z) d(\mathbf{v} - \mu)(x) d(\mathbf{v} - \mu)(y)$$
$$= \int_{\Omega} \left| \int_{\Omega} k(x, z) d(\mathbf{v} - \mu)(x) \right|^{2} d\mu(z) = D_{L^{2}, k, \mu}^{2}(\mathbf{v}).$$
(5.38)

Remark: Observe that, for $k \in L^2(\Omega \times \Omega, \mu \times \mu)$, it is not technically obvious that the definition of $D^2_{L^2,k,\mu}(v)$ in (5.34) is properly justified: we do not know a priori that k is integrable with respect to v, only with respect to μ . (This problem does not occur in the discrete case since we know that $k(\cdot, z_i) \in L^2(\Omega, \mu)$ for each i = 1, ..., N.) However, the proof of Stolarsky principle (5.36) demonstrates that the L^2 discrepancy $D^2_{L^2,k,\mu}$ is well-defined for any Borel measure v. Indeed, the inner integral with respect to $d\mu(z)$ in (5.38) is defined according to part (5) of Theorem 3.3.2, and, moreover, produces the function K(x,y), which is continuous and therefore integrable with respect to the finite Borel measure sure $(v - \mu) \times (v - \mu)$ on $\Omega \times \Omega$. Hence Fubini's theorem applies and

$$\int_{\Omega} \int_{\Omega} \int_{\Omega} k(x,z)k(y,z)d(v-\mu)(x)d(v-\mu)(y) = \left| \int_{\Omega} k(x,z)d(v-\mu)(x) \right|^{2}$$

is finite for μ -a.e. z and is integrable with respect to $d\mu(z)$, i.e. $D^2_{L^2,k,\mu}(v)$ is well-defined.

Chapter 6

P-frame Energy

We now turn our attention to a specific family of energies: the p-frame energies, given by kernels of the form

$$F_p(\langle x, y \rangle) := |\langle x, y \rangle|^p \qquad p > 0 \tag{6.1}$$

on $\mathbb{S}_{\mathbb{F}}^{d-1}$, for $\mathbb{F} = \mathbb{R}, \mathbb{C}, \mathbb{H}$. Of course, since this energy does not depend on unitary transformations, the analysis of such energies naturally lends itself to the projective spaces \mathbb{FP}^{d-1} , where the *p*-frame energies correspond to taking kernels of the form (using the notation from Section 2.4, e.g. $\tau(x, y) = \cos(\vartheta(x, y))$)

$$F_p^*(\tau(x,y)) = \left(\frac{1+\tau(x,y)}{2}\right)^{\frac{p}{2}},\tag{6.2}$$

as, for any $w, z \in \mathbb{S}_{\mathbb{F}}^{d-1}$, we have

$$F_p^*(\tau(w\mathbb{F}, z\mathbb{F})) = F_p^*(2|\langle w, z \rangle|^2 - 1) = |\langle w, z \rangle|^p.$$

For $\mathbb{F} = \mathbb{R}$ or \mathbb{C} , the *p*-frame energies have a rich history. The case of p = 2 and $\mathbb{F} = \mathbb{R}$ was studied in [Sid74] and later again in [BF03], In the latter paper, it was proved that the minimizers of the discrete energy are precisely the finite unit norm tight frames (FUNTFs), resulting in a recent increase in interest of this energy. We will discuss this in greater detail

in Section 6.1.

In the case p = 4, the *p*-frame energy is closely connected to the maximal equiangular tight frames, which in the complex case are tight projective 2-designs, which are best known by their alternative name *symmetric informationally complete positive operatorvalued measures* (SIC-POVMs). These are unit norm tight frames $\{z_j\}_{j=1}^N$ with the property that $|\langle z_i, z_j \rangle|^2 = \frac{1}{d+2}$ or $\frac{1}{d+1}$ for $i \neq j$, in the real and complex case respectively. In \mathbb{C}^d , Zauner's conjecture [Zau11] states that SIC-POVMs exist in all dimensions $d \geq 2$, which is supported by extensive numerical evidence [SG10, RBKSC04]. In the real case, the existence of analogous objects (i.e. tight projective 2-designs) is also mysterious: they may exist only in dimensions $d = (2m - 1)^2 - 2$ [BD79, BD80, DGS77, LS73], but do not exist for infinitely many values of *m* [BMV04, Mak02]. When these objects do exist, they minimize the 4-frame energy.

More generally, for even integers p, these energies were considered in [Sid74, Wel74, Ven01], and it is known that for $\mathbb{F} = \mathbb{R}$ or \mathbb{C} projective k-designs minimize the p = 2k energy. In this terminology, FUNTFs are equivalent to projective 1-designs, while spherical 2-designs are exactly the FUNTFs with center of mass at the origin. Spherical 2-designs were constructively shown to exist for $d \ge 2$ precisely when the number of points N satisfies $N \ge d + 1$ and $N \ne d + 2$ when d is odd [Mim90], whereas FUNTFs of cardinality N exist for all $N \ge d$ [BF03]. Surface measure is also known to be a minimizer for $p \in 2\mathbb{N}$: this can be seen either from the definition of k-designs, or from the fact that the function F_p^* is positive definite in this case, and was originally proved in the real case in [Sid74].

Conversely, when $p \notin 2\mathbb{N}$, the situation is much less studied. The problem of minimizing general *p*-frame energies on $\mathbb{S}_{\mathbb{R}}^{d-1}$ was first posed by Ehler and Okoudjou in [EO12], where they also provided some initial results on both the discrete and continuous energies, in particular showing that distributing mass equally on the vertices of the cross-polytope yields the unique symmetric minimizer of the continuous *p*-frame energy for $p \in (0,2)$, up to orthogonal transformations. There have since been developments in determining minimizers of the discrete energy [CGG⁺20, GP20, XX21] and continuous energy [BGM⁺a, BGM⁺b], the latter of which will be the main focus of this chapter.

In addition to the connections and applications in other areas of mathematics, the *p*-frame energy is an interesting subject to study from a Potential Theoretical point of view. In the introduction of Chapter 4, we discussed how, in the Euclidean setting, the discreteness of minimizers follows from the potential *W* being mildly repulsive, i.e. that the potential, as a function of r = ||x - y||, behaves as $W(0) - Cr^{\gamma}$ for small *r*, with $\gamma > 2$ and *C* some positive constant. Since, on the sphere \mathbb{S}^{d-1} ,

$$|\langle x,y \rangle||^{p} = \left(1 - \frac{||x-y||^{2}}{2}\right)^{p} = \sum_{m=0}^{\infty} {p \choose m} \frac{(-r^{2})^{m}}{2^{m}} \approx 1 - \frac{pr^{2}}{2},$$

the *p*-frame potential corresponds to the to the endpoint case $\gamma = 2$ and thus is quite delicate. Indeed, we know for some values of *p*, there exist non-discrete minimizers, while for others all minimizers are discrete, as we will show in Section 6.2. On the sphere, for $p \notin 2\mathbb{N}$, the *p*-frame potentials also have Gegenbauer expansions with infinitely many positive and negative coefficients (see 6.7), so they fall outside the scope of Theorem 4.1.3. This means that there is no guarantee that discrete minimizers exist, though we conjecture that all minimizers are discrete in these cases (see Conjecture 6.2.5). Thus, characterizing the behavior of minimizers of the *p*-frame energy may provide some insights on minimizing energies with similar behaviors, something that the current theory is not equipped to handle.

We discuss the 2-frame energy, or simply "frame energy", in Section 6.1 and the more general *p*-frame energy in Section 6.2, in particular applying the results of Section 4.3 to show that in certain circumstances, tight designs are minimizers of the *p*-frame energy, and conjecture that for $p \notin 2\mathbb{N}$, all minimizers of $I_{F_p^*}$ are discrete. In Section 6.3, we show that the 600-cell is minimizer of certain *p*-frame energies as well, though it is not a tight design. While the discreteness of minimizers claimed in Conjecture 6.2.5 remains out of reach, we establish that the support of measures minimizing the *p*-frame energy I_{F_p} , with $p \notin 2\mathbb{N}$, on the real sphere must have empty interior in Section 6.4. Section 6.5 extends some of our results to the non-compact setting. In Section 6.6, we show a connection to Convex Geometry, applying our results to the problem of minimizing mixed volumes of convex bodies.

6.1 Frame Energy

For $\mathbb{F} = \mathbb{R}$, the 2-frame energy, or simply *frame energy*, with the potential $F_2(t) = t^2$, was studied in [Sid74] and later again in [BF03]. In the latter paper, which coined the name for this energy, Benedetto and Fickus studied the discrete frame energy E_{F_2} in order to characterize all finite unit norm tight frames.

Given a finite dimensional Hilbert space *H*, a sequence $\{z_j\}_{j=1}^N$ in *H* is called a *frame* for *H* if there exist constants $0 < A \le B < \infty$ such that for all $y \in H$

$$A||y||_{H}^{2} \leq \sum_{j=1}^{N} |\langle y, z_{j} \rangle|^{2} \leq B||y||_{H}^{2}.$$
(6.3)

A frame is *tight* if A = B in (6.3), and a tight frame is called a *unit norm tight frame* if $||z_j||_H = 1$ for $1 \le j \le N$. While it is possible to define frames for infinite-dimensional Hilbert spaces as well, in which case the frames may be infinite sequences, if one focuses on finite dimensional spaces, then it makes sense to only consider frames that are finite sequences, thus the term finite unit norm tight frames (FUNTFs).

Frames play an important role in signal processing and other branches of applied mathematics, as they act as overcomplete (or redundant) spanning sets of vectors that provide stable signal representations and allow modeling for noisy environments [BL98, BT93]. They were introduced in 1952 by Duffin and Schaeffer [DS52] as part of an on-going development of non-harmonic Fourier series, and have since been the subject of much research (see, e.g., [BCHL03, DGM86, SH03]). FUNTFs are of particular interest, as their characterization means that they provide an analogue of Parseval's Identity: If $\{z_j\}_{j=1}^N$ is a FUNTF, then there exists some A > 0 such that

$$y = \frac{1}{A} \sum_{j=1}^{N} \langle y, z_j \rangle z_j \qquad \forall y \in H.$$
(6.4)

This property has led to various applications (see, e.g., [CK03, Tyl87]), and connects FUNTFs to problems such as the Kadison-Singer conjecture [CFTW06]. For a more complete exposition on Frame Theory, see [Chr03].

In [BF03], Benedetto and Fickus provided a means of characterizing FUNTFs that connected them to energy minimization:

Theorem 6.1.1. Let $N \in \mathbb{N}$ and $\omega_N = \{z_1, ..., z_N\} \subset \mathbb{S}^{d-1}$. Then

- 1. Treating $E_{F_2}(\omega_N)$ as a function of $z_1, ..., z_N$, every local minimizer of the frame energy is a global minimizer.
- 2. If $N \leq d$, then

$$\mathscr{E}_{F_2}(\mathbb{S}^{d-1},N) = \frac{1}{N}$$

and the minimizers are exactly the orthonormal sequences in \mathbb{R}^d .

3. If $N \ge d$, *then*

$$\mathscr{E}_{F_2}(\mathbb{S}^{d-1}, N) = \frac{1}{d} \tag{6.5}$$

and the minimizers are exactly the N-element FUNFT's.

We now provide a few remarks about this theorem. The first is that it shows that for all $N \ge d$, a FUNTF, or a spherical {2}-design as defined in Section 4.2, of N points exists. Though less wide in scope than Theorem 2.6.1, this result has the advantage of giving an explicit value after which such objects exist, rather than just the order. In addition, part (1) is an interesting and unusual result. As discussed in Section 3.3 we know that whenever F is conditionally positive definite on \mathbb{S}^{d-1} , all local minimizers of I_F are in fact global minimizers. Such a result rarely holds in the discrete setting, where the definition of local minimizer is necessarily quite different, and is one aspect that makes discrete energy optimization so difficult.

Since the minimum in (6.5) is independent of N, the weak*-density of discrete probability measures shows that the result of Benedetto and Fickus also implies (6.6), below. Moreover, inequality (6.6) for arbitrary measures $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ was stated in [Sid74, EO12], and Ehler and Okoudjou also characterized all minimizers of the continuous frame energy, I_{F_2} , as probabilistic tight frames. The author and coauthor provided a new proof of these results in [BM19], as well as an alternate characterization the minimizers of I_{F_2} (Lemma 6.1.3), which uses the following definition.

Definition 6.1.2. A probability measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1} \text{ is called isotropic if its second moment}$ matrix is a multiple of the identity, i.e. $\left[\int_{\mathbb{S}^{d-1}} x_i x_j d\mu(x)\right]_{i,j=1}^d = c \mathbb{I}_d$, with $x_i = \langle x, e_j \rangle$ for $1 \leq j \leq d$ and $\{e_1, \dots, e_d\}$ and orthonormal basis of \mathbb{R}^d .

Lemma 6.1.3. *For any* $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ *, we have*

$$I_{F_2}(\mu) \ge \frac{1}{d}.\tag{6.6}$$

Equality is achieved precisely for isotropic measures.

Proof. We expand the square, throw away off-diagonal terms, and apply the Cauchy–Schwartz inequality:

$$I_{F_{2}}(\mu) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} (\langle x, y \rangle)^{2} d\mu(x) d\mu(y) = \sum_{i,j=1}^{d} \left(\int_{\mathbb{S}^{d-1}} x_{i}x_{j} d\mu(x) \right)^{2}$$
$$\geq \sum_{i=1}^{d} \left(\int_{\mathbb{S}^{d-1}} x_{i}^{2} d\mu(x) \right)^{2} \geq \frac{1}{d} \left(\sum_{i=1}^{d} \int_{\mathbb{S}^{d-1}} x_{i}^{2} d\mu(x) \right)^{2} = \frac{1}{d}.$$

From this, the characterization of the minimizers follows immediately.

6.2 P-frame Energy

When p = 2k and $\Phi = \mathbb{FP}^{d-1}$ ($\mathbb{F} = \mathbb{R}, \mathbb{C}$, or \mathbb{H}), we have that $F_{2k}^*(t) = 2^{-k} \cdot (1+t)^k$ is a polynomial. It is standard to check that this polynomial is positive definite on Ω : this could be done by checking that the coefficients in its Jacobi expansion are non-negative, but it would be perhaps simpler to prove it as follows. Observe that, since $C_0^{(\alpha,\beta)}(t) = 1$ and $C_1^{(\alpha,\beta)}(t) = \frac{\alpha-\beta}{2(\alpha+1)} + \frac{\alpha+\beta+2}{2(\alpha+1)} \cdot t$, we have that

$$1+t=\frac{2(\alpha+1)}{(\alpha+\beta+2)}C_1^{(\alpha,\beta)}(t)+\frac{2(\beta+1)}{\alpha+\beta+2}C_0^{(\alpha,\beta)}(t).$$

Since $\alpha + 1 = \frac{d-1}{2} \cdot \dim_{\mathbb{R}}(\mathbb{F}) > 0$ and $\beta + 1 = \frac{1}{2} \cdot \dim_{\mathbb{R}}(\mathbb{F}) > 0$, we see that the function 1 + t is positive definite on Ω . It easily follows from repeated application of Schur's Theorem that $F_{2k}^*(t) = 2^{-k} \cdot (1+t)^k$ is positive definite on Φ , and therefore $I_{F_{2k}^*}$ is minimized by the surface measure η . This also means that on the sphere $\mathbb{S}_{\mathbb{F}}^{d-1}$, F_{2k} is positive definite and minimized by the normalized uniform surface measure on $\mathbb{S}_{\mathbb{F}}^{d-1}$, which we denote $\sigma_{\mathbb{F}}$.

The minimal values of the p = 2k energy, in either the spherical or projective setting, may be expressed in elementary functions for each \mathbb{F} . The values for the real case were given by Sidel'nikov in [Sid74] and all values were determined by Shatalov in [Sha01].

Proposition 6.2.1. *Let* $k \in \mathbb{N}$ *. Then*

$$\begin{aligned} \mathscr{I}_{F_{2k}}(\mathbb{S}_{\mathbb{R}}^{d-1}) &= \mathscr{I}_{F_{2k}^{*}}(\mathbb{R}\mathbb{P}^{d-1}) = \frac{1 \cdot 3 \cdot 5 \dots (2k-1)}{d \cdot (d+2) \dots (d+2(k-1))}, \\ \mathscr{I}_{F_{2k}}(\mathbb{S}_{\mathbb{C}}^{d-1}) &= \mathscr{I}_{F_{2k}^{*}}(\mathbb{C}\mathbb{P}^{d-1}) = 1/\binom{d+k-1}{k}, \\ \mathscr{I}_{F_{2k}}(\mathbb{S}_{\mathbb{H}}^{d-1}) &= \mathscr{I}_{F_{2k}^{*}}(\mathbb{H}\mathbb{P}^{d-1}) = (k+1)/\binom{2d+k-1}{k}. \end{aligned}$$

When p is not an even integer, the p-frame energies are not positive definite. While this could be checked by showing that the Jacobi expansion of F_p^* has negative coefficients in each setting, we may show this without these computations. The kernel F_p^* is clearly positive definite on \mathbb{FP}^{d-1} if and only if F_p is positive definite on $\mathbb{S}^{d-1}_{\mathbb{F}}$. Since $\mathbb{S}^{d-1}_{\mathbb{H}}$ and $\mathbb{S}^{d-1}_{\mathbb{C}}$ each contain a copy of $\mathbb{S}^{d-1}_{\mathbb{R}}$, it is enough to show that F_p is not positive definite on the real sphere, which follows from its Gegenbauer expansion. Since it is an even function, all of F_p 's odd coefficients are zero, so we need only calculate the even ones.

Lemma 6.2.2. Let $k \in \mathbb{N}$, $p \in (0, \infty)$ and $\lambda = \frac{d-2}{2}$. Then

$$\widehat{F}_p(2k,\lambda) = \frac{\Gamma(\lambda+1)\Gamma(\frac{p+1}{2})}{\sqrt{\pi}\Gamma(k+\frac{p}{2}+\lambda+1)} \prod_{l=0}^{k-1} \left(\frac{p}{2}-j\right).$$
(6.7)

In particular, the signs of the even coefficients alternate in sign for k sufficiently large.

Proof. Let us define, for all $n \in \mathbb{N}_0$, p > 0 and $\lambda > \frac{-1}{2}$

$$G(n,\lambda,p) = \int_{0}^{1} t^{p} C_{n}^{\lambda}(t) (1-t^{2})^{\lambda-\frac{1}{2}} dt.$$
 (6.8)

From (2.26) and the fact that F_p is even, we see that

$$\widehat{F}_p(2k,\lambda) = \frac{\Gamma(\lambda+1)G(2k,\lambda,p)}{\sqrt{\pi}\Gamma(\lambda+\frac{1}{2})C_{2k}^{\lambda}(1)}.$$
(6.9)

We claim that for all $k \in \mathbb{N}_0$,

$$G(2k,\lambda,p) = \frac{2^{k-1}\Gamma(\lambda+k+\frac{1}{2})\Gamma(\frac{p+1}{2})}{(2k)!\Gamma(k+\frac{p}{2}+\lambda+1)} \left(\prod_{l=0}^{k-1}(\lambda+l)\right) \left(\prod_{j=0}^{k-1}(p-2j)\right)$$
(6.10)

and

$$G(2k+1,\lambda,p) = \frac{2^{k}\Gamma(\lambda+k+\frac{1}{2})\Gamma(\frac{p}{2}+1)}{(2k+1)!\Gamma(k+\frac{p}{2}+\lambda+\frac{3}{2})} \left(\prod_{l=0}^{k} (\lambda+l)\right) \left(\prod_{j=0}^{k-1} (p-2j-1)\right).$$
(6.11)

A quick computation shows that these hold for k = 0.

Now assume that (6.10) and (6.11) hold for some $k \ge 0$. Using the recurrence relation

$$C_{n}^{\lambda}(t) = \frac{2(n+\lambda-1)}{n} t C_{n-1}^{\lambda}(t) - \frac{n+2\lambda-2}{n} C_{n-2}^{\lambda}(t)$$
(6.12)

for $n \ge 2$, we find that

$$\begin{split} G(2k+2,\lambda,p) &= \frac{2(2k+\lambda+1)}{2k+2} G(2k+1,\lambda,p+1) - \frac{2k+2\lambda}{2k+2} G(2k,\lambda,p) \\ &= \frac{1}{2k+2} \Biggl[\frac{2^{k-1} \Gamma(\lambda+k+\frac{1}{2}) \Gamma(\frac{p+1}{2})}{(2k+1)! \Gamma(k+\frac{p}{2}+\lambda+2)} \Bigl(\prod_{l=0}^{k-1} (\lambda+l) \Bigr) \Bigl(\prod_{j=0}^{k-1} (p-2j) \Bigr) \Biggr] \\ &\times \Biggl[4(2k+\lambda+1) (\frac{p+1}{2}) (\lambda+k) - (2k+1) (2k+2\lambda) (k+\frac{p}{2}+\lambda+1) \Biggr] \\ &= \Biggl[\frac{2^{k-1} \Gamma(\lambda+k+\frac{1}{2}) \Gamma(\frac{p+1}{2})}{(2k+2)! \Gamma(k+\frac{p}{2}+\lambda+2)} \Bigl(\prod_{l=0}^{k-1} (\lambda+l) \Bigr) \Bigl(\prod_{j=0}^{k-1} (p-2j) \Bigr) \Biggr] \\ &\times \Biggl[2(\lambda+k+\frac{1}{2}) (\lambda+k) (p-2k) \Biggr] \\ &= \frac{2^k \Gamma(\lambda+k+\frac{3}{2}) \Gamma(\frac{p+1}{2})}{(2k+2)! \Gamma(k+\frac{p}{2}+\lambda+2)} \Bigl(\prod_{l=0}^k (\lambda+l) \Bigr) \Bigl(\prod_{j=0}^k (p-2j) \Bigr) \end{split}$$

and so

$$\begin{split} G(2k+3,\lambda,p) &= \frac{2(2k+\lambda+2)}{2k+3} G(2k+2,\lambda,p+1) - \frac{2k+2\lambda+1}{2k+3} G(2k+1,\lambda,p) \\ &= \frac{1}{2k+3} \Biggl[\frac{2^k \Gamma(\lambda+k+\frac{1}{2}) \Gamma(\frac{p}{2}+1)}{(2k+2)! \Gamma(k+\frac{p}{2}+\lambda+\frac{5}{2})} \Bigl(\prod_{l=0}^k (\lambda+l) \Bigr) \Bigl(\prod_{j=0}^{k-1} (p-2j-1) \Bigr) \Biggr] \\ &\quad \times (2k+2\lambda+1) \Biggl[(2k+\lambda+2)(p+1) - (2k+2)(k+\frac{p}{2}+\lambda+\frac{3}{2}) \Biggr] \\ &= \Biggl[\frac{2^k \Gamma(\lambda+k+\frac{1}{2}) \Gamma(\frac{p}{2}+1)}{(2k+3)! \Gamma(k+\frac{p}{2}+\lambda+\frac{5}{2})} \Bigl(\prod_{l=0}^k (\lambda+l) \Bigr) \Bigl(\prod_{j=0}^{k-1} (p-2j-1) \Bigr) \Biggr] \\ &\quad \times 2(k+\lambda+\frac{1}{2})(k+\lambda+1)(p-2k-1) \\ &= \frac{2^{k+1} \Gamma(\lambda+k+\frac{3}{2}) \Gamma(\frac{p}{2}+1)}{(2k+3)! \Gamma(k+\frac{p}{2}+\lambda+\frac{5}{2})} \biggl(\prod_{l=0}^{k-1} (\lambda+l) \biggr) \biggl(\prod_{j=0}^k (p-2j-1) \Bigr) \Biggr] \end{split}$$

Thus, inductively, we have that (6.10) and (6.11) hold for all $k \in \mathbb{N}_0$. From (6.9) and (6.10), we can now compute $\widehat{F}_p(2k,\lambda)$, giving us our claim.

Thus, the p-frame energy is not positive definite when p is not even, so the uniform measure is not a minimizer. Moreover, by applying our linear programming results, we find that in certain instances, only discrete minimizers exist.

Theorem 6.2.3. With $F_p(t) = |t|^p$ for $t \in [-1, 1]$, we have the following.

1. Suppose there exists a tight spherical (2t+1)-design $\mathscr{C} \subset \mathbb{S}^{d-1}$, then the measure

$$\mu_{\mathscr{C}} = rac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} \delta_x$$

is a minimizer of the p-frame energy I_{F_p} with $2t - 2 \le p \le 2t$ over $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$.

2. Let $\mathbb{F} = \mathbb{R}$, \mathbb{C} or \mathbb{H} . Assume that there exists a tight projective t-design $\widetilde{\mathscr{C}} \subset \mathbb{FP}^{d-1}$, and let the code $\mathscr{C} \subset \mathbb{S}^{d-1}_{\mathbb{F}}$ consist of the representers of $\widetilde{\mathscr{C}}$ in $\mathbb{S}^{d-1}_{\mathbb{F}}$ according to (2.16). Then the measure

$$\mu_{\mathscr{C}} = rac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} \delta_x$$

is a minimizer of the p-frame energy I_{F_p} with $2t - 2 \le p \le 2t$ over $\mu \in \mathbb{P}(\mathbb{S}^{d-1}_{\mathbb{F}})$. Furthermore, if the inequalities are strict, then $\mu \in \mathbb{P}(\mathbb{S}^{d-1}_{\mathbb{F}})$ is a minimizer of I_{F_p} only if the push-forward of μ , $(p_{\mathbb{F}})_*\mu$ is a tight projective t-design.

3. If $p \notin 2\mathbb{N}$, then the minimizers of I_{F_p} on the circle \mathbb{S}^1 are discrete.

Proof. We observe that part (3) is a special case of part (1), as on the circle \mathbb{S}^1 , there exists a tight (2t + 1)-design for all $t \in \mathbb{N}_0$: the vertices of a regular (2t + 2)-gon. In addition, part (1) is itself essentially contained in part (2) with $\mathbb{F} = \mathbb{R}$: indeed, odd-strength tight spherical designs are necessarily symmetric, and by taking one point in each antipodal pair one obtains a tight projective design, as discussed in Section 2.6. Since tight projective t-designs on \mathbb{RP}^1 exist for all $t \in \mathbb{N}$ (i.e. t + 1 equally spaced points on \mathbb{RP}^1), part 2 fully characterizes the minimizers of the *p*-frame energy. Thus, it is enough to prove part 2, which follows from Theorem 4.3.1 and the fact that $F_p^*(t) = (\frac{1+t}{2})^{\frac{p}{2}}$ is strictly absolutely monotonic of degree $\lceil p/2 \rceil$ and has as its first nonpositive derivative $(F_p^*)^{(\lceil p/2 \rceil+1)}(t), -1 < t < 1$. If *p* is not an even integer then this derivative is in fact negative, so Theorem 4.3.12 applies, finishing the proof.

Minimizing the continuous energy over all probability measures and obtaining discrete minimizers allows us to make new conclusions about the minimizing configurations of the discrete p-frame energies for certain values of the cardinality N. One directly obtains the following corollary:

Corollary 6.2.4. Let \mathbb{F} , d, p, and $\mathscr{C} = \{x_1, ..., x_N\}$ be as in any of the parts of Theorem 6.2.3. Let $k \in \mathbb{N}$ and $\omega_{kN} = \{z_1, ..., z_{kN}\} \subset \mathbb{S}_{\mathbb{F}}^{d-1}$ such that for $0 \le j \le k-1$ and $1 \le m \le N$, $z_{kN+m} = x_m$. Then N-point discrete p-frame energy is minimized by ω_{kN} , i.e. the configuration \mathscr{C} repeated k times. In other words

$$E_{F_n}(\boldsymbol{\omega}_{kN}) = \mathscr{E}_{F_n}(\mathbb{S}_{\mathbb{F}}^{d-1}, N).$$
(6.13)

Thus, for example, if *N* is a multiple of 6, then repeated copies of a "half" of the icosahedron minimize the *N*-point *p*-frame energy on \mathbb{S}^2 for $p \in [2,4]$.

Extensive numerical experiments, the results of which were compiled and discussed in $[BGM^+b]$, together with Theorem 6.2.3, the fact that the *p*-frame energy is almost mildly repulsive, and the lack of positive definiteness leads us to the following conjecture:

Conjecture 6.2.5 ([BGM⁺b]). For all $d \ge 2$ and p > 0 such that $p \notin 2\mathbb{N}$, the minimizing measures of $I_{F_p^*}$ on \mathbb{FP}^{d-1} are discrete.

While we have yet to prove this conjecture, in Section 6.4, we show that on \mathbb{S}^{d-1} , whenever *p* is not even, the support of any minimizer of the *p*-frame energy I_{F_p} necessarily has empty interior.

6.3 Optimality of the 600-cell

In addition to tight designs, the vertices of the 600-cell also provide a minimizer for certain p-frame energies, specifically those on \mathbb{S}^3 for 8 . The 600-cell is one of the six 4-dimensional convex regular polytopes; it has 600 tetrahedral faces, which explains the origin of its name. When its 120 vertices are identified with unit quaternions, they give a representation of the elements of a group known as the binary icosahedral group [Sti01].

As discussed above, optimization of *p*-frame energy I_{F_p} on the sphere \mathbb{S}^3 is equivalent to optimization of the expression

$$I_{F_p^*}(\mu) = \int_{\mathbb{RP}^3} \int_{\mathbb{RP}^3} F_p^*(\tau(x, y)) d\mu(x) d\mu(u)$$

over probability measures μ on \mathbb{RP}^3 , where the kernel F_p^* is given by

$$F_p^*(t) = \left(\frac{1+t}{2}\right)^{\frac{p}{2}}.$$

We therefore assume for the rest of this section the underlying space to be \mathbb{RP}^3 , and use the corresponding normalized Jacobi polynomials $C_n^{(-1/2,1/2)}(t)$. Following the approach in Section 4.3, we will establish a sequence of inequalities similar to (4.13).

The 600-cell is only a projective 5-design and therefore not tight. The authors in [CK07], motivated by an approach found in the paper [And99], found means to prove universal optimality of the 600-cell by using a higher degree interpolating polynomial. The 600-cell has the notable property that the averages of the 7th, 8th, and 9th degree Jacobi polynomials vanish over it, although this is not the case for 6th degree Jacobi polynomial. This allows for constructing a degree 8 polynomial h which is less than or equal to F_p^* , positive definite, and agrees with F_p^* at the distances appearing in the 600-cell, and which finally has the property that its 6th Jacobi coefficient vanishes.

For a polynomial *h* of the form,

$$h = \sum_{\substack{n=0\\n\neq 6}}^{8} \widehat{h}_n C_n^{(1/2, -1/2)}(t), \tag{6.14}$$

the coefficients \hat{h}_n can be uniquely determined as functions of p by setting

$$h(t_i) = f(t_i),$$
 $1 \le i \le 5$
 $h'(t_i) = f'(t_i),$ $2 \le i \le 4,$

where $-1 = t_1 < t_2 < ... < t_5 = 1$ are the elements of $\mathscr{A}(\mathscr{C})$, for \mathscr{C} the projective 600-cell. It turns out that for all $p \in [8, 10]$, the coefficients $\hat{h}_n(p)$ are nonnegative when $0 \le n \le 8$, $n \ne 6$. This was shown through a computer-assisted approach carried out by Glazyrin, Park, and Vlasiuk; specifically, using interval arithmetic, we compute values of $\hat{h}_n(p)$ on a grid fine enough to guarantee that $\hat{h}_n(p) \ge 0$. The details of these computations are available in the auxiliary files of the arXiv submission [BGM⁺b].

Lemma 6.3.1. If $p \in [8, 10]$ and the polynomial h is constructed as above, the coefficients \hat{h}_n in the Jacobi expansion (6.14) satisfy $\hat{h}_n(p) \ge 0$.

Using this fact we show optimality of the 600-cell on the range $p \in [8, 10]$.

Theorem 6.3.2. For $p \in [8, 10]$, the 600-cell minimizes the *p*-frame energy $I_{F_p^*}$ over Borel probability measures on \mathbb{RP}^3 .

Note that this also means that the 600-cell is a minimizer of I_{F_p} on the sphere \mathbb{S}^3 .

Proof. Let $F_p^*(t) = \left(\frac{t+1}{2}\right)^{p/2}$ for some $8 , <math>t_1 = -1$, $t_2 = \frac{-\sqrt{5}-1}{4}$, $t_3 = -\frac{1}{2}$, $t_4 = \frac{\sqrt{5}-1}{4}$, and $t_5 = 1$. Let h(t) be the 8th degree polynomial given by (6.14), such that $h(t_i) = p(t_i)$ for $1 \le i \le 5$, and $h'(t_i) = p'(t_i)$ for $2 \le i \le 4$. By Lemma 6.3.1, the coefficients \hat{h}_n are non-negative for $p \in [8, 10]$.

Let $g(t) = (t^2 - 1) \prod_{i=2}^{4} (t - t_i)^2$ and $\tilde{h}(t) = H[F_p^*(t), g](t)$. Then we also have $\tilde{h}(t) = H[h, g](t)$. By Lemma 4.3.6, this gives that for all $t \in [-1, 1]$ there exist some $\xi, s \in (\min(t, t_1), \max(t, t_5))$ such that

$$F_p^*(t)(t) - \tilde{h}(t) = \frac{(F_p^*)^{(8)}(\xi)}{8!}g(t) \ge 0,$$

and

$$h(t) - \tilde{h}(t) = \frac{h^{(8)}(v)}{8!}g(t) \le 0.$$

We thus have $F_p^*(t)(t) - h(t) = F_p^*(t)(t) - \tilde{h}(t) + \tilde{h}(t) - h(t) \ge 0$. Since h(t) is positive definite and $\hat{h}_6 = 0$, for the 600-cell \mathscr{C} , we have the following sequence of inequalities

$$I_{F_p^*(t)}(\boldsymbol{\mu}) \geq I_h(\boldsymbol{\mu}) \geq I_h(\boldsymbol{\sigma}) = I_h(\boldsymbol{\mu}_{\mathscr{C}}) = I_F(\boldsymbol{\mu}_{\mathscr{C}}),$$

implying that equally weighted vertices of \mathscr{C} minimize *p*-frame energy.

6.4 Empty Interior of *p*-frame Energy Minimizers

Outside of some specific cases covered by Theorem 6.2.3, Conjecture 6.2.5, which states that the minimizers of the *p*-frame energy with $p \notin 2\mathbb{N}$ on the real sphere (and on projective spaces) are necessarily discrete, remains open. In the present section, we prove a weaker statement for the real sphere: namely that the support of every minimizer of such energies has empty interior.

Theorem 6.4.1. Let $F_p(t) = |t|^p$, for $p \in \mathbb{R}_+ \setminus 2\mathbb{N}$. Let μ be a minimizer of the p-frame energy I_{F_p} . Then $\operatorname{supp}(\mu)$ has empty interior.

While we proved a similar result in Theorem 4.2.1, it is clear that when $p \notin 2\mathbb{N}$, the *p*-frame potential (given on the sphere or on a projective space) is not analytic, meaning that result does not apply. A similar result to Theorem 6.4.1 was proven in [FS13] for the Causal

Variational Principle, given by the kernel (4.19), on \mathbb{S}^2 . While our approach is inspired by theirs and the main line of reasoning follows an analogous path, specific constructions and arguments in the proofs of Propositions 6.4.2 and 6.4.3 below are much more peculiar and significantly more involved in the case of the *p*-frame energy.

The proof of Theorem 6.4.1 is based on two properties of interior points of $supp(\mu)$. The following statements hold:

Proposition 6.4.2. Let $p \in \mathbb{R}_+ \setminus 2\mathbb{N}$, $F_p(t) = |t|^p$, and μ be a minimizer of I_{F_p} . Then for $z \in (\operatorname{supp}(\mu))^\circ$,

$$\operatorname{supp}(\boldsymbol{\mu}) \cap \boldsymbol{z}^{\perp} = \boldsymbol{\emptyset}.$$

Proposition 6.4.3. Let the same conditions as in Proposition 6.4.2 hold. Then for $z \in (\operatorname{supp}(\mu))^{\circ}$,

$$\operatorname{supp}(\mu) \cap z^{\perp} \neq \emptyset.$$

Since these two statements are clearly mutually exclusive whenever $(\operatorname{supp}(\mu))^{\circ}$ is nonempty, their validity proves Theorem 6.4.1, i.e. that there are no interior points in the support of a minimizer. The remainder of this section is dedicated to the proof of these propositions.

We now sketch the argument for the first proposition. In short, the idea of the proof is the following: Let $z \in (\operatorname{supp}(\mu))^{\circ}$ and assume that there exists a point $y \in \operatorname{supp}(\mu)$ such that $\langle y, z \rangle = 0$. We shall construct a finite set of points $X = \{x_i\}_{i=1}^N \subset \operatorname{supp}(\mu)$, such that the matrix $[F(\langle x_i, x_j \rangle)]_{i,j}$ is not positive semidefinite, thus violating Lemma 3.1.11. The set X will consist of the points z, y, and a number (depending on p) of points equidistantly spaced around z on the great circle connecting y and z. We now make this precise.

Proof of Proposition 6.4.2. Fix *z* in the interior of supp(μ) and let $y \in \mathbb{S}^{d-1}$ be any point such that $\langle y, z \rangle = 0$. Setting $k \in \mathbb{N}$ so that $2k - 2 , we shall construct a set <math>\{x_0, \ldots, x_{N-1}\}$ of N = 2k + 2 points, all of which lie on the great circle connecting *z* and *y*. The points x_0, \ldots, x_{2k} are chosen in such a way that the angle between x_j and *z* is $(j - k)\varepsilon$

for some small $\varepsilon > 0$. Thus $x_k = z$, and the points x_0 and x_{2k} make angles $-k\varepsilon$ and $k\varepsilon$ with z, respectively. Observe that when ε is small enough, all of these points x_0, \ldots, x_{2k} belong to supp (μ) , since z is an interior point. Finally, we set $x_{2k+1} = y$. Then the angle between $x_{2k+1} = y$ and x_j , $j = 0, \ldots, 2k$, is $\frac{\pi}{2} - (j-k)\varepsilon$. In order to apply Lemma 3.1.11, we consider the matrix $A = [F_p(\langle x_i, x_j \rangle)]_{i,j=0}^{2k+1}$.

We will show that the matrix A is not positive semidefinite. To this end, we first construct an auxiliary vector $v \in \mathbb{R}^{2k+1} \setminus \{0\}$ such that for $m \in \{0, 1, ..., 2k-1\}$,

$$\sum_{j=0}^{2k} j^m v_j = 0, (6.15)$$

i.e. this vector must be in the (right) kernel of the Vandermonde matrix

$$\begin{pmatrix} 1 & 1 & 1 & 1 & \cdots & 1 \\ 0 & 1 & 2 & 3 & \cdots & 2k \\ 0 & 1 & 2^2 & 3^2 & \cdots & (2k)^2 \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 1 & 2^{2k-1} & 3^{2k-1} & \cdots & (2k)^{2k-1} \end{pmatrix}$$

We can take the entries of *v* to be

$$v_j = \prod_{\substack{i=0\\i\neq j}}^{2k} \frac{1}{j-i} = \frac{(-1)^j}{(2k-j)!j!}.$$
(6.16)

Such a vector can be seen to be in the kernel of the matrix above by use of the formula for the inverse of the square Vandermonde matrix (see Ex. 40 on page 38 of [Knu97]).

Consider a vector $u = [av_0, av_1, ..., av_{2k}, b]^T \in \mathbb{R}^{2k+2}$, where $a, b \in \mathbb{R}$. Then we have

$$\langle Au, u \rangle = a^2 \left(\sum_{i,j=0}^{2k} v_i v_j F_p(\langle x_i, x_j \rangle) \right) + 2ab \left(\sum_{j=0}^{2k} v_j F_p(\langle x_{2k+1}, x_j \rangle) \right) + b^2.$$
(6.17)

We shall show that the real numbers a and b can be chosen in such a way that the expression

above is negative, for ε sufficiently small.

Observe that for i, j = 0, ..., 2k we have

$$F_p(\langle x_i, x_j \rangle) = \cos^p((i-j)\varepsilon).$$

Since $\cos^{p}(t)$ is even, smooth near zero, and $\cos^{p}(0) = 1$, we can use its Taylor expansion to estimate the first term of (6.17) as follows

$$\begin{split} \sum_{i,j=0}^{2k} v_i v_j F_p(\langle x_i, x_j \rangle) &= \sum_{i,j=0}^{2k} v_i v_j \cos^p \left((i-j) \varepsilon \right) \\ &= \sum_{i,j=0}^{2k} v_i v_j \left(1 + \sum_{m=1}^{2k-1} c_m \varepsilon^{2m} (i-j)^{2m} + O(\varepsilon^{4k}) \right) \\ &= \left(\sum_{j=0}^{2k} v_j \right) \left(\sum_{i=0}^{2k} v_i \right) + \sum_{m=1}^{2k-1} c_m \varepsilon^{2m} \left(\sum_{i,j=0}^{2k} v_i v_j (i-j)^{2m} \right) + O(\varepsilon^{4k}) \\ &= \sum_{m=1}^{2k-1} c_m \varepsilon^{2m} \sum_{i,j=0}^{2k} v_i v_j \sum_{l=0}^{2m} \binom{2m}{l} i^l j^{2m-l} + O(\varepsilon^{4k}) \\ &= \sum_{m=1}^{2k-1} c_m \varepsilon^{2m} \sum_{l=0}^{2m} \binom{2m}{l} \left(\sum_{i=0}^{2k} v_i i^l \right) \left(\sum_{j=0}^{2k} v_j j^{2m-l} \right) + O(\varepsilon^{4k}) \\ &= O(\varepsilon^{4k}), \end{split}$$
(6.18)

where we have used the fact that for all values of l = 0, 1, ..., 2m, either $l \le 2k - 1$ or $2m - l \le 2k - 1$.

We now turn to the second term of (6.17). Observe that for j = 0, ..., 2k we have

$$F_p(\langle x_{2k+1}, x_j \rangle) = F_p(\langle y, x_j \rangle) = \cos^p\left(\frac{\pi}{2} - (j-k)\varepsilon\right) = \left|\sin\left((j-k)\varepsilon\right)\right|^p.$$

We then find that

$$\sum_{j=0}^{2k} v_j F_p(\langle y, x_j \rangle) = \sum_{j=0}^{2k} v_j \sin^p (|k-j|\varepsilon) = \sum_{j=0}^{2k} v_j (|k-j|\varepsilon + O(\varepsilon^3))^p$$

$$= \sum_{j=0}^{2k} v_j (|k-j|\varepsilon)^p (1+O(\varepsilon^2))^p = \varepsilon^p \sum_{j=0}^{2k} v_j |k-j|^p + O(\varepsilon^{p+2}).$$
(6.19)

We now analyze the coefficient of ε^p in the above expression using (6.16)

$$\sum_{j=0}^{2k} v_j |k-j|^p = \sum_{j=0}^{2k} (-1)^j \frac{|k-j|^p}{(2k-j)!j!} = 2\sum_{j=0}^{k-1} (-1)^j \frac{(k-j)^p}{(2k-j)!j!}.$$
(6.20)

Since the right-hand side of (6.20) is a sum of k exponential functions of p, we know that $\sum_{j=0}^{k-1} (-1)^j \frac{(k-j)^p}{(2k-j)!j!}$ has at most k-1 zeros, see e.g. Ex. 75 from [PS76, pg. 46]. We will show that these zeros are exhausted by the even integer values p = 2, 4, ..., 2k-2. Indeed, assume indirectly that

$$2\sum_{j=0}^{k-1} (-1)^j \frac{(k-j)^p}{(2k-j)! j!} = b \neq 0$$

for some even integer 0 . Then according to (6.17), (6.18), and (6.19) we have

$$\langle Au, u \rangle = a^2 O(\varepsilon^{4k}) + 2ab (b\varepsilon^p + O(\varepsilon^{p+2})) + b^2.$$

Since p < 2k, for ε sufficiently small, the discriminant of this quadratic form is positive, hence we can choose *a* and *b* so that $\langle Au, u \rangle < 0$. However, since $F_p(t) = |t|^p$ is a positive definite function on \mathbb{S}^{d-1} for any even integer *p*, this is a contradiction, as the matrix *A* must be positive semidefinite for any collection $\{x_i\}$. Therefore

$$2\sum_{j=0}^{k-1} (-1)^j \frac{(k-j)^p}{(2k-j)! j!} = 0$$

for all $p \in \{2, 4, ..., 2k - 2\}$. Since there are at most k - 1 zeros of this function, we then

know that

$$b_p := 2\sum_{j=0}^{k-1} (-1)^j \frac{(k-j)^p}{(2k-j)! j!} \neq 0$$

for all other values of p. Let $p \in (0, 2k) \setminus \{2, 4, ..., 2k - 2\}$. Then

$$\langle Au, u \rangle = a^2 O(\varepsilon^{4k}) + 2ab \left(b_p \varepsilon^p + O(\varepsilon^{p+2}) \right) + b^2,$$

and by the previous argument, for ε sufficiently small, we could choose a and b so that $\langle Au, u \rangle < 0$, i.e. A is not positive definite. Thus, according to Lemma 3.1.11, $\{x_0, ..., x_{2k}, y\}$ is not a subset of supp (μ) . Since, by assumption, for small $\varepsilon > 0$ the points $x_0, x_1, ..., x_{2k}$ all lie in a neighborhood of z and hence in supp (μ) , this implies that $y \notin \text{supp}(\mu)$ and so $\text{supp}(\mu) \cap z^{\perp} = \emptyset$.

We would like to make the following remark. Observe that for $p \notin 2\mathbb{N}$, the number of points used to disprove positive definiteness of $F_p(t) = |t|^p$ in the argument above is of the order p. A restriction of this type is actually necessary. Indeed, according to the result of Fitzgerald and Horn [FH77], for any positive definite matrix $C = [c_{ij}]_{i,j=1}^N$ with non-negative entries $c_{ij} \ge 0$, its Hadamard powers $C^{(a)} = [c_{ij}^a]_{i,j=1}^N$ are also positive definite when $a \ge N - 2$. Let $G = [\langle x_i, x_j \rangle]_{i,j=1}^N$ be the Gram matrix of the set $X = \{x_1, \dots, x_N\} \subset \mathbb{S}^{d-1}$. Since the matrix $G^{(2)} = [|\langle x_i, x_j \rangle|^2]_{i,j=1}^N$ is positive definite and has non-negative entries, we have that the matrix

$$G^{(p)} = \left[|\langle x_i, x_j \rangle|^p \right]_{i,j=1}^N = \left(G^{(2)} \right)^{(p/2)}$$

is positive definite whenever $p/2 \ge N-2$. Therefore, to obtain a non-positive definite matrix $G^{(p)}$, we must take $N \ge 2 + p/2$ points.

We now prove Proposition 6.4.3, completing the proof of Theorem 6.4.1.

The proof of Proposition 6.4.3. Suppose a neighborhood of a point $z \in \mathbb{S}^{d-1}$ is contained in the support of μ . We shall demonstrate that $\operatorname{supp}(\mu)$ must intersect the hyperplane z^{\perp} .

Let us assume the contrary, i.e. $supp(\mu) \cap z^{\perp} = \emptyset$. We may move all the mass of μ to

the hemisphere centered at z by defining a new measure $\mu_z \in \mathbb{P}(\mathbb{S}^{d-1})$:

$$\mu_{z}(E) = \begin{cases} \mu(-E \cup E), & \text{if } E \subseteq \{x \in \mathbb{S}^{d-1} : \langle z, x \rangle > 0\}, \\ \mu(E), & \text{if } E \subseteq z^{\perp}, \\ 0, & \text{if } E \subseteq \{x \in \mathbb{S}^{d-1} : \langle z, x \rangle < 0\}. \end{cases}$$

As our discussion in Section 2.4 about kernels that depend on $|\langle x, y \rangle|$ shows,

$$I_{F_p}(\mu) = I_{F_p}(\mu_{U(\mathbb{R})}) = I_{F_p}((\mu_z)_{U(\mathbb{R})}) = I_{F_p}(\mu_z),$$

so that μ_z is also a minimizer.

Since $\operatorname{supp}(\mu) \cap z^{\perp} = \emptyset$, we also have that $\operatorname{supp}(\mu_z) \cap z^{\perp} = \emptyset$, i.e. $\operatorname{supp}(\mu_z) \subset \{x \in \mathbb{S}^{d-1} : \langle z, x \rangle > 0\}$. Compactness of the support of μ_z then implies that it is separated from z^{\perp} , i.e. for some $\delta > 0$ we have $\langle y, z \rangle > \delta$ for each $y \in \operatorname{supp}(\mu_z)$. Let us choose an open neighborhood U_z of z, small enough so that $U_z \subseteq \operatorname{supp}(\mu_z)$ and so that for each $x \in U_z$ and each $y \in \operatorname{supp}(\mu_z)$, $\langle y, x \rangle > \delta > 0$.

We can now write the potential of μ_z at the point $x \in U_z$ as

$$U_{F_p}^{\mu_z}(x) = \int_{\mathbb{S}^{d-1}} |\langle x, y \rangle|^p \, d\mu_z(y) = \int_{\operatorname{supp}(\mu_z)} \langle x, y \rangle^p \, d\mu_z(y).$$
(6.21)

The discussion above implies that the last expression is well-defined for all p > 0. According to Theorem 3.1.7, the potential $U_{F_p}^{\mu_z}(x)$ is constant on $U_z \subseteq \text{supp}(\mu_z)$.

When p is an odd integer, the proof can be finished very quickly. Indeed, in this case the expression

$$g(x) = \int_{\operatorname{supp}(\mu_z)} \langle x, y \rangle^p \, d\mu_z(y)$$

is well-defined for each $x \in \mathbb{S}^{d-1}$ and yields an analytic function on the sphere (actually, a polynomial). Hence, being constant on an open set implies that it is constant on all of

 \mathbb{S}^{d-1} , by Lemma 4.2.2, which is not possible since, obviously, $g(-z) = -g(z) = -U_{F_p}^{\mu_z}(z) = -I_{F_p}(\mu_z) \neq 0$. Compare this argument to Theorem 4.2.1.

We now will present an approach which works for all $p \in \mathbb{R}_+ \setminus 2\mathbb{N}$. Assume that there exists a differential operator *D* acting on functions on the sphere with the following two properties:

- 1. *D* locally annihilates constants, i.e. if u(x) is constant on some open set *U*, then $D_x u = 0$ on *U*;
- 2. $D_x(\langle x, y \rangle^p) < 0$ for all $x \in U_z$ and $y \in \text{supp}(\mu_z)$.

Existence of such an operator would finish the proof since we would then have for each $x \in U_z$

$$0 = D_x U_{F_p}^{\mu_z}(x) = \int_{\text{supp}(\mu_z)} D_x \left(\langle x, y \rangle^p \right) d\mu_z(y) < 0,$$
(6.22)

which is a contradiction. Note that that switching to $D_x(\langle x, y \rangle^p) > 0$ in condition (2) does not affect the proof.

We now construct such an operator *D*. Let Δ denote the Laplace–Beltrami operator on \mathbb{S}^{d-1} . Writing it in the standard spherical coordinates $\theta_1, \ldots, \theta_d$ one obtains (see, e.g., equation (2.2.4) in [KMR01])

$$\Delta = \sum_{j=1}^{d-1} \frac{1}{q_j (\sin \theta_j)^{d-1-j}} \frac{\partial}{\partial \theta_j} \left((\sin \theta_j)^{d-1-j} \frac{\partial}{\partial \theta_j} \right), \tag{6.23}$$

where $q_1 = 1$ and $q_j = (\sin \theta_1 \dots \sin \theta_{j-1})^2$ for j > 1.

For a fixed $y \in \mathbb{S}^{d-1}$, choose the coordinates so that $\cos \theta_1 = \langle y, x \rangle$. Then $\langle y, x \rangle^p = \cos^p \theta_1$, effectively leaving just one term in the formula above, and a direct computation shows that

$$\Delta_x(\langle x, y \rangle^p) = p(p-1)\langle x, y \rangle^{p-2} - p(p+d-1)\langle x, y \rangle^p.$$
(6.24)

Observe that if $p \in (0, 1]$, then the operator Δ_x satisfies conditions (1) and (2), hence com-

pleting the proof for this range of *p*.

Now consider the operator $D = \Delta (\Delta + p(p+d-1))$. It is easy to see that

$$D_{x}(\langle x, y \rangle^{p}) = p(p-1)\Delta_{x}(\langle x, y \rangle^{p-2})$$

= $p(p-1)(p-2)\langle x, y \rangle^{p-4} \cdot ((p-3) - (p+d-3)\langle x, y \rangle^{2}).$ (6.25)

If $p \in (2,3]$, then $p-3 \le 0$ and $p+d-3 > d-1 \ge 0$, so the expression above is strictly negative. Hence this operator satisfies conditions (1) and (2) for 2 .

Moreover, if $p \in (1,2)$, the expression above is strictly positive. Indeed, the function $g_p(t) = (p-3) - (p+d-3)t$ is monotone on [0,1] with $g_p(0) = p-3 < 0$ and $g_p(1) = -d < 0$. Therefore, condition (2) holds with the inequality reversed, so the case 1 is also covered.

It is now clear how to iterate this process. Define the operator $D^{(0)} := \Delta$, $D^{(1)} = \Delta (\Delta + p(p+d-1))$, and, more generally, for $k \in \mathbb{N}$, define the differential operator of order 2k+2

$$D^{(k)} = \Delta \left(\Delta + (p+d-2k+1) \prod_{j=0}^{2k-2} (p-j) \right) \left(\Delta + (p+d-2k-1) \prod_{j=0}^{2k-4} (p-j) \right)$$
$$\cdots \left(\Delta + p(p-1)(p-2)(p+d-3) \right) \left(\Delta + p(p+d-1) \right).$$

Let $p \in \mathbb{R}_+ \setminus 2\mathbb{N}$ and choose $k \in \mathbb{N}_0$ so that 2k - 1 . An iterative computation similar to (6.25) gives us

$$\begin{split} D_x^{(k)}\left(\langle x, y \rangle^p\right) &= \left(\prod_{j=0}^{2k+1} (p-j)\right) \langle x, y \rangle^{p-2k-2} - \left(\prod_{j=0}^{2k} (p-j)\right) (p+d-2k-1) \langle x, y \rangle^{p-2k} \\ &= \left(\prod_{j=0}^{2k} (p-j)\right) \langle x, y \rangle^{p-2k-2} \cdot \left((p-2k-1) - (p+d-2k-1) \langle x, y \rangle^2\right). \end{split}$$

For $p \in (2k, 2k+1]$, the expression above is strictly negative, since $p - 2k - 1 \le 0$ and $p + d - 2k - 1 > d - 1 \ge 0$.

At the same time, for $p \in (2k - 1, 2k)$, this expression is strictly positive, because

 $\Pi_{j=0}^{2k}(p-j) < 0 \text{ and the monotone function } g_p(t) = (p-2k-1) - (p+d-2k-1)t \text{ takes}$ values $g_p(0) = p - 2k - 1 < 0$ and $g_p(1) = -d < 0$. Thus, operator $D^{(k)}$ allows us to prove Proposition 6.4.3 for p in the range $(2k-1,2k) \cup (2k,2k+1]$.

6.5 *p*-frame Energies in Non-compact Spaces

As discussed in Section 2.1, energy optimization on non-compact spaces is possible so long as there is some constraint that forces minimizing measures to have bounded support. This can be done by placing certain constraints on the kernel, as was discussed in the introduction of Chapter 4. Another possibility is to place constraints on the set of probability measures. Just as above, we consider $\mathbb{F} = \mathbb{R}, \mathbb{C}, \mathbb{H}$. In this setting, we consider the set of probability measures $\mu \in \mathbb{P}(\mathbb{F}^d)$ with the additional restriction

$$\int_{\mathbb{F}^d} ||x||^2 d\mu(x) = 1,$$
(6.26)

and denote this set by $\mathbb{P}^*(\mathbb{F}^d)$. Note that $\mu(\{0\}) = 0$ for all $\mu \in \mathbb{P}^*(\mathbb{F}^d)$.

We have previously used linear programming to bound energies and determine minimizers on compact, connected, two-point homogeneous spaces. This approach can be extended to *p*-frame energies in non-compact spaces as well. The normalization from (6.26) allows us to obtain a direct extension of above results for the spherical case, and by scaling, solutions to more general problems can be obtained from these results. A similar problem of finding minimizers for *p*-frame energies for $p \le 2$, subject to the condition that measures be isotropic, was investigated in [Gla].

In what follows, we shall assume $F_p : \mathbb{F} \to [0,\infty)$ is the *p*-frame potential $F(\langle x, y \rangle) = |\langle x, y \rangle|^p$ on \mathbb{F}^d , $F_p^*(t) = \left(\frac{t+1}{2}\right)^p$ is the projective *p*-frame kernel on $\mathbb{FP}^{d-1} \times \mathbb{FP}^{d-1}$

$$I_{F_p}(\mu) = \int_{\mathbb{F}^d} \int_{\mathbb{F}^d} F_p(\tau(x, y)) d\mu(x) d\mu(y)$$

We shall also define $q_{\mathbb{F}} : \mathbb{F}^d \setminus \{0\} \to \mathbb{FP}^{d-1}$ to be the projection from \mathbb{F}^d to \mathbb{FP}^{d-1} . From the discussion in Section 2.4 about the relationships between energies on the spheres and projective spaces, we can see that for all $x, y \in \mathbb{F}^d \setminus \{0\}$,

$$\begin{split} F_p(\langle x, y \rangle) &= ||x||^p ||y||^p \Big| \Big\langle \frac{x}{||x||}, \frac{y}{||y||} \Big\rangle \Big|^p \\ &= ||x||^p ||y||^p F_p^* \Big(\tau \Big(p_{\mathbb{F}} \Big(\frac{x}{||x||} \Big), p_{\mathbb{F}} \Big(\frac{y}{||y||} \Big) \Big) \Big) \\ &= ||x||^p ||y||^p F_p^* \Big(\tau \big(q_{\mathbb{F}}(x), q_{\mathbb{F}}(x) \big) \Big). \end{split}$$

As above, the normalized Jacobi polynomials for the projective spaces \mathbb{FP}^{d-1} are denoted C_m .

Lemma 6.5.1. Let $p \ge 2$, and assume $F_p^*(t) \ge h(t) = \sum_{m=0}^{\infty} \widehat{h}_m C_m(t)$ for all $t \in [-1, 1]$, where $\widehat{h}_m \ge 0$ for all $m \ge 0$. Then $I_{F_p}(\mu) \ge \widehat{h}_0$ for all $\mu \in \mathbb{P}^*(\mathbb{F}^d)$.

Proof. Since discrete measures are weak^{*} dense in $\mathbb{P}(\mathbb{F}^d)$, it is sufficient to prove the inequality for them only. Let μ take the form $\mu = \frac{1}{N} \sum_{i=1}^{N} \delta_{x_i}, x_i \in \mathbb{F}^d$. Then,

$$\begin{split} I_{F_p}(\mu) &= \frac{1}{N^2} \sum_{i,j=1}^N ||x_i||^p ||x_j||^p F_p^*(\tau(q_{\mathbb{F}}(x_i), q_{\mathbb{F}}(x_j))) \\ &\geq \frac{1}{N^2} \sum_{i,j=1}^N ||x_i||^p ||x_j||^p h(\tau(q_{\mathbb{F}}(x_i), q_{\mathbb{F}}(x_j))) \\ &= \frac{1}{N^2} \sum_{m=0}^\infty \widehat{h}_m \sum_{i,j=1}^N ||x_i||^p ||x_j||^p C_m(\tau(q_{\mathbb{F}}(x_i), q_{\mathbb{F}}(x_j))). \end{split}$$

For any $m \ge 1$, C_m is positive definite on \mathbb{FP}^{d-1} so that each sum

$$\sum_{i,j=1}^N ||x_i||^p ||x_j||^p C_m(\tau(q_{\mathbb{F}}(x_i), q_{\mathbb{F}}(x_j)))$$

is non-negative. Thus,

$$I_{F_p}(\mu) \ge \widehat{h}_0 \frac{1}{N^2} \sum_{i,j=1}^N ||x_i||^p ||x_j||^p C_0(\tau(q_{\mathbb{F}}(x_i), q_{\mathbb{F}}(x_j))) = \widehat{h}_0\left(\frac{1}{N} \sum_{i=1}^N ||x_i||^p\right)^2.$$

Since $p \ge 2$,

$$\frac{1}{N}\sum_{i=1}^{N}||x_i||^p \ge \left(\frac{1}{N}\sum_{i=1}^{N}||x_i||^2\right)^{\frac{p}{2}},$$

holds by Jensen's inequality. The constraint (6.26) is equivalent to $\frac{1}{N}\sum_{i=1}^{N} ||x_i||^2 = 1$, and so combining all inequalities, we complete the proof of the lemma.

Lemma 6.5.1 gives that any linear programming bounds for p-frame energies for the spherical/projective case will work in the non-compact setting as well. As a consequence of this approach we obtain the following result.

Theorem 6.5.2. Let \mathscr{C} be a set of arbitrary unit representatives of a tight projective *M*design, $M \ge 2$, in \mathbb{FP}^{d-1} and $p \in [2M - 2, 2M]$. Alternatively, let \mathscr{C} be a set of arbitrary unit representatives of the projective 600-cell in \mathbb{RP}^3 and $p \in [8, 10]$. Then

$$\mu_{\mathscr{C}} = \frac{1}{|\mathscr{C}|} \sum_{x \in \mathscr{C}} \delta_x$$

is a minimizer of

$$I_{F_p}(\mu) = \int_{\mathbb{F}^d} \int_{\mathbb{F}^d} F_p(\langle x, y \rangle) d\mu(x) d\mu(y)$$

over $\mathbb{P}^*(\mathbb{F}^d)$.

Proof. Set *h* to be the interpolating polynomial $H[F_p^*, g]$ used in the proof of Theorem 4.3.1 or the polynomial *h* from Theorem 6.3.2, depending on \mathscr{C} . Let us define

$$h_p(x,y) := ||x||^p ||y||^p h(\tau(q_{\mathbb{F}}(x), q_{\mathbb{F}}(y))),$$
(6.27)

for $x, y \in \mathbb{F}^d \setminus \{0\}$.

By construction, we see that for all $\mu \in \mathbb{P}^*(\mathbb{F}^d)$,

$$I_{F_p}(\mu) \geq I_{h_p}(\mu)$$

with equality if $\mu = \mu_{\mathscr{C}}$. From the proof of Lemma 6.5.1, it is also clear that

$$I_{h_p}(\mu) \geq \widehat{h}_0$$

with equality if $\mu(\mathbb{S}_F^{d-1}) = 1$ and $I_h((q_{\mathbb{F}})_*\mu) = \hat{h}_0$. Once again, this condition is satisfied if $\mu = \mu_{\mathscr{C}}$, since *h* is a positive definite polynomial and $(q_{\mathbb{F}})_*\mu_{\mathscr{C}}$ is a projective design of the appropriate strength.

6.6 Mixed Volume Inequalities

In this section we demonstrate an intriguing connection between the *p*-frame energy and convex geometry. We begin by briefly recalling some of the basic notions from convex geometry. See [Kol05, Ch. 2] for a more thorough development.

Let $K \subset \mathbb{R}^d$ be a convex body and σ_K be the surface measure of K, that is, a measure supported on the unit sphere \mathbb{S}^{d-1} , satisfying

 $\sigma_K(B) = |\{x \in \partial K : \text{ the outer unit normal to } K \text{ at } x \text{ belongs to } B\}|_{d-1}$

for all Borel sets $B \subseteq \mathbb{S}^{d-1}$, where $|\cdot|_{d-1}$ denotes the (d-1)-dimensional Hausdorff measure. For example, if *K* is a polytope with faces $\{K_i\}_{i=1}^m$ and normals $\{n_i\}_{i=1}^m$, σ_K is atomic with mass $|K_i|_{d-1}$ at each n_i ,

$$\sigma_K = \sum_{i=1}^m |K_i|_{d-1} \delta_{n_i},$$

and if $K = \mathbb{B}^d$ is the *d*-dimensional unit ball, then σ_K simply coincides with the standard (unnormalized) uniform surface area measure $\sigma_K(B) = |B|_{d-1} = A_{d-1}\sigma(B) = \frac{2\pi^{d/2}}{\Gamma(d/2)}\sigma(B)$.

Recall that for a convex body, $K \subset \mathbb{R}^d$, the *support function* $h_K(u)$ of *K* takes the form

$$h_K(u) = \sup_{v \in K} \langle u, v \rangle,$$

and uniquely determines K.

Given two convex bodies K and L, and $p \ge 1$, define the L_p -mixed volume

$$V_p(K,L) = \frac{p}{d} \lim_{\varepsilon \to 0} \frac{|K + p \varepsilon L| - |K|}{\varepsilon}$$

where $K +_p \mathcal{E}L$ is the convex body with support function $h_{K+_p \mathcal{E}L}(u)$ satisfying

$$h_{K+_p \in L}(u)^p = h_K(u)^p + \varepsilon h_L(u)^p.$$

Note that if *L* is the unit ball \mathbb{B}^d and p = 1, the above quantity is just the definition of the surface area of *K*. In general, $V_p(K,L)$ is known as the L_p -mixed volume of *K* and *L*. The following alternative integral representation for $V_p(K,L)$ is known

$$V_p(K,L) = \frac{1}{d} \int_{\mathbb{S}^{d-1}} h_L(u)^p d\sigma_K^p(u),$$

where $d\sigma_K^p(u) = h_K(u)^{1-p} d\sigma_K(u)$, so that in particular $d\sigma_K^1(u) = d\sigma_K(u)$.

Now, let us call a probability measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ admissible, if it is symmetric and not concentrated on a proper subspace. A classical result, which follows from Minkowski's theorem, says that any admissible measure can be realized as the surface area measure of a symmetric convex body; see more in [Sch93, Ch. 7].

The L_p -projection body $\Pi_p K$ of a convex body K is defined to be the origin-symmetric body whose support function is given by

$$h_{\Pi_p K}(u) = \left(c_{d,p} \int_{\mathbb{S}^{d-1}} |\langle u, v \rangle|^p d\sigma_K^p(u)\right)^{\frac{1}{p}},\tag{6.28}$$

where $c_{d,p}$ is the normalization chosen so that for the unit ball \mathbb{B}^d , $\Pi_p \mathbb{B}^d = \mathbb{B}^d$ (see [LYZ00, LZ97]). Thus, the identities

$$I_{F_p}(\sigma_K^p) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} |\langle u, v \rangle| \, d\sigma_K^p(u) d\sigma_K^p(v) = \int_{\mathbb{S}^{d-1}} h_{\Pi_p K}(u)^p \, d\sigma_K^p(u) = dV_p(K, \Pi K)$$

finally establish the connection between L_p -mixed volumes and p-frame energies.

Theorems 6.2.3 and 6.3.2 show that minimizers of $I_{F_p}(\mu)$ over probability measures are admissible when a corresponding tight design (or the 600-cell) exists, as this measure is symmetric and the support spans \mathbb{R}^d . Therefore, minimizing the *p*-frame energy in these cases is equivalent to minimizing the L_p mixed volume of a convex body *K* with its L_p projection body $\Pi_p K$ over all convex bodies satisfying $\sigma_K^p(\mathbb{S}^{d-1}) = 1$. In the case that the design is supported on the vertices of a polyhedron *C*, the minimizing *K* will be the dual of *C*, scaled so that σ_K^p is a probability measure. In particular, due to the fact that $\sigma_K^1 = \sigma_K$ we obtain the following:

Proposition 6.6.1. *The minimum of the quantity*

$$\frac{V_1(K,\Pi_1K)}{|\partial K|^2}$$

over all symmetric convex bodies in \mathbb{R}^d is achieved when K is a cube.

Indeed, it is easy to see that, when *K* is a cube, the surface measure σ_K is equally distributed on the vertices of a cross-polytope, which minimizes the *p*-frame energy for p = 1.

In accordance with Conjecture 6.2.5, we also anticipate that whenever p is not an even integer, the L_p -mixed volume $V_p(K, \Pi_p K)$ is always minimized by a convex body which is polyhedral (with discrete surface measure).

Chapter 7

Acute Angle Energy

The famous Hungarian geometer László Fejes Tóth formulated a variety of problems and conjectures about point distributions on the sphere, two of which, pertaining to the maximizers of the discrete Euclidean and geodesic Riesz energies, were discussed in Section 5.2. In this chapter, we discuss another conjecture of his that is currently open. In 1959, Fejes Tóth posed the following question [FT59]: what is the maximal value of the sum of pairwise acute angles defined by *N* vectors in the sphere S^2 ? More precisely, setting the kernel

$$\varphi(\langle x, y \rangle) := \arccos(|\langle x, y \rangle|) = \min\left\{\arccos(\langle z_i, z_j \rangle), \pi - \arccos(\langle z_i, z_j \rangle)\right\},$$
(7.1)

determine which *N*-element point sets $\omega_N = \{z_1, \dots, z_N\} \subset \mathbb{S}^2$ maximize the discrete energy E_{φ} .

He conjectured that this sum is maximized by the periodically repeated copies of the standard orthonormal basis. Though he only stated this for $d \le 3$, the generalization is clear, and has been independently stated in [Pet14] for all $d \ge 2$.

Conjecture 7.0.1 (Fejes Tóth, 1959 [FT59]). Let $d \ge 2$ and N = md + k with $m \in \mathbb{N}_0$ and $0 \le k \le d - 1$. Then the discrete energy E_{φ} on \mathbb{S}^{d-1} is maximized by the point set $\omega_N = \{z_1, \ldots, z_N\} \subset \mathbb{S}^{d-1}$ with $z_{pd+i} = e_i$, where $\{e_i\}_{i=1}^d$ is the standard orthonormal basis of \mathbb{R}^d , i.e. ω_N consists of m + 1 copies of k elements of the orthonormal basis of \mathbb{R}^d and m copies of the remaining d - k basis elements. In this case,

$$E_{\varphi}(\omega_N) = \frac{\pi}{2N^2} \left(k(k-1)(m+1)^2 + 2km(d-k)(m+1) + (d-k-1)(d-k)m^2 \right).$$
(7.2)

In particular, if N = md, the sum is maximized by m copies of the orthonormal basis:

$$\max_{\boldsymbol{\omega}_N \subset \mathbb{S}^{d-1}} E_{\boldsymbol{\varphi}}(\boldsymbol{\omega}_N) = \frac{\pi}{2} \cdot \frac{d-1}{d}.$$
(7.3)

Due to Lemma 2.1.1 and because (7.3) is independent of N, we may also formulate a continuous version of the conjecture: that the continuous energy I_{φ} is maximized by a measure whose mass is equally distributed between elements of the standard orthonormal basis, i.e.

$$\mu_{ONB} := \frac{1}{d} \sum_{i=1}^{d} \delta_{e_i}.$$
(7.4)

Conjecture 7.0.2. *The energy integral* I_{φ} *is maximized by* μ_{ONB} *:*

$$\max_{\boldsymbol{\mu} \in \mathbb{P}(\mathbb{S}^{d-1})} I_{\boldsymbol{\varphi}}(\boldsymbol{\mu}) = I(\boldsymbol{\mu}_{ONB}) = \frac{\pi}{2} \cdot \frac{d-1}{d}.$$
(7.5)

Since the maximal value in (7.3) is independent of N, Lemma 2.1.1 shows that the case N = md of Conjecture 7.0.1 implies Conjecture 7.0.2, and the converse implication is obvious.

The case d = 2 of Conjecture 7.0.1 has been settled in [FVZ16, Pet14]. We shall return to this case in Section 7.2 and shall give several alternative proofs.

In dimension d = 3, Fejes Tóth confirmed Conjecture 7.0.1 for $N \le 6$ and established an asymptotic upper bound $E_{\varphi}(\omega_N) \le \frac{2\pi}{5}$ for large *N*. In [FVZ16] Fodor, Vígh, and Zarnócz
proved that for any point distribution $\omega_N \subset \mathbb{S}^2$

$$E_{\varphi}(\omega_N) \le \frac{3\pi}{8}$$
 when N is even, (7.6)

with a small correction for N odd.

In their recent papers [LM21, LMb], Lim and McCann considered a different approach to this problem, studying the kernel $\varphi_s^*(x, y) := \left(\frac{2}{\pi} \arccos(|\langle x, y \rangle|)\right)^{-s}$ for s < 0 (the Riesz *s*-energy for the normalized acute angle potential φ^*). Among other results, they showed that there exists some $s_d \leq -1$ such that for all $s < s_d$, the set

$$\mathbb{P}_{ONB}(\mathbb{S}^{d-1}) := \{ \mu \in \mathbb{P}(\mathbb{S}^{d-1}) : \mu(\{e_i, -e_i\}) = \frac{1}{d},$$

$$i \in \{1, \dots, d\}, \{e_i\}_{i=1}^d \text{ an orthonormal basis of } \mathbb{R}^d \}$$
(7.7)

characterizes the maximizers of $I_{\varphi_s^*}$ up to rotation. The also found that at this endpoint s_d , there must exist a maximizer which is not in $\mathbb{P}_{ONB}(\mathbb{S}^{d-1})$. In response to an earlier draft of Lim and McCann's work, the author and coauthors Bilyk, Glazyrin, Park, and Vlasiuk provided a proof that $s_d > -2$, improving a previously discovered bound. This result now appears in the appendix of [LMb], but we provide it here as well.

Proposition 7.0.3. *If* $s \le -2$, *then the the maximizers of*

$$I_{\varphi_s^*}(\mu) := \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \left(\frac{2}{\pi} \arccos(|\langle x, y \rangle|) \right)^{-s} d\mu(x) d\mu(y)$$
(7.8)

are precisely the elements of $\mathbb{P}_{ONB}(\mathbb{S}^{d-1})$.

Proof. For $t \in [-1,1]$ and s < 0, let $F_s(t) = \left(\frac{2}{\pi}\arccos(|t|)\right)^{-s}$ and $g(t) = 1 - t^2$. We first show that $s \le -2$ if and only if $g(t) \ge F_s(t)$ on [-1,1] and $g(t) = F_s(t)$ exactly when $t \in \{-1,0,1\}$.

Setting $h_s(t) := g(t)^{\frac{1}{-s}} - F_s(t)^{\frac{1}{-s}}$, the computation $h''_{-2}(t) = (2t\pi^{-1} - 1)(1 - t^2)^{-3/2}$ shows $h_{-2}(t)$ to be strictly concave on the interval [0, 1] and to vanish at both endpoints. This establishes the above equivalence for s = -2. For s < -2, the same conclusion then follows from $\frac{dh_s}{ds}(t) \le 0$. For s > -2 we have that

$$\lim_{t \to 1} F_s'(t) = -\infty < g'(t)$$

and $F_s(1) = 0 = g(1)$, so domination of F_s by g fails. Thus we indeed have that $s \le -2$ if and only if $g(t) \ge F_s(t)$ on [-1,1] and $g(t) = F_s(t)$ exactly when $t \in \{-1,0,1\}$.

We now use linear programming to complete the proof. Using the fact that *g* bounds F_s from above, with equality on $\{-1, 0, 1\}$, and our knowledge of frame energy (see Lemma 6.1.3), we find that for all $s \leq -2$ and $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$

$$I_{F_s}(\mu) \le I_g(\mu) \le I_g(\mu_{ONB}) = I_{F_s}(\mu_{ONB}).$$
 (7.9)

The first inequality becomes an equality precisely when $\mathscr{A}(\operatorname{supp}(\mu)) \subseteq \{-1,0,1\}$ and the second when μ is an isotropic measure. The two properties characterize $\mathbb{P}_{ONB}(\mathbb{S}^{d-1})$, and our claim now follows.

Though Proposition 7.0.3 clearly does not prove Conjecture 7.0.2, it may be seen as some further evidence to suggest the conjecture is correct: in [LM21], the Lim and McCann showed that Conjecture 7.0.2 is true if and only if $s_d = -1$.

The kernel

$$\tilde{\varphi}(t) := \frac{\pi}{2} - \varphi(t) = \arcsin(|t|)$$

(for which minimization of $I_{\tilde{\varphi}}$ is naturally equivalent to the maximization of I_{φ}) has quite a bit in common with the *p*-frame potentials discussed in Chapter 6. It is clearly even and repulsive-attractive, with a minimum at t = 0, and for $d \ge 3$, $\tilde{\varphi}$ is not positive definite on the sphere \mathbb{S}^{d-1} . This can be seen by the fact that on \mathbb{S}^2 , σ is not a minimizer of the energy:

$$I_{\tilde{\varphi}}(\sigma) = \frac{\pi}{2} - 1 > \frac{\pi}{6} = I_{\tilde{\varphi}}(\mu_{ONB}).$$

Since spheres of higher dimensions contain a copy of \mathbb{S}^2 within them, $\tilde{\varphi}$ cannot be positive definite on them either. However, $\tilde{\varphi}$ defers from the *p*-frame potentials in two interesting ways. Unlike the *p*-frame potentials, $\tilde{\varphi}$ is positive definite on the circle \mathbb{S}^1 , as will be shown in Section 7.2. As we will discuss in that section, this suggests that there exists onedimensional (in the sense of Hausdorff dimension) minimizers of $I_{\tilde{\varphi}}$, rather than exclusively discrete ones, as we suspect for the *p*-frame energy. The second is that $\lim_{t\to\pm 1} \tilde{\varphi}'(t) = \infty$, which prevents us from creating a Hermite interpolant as we did in Section 4.3.

In Section 7.1, we prove a new upper bound for I_{φ} in all dimensions, and show that the validity of Conjecture 7.0.2 in some dimension $d \ge 3$ implies its validity in all lower dimensions. In Section 7.2, we provide several proofs of Conjectures 7.0.1 and 7.0.2 on S¹: two based on orthogonal expansions, and one based on a variant of the Stolarsky Invariance Principle.

7.1 New Results in all Dimensions

New Bound

As discussed in Section 4.3, when a suitable candidate for an interpolant does not present itself, we may find bounds on the energy I_{φ} by determining an auxiliary polynomial h that bounds φ from above and for which we know the maximizers. In the present section, we prove a new upper bound for I_{φ} in all dimensions $d \ge 3$ through the use of a quadratic auxiliary function. In addition to being the first result which determines a bound for all dimension, this result also provides a stronger bound than (7.6) when restricted to \mathbb{S}^2 .

Theorem 7.1.1. *In all dimensions* $d \ge 3$

$$\max_{\mu \in \mathbb{P}(\mathbb{S}^{d-1})} I_{\varphi}(\mu) \le \frac{\pi}{2} - \frac{69}{50d}.$$
(7.10)

In particular, for d = 3,

$$\max_{\mu \in \mathbb{P}(\mathbb{S}^2)} I_{\varphi}(\mu) \le \frac{\pi}{2} - \frac{69}{150} = 1.110796... < \frac{3\pi}{8} = 1.178097...,$$
(7.11)

thus improving upon (7.6).

We recall that, according to (7.3) and (7.5), the conjectured maximal value in dimension d = 3 is $\frac{\pi}{3} = 1.047198...$

Proof. It suffices to demonstrate that the inequality

$$\arccos|t| \le \frac{\pi}{2} - \frac{69}{50}t^2$$
 (7.12)

holds for all $t \in [-1, 1]$, as then Lemma 6.1.3 would imply that, for all $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$,

$$I_{\varphi}(\mu) \le \frac{\pi}{2} - \frac{69}{50d}.$$
(7.13)

For $b \ge 1$, $\frac{\pi}{2} - bt^2 - \arccos(|t|)$ is an even function, increasing on $\left[0, \sqrt{\frac{b-\sqrt{b^2-1}}{2b}}\right)$ and $\left(\sqrt{\frac{b+\sqrt{b^2-1}}{2b}}, 1\right)$, and decreasing on $\left(\sqrt{\frac{b-\sqrt{b^2-1}}{2b}}, \sqrt{\frac{b+\sqrt{b^2-1}}{2b}}\right)$. Thus, $\arccos|t| \le \frac{\pi}{2} - bt^2$ for all $t \in [-1, 1]$ if and only if

$$\frac{\pi}{2} - \frac{b + \sqrt{b^2 - 1}}{2} - \arccos\left(\sqrt{\frac{b + \sqrt{b^2 - 1}}{2b}}\right) \ge 0$$

which holds for $b \le \frac{69}{50}$, completing our proof. Figure 7.1 illustrates inequality (7.12).

Figure 7.1 indicates that there is very little room for improvement via this method (one can check that $\frac{\pi}{2} - bt^2 + \arccos(|t|)$ takes negative values if $b \ge \frac{7}{5}$), and therefore it cannot yield Conjecture 7.0.2.



Figure 7.1: An illustration of inequality (7.12): the graph of the function $F(t) = \frac{\pi}{2} - \frac{69}{50}t^2 - \arccos(|t|)$ for $0 \le t \le 1$.

Dimension Reduction Argument

Here we show generally that if a measure's components are orthogonal and that measure is a maximizer of the energy integral of some function, then the components are maximizers of this energy in lower dimensions. Naturally, the simplest case of this occurs when the measure is an orthonormal basis.

Let $F : [-1,1] \to \mathbb{R}$ be a bounded, measurable function that achieves its maximum at 0. Let $v \in \mathbb{P}(\mathbb{S}^k)$ and $\lambda \in \mathbb{P}(\mathbb{S}^l)$, with k + l = d - 2. Construct a measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ as follows:

$$\mu = a\widetilde{\mathbf{v}} + b\widetilde{\lambda},\tag{7.14}$$

where $0 \le a, b \le 1$, a + b = 1, and \tilde{v} and $\tilde{\lambda}$ are copies of v and λ , supported on mutually orthogonal subspheres of \mathbb{S}^{d-1} of dimensions k and l, respectively. It is easy to see that

$$I_F(\mu) = a^2 I_F(\nu) + b^2 I_F(\lambda) + 2abF(0), \qquad (7.15)$$

which implies that if μ is a maximizer of I_F over $\mathbb{P}(\mathbb{S}^{d-1})$, then ν and λ maximize I_F respectively over $\mathbb{P}(\mathbb{S}^k)$ and $\mathbb{P}(\mathbb{S}^l)$.

The orthonormal basis measure μ_{ONB} is precisely of the form (7.14), with ν and λ as orthonormal basis measures in lower dimensions. Therefore, in particular, we have just proved that the validity of Conjecture 7.0.2 on \mathbb{S}^{d-1} for some dimension $d \ge 3$ implies its validity on \mathbb{S}^k for k < d - 1, i.e. in all lower dimensions.

Moreover, in our case, for $\varphi(t) = \arccos(|t|)$, we have $\varphi(0) = \frac{\pi}{2}$. Let l = 0 and k = d-2. Then $I_{\varphi}(\lambda) = \varphi(1) = 0$, and thus (7.15) becomes

$$I_{\varphi}(\mu) = a^2 I_{\varphi}(\nu) + \pi a(1-a).$$

Optimizing this quadratic polynomial in *a* we find that for $a = \frac{\pi}{2(\pi - I_{\varphi}(\nu))}$,

$$I_{\varphi}(\mathbf{v}) = \pi - \frac{\pi^2}{4I_{\varphi}(\mu)}$$

This discussion leads to the following conclusion.

Proposition 7.1.2 (Dimension reduction). *Denote* $M_{d-1} = \max\{I_{\varphi}(\mu) : \mu \in \mathbb{P}(\mathbb{S}^{d-1})\}$. *We have the following:*

1. For $d \ge 3$ *,*

$$M_{d-2} \le \pi - \frac{\pi^2}{4M_{d-1}}$$

2. Assume that Conjecture 7.0.2 holds on \mathbb{S}^{d-1} , i.e. $M_{d-1} = \frac{\pi(d-1)}{2d}$. Then it also holds on \mathbb{S}^{d-2} , and consequently in all lower dimensions.

Observe, that part (2) follows both from part (1), and, in a more general setting, from the discussion in the beginning of this section. We also see that Proposition 7.1.2 implies that, in order to prove Conjecture 7.0.2 in all dimensions $d \ge 3$, it is enough to establish its validity for infinitely many values of d.

7.2 The Case of \mathbb{S}^1 Revisited

We now revisit the case d = 2, in which Conjecture 7.0.2 (and hence also Conjecture 7.0.1) has been settled in [FVZ16, Jia08] through a geometric proof. We shall provide three analytic approaches to this case (one based on a Chebyshev expansion, one based on a Fourier expansion, and one based on connections to discrepancy theory), which first appeared in [BM19].

Before we proceed, we observe that on \mathbb{S}^1

$$I_{\varphi}(\mu_{ONB}) = I_{\varphi}(\sigma) = I_{\varphi}(\sigma_{4N}) = \frac{\pi}{4}, \qquad (7.16)$$

where σ_N is the measure with mass equally concentrated at *N* equally spaced points, i.e. with $x_{N,k} = \begin{pmatrix} \cos(2\pi k/N) \\ \sin(2\pi k/N) \end{pmatrix}$, $\sigma_N = \frac{1}{N} \sum_{k=1}^N \delta_{x_{N,k}}$.

Hence, to prove Conjecture 7.0.2 it suffices to prove that any of these measures is a maximizer.

It is worth noting the implications of the uniform measure σ being a minimizer for $I_{\tilde{\varphi}}$ (maximizer for I_{φ}) on \mathbb{S}^1 . If the Conjecture 7.0.2 is true, then $\mu = \frac{1}{4} \sum_{j=1}^4 \delta_{e_j}$ is a minimizer of $I_{\tilde{\varphi}}$ on \mathbb{S}^3 . Following the dimension-reduction discussion above, the normalized uniform 1-dimensional Hausdorff measure on two orthogonal copies of \mathbb{S}^1 , i.e. on the set

$$\{(x_1, x_2, 0, 0) : x_1^2 + x_2^2 = 1\} \cup \{(0, 0, x_3, x_4) : x_3^2 + x_4^2 = 1\},\$$

would also yield a minimizer. This naturally generalizes to higher dimensions. Thus, assuming the validity of the conjecture, this energy has non-discrete minimizers.

Chebyshev Polynomial Expansion

From Proposition 3.5.2, we know that σ is a maximizer of the energy integral I_F on \mathbb{S}^1 if and only if *F* is a negative definite function on \mathbb{S}^1 (up to the constant term), which is equivalent to the fact that in the orthogonal expansion of *F* into Chebyshev polynomials,

 $F(t) \sim \sum_{n=0}^{\infty} \widehat{F}(n,0)T_n(t)$, the coefficients of all non-constant terms are non-positive, i.e.

$$\widehat{F}(n,0) \leq 0$$
 for $n \geq 1$.

Since the Chebyshev coefficients of $\varphi(t) = \arccos |t|$,

$$\widehat{\varphi}(n,0) = \frac{1}{\pi} \int_{-1}^{1} F(t) T_n(t) (1-t^2)^{-\frac{1}{2}} dt = \begin{cases} \frac{\pi}{4}, & \text{if } n = 0, \\ \frac{-4}{\pi n^2}, & \text{if } n \equiv 2 \mod 4, \\ 0, & \text{otherwise} \end{cases}$$
(7.17)

1

 σ clearly maximizes I_{φ} on \mathbb{S}^1 , which together with (7.16) implies Conjecture 7.0.2.

Fourier Series

Our second orthogonal expansion is a Fourier series. Every point in \mathbb{S}^1 can be defined by its angle, so setting $\mathbb{T} = \mathbb{R}/(2\pi\mathbb{Z})$ and defining, for every $\mu \in \mathscr{M}(\mathbb{S}^1)$, a corresponding measure $\tilde{\mu} \in \mathscr{M}(\mathbb{T})$ by

$$\tilde{\mu}(B) = \mu\left(\left\{ \begin{pmatrix} \cos(\theta) \\ \sin(\theta) \end{pmatrix} : \theta \in B \right\}\right)$$

for all Borel sets $B \subseteq \mathbb{T}$, we see that for any $F \in C([-1,1])$ and $\mu \in \mathscr{M}(\mathbb{S}^1)$,

$$I_F(\mu) = \int_{\mathbb{T}} \int_{\mathbb{T}} F(\cos(\theta - \phi)) d\tilde{\mu}(\theta) \tilde{\mu}(\phi) = \int_{\mathbb{T}} \int_{\mathbb{T}} G(\theta - \phi) d\tilde{\mu}(\theta) d\tilde{\mu}(\phi).$$

Let $G(\theta) = F(\cos(\theta))$ be an even function. Then for $v, \mu \in \mathbb{P}(\mathbb{S}^{d-1})$ such that $d\tilde{v}(\theta) = \frac{d\tilde{\mu}(\theta) + d\tilde{\mu}(-\theta)}{2}$ for all $\theta \in \mathbb{T}$, we have $I_F(\mu) = I_F(v)$. Thus, for the rest of this subsection, we may assume that $\mu \in \mathbb{P}(\mathbb{S}^1)$ such that $\tilde{\mu}$ is an even measure. We note that since *G* is an

even function, its Fourier series is a cosine series, i.e. $G(\theta) \sim \sum_{n=0}^{\infty} \widehat{G}(n) \cos(n\theta)$. As long as this series converges absolutely, we can use it to compute the energy:

$$I_{F}(\mu) = \sum_{n=0}^{\infty} \widehat{G}(n) \int_{\mathbb{T}} \int_{\mathbb{T}} \cos(n(\theta - \phi)) d\tilde{\mu}(\theta) d\tilde{\mu}(\phi)$$

$$= \sum_{n=0}^{\infty} \widehat{G}(n) \int_{\mathbb{T}} \int_{\mathbb{T}} \left(\cos(n\theta) \cos(n\phi) + \sin(n\theta) \sin(n\phi) \right) d\tilde{\mu}(\theta) d\tilde{\mu}(\phi).$$
(7.18)

Since $\tilde{\mu}$ is an even measure, the sines do not contribute to the integral, and we have

$$I_F(\mu) = \sum_{n=0}^{\infty} \widehat{G}(n) \left(\int_{\mathbb{T}} \cos(n\theta) d\tilde{\mu}(\theta) \right)^2 \le \widehat{G}(0) + \sum_{n \ge 1: \ \widehat{G}(n) \ge 0} \widehat{G}(n).$$
(7.19)

The equality above is achieved if and only if $\int_{\mathbb{T}} \cos(n\theta) d\tilde{\mu}(\theta) = 0$ for each value of $n \ge 1$ for which $\hat{G}(n) < 0$ and $\int_{\mathbb{T}} \cos(n\theta) d\tilde{\mu}(\theta) = 1$ for each value of $n \ge 1$ such that $\hat{G}(n) > 0$.

Lemma 7.2.1. Let G be even and have an absolutely convergent Fourier (cosine) series and let $\tilde{\sigma}_N$ be the probability measure generated by point masses at N equally spaced points, i.e.

$$ilde{\sigma}_N := rac{1}{N} \sum_{k=0}^{N-1} \delta_{2\pi k/N}.$$

Assume that for all $n \ge 1$ we have $\widehat{G}(n) \ge 0$ if n is a multiple of N, and $\widehat{G}(n) \le 0$ otherwise. Then the measure σ_N maximizes I_F over \mathbb{T} .

The proof easily follows from the discussion above and the fact that

$$\int_{\mathbb{T}} \cos(n\theta) d\tilde{\sigma}_N(\theta) = \frac{1}{N} \sum_{k=0}^{N-1} \cos \frac{2\pi nk}{N} = \begin{cases} 1, & \text{if } n \text{ is a multiple of } N\\ 0, & \text{otherwise.} \end{cases}$$

Remark: The conditions on $\widehat{G}(n)$ in Lemma 7.2.1 also become necessary if we assume that σ_N maximizes I_F over all signed probability measures, $\widetilde{\mathbb{P}}(\mathbb{S}^1)$. This can be proved by

considering signed measures of the form $d\tilde{\mu} = d\tilde{\sigma} - \cos n\theta d\tilde{\sigma}$.

Returning to our specific case, $G(\theta) = \varphi(\cos(\theta)) = \min\{|\theta|, \pi - |\theta|\}$, we observe that according to definition (2.27) and relation (7.17), we have $\widehat{G}(n) = \widehat{\varphi}(n,0) = -\frac{4}{\pi n^2}$ whenever $n \equiv 2 \mod 4$, and $\widehat{G}(n) = 0$ for all other values of $n \ge 1$. Therefore, Lemma 7.2.1 with N = 4 applies and σ_4 maximizes I_{φ} . Since φ is an even function, symmetry implies that μ_{ONB} is also a maximizer, i.e. Conjecture 7.0.2 holds on \mathbb{S}^1 .

Notice that alternatively, since $\widehat{G}(n) \leq 0$ for all $n \geq 1$, one could deduce from (7.19) that $I_{\varphi}(\mu) \leq \widehat{G}(0) = I_{\varphi}(\sigma)$. Therefore σ is a maximizer of the energy integral I_{φ} , and, due to (7.16), so is μ_{ONB} , leading to another proof of Conjecture 7.0.2 (almost identical to our proof involving the Chebyshev expansion). However, Lemma 7.2.1 could be applied to determine a maximizer for an energy with both positive and negative Fourier coefficients, allowing one to address a wider range of energy optimization problems, so we wanted to include it here.

Stolarsky Principle

Finally, our last approach is yet another analogue of the Stolarsky Invariance Principle (Theorem 5.1.1).

For $x \in \mathbb{S}^1$, define the *antipodal quadrants* in the direction of x as $Q(x) = \{y : |\langle x, y \rangle| > \frac{\sqrt{2}}{2}\}$, i.e. Q(x) is a union of two symmetric quarter-circle arcs with midpoints at x and -x. The size of the intersection of two such sets is given by

$$\sigma(Q(y) \cap Q(z)) = \frac{1}{2} - \frac{1}{\pi}\arccos|y \cdot z|$$
(7.20)

as the left-hand side is $\frac{1}{2}$ if z = y and 0 if z and y are orthogonal, and clearly changes linearly with respect to the acute angle between the points.

We define the L^2 discrepancy of a point set $\omega_N = \{z_1, ..., z_N\} \subset \mathbb{S}^1$ with respect to these antipodal quadrants as the quadratic average of the difference between the empirical measure $\frac{|Q(x) \cap \omega_N|}{N}$ and the uniform measure $\sigma(Q(x)) = \frac{1}{2}$.

$$D_{L^{2},\text{quad}}^{2}(\omega_{N}) = \int_{\mathbb{S}^{1}} \left| \frac{1}{N} \sum_{i=1}^{N} \mathbb{1}_{Q(x)}(z_{i}) - \sigma(Q(x)) \right|^{2} d\sigma(x).$$
(7.21)

More generally, for a measure $\mu \in \mathbb{P}(\mathbb{S}^1)$ we can define its discrepancy as

$$D_{L^2,\text{quad}}^2(\mu) = \int_{\mathbb{S}^1} \left| \mu\left(Q(x)\right) - \sigma\left(Q(x)\right) \right|^2 d\sigma(x), \tag{7.22}$$

i.e. $D^2_{L^2,\text{quad}}(\omega_N) = D^2_{L^2,\text{quad}}(\frac{1}{N}\sum_{z\in\omega_N}\delta_z).$

The following version of the Stolarsky principle holds:

Proposition 7.2.2 (Stolarsky principle for quadrants). *For any measure* $\mu \in \mathbb{P}(\mathbb{S}^1)$ *we have*

$$D_{L^2,\text{quad}}^2(\mu) = \frac{1}{\pi} \left(I_{\varphi}(\sigma) - I_{\varphi}(\mu) \right) = \frac{1}{4} - \frac{1}{\pi} I_{\varphi}(\mu).$$
(7.23)

In particular, for a discrete point set $\omega_N = \{z_1, ..., z_N\} \subset \mathbb{S}^1$

$$D_{L^{2},\text{quad}}^{2}(\omega_{N}) = \frac{1}{4} - \frac{1}{\pi} E_{\varphi}(\omega_{N}).$$
(7.24)

Proof. We use relations (7.22) and (7.20) to obtain

$$D_{L^{2},\text{quad}}^{2}(\mu) = \iint_{\mathbb{S}^{1}} \iint_{\mathbb{S}^{1}} \mathbb{1}_{Q(x)}(y) \cdot \mathbb{1}_{Q(x)}(z) d\sigma(x) d\mu(y) d\mu(z)$$

$$-\iint_{\mathbb{S}^{1}} \iint_{\mathbb{S}^{1}} \mathbb{1}_{Q(x)}(y) d\sigma(x) d\mu(y) + \frac{1}{4}$$

$$= \iint_{\mathbb{S}^{1}} \iint_{\mathbb{S}^{1}} \sigma(Q(y) \cap Q(z)) - \frac{1}{4} = \frac{1}{4} - \frac{1}{\pi} \iint_{\mathbb{S}^{1}} \inf_{\mathbb{S}^{1}} \operatorname{arccos}(|\langle y, z \rangle|) d\mu(y) d\mu(z),$$

which proves (7.23).

Remarks: Discrepancy with respect to similar sets, but with arbitrary apertures, ("wedges")

on \mathbb{S}^{d-1} arose in [BL17] in connection to sphere tessellations and one-bit sensing, which led to a Stolarsky principle involving energies with the potential $F(t) = \arcsin(t)^2$. Observe also that Proposition 7.2.2 follows from our Generalized Stolarsky Principle (Theorem 5.3.1), with $f(t) = \mathbb{1}_{[-1,\frac{-\sqrt{2}}{2})\cup(\frac{\sqrt{2}}{2},1]}(t)$ and $F(t) = \frac{1}{2} - \frac{1}{\pi} \arccos(|t|)$. However, we have provided a proof here that does not assume any positive definiteness.

Our Stolarsky principle (7.23) proves that $I_{\varphi}(\mu) \leq \frac{\pi}{4}$ and σ maximizes this energy, and by (7.16) so does v_{ONB} proving Conjecture 7.0.2. In the discrete case, (7.24) shows that $E_{\varphi}(\omega_N) \leq \frac{\pi}{4}$ for even N and $E_{\varphi}(\omega_N) \leq \frac{\pi}{4} \cdot \frac{N^2 - 1}{N^2}$ for odd N. It also allows one to characterize the maximizers of $E_{\varphi}(\omega_N)$ and prove Conjecture 7.0.1: (7.24) implies that maximizing $E_{\varphi}(\omega_N)$ is equivalent to minimizing the discrepancy $D_{L^2,\text{quad}}^2(\omega_N)$. It is easy to see from (7.21) that this happens exactly when the following holds: for σ -almost every $x \in \mathbb{S}^1$ the difference between the number of points of ω_N in Q(x) and in its complement $(Q(x))^c$ is zero (when N is even) or ± 1 (when N is odd). Equivalently, this should hold for any x such that the boundary of Q(x) doesn't intersect ω_N , which recovers the characterization in [FVZ16]. The extremizing configurations in Conjecture 7.0.1 obviously satisfy this condition.

This method unfortunately does not hold for dimensions $d \ge 3$, as $\frac{1}{2} - \frac{1}{\pi} \arccos(|t|)$ is not positive definite on these spheres, so we cannot create an appropriate Stolarsky-type invariance principle.

Chapter 8

Energy with Multivariate Potentials

In this final chapter, we discuss optimization problems for more complicated energies, defined by interactions of triples, quadruples, or even higher numbers of particles, i.e. energies of the type

$$E_{K}(\omega_{N}) = \frac{1}{N^{n}} \sum_{x_{1},...,x_{n} \in \omega_{N}} K(x_{1},...,x_{n}),$$
(8.1)

$$I_K(\mu) = \int_{\Omega} \cdots \int_{\Omega} K(x_1, \dots, x_n) d\mu(x_1) \dots d\mu(x_n), \qquad (8.2)$$

with $n \ge 3$. Energies of this type arise naturally in various fields:

In different branches of *physics* (nuclear, quantum, chemical, condensed matter, material science, etc.), it has been suggested that, if the behavior of the system cannot be accurately modeled by two-body interactions, more precise information may be obtained from three-body or many-body interactions. Such forces are observed among nucleons in atomic nuclei (three-nucleon force) [Zel09], in carbon nanostructures [MS14], crystallization of atomistic configurations [FT15], cold polar molecules in optical lattices [BMZ07], interactions of solid and liquid forms of silicon [SW85], interactions between atoms [AT43], in "perfect glass" potentials [ZST16], and many other areas.

- Energy integrals with multivariate kernels defined in (8.2) play the role of polynomials on the space P(Ω) of probability measures on Ω e.g., their linear span over all n ∈ N is dense in the space of continuous functions on P(Ω), according to the Stone–Weierstrass theorem. Such functionals on the space of measures appear in optimal transport [San15] and mean field games [Lio08].
- 3. A classical example of a three-input energy, coming from geometric measure theory, is given by the total Menger curvature of a measure μ

$$c^{2}(\boldsymbol{\mu}) = \int_{\Omega} \int_{\Omega} \int_{\Omega} \int_{\Omega} c^{2}(x, y, z) d\boldsymbol{\mu}(x) d\boldsymbol{\mu}(y) d\boldsymbol{\mu}(z), \qquad (8.3)$$

where $c(x, y, z) = \frac{1}{R(x, y, z)}$ and R(x, y, z) is the circumradius of the triangle *xyz*. This object plays an important role in the study of the L^2 -boundedness of the Cauchy integral, analytic capacity, and uniform rectifiability [Dav99, MMV96].

- 4. Some questions in *probabilistic geometry* admit natural reformulations in terms of multi-input energies (8.1) or (8.2). For example, assume that three points are chosen in a domain Ω, e.g. Ω = S², independently at random, according to the probability distribution *μ*. Which probability distribution maximizes the expected area of the triangle generated by these random points or the volume of the parallelepiped spanned by the random vectors? These quantities can be written as energy integrals (8.1) with *n* = 3, and higher dimensional versions of such questions call for energies with more inputs, which may be viewed as natural extensions of the classical Riesz energy. Questions of this type are discussed in Section 8.8 and are explored in more detail in [BFG⁺b].
- 5. Energies with more than two inputs akin to (8.1) appear in three-point bounds [CW12] and, more generally, *k*-point bounds [dLMdOFV, Mus14] in semidefinite programming [BV08] a very fruitful method, which led to numerous breakthroughs in dis-

crete geometry. We shall briefly revisit this method in Section 8.7 and show some applications to the aforementioned geometric problems in Section 8.8. A more complete discussion of this method in the context of the multivariate energy optimization and, in particular, applications to the geometric problems described in Section 8.8 can be found in [BFG⁺b].

6. Relations between the L^2 -discrepancy and the two-input energies, in particular, the Stolarsky principle [Sto73], are well known [BDM18, Skr20]. In a similar spirit, other L^n -norms of the discrepancy or "number variance" with integer values of *n* lead to *n*-particle interaction energies (8.1). Some similar ideas have been put forward in [Tor10].

Despite the abundance of applications, there seems to have been no systematic development of a general theory of multi-input energies preceding [BFG⁺a, BFG⁺b], unlike the case of classical two-input energies, which has been deeply and extensively explored. This chapter covers the results of [BFG⁺a] (as well as some from [BFG⁺b]) which made the first attempt to remedy this shortcoming. In these works, the author and coauthors study the general properties of point configurations and measures, minimization of the multiinput energies (8.1) and (8.2), and the relations between the structure of the multivariate kernel *K* and the energy minimizers. This theory presents many intrinsic obstacles and is far from a straightforward generalization of the two-input case. For instance, in the spherical case $\Omega = \mathbb{S}^{d-1}$ with rotationally invariant two-input kernel $F(\langle x, y \rangle)$, we know that the uniform surface measure σ minimizes I_F if and only if the *F* is conditionally positive definite 3.3.1. However, in the multi-input case, such a characterization is still elusive: while we obtain various natural sufficient conditions for the surface measure σ to minimize the energy (8.1) in Section 8.4, counterexamples presented in Section 8.5 show that none of them are necessary.

In Section 8.1 we introduce the notation and some of the main definitions, including the notion of n-positive definiteness. In Section 8.2 we explore some basic properties of mul-

tivariate energies. In particular, we analyze the connections between (conditional) positive definiteness of the kernel *K*, convexity of the energy functional $I_K(\mu)$, and arithmetic and geometric mean inequalities for the mixed energies.

Section 8.3 deals with analogues of various results from Section 3.3, which provide certain necessary (Theorem 8.3.2) and sufficient (Theorem 8.3.3) conditions for a measure μ to be a minimizer of the *n*-input energy integral in terms of the (n-1)-fold potential of the kernel *K* with respect to μ . Even though some of these results by themselves are clear-cut generalizations of standard statements for two-input energies, they yield several interesting consequences in the *n*-input case. In particular, Theorem 8.3.7 states that, under some additional assumptions (e.g., if *K* is *n*-positive definite), for any $1 \le k \le n-2$, if the measure μ minimizes the (n-k)-input energy $I_{U_K^{\mu k}}$, where $U_K^{\mu k}$ is the *k*-fold integral of *K* with respect to μ , then μ also minimizes the *n*-input energy I_K . This statement allows one to simplify proving that a given measure is a minimizer of a multi-input energy by considering energies with a lower number of inputs. A partial converse to Theorem 8.3.7, for k = n-2, is given in Theorem 8.3.8. In addition, in Lemma 8.3.5, we show that, for *n*-positive definite kernels, every local minimizer of I_K is necessarily a global minimizer.

In Section 8.4 we adapt the methods of Section 8.3 to energies with rotationally invariant kernels on the sphere \mathbb{S}^{d-1} , where symmetries allow for a more delicate analysis, and one has a natural candidate for a minimizer: the uniform surface measure σ . Theorem 8.4.1 states that energies with conditionally *n*-positive definite rotationally invariant kernels on the sphere are minimized by the surface measure σ (without any additional assumptions). As mentioned above, it turns out that, in contrast to the classical case n = 2, conditional *n*positive definiteness is not necessary for σ to minimize the *n*-input energy, which is shown by examples presented in Propositions 8.6.3 and 8.6.4. Nevertheless, Theorem 8.4.1 allows one to prove that σ minimizes a variety of interesting energies, which did not seem to be accessible by different methods, see e.g. Corollary 8.4.2. In Theorem 8.4.3 we obtain very close necessary and sufficient conditions for σ to be a *local* minimizer of the *n*-input energy I_K in terms of the minimization properties of the two-input energy with the kernel given by the (n-2)-fold integral of K (or the conditional positive definiteness of this kernel). We also conjecture these are the correct conditions for σ to be a *global* minimizer of I_K .

Section 8.5 is dedicated to constructing various classes of *n*-positive definite kernels, proving that certain kernels of interest are (conditionally) *n*-positive definite, whereas Section 8.6 exhibits some naturally arising 3-input kernels on the sphere which are not conditionally 3-positive definite, yet the corresponding energies are minimized by the surface measure σ . These examples are presented in Propositions 8.6.1, 8.6.2, 8.6.3, and 8.6.4. The first two are closely related to the semidefinite programming method, while the last two are geometric.

In Section 8.7 we adapt the powerful method of semidefinite programming bounds [BV08, CW12, Mus14] to our setting. While multi-input energies arise naturally in this method, its applications were mainly focused on discrete configurations and energies, rather than on minimizing measures and energy integrals. By adapting the results of [BV08] to our setting, the formulation of Theorem 8.7.1 provides a way to determine that σ minimizes the three-input energy for a wide class of functions.

Section 8.8 addresses some problems from probabilistic discrete geometry, which deal with objects that can be viewed as multi-input analogues of the classical Riesz energies. We show that if three random vectors are chosen in the sphere \mathbb{S}^{d-1} independently according to the probability distribution μ , then the expected volume squared of the tetrahedron generated by these vectors (Theorem 8.8.1) as well as the square of the area of the triangle defined by these points (Theorem 8.8.4) are maximized if the distribution is uniform, $\mu = \sigma$. Both statements can be proved either by semidefinite programming or directly, using ideas from linear algebra. While for the volume and area (without the square) these questions remain open, we use our results to show that, for N = d + 1 points, the corresponding discrete energy (the average area of the triangle or the average volume of the tetrahedron) is maximized when the point configuration consists of the vertices of a regular

simplex in \mathbb{S}^{d-1} , see Corollary 8.8.6.

8.1 Background and Definitions

In what follows, we always assume that (Ω, ρ) is a compact metric space, $n \in \mathbb{N} \setminus \{1\}$, and the kernel $K : \Omega^n \to \mathbb{R}$ is continuous and symmetric, i.e. for any permutation $\pi \in S_n$ and $x_1, ..., x_n \in \Omega$, $K(x_1, ..., x_n) = K(x_{\pi(1)}, ..., x_{\pi(n)})$. Let $\omega_N = \{z_1, z_2, ..., z_N\}$ be an *N*-point configuration (multiset) in Ω for $N \ge n$. We define the discrete *K*-energy of ω_N to be

$$E_K(\omega_N) := \frac{1}{N^n} \sum_{j_1=1}^N \cdots \sum_{j_n=1}^N K(z_{j_1}, \dots, z_{j_n}),$$
(8.4)

and the minimal discrete N-point K-energy of Ω as

$$\mathscr{E}_{K}(\Omega, N) := \inf_{\omega_{N} \subseteq \Omega} E_{K}(\omega_{N}).$$
(8.5)

Let $\mu_1, ..., \mu_n \in \mathscr{M}(\Omega)$, then we define their mutual energy as

$$I_K(\mu_1,...,\mu_n) = \int_{\Omega} \cdots \int_{\Omega} K(x_1,...,x_n) d\mu_1(x_1) \cdots d\mu_n(x_n), \qquad (8.6)$$

and, for j < n, the *j*-th potential function as

$$U_{K}^{\mu_{1},...,\mu_{j}}(x_{j+1},...,x_{n}) = \int_{\Omega} \cdots \int_{\Omega} K(x_{1},...,x_{n})d\mu_{1}(x_{1})\cdots d\mu_{j}(x_{j}).$$
(8.7)

Note that since we are working with continuous *K*, the energy is well-defined for all finite signed Borel measures. We will abuse notation by writing μ^k if *k* of the measures are the same and define the *K*-energy of a measure $\mu \in \mathcal{M}(\Omega)$ by

$$I_K(\mu) = I_K(\mu^n) = I_K(\mu, ..., \mu),$$
(8.8)

and the minimal K-energy over all probability measures by

$$\mathscr{I}_{K}(\Omega) = \inf_{\mu \in \mathbb{P}(\Omega)} I_{K}(\mu).$$
(8.9)

The definitions of discrete (8.4) and continuous (8.8) energies are compatible in the sense that

$$E_K(\omega_N) = I_K(\mu_{\omega_N}), \text{ where } \mu_{\omega_N} = \frac{1}{N} \sum_{j=1}^N \delta_{z_j}$$
 (8.10)

and due to the weak^{*} density of the linear span of Dirac masses in $\mathbb{P}(\Omega)$

$$\lim_{N \to \infty} \mathscr{E}_K(\Omega, N) = \mathscr{I}_K(\mu).$$
(8.11)

We can extend the classical notion of positive definiteness for two-input kernels to *n*-input kernels by demanding that, if one fixes arbitrary values of all but two variables, the resulting two-input kernel is positive definite in the classical sense. For every m < n and $z_1, z_2, ..., z_m \in \Omega$, we define

$$K_{z_1, z_2, \dots, z_m}(x_1, \dots, x_{n-m}) := K(z_1, \dots, z_m, x_1, \dots, x_{n-m}).$$
(8.12)

Definition 8.1.1. We shall say that a continuous symmetric kernel $K : \Omega^n \to \mathbb{R}$ is (conditionally) *n*-positive definite if, for all $z_1, z_2, ..., z_{n-2} \in \Omega$, the two-input kernel $K_{z_1,...,z_{n-2}}$ is (conditionally) positive definite in the sense of Definition 2.2.1.

We would like to emphasize that this definition relies more on the pointwise twovariable structure, rather than the full set of variables. In particular, it does not have any connection to positive definite tensors [Qi05]. Thus, it may appear that the name *n-positive definite* might be somewhat misleading. However, from the point of view of energy optimization, which is the main theme of this work, this nomenclature seems absolutely justified. Indeed, in various statements about minimal energy (e.g., Theorem 8.3.3, Corollary 8.3.4, Theorem 8.4.1), this condition naturally replaces positive definiteness of classical two-input kernels. In addition, non-symmetric multivariate kernels of similar flavor have been considered in the context of *k*-point bounds in semidefinite programming [dLMdOFV, Mus14]. The class of *n*-positive definite kernels is rather rich: throughout this chapter, particularly in Section 8.5, we present numerous examples of functions with this property.

We immediately observe that this property is inherited by kernels with a lower number of inputs, which are obtained as potentials of K with respect to arbitrary probability measures.

Lemma 8.1.2. Let n > 2 and assume that K is (conditionally) n-positive definite. Then for every $\mu \in \mathbb{P}(\Omega)$, the potential $U_K^{\mu}(x_1, \ldots, x_{n-1})$ is (conditionally) (n-1)-positive definite.

Proof. Let *v* be a finite signed Borel measure on Ω (with $v(\Omega) = 0$ if *K* is conditionally *n*-positive definite). Then by Fubini–Tonelli

$$I_{\left(U_{K}^{\mu}\right)_{z_{2},\dots,z_{n-2}}}(\mathbf{v}) = \int_{\Omega} \int_{\Omega} \int_{\Omega} K(z_{1},z_{2},\dots,z_{n-3},z_{n-2},x,y)d\mu(z_{1}) d\mathbf{v}(x)d\mathbf{v}(y)$$
$$= \int_{\Omega} \int_{\Omega} \int_{\Omega} \int_{\Omega} K_{z_{1},\dots,z_{n-2}}(x,y)d\mathbf{v}(x)d\mathbf{v}(y) d\mu(z_{1}) \ge 0,$$

since $K_{z_1,...,z_{n-2}}$ is (conditionally) positive definite for all $z_1,...,z_{n-2} \in \Omega$.

As a corollary of Lemma 8.1.2, we observe that if $K : \Omega^n \to \mathbb{R}$ is (conditionally) *n*-positive definite, then for all $\mu_1, ..., \mu_k \in \mathbb{P}(\Omega)$, with $k \le n-2$, $U_K^{\mu_1,...,\mu_k}(x_{k+1},...,x_n)$ is (conditionally) (n-k)-positive definite.

Naturally, (conditionally) *n*-positive definite kernels enjoy the same basic properties as their classical two-variable counterparts. The following result generalizes Lemma 2.2.3 to *n*-input kernels and immediately follows this lemma as well.

Lemma 8.1.3. If K and L are n-positive definite, then so are K + L and KL. If $K_1, K_2, ...$ are n-positive definite and $\lim_{n\to\infty} K_n = K$ uniformly, then K is n-positive definite. The statements about the sum and the limit (but not about the product) continue to hold if we replace n-positive definite with conditionally n-positive definite.

8.2 First Principles

In this section we explore some basic properties related to (conditional) *n*-positive definiteness, such as inequalities for mixed energies and convexity of the energy functionals, as well as connections between these notions, much as we did in the two input case in Section 3.3. All the kernels in this section are assumed to be continuous and symmetric.

Bounds on Mutual Energies

The arithmetic and geometric mean inequalities (3.1) and (3.2) can be extended to *n*-input energies with (conditionally) *n*-positive definite kernels.

Lemma 8.2.1. Suppose K is a conditionally n-positive definite kernel on Ω^n . Then for every *n*-tuple of Borel probability measures μ_1, \ldots, μ_n on Ω , the mutual energy $I_K(\mu_1, \ldots, \mu_n)$ satisfies

$$I_K(\mu_1,...,\mu_n) \le \frac{1}{n} \sum_{j=1}^n I_K(\mu_j).$$
 (8.13)

If, moreover, K is n-positive definite,

$$I_K(\mu_1,\ldots,\mu_n) \le \prod_{j=1}^n \sqrt[n]{I_K(\mu_j)}.$$
(8.14)

Proof. We only prove (8.14), as one could repeat the proof below verbatim, with the multiplicative notation replaced by the additive, to arrive at (8.13) (when K is n-positive definite, it would also follow from the arithmetic–geometric mean inequality).

By Lemma 3.1.2, our claim holds for n = 2. Now, suppose our claim holds for some $k \ge 2$, and let $\mu_1, ..., \mu_{k+1} \in \mathbb{P}(\Omega)$. Lemma 8.1.2 tells us that for $1 \le j \le k+1$, $U_K^{\mu_j}$ is

k-positive definite, so by our inductive hypothesis

$$I_{K}(\mu_{1},...,\mu_{k+1}) = I_{U_{K}^{\mu_{1}}}(\mu_{2},...,\mu_{k+1}) \le \prod_{j=1}^{k} \sqrt[k]{I_{K}(\mu_{1},\mu_{j+1}^{k})}.$$
(8.15)

Again using the inductive hypothesis, and the fact that *K* is symmetric, we have that for $1 \le j \le k$,

$$\begin{split} I_{K}(\mu_{1},\mu_{j+1}^{k}) &= I_{K}(\mu_{j+1},\mu_{1},\mu_{j+1}^{k-1}) \\ &\leq \sqrt[k]{I_{K}(\mu_{j+1},\mu_{1}^{k})} \sqrt[k]{I_{K}(\mu_{j+1})^{k-1}} \\ &= \sqrt[k]{I_{K}(\mu_{1},\mu_{j+1},\mu_{1}^{k-1})} \sqrt[k]{I_{K}(\mu_{j+1})^{k-1}} \\ &\leq \sqrt[k]{\sqrt[k]{I_{K}(\mu_{1})^{k-1}}} \sqrt[k]{\sqrt[k]{I_{K}(\mu_{1},\mu_{j+1}^{k})}} \sqrt[k]{I_{K}(\mu_{j+1})^{k-1}} \end{split}$$

where in the second and last lines we have used (8.15). Rearranging the terms, we have

$$\left(I_{K}(\mu_{1},\mu_{j+1}^{k})\right)^{\frac{k^{2}-1}{k^{2}}} \leq I_{K}(\mu_{1})^{\frac{k-1}{k^{2}}}I_{K}(\mu_{j+1})^{\frac{k-1}{k}},$$

so that

$$\sqrt[k]{I_K(\mu_1,\mu_{j+1}^k)} \le I_K(\mu_1)^{\frac{1}{k(k+1)}} I_K(\mu_{j+1})^{\frac{1}{k+1}}$$

Plugging this back into (8.15), we have

$$I_{K}(\mu_{1},...,\mu_{k+1}) \leq \prod_{j=1}^{k} \sqrt[k]{I_{K}(\mu_{1},\mu_{j+1}^{k})} \leq \prod_{j=1}^{k+1} \sqrt[k+1]{I_{K}(\mu_{j})}.$$
(8.16)

Our claim then follows via induction.

The upper bound (8.13) allows us to prove a corresponding lower bound for the mixed energy:

Corollary 8.2.2. If K is n-positive definite on Ω^n , then for all $\mu_1, ..., \mu_n \in \mathbb{P}(\Omega)$,

$$-\frac{1}{n}\sum_{j=1}^{n}I_{K}(\mu_{j})\leq I_{K}(\mu_{1},...,\mu_{n}).$$
(8.17)

Proof. Suppose n = 2, and let $\mu_1, \mu_2 \in \mathbb{P}(\Omega)$. Setting $\mu = \frac{1}{2}(\mu_1 + \mu_2)$, we have

$$0 \leq 4I_K(\mu) = I_K(\mu_1) + I_K(\mu_2) + 2I_K(\mu_1, \mu_2),$$

since K is positive definite, and (8.17) follows.

Now suppose our claim holds for some $k \ge 2$, and let $\mu_1, ..., \mu_{k+1} \in \mathbb{P}(\Omega)$. Since by Lemma 8.1.2 the potential $U_K^{\mu_1}$ is *k*-positive definite, the inductive hypothesis implies that

$$-\frac{1}{k}\sum_{j=1}^{k}I_{U_{K}^{\mu_{1}}}(\mu_{j+1}) \leq I_{U_{K}^{\mu_{1}}}(\mu_{2},...,\mu_{k+1}) = I_{K}(\mu_{1},...,\mu_{k+1}).$$

For $1 \le j \le k$, Lemma 8.2.1 gives us that

$$I_{U_{K}^{\mu_{1}}}(\mu_{j+1}) = I_{K}(\mu_{1}, \mu_{j+1}^{k}) \leq \frac{1}{k+1} \Big(I_{K}(\mu_{1}) + kI_{K}(\mu_{j+1}) \Big),$$

leading to

$$-\frac{1}{k+1}\sum_{j=1}^{k+1}I_{K}(\mu_{j})\leq I_{K}(\mu_{1},...,\mu_{k+1}),$$

which finishes the proof of the claim.

		L
		L
-	-	

Lemma 8.2.1 and Corollary 8.2.2 imply that if *K* is *n*-positive definite on Ω^n and $\mu_1, ..., \mu_n \in \mathbb{P}(\Omega)$, then

$$|I_K(\mu_1,...,\mu_n)| \le \frac{1}{n} \sum_{j=1}^n I_K(\mu_j).$$
 (8.18)

Of course, since we can choose the probability measures μ_k to be Dirac masses, inequality (8.18) yields pointwise bounds on *K*. For instance, if *K* is *n*-positive definite, then for all

 $z_1,\ldots,z_n\in\Omega$,

$$|K(z_1,...,z_n)| \le \frac{1}{n} \sum_{j=1}^n K(z_j,...,z_j)$$

and for conditionally *n*-positive definite kernels K, this inequality holds without the absolute value. Clearly then, K must achieve its maximum value on its diagonal, something that is already known for the two-input case.

Corollary 8.2.3. Suppose K is a conditionally n-positive definite kernel. Then

$$K(z_1,\ldots,z_n) \le \max_{z\in\Omega} \{K(z,\ldots,z)\}.$$
(8.19)

Convexity

As in the two-input case, convexity of the underlying energy functionals naturally plays an important role in energy minimization.

Definition 8.2.4. Suppose $K : \Omega^n \to \mathbb{R}$. We say that I_K is **convex at** $\mu \in \mathbb{P}(\Omega)$ if for every $v \in \mathbb{P}(\Omega)$ there exists some $t_v \in (0,1]$, such that for all $t \in [0,t_v)$,

$$I_K((1-t)\mu + t\nu) \le (1-t)I_K(\mu) + tI_K(\nu).$$
(8.20)

We say I_K is **convex** on $\mathbb{P}(\Omega)$ if inequality (8.20) holds for every μ , $\nu \in \mathbb{P}(\Omega)$ and all $t \in [0,1]$.

As in the two-input case, we observe that convexity of I_K on $\mathbb{P}(\Omega)$ is equivalent to the fact that I_K is convex at all $\mu \in \mathbb{P}(\Omega)$. Indeed, if (8.20) fails for some μ , $\nu \in \mathbb{P}(\Omega)$, then the polynomial $f(t) = I_K((1-t)\mu + t\nu)$ is not convex on the interval [0,1], i.e. f''(t) < 0 on some subinterval $[a,b] \subset [0,1]$. But in this case, one can easily see that I_K fails to be convex at $\mu_a = (1-a)\mu + a\nu$.

We know that conditional positive definiteness of the kernel *K* is equivalent to convexity of the corresponding energy functional I_K in the case n = 2 (Proposition 3.1.6). As the next proposition shows, a one-sided implication holds for $n \ge 3$: convexity of I_K can be deduced from relaxed arithmetic or geometric mean inequalities akin to (8.13) and (8.14). This implies, due to Lemma 8.2.1, that conditionally *n*-positive definite kernels *K* give rise to convex energies.

Proposition 8.2.5. Let $K : \Omega^n \to \mathbb{R}$ be continuous and symmetric and fix $\mu \in \mathbb{P}(\Omega)$. Suppose that for all $v \in \mathbb{P}(\Omega)$ and $0 \le k \le n$,

$$I_{K}(\mu^{k}, \nu^{n-k}) \leq \frac{k}{n} I_{K}(\mu) + \frac{n-k}{n} I_{K}(\nu).$$
(8.21)

Alternatively, assume that for all $v \in \mathbb{P}(\Omega)$ we have $I_K(v) \ge 0$ and for all $0 \le k \le n$

$$I_K(\mu^k, \nu^{n-k}) \le \left(I_K(\mu)\right)^{\frac{k}{n}} \cdot \left(I_K(\nu)\right)^{\frac{n-k}{n}}.$$
(8.22)

Then I_K is convex at μ . If (8.21) or (8.22) holds for all $\mu \in \mathbb{P}(\Omega)$, then I_K is convex on $\mathbb{P}(\Omega)$.

Proof. Assume that (8.21) holds. For all $t \in [0, 1]$, we have

$$\begin{split} I_{K}((1-t)\mu+t\nu) &= \sum_{k=0}^{n} (1-t)^{k} t^{n-k} \binom{n}{k} I_{K}(\mu^{k},\nu^{n-k}) \\ &\leq \sum_{k=0}^{n} (1-t)^{k} t^{n-k} \binom{n}{k} \binom{k}{n} I_{K}(\mu) + \frac{n-k}{n} I_{K}(\nu) \Biggr) \\ &= \sum_{k=1}^{n} (1-t)^{k} t^{n-k} \binom{n-1}{k-1} I_{K}(\mu) + \sum_{k=0}^{n-1} (1-t)^{k} t^{n-k} \binom{n-1}{k} I_{K}(\nu) \\ &= (1-t) I_{K}(\mu) + t I_{K}(\nu), \end{split}$$

which proves convexity of the energy functional. The multiplicative inequality (8.22) implies (8.21) by the arithmetic-geometric mean inequality, leading to convexity of K in this case.

Observe that Proposition 8.2.5 admits a partial converse:

Lemma 8.2.6. Suppose $\mu \in \mathbb{P}(\Omega)$ is such that I_K is convex at μ . Then for all $\nu \in \mathbb{P}(\Omega)$,

$$I_{K}(\mu^{n-1},\nu) \leq \frac{n-1}{n} I_{K}(\mu) + \frac{1}{n} I_{K}(\nu).$$
(8.23)

Proof. Let $v \in \mathbb{P}(\Omega)$. Assume $t \in (0, 1)$ such that (8.20) holds. Then

$$tI_{K}(\mathbf{v}) + (1-t)I_{K}(\mu) \ge I_{K}(t\mathbf{v} + (1-t)\mu)$$

= $\sum_{j=0}^{n} (1-t)^{j} t^{n-j} {n \choose j} I_{K}(\mu^{j}, \mathbf{v}^{n-j}).$

Clearly then

$$\left(t-t^{n}\right)I_{K}(\mathbf{v})+\left((1-t)-(1-t)^{n}\right)I_{K}(\mu)\geq\sum_{j=1}^{n-1}(1-t)^{j}t^{n-j}\binom{n}{j}I_{K}(\mu^{j},\mathbf{v}^{n-j}),$$

and dividing by t(1-t), we obtain

$$\Big(\sum_{k=0}^{n-2} t^k\Big)I_K(\mathbf{v}) + \Big(\sum_{l=0}^{n-2} (1-t)^l\Big)I_K(\mu) \ge \sum_{j=1}^{n-1} (1-t)^{j-1} t^{n-j-1} \binom{n}{j}I_K(\mu^j, \mathbf{v}^{n-j}).$$

If I_K is convex at μ , then we may take the limit as t goes to 0, which gives us

$$I_K(\mathbf{v}) + (n-1)I_K(\boldsymbol{\mu}) \ge nI_K(\boldsymbol{\mu}^{n-1}, \mathbf{v}).$$

We note that if I_K is convex (in particular, convex at v), then by switching the roles of μ and v we obtain

$$(n-1)I_K(\mathbf{v})+I_K(\boldsymbol{\mu})\geq nI_K(\boldsymbol{\mu},\mathbf{v}^{n-1}).$$

Therefore, in the case n = 2, 3, Lemma 8.2.6 provides the converse of Proposition 8.2.5, in other words, I_K is convex if and only if it satisfies the arithmetic mean inequalities (8.21).

Lemma 8.2.1 with $\mu_1 = \cdots = \mu_k = \mu$ and $\mu_{k+1} = \cdots = \mu_n = \nu$ shows that inequality

(8.21) holds if K is conditionally *n*-positive definite. This leads to the following corollary.

Corollary 8.2.7. If K is conditionally n-positive definite, then I_K is convex on $\mathbb{P}(\Omega)$.

It is not completely clear whether the equivalence from Proposition 3.1.6 holds for $n \ge 3$ in the equivalence holds for $n \ge 3$, but evidence suggests that it does not. Indeed, Proposition 8.6.1 provides an example of a rotationally invariant three-input kernel with $\Omega = \mathbb{S}^{d-1}$, which is not conditionally 3-positive definite, but at the same time the energy functional is convex at σ (although we do not know if it is convex at *all* measures in $\mathbb{P}(\mathbb{S}^{d-1})$) and is minimized by σ , which we know would be impossible in the two-input case.

In this regard, we would also like to point out that a number of our results about energy minimizers do not require the full power of convexity of I_K on $\mathbb{P}(\Omega)$, but rather just the convexity at the presumptive minimizer μ . In particular, condition (8.26), which appears in Theorems 8.3.3 and 8.3.7, is implied by inequality (8.23) of Lemma 8.2.6, and hence it holds if I_K is convex at μ .

Using convexity of the energy functional, one can draw a connection between minimizing the *n*-input energy I_K and the (n-1)-input energy $I_{U_K^{\mu}}$, thus obtaining our first result about minimizers of multi-input energies.

Proposition 8.2.8. Let $n \ge 3$. Assume that $K : \Omega^n \to \mathbb{R}$ is continuous and symmetric, I_K is convex, and that $\mu \in \mathbb{P}(\Omega)$ is a minimizer of $I_{U_K^{\mu}}$. Then μ is a minimizer of I_K .

Proof. We first prove that if the energy I_K is convex and μ , $\nu \in \mathbb{P}(\Omega)$, then

$$I_{K}(\mathbf{v}) - I_{K}(\mu) \ge \frac{n}{n-1} \Big(I_{U_{K}^{\mu}}(\mathbf{v}) - I_{U_{K}^{\mu}}(\mu) \Big).$$
(8.24)

Indeed, we have $I_{U_{K}^{\mu}}(\mu) = I_{K}(\mu)$ and, by Lemma 8.2.6, $I_{U_{K}^{\mu}}(\nu) = I_{K}(\mu, \nu^{n-1}) \leq \frac{1}{n}I_{K}(\mu) + I_{K}(\mu)$

 $\frac{n-1}{n}I_K(v)$. Thus,

$$I_{K}(\mathbf{v}) - I_{K}(\mu) - n \Big(I_{U_{K}^{\mu}}(\mathbf{v}) - I_{U_{K}^{\mu}}(\mu) \Big) = I_{K}(\mathbf{v}) - n I_{K}(\mu, \mathbf{v}^{n-1}) + (n-1) I_{K}(\mu)$$

$$\geq (n-2) \Big(I_{K}(\mu) - I_{K}(\mathbf{v}) \Big),$$

which implies inequality (8.24).

Inequality (8.24), together with the fact that μ is a minimizer of $I_{U_K^{\mu}}$, implies that for all $v \in \mathbb{P}(\Omega)$, we have

$$I_{K}(\mathbf{v}) - I_{K}(\mu) \geq \frac{n}{n-1} \Big(I_{U_{K}^{\mu}}(\mathbf{v}) - I_{U_{K}^{\mu}}(\mu) \Big) \geq 0,$$

hence μ minimizes I_K .

Proposition 8.2.8 can be viewed as a precursor of some of our more advanced results from Section 8.3 which show that there is a strong relation between μ minimizing the *n*input energy I_K and the energy functional $I_{U_K^{\mu k}}$ with a lower number of inputs. In fact, Theorem 8.3.7 contains Proposition 8.2.8 as a special case. We have nevertheless decided to include this proposition, as it admits a very transparent and elementary proof, which also provides a quantitative relation between the minimization of I_K and $I_{U_K^{\mu}}$.

8.3 Minimizers of the Energy Functional

We finally turn to some of the general results about minimizers of energies with multivariate kernels. It is clear from various results in Chapter 3 that in the classical two-input case, properties of minimizing measures are closely related to their potentials. Direct analogues of these statements can be obtained for multi-input energies. We start with the necessary condition, which is a generalization of Theorem 3.1.7 and Corollary 3.1.9, stating that the potential of a local minimizer is constant on its support. As before, in all of the statements of this section we assume that $K : \Omega^n \to \mathbb{R}$ is continuous and symmetric, even if not explicitly stated.

We define a local minimizer the same way we did for the two-input case:

Definition 8.3.1. We shall say that μ is a **local minimizer** of I_K if it is a local minimizer in every direction, in other words, if for each $\nu \in \mathbb{P}(\Omega)$, there exists $t_{\nu} \in (0,1]$ such that for all $t \in [0, t_{\nu}]$ we have

$$I_K((1-t)\mu+t\nu)\geq I_K(\mu).$$

Observe that this definition differs from the definition of local minimizers with respect to some metric, such as the Wasserstein d_{∞} metric or the total variation norm (the difference is similar to that between the Gateaux and Fréchet derivatives).

Theorem 8.3.2. Let $K : \Omega^n \to R$ be continuous and symmetric. Suppose that μ is a local minimizer of I_K over $\mathbb{P}(\Omega)$. Then $U_K^{\mu^{n-1}}(x) = I_K(\mu)$ on $\operatorname{supp}(\mu)$ and $U_K^{\mu^{n-1}}(x) \ge I_K(\mu)$ on Ω .

Proof. The proof is a simple extension of the proof of Theorem 3.1.7 and we include it for the sake of completeness. Let $v \in \mathbb{P}(\Omega)$, t_v be as in Definition 8.3.1, and $\tilde{v} = v - \mu \in \mathscr{Z}(\Omega)$. Then for all $0 \le t \le 1$ and $\mu + t\tilde{v} = (1-t)\mu + tv \in \mathbb{P}(\Omega)$, and satisfies

$$I_K(\mu) \leq I_K(\mu + t\tilde{\nu}) = \sum_{k=0}^n \binom{n}{k} t^k I_K(\mu^{n-k}, \tilde{\nu}^k).$$

Thus, for $0 \le t \le t_V$,

$$0 \leq t \left(\sum_{k=1}^n \binom{n}{k} t^{k-1} I_K(\mu^{n-k}, \tilde{\nu}^k) \right).$$

This means that $I_K(\mu^{n-1}, \tilde{v}) \ge 0$.

Suppose, indirectly, that there exist $a, b \in \mathbb{R}$, $z \in \text{supp}(\mu)$ and $y \in \Omega$ such that

$$a = U_K^{\mu^{n-1}}(z) > U_K^{\mu^{n-1}}(y) = b.$$

Let *B* be a ball centered at *z*, small enough so that $y \notin B$ and oscillation of $U_K^{\mu^{n-1}}(x)$ is at

most $\frac{a-b}{2}$, and let $m = \mu(B)$. Let v be defined by

$$\mathbf{v}(A) = \boldsymbol{\mu}(A) + \boldsymbol{m}\boldsymbol{\delta}_{\mathbf{y}}(A) - \boldsymbol{\mu}(A \cap B). \tag{8.25}$$

Then

$$I_{K}(\mu^{n-1},\tilde{v}) = U_{K}^{\mu^{n-1}}(y) \cdot m - \int_{B} U_{K}^{\mu^{n-1}}(x) d\mu(x) \le bm - \left(a - \frac{a-b}{2}\right)m < 0,$$

which is a contradiction. Thus, if $U_K^{\mu^{n-1}}(z) = a$ for some $z \in \text{supp}(\mu)$, then $U_K^{\mu^{n-1}}(x) \ge a$ for all $x \in \Omega$. Our claim then follows.

In general, the converse to Theorem 8.3.2 is not true. However, with some additional convexity assumptions, the necessary condition for μ to be a global minimizer also becomes sufficient.

Theorem 8.3.3. Let $K : \Omega^n \to \mathbb{R}$ be symmetric and continuous. Suppose that for some $\mu \in \mathbb{P}(\Omega)$, there exists a finite constant M such that $U_K^{\mu^{n-1}}(x) \ge M$ on Ω and $U_K^{\mu^{n-1}}(x) = M$ on $\operatorname{supp}(\mu)$. Suppose further that for all $v \in \mathbb{P}(\Omega)$, there exists some $a \in (0,1)$, possibly depending on v, such that

$$I_K(\mu^{n-1}, \nu) \le aI_K(\nu) + (1-a)I_K(\mu).$$
(8.26)

Then μ is a minimizer of I_K .

Proof. For any $v \in \mathbb{P}(\Omega)$, for some $a \in (0, 1)$, we have

$$I_{K}(\mu) = \int_{\Omega} U_{K}^{\mu^{n-1}}(x) d\mu(x) \leq \int_{\Omega} U_{K}^{\mu^{n-1}}(x) d\nu(x) = I_{K}(\mu^{n-1}, \nu) \leq aI_{K}(\nu) + (1-a)I_{K}(\mu),$$

hence $I_K(\mu) \leq I_K(\nu)$.

Some remarks concerning the assumptions of Theorem 8.3.3, i.e. condition (8.26), are

in order. Due to Lemma 8.2.6, convexity of the energy functional I_K at μ implies condition (8.26) with $a = \frac{1}{n}$. In turn, if *K* is conditionally *n*-positive definite, Corollary 8.2.7 states that I_K is convex, and hence again condition (8.26) is satisfied (alternatively, Lemma 8.2.1 shows directly that conditional *n*-positive definiteness of *K* implies the convexity condition (8.26) of Theorem 8.3.3 with $a = \frac{1}{n}$). The hierarchy of these conditions can be summarized in the following diagram:

K is *n*-positive definite
$$\implies$$
 K is conditionally *n*-positive definite \implies (8.27)
 $\implies I_K$ is convex $\implies I_K$ is convex at $\mu \implies$ condition (8.26) holds.

Therefore, Theorem 8.3.3 (as well as other statements relying on (8.26), e.g. Lemma 8.3.5 or Theorem 8.3.7) may be applied under the assumptions that *K* is (conditionally) *n*-positive definite or that I_K is convex at μ .

We also make the following remark: in the case when μ has full support, i.e. $\supp(\mu) = \Omega$, if the first condition of Theorem 8.3.3 holds, i.e. $U_K^{\mu^{n-1}}(x) = M$ for all $x \in \Omega$, then $I_K(\mu^{n-1}, \nu) = I_K(\mu)$, and the assumption (8.26) is obviously the same as the conclusion of Theorem 8.3.3. This does not, however, render this case of the theorem useless – on the contrary, if one replaces (8.26) with one of the stronger conditions in (8.27), one obtains an interesting and meaningful statement. (This shows that most of the content is hidden in the implications presented in (8.27).) We summarize this case in a separate corollary, as it will be of use later.

Corollary 8.3.4. Let $K : \Omega^n \to \mathbb{R}$ be symmetric and continuous. Suppose that $\mu \in \mathbb{P}(\Omega)$ has full support ($\operatorname{supp}(\mu) = \Omega$) and that there exists a constant M such that $U_K^{\mu^{n-1}}(x) = M$ on Ω . Assume also that any of the conditions in (8.27) holds (e.g., K is n-positive definite or I_K is convex). Then μ is a minimizer of I_K .

We also observe that Theorems 8.3.2 and 8.3.3 imply the following local-to-global principle for minimizers of I_K under convexity assumptions.

Lemma 8.3.5. Let $n \ge 2$ and let μ be a local minimizer of the energy functional I_K . Assume also that condition (8.26) is satisfied. Then μ is a global minimizer of I_K over $\mathbb{P}(\Omega)$.

Proof. Theorem 8.3.2 shows that the first condition of Theorem 8.3.3 holds. Together with condition (8.26), this implies that μ is a global minimizer of I_K .

Naturally, the set of minimizers of a convex functional is convex. By Corollary 8.2.7, for conditionally *n*-positive definite kernels, the energy I_K is convex, i.e. minimizers of I_K form a convex set in this case.

Proposition 8.3.6. Let K be a conditionally n-positive definite kernel. Then the set of minimizers of the energy I_K is convex.

Proof. For two minimizers μ , ν of I_K , and all $t \in [0, 1]$, by convexity of energy,

$$I_K(\mathbf{v}) \leq I_K((1-t)\mu + t\mathbf{v}) \leq (1-t)I_K(\mu) + tI_K(\mathbf{v}) = I_K(\mathbf{v}).$$

Hence,
$$I_K((1-t)\mu + t\nu) = I_K(\nu)$$
 and $(1-t)\mu + t\nu$ minimizes I_K .

While Theorems 8.3.2 and 8.3.3 are straightforward generalizations of the corresponding facts for the classical two-input energies, they lead to some interesting consequences for energies with multivariate kernels. In particular, we start by showing that under condition (8.26), if μ minimizes the lower input energy with the kernel $U_K^{\mu k}$, then it also minimizes the original *n*-input energy I_K .

Theorem 8.3.7. Let $K : \Omega^n \to \mathbb{R}$, $n \ge 3$, be symmetric and continuous. Assume that for some $1 \le k \le n-2$, the measure $\mu \in \mathbb{P}(\Omega)$ (locally) minimizes the (n-k)-input energy $I_{U_K^{\mu^k}}$. Assume also that μ satisfies condition (8.26) of Theorem 8.3.3. Then μ minimizes the *n*-input energy I_K .

Proof. Theorem 8.3.2 applied to the kernel $U_K^{\mu^k}$ implies that for all $x \in \Omega$

$$U_{K}^{\mu^{n-1}}(x) = U_{U_{K}^{\mu^{k}}}^{\mu^{n-k-1}}(x) \ge I_{U_{K}^{\mu^{k}}}(\mu) = I_{K}(\mu)$$

with equality for $x \in \text{supp}(\mu)$. Condition (8.26) then allows one to invoke Theorem 8.3.3, which shows that μ minimizes I_K .

The converse to Theorem 8.3.7 holds for k = n - 2 even without any convexity assumptions: in this case, if μ locally minimizes I_K , it also locally minimizes the two-input energy $I_{U_K^{\mu^{n-2}}}$. Moreover, under the additional condition that μ has full support, one can deduce that the measure μ is a *global* minimizer of $I_{U_K^{\mu^{n-2}}}$, which follows from Theorem 3.3.1, though we prove it directly here. Furthermore, this implication may be reversed, if one additionally assumes that μ uniquely minimizes $I_{U_K^{\mu^{n-2}}}$. Observe that, unlike Theorem 8.3.7, part (3) of Theorem 8.3.8 does not require any of the conditions of (8.27) and, unlike part (2), it does not require the condition supp(μ) = Ω .

Theorem 8.3.8. Let $K : \Omega^n \to \mathbb{R}$, $n \ge 3$, be symmetric and continuous and let $\mu \in \mathbb{P}(\Omega)$.

- 1. Let μ be a local minimizer of I_K . Then μ is a local minimizer of the two-input energy $I_{U_{\nu}^{\mu^{n-2}}}$.
- 2. Let μ be a local minimizer of I_K and assume, in addition, that μ has full support, i.e. supp $(\mu) = \Omega$. Then μ minimizes the two-input energy $I_{U_{\nu}^{\mu^{n-2}}}$ over $\mathbb{P}(\Omega)$.
- 3. If μ is the unique minimizer of $I_{U_{\kappa}^{\mu^{n-2}}}$ in $\mathbb{P}(\Omega)$, then μ is a local minimizer of I_{K} .

Proof. Fix an arbitrary measure $v \in \mathbb{P}(\Omega)$. For $t \in [0, 1]$, let us define two functions $g_v(t) = I_K((1-t)\mu + tv)$ and $h_v(t) = I_{U_K^{\mu^{n-2}}}((1-t)\mu + tv) = I_K(\mu^{n-2}, ((1-t)\mu + tv)^2)$. We have

$$g_{\nu}(t) = (1-t)^{n} I_{K}(\mu) + nt(1-t)^{n-1} I_{K}(\mu^{n-1},\nu) + {\binom{n}{2}} t^{2} (1-t)^{n-2} I_{K}(\mu^{n-2},\nu^{2}) + R_{\nu}(t),$$

where each term in $R_v(t)$ contains a factor of the form t^k with $k \ge 3$ and, therefore, $R'_v(0) = R''_v(0) = 0$,

$$h_{\mathbf{v}}(t) = (1-t)^2 I_K(\mu) + 2t(1-t)I_K(\mu^{n-1}, \mathbf{v}) + t^2 I_K(\mu^{n-2}, \mathbf{v}^2).$$

A direct (elementary, but lengthy) computation, which we omit, shows that

$$h'_{\nu}(0) = \frac{2}{n}g'_{\nu}(0) = 2(I_{K}(\mu^{n-1},\nu) - I_{K}(\mu)), \qquad (8.28)$$

$$h_{\nu}''(0) = \frac{2}{n(n-1)}g_{\nu}''(0) = 2(I_K(\mu) - 2I_K(\mu^{n-1}, \nu) + I_K(\mu^{n-2}, \nu^2)).$$
(8.29)

We now start by proving (1). Let μ be a local minimizer of I_K . According to Corollary 3.1.9, we have that $U_K^{\mu^{n-1}}(x) \ge I_K(\mu)$ on Ω and therefore, $I_K(\mu^{n-1}, v) \ge I_K(\mu)$ for any $v \in \mathbb{P}(\Omega)$. Since g_v has a local minimum at t = 0, either $g'_v(0) > 0$, or $g'_v(0) = 0$ and $g''_v(0) \ge 0$. In the first case, we also have $h'_v(0) > 0$. In the second case, $h'_v(0) = 0$ and $h''_v(0) \ge 0$, and since h_v is quadratic, this implies that $h_v(t) = at^2 + b$ with $a \ge 0$. Thus, h_v has a local minimum at t = 0 for each $v \in \mathbb{P}(\Omega)$, i.e. μ is a local minimizer of $I_{U_k}^{\mu^{n-2}}$.

If in addition μ has full support, then Corollary 3.1.9 implies that for any $v \in \mathbb{P}(\Omega)$, we have $I_K(\mu^{n-1}, v) = I_K(\mu)$. Therefore, relations (8.28)-(8.29), together with the fact that g_v has a local minimum at t = 0, show that $g'_v(0) = 0$, hence $g''_v(0) \ge 0$, and at the same time

$$g_{\nu}''(0) = n(n-1) \left(I_{K}(\mu^{n-2},\nu^{2}) - I_{K}(\mu) \right) = n(n-1) \left(I_{U_{K}}^{\mu^{n-2}}(\nu) - I_{U_{K}}^{\mu^{n-2}}(\mu) \right).$$
(8.30)

Hence, μ is a global minimizer of $I_{U_{\nu}^{\mu^{n-2}}}$, which proves part (2).

To prove (3), assume that μ is the unique global minimizer of $I_{U_K^{\mu^{n-2}}}$. Observe that, since the potential of $U_K^{\mu^{n-2}}$ with respect to μ is $U_K^{\mu^{n-1}}$, Theorem 8.3.2 applied to $U_K^{\mu^{n-2}}$ implies that, just like in part (1), we have $I_K(\mu^{n-1}, \mathbf{v}) \ge I_K(\mu)$. Thus, $g'_V(0) \ge 0$ by (8.28). If $g'_V(0) > 0$, there is a local minimum at t = 0. If, however, $g'_V(0) = 0$, then $I_K(\mu^{n-1}, \mathbf{v}) = I_K(\mu)$ and relation (8.30) holds. Since μ uniquely minimizes $I_{U_K^{\mu^{n-2}}}$, this proves that $g''_V(0) > 0$ for $\mathbf{v} \ne \mu$. Hence, in each case, g_V has a local minimum at t = 0, i.e. μ is a local minimizer of I_K .

Applying Theorem 3.3.1, we obtain the following corollary to part (2) of Theorem 8.3.8:

Corollary 8.3.9. Assume that $\mu \in \mathbb{P}(\Omega)$ with $\operatorname{supp}(\mu) = \Omega$ is a local minimizer of I_K . Then the (n-2)-fold potential of K with respect to μ , i.e. the two-variable function $U_K^{\mu^{n-2}}(x,y)$, is conditionally positive definite on Ω .

Observe that, if the kernel *K* is conditionally *n*-positive definite, then, according to Lemma 8.1.2, $U_K^{\mu^{n-2}}(x,y)$ is conditionally positive definite. Moreover, Theorem 8.3.7 applies for conditionally positive definite kernels *K*. Therefore, the statement of Corollary 8.3.9 may be viewed as a partial converse of Theorem 8.3.7 for conditionally positive definite kernels. This interplay will manifest itself in an even stronger fashion on the sphere, the situation to be explored in Section 8.4.

8.4 Multi-input Energy on the Sphere

We now restrict our attention to the unit sphere \mathbb{S}^{d-1} , where the symmetries and structure of the domain allow one to deduce additional information about energy minimization. One of the most natural questions is whether the normalized uniform surface measure σ minimizes the energy functional over $\mathbb{P}(\mathbb{S}^{d-1})$, or, in other words, whether energy minimization induces uniform distribution.

In this section, we shall be interested in kernels, which (in addition to being continuous and symmetric) are *rotationally invariant*, i.e. have the form

$$K(x_1,\ldots,x_n) = F\left((\langle x_i,x_j \rangle)_{i,j=1}^n\right),\tag{8.31}$$

in other words, they depend only on the Gram matrix of $\{x_1, ..., x_n\} \subset \mathbb{S}^{d-1}$. When n = 2, one obtains classical pairwise interaction kernels of the form $K(x,y) = F(\langle x, y \rangle)$, which has been thoroughly discussed in previous chapters. In the case n = 3 rotationally invariant kernels are functions of the form

$$K(x, y, z) = F(\langle x, y \rangle, \langle x, z \rangle, \langle z, y \rangle) = F(t, u, v),$$
(8.32)

where we set $t = \langle x, y \rangle$, $u = \langle x, z \rangle$, $v = \langle z, y \rangle$, and we shall keep this notation, which comes from [BV08], throughout the text.

Observe that, if the *n*-input kernel *K* is rotationally invariant, its potential with respect to σ is again rotationally invariant. Indeed, for any $V \in SO(d)$, we have

$$U_K^{\sigma}(Vx_1, \dots, Vx_{n-1}) = U_K^{\sigma}(x_1, \dots, x_{n-1}),$$
(8.33)

which easily follows from (8.31) and the facts that $\langle x_i, x_j \rangle = \langle Vx_i, Vx_j \rangle$ and $\langle Vx_i, x_n \rangle = \langle x_i, V^{-1}x_n \rangle$, $1 \le i, j \le n-1$, together with the rotational invariance of σ , i.e. $d\sigma(x_n) = d\sigma(V^{-1}x_n)$. Iterating this observation, one finds that all *k*-fold potentials of *K* with respect to σ , i.e. functions $U_K^{\sigma^k}$ with $1 \le k \le n-1$, are rotationally invariant. In particular, when k = n-2, the two-input kernel $U_K^{\sigma^{n-2}}$ depends only on the inner product of the inputs, and for k = n-1, the potential $U_K^{\sigma^{n-1}}$ is just a constant:

$$U_K^{\sigma^{n-2}}(x,y) = G(\langle x,y \rangle) = G(t) \text{ and } U_K^{\sigma^{n-1}}(x) = \text{const} = I_K(\sigma).$$
(8.34)

Recall that Theorem 8.3.2 would guarantee the latter condition in the case when σ is a minimizer of I_K . However, for rotationally invariant kernels, this is automatically satisfied, which facilitates the application of the results of Section 8.3 and will play an important role later, in Theorem 8.4.1.

Turning to the primary task of understanding when σ minimizes I_K , we first remind ourselves that in the classical case of a two-input energy with a rotationally invariant kernel $G(\langle x, y \rangle)$ on \mathbb{S}^{d-1} , the answer to this question is well understood, see, e.g. Theorem 3.3.1 and Proposition 3.5.1. In particular, the following three conditions are equivalent:

- 1. The uniform surface measure σ minimizes I_G over $\mathbb{P}(\mathbb{S}^{d-1})$.
- 2. The kernel *G* is conditionally positive definite on \mathbb{S}^{d-1} .
- 3. The kernel G is positive definite on \mathbb{S}^{d-1} up to a constant term, i.e. there exists a
constant $c \in \mathbb{R}$ such that G + c is positive definite on \mathbb{S}^{d-1} (in fact, one can take $c = -I_G(\sigma)$).

Our goal is to generalize these statements (at least partially) to the case of multi-input energies. We observe that, if a symmetric rotationally invariant kernel *K* is conditionally *n*-positive definite on \mathbb{S}^{d-1} , then, according to Lemma 8.1.2, the potential $G(\langle x, y \rangle) = U_K^{\sigma^{n-2}}(x,y)$ is also conditionally positive definite, and hence, by the discussion above, σ is a minimizer of the two-input energy $I_{U_K^{\sigma^{n-2}}}$. Therefore, since conditionally *n*-positive definite kernels satisfy condition (8.26), Theorem 8.3.7 with k = n - 2 applies and we obtain the following statement:

Theorem 8.4.1. Suppose that $K : (\mathbb{S}^{d-1})^n \to \mathbb{R}$ is continuous, symmetric, rotationally invariant, and conditionally n-positive definite on \mathbb{S}^{d-1} . Then σ is a minimizer of I_K over $\mathbb{P}(\Omega)$.

This theorem also easily follows from Theorem 8.3.3 and the remarks thereafter (or, more precisely, from Corollary 8.3.4), since, as explained above, the potential $U_K^{\sigma^{n-1}}$ is constant on \mathbb{S}^{d-1} . In addition, we notice that, unlike some statements of Section 8.3, such as Theorem 8.3.7, for rotationally invariant kernels in the theorem above, one does not need to assume anything about energies with a lower number of inputs – conditional *n*-positive definiteness alone suffices.

Theorem 8.4.1 immediately yields some interesting examples:

Corollary 8.4.2. Let $f : [-1,1] \to \mathbb{R}$ be a real-analytic function with nonnegative Maclaurin coefficients and let F(t,u,v) = f(tuv). Then, for K defined as in (8.32), the uniform surface measure σ minimizes the energy I_K over $\mathbb{P}(\mathbb{S}^{d-1})$.

Proof. Observe first that in this setup, if K_z is positive definite for one point $z \in \mathbb{S}^{d-1}$, it is also positive definite for each $z \in \mathbb{S}^{d-1}$ due to rotational invariance, i.e. Definition 8.1.1 only needs to be checked at one point. Consider first F(t, u, v) = tuv and fix any $z \in \mathbb{S}^{d-1}$,

e.g., $z = e_1$. Then for any $v \in \mathcal{M}(\mathbb{S}^{d-1})$,

$$I_{K_{e_1}}(\mathbf{v}) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \langle x, y \rangle x_1 y_1 d\mathbf{v}(x) d\mathbf{v}(y) = \sum_{i=1}^d \left(\int_{\mathbb{S}^{d-1}} x_1 x_i d\mathbf{v}(x) \right)^2 \ge 0,$$

i.e. the kernel $K(x, y, z) = \langle x, y \rangle \langle x, z \rangle \langle z, y \rangle = tuv$ is 3-positive definite, and hence, by Lemma 8.1.3, so are all of its integer powers, positive linear combinations and their limits. The conclusion now follows from Theorem 8.4.1.

This corollary provides a whole array of examples: for instance, three-input energies with kernels K(x, y, z) = tuv, or $(tuv)^n$, or e^{tuv} are all minimized by σ . We remark that, while for K = tuv this statement could be proved using semidefinite programming methods, for higher powers $(tuv)^n$ this would be extremely difficult technically, and for kernels like e^{tuv} almost impossible.

For even exponents, the energies with the kernels $K = (uvt)^{2k}$ can be viewed as threeinput generalizations of the well-known *p*-frame potentials discussed in Chapter 6. We also point out that Proposition 8.5.2 provides a more general class of *n*-positive definite kernels, which contains K = tuv as a special case.

Unfortunately, unlike the classical two-input case, the converse to Theorem 8.4.1 is not true: Propositions 8.6.1, 8.6.2, 8.6.3, and 8.6.4 show that some kernels, naturally arising in semidefinite programming and geometry, fail to be conditionally *n*-positive definite, even though σ minimizes corresponding energies (see Theorems 8.7.1, 8.8.1, and 8.8.3). In other words, conditional *n*-positive definiteness of the kernel is not equivalent to the fact that σ minimizes the energy.

We suspect that the property that σ minimizes I_K is equivalent to $U_K^{\sigma^{n-2}}$ being conditionally positive definite, i.e. the two-input energy $I_{U_K^{\sigma^{n-2}}}$ is minimized by σ . This conjecture is supported by all examples currently known to us. Conditional positive-definiteness of $U_K^{\sigma^{n-2}}$ obviously follows from conditional *n*-positive definiteness of *K*, due to Lemma 8.1.2, but the converse implication is not true. In fact, all the kernels discussed in Section 8.6 possess this property: they are not 3-positive definite, but their potentials U_K^{σ} with respect to σ are (conditionally) positive definite, and the corresponding energies I_K are minimized by σ .

Theorem 8.4.3 below (which is essentially a restatement of Theorem 8.3.8 for the spherical case, along with the fact that σ has full support) shows that conditional positive definiteness of $U_K^{\sigma^{n-2}}$ is implied if σ is a *local* minimizer of I_K , and a partial converse to this statement also holds. Observe that, if the conjecture above is true, then being a local and global minimizer are equivalent for σ , a fact is indeed true for the two-input energies (Theorem 3.3.1).

Theorem 8.4.3. Let $K : (\mathbb{S}^{d-1})^n \to \mathbb{R}$ be a continuous, symmetric, and rotationally invariant kernel.

- 1. Assume that σ is a local minimizer of I_K in $\mathbb{P}(\mathbb{S}^{d-1})$. Then the uniform measure σ is a global minimizer of the two-input energy $I_{U_K^{\sigma^{n-2}}}$, or, equivalently, $U_K^{\sigma^{n-2}}$ is conditionally positive definite on the sphere \mathbb{S}^{d-1} .
- 2. Assume that σ is the unique global minimizer of $I_{U_K^{\sigma^{n-2}}}$ over $\mathbb{P}(\mathbb{S}^{d-1})$. Then σ is a local minimizer of the n-input energy I_K .

Theorem 8.4.3 above shows that if σ is a global minimizer of I_K , then the potential $U_K^{\sigma^{n-2}}$ is conditionally positive definite. We do not know whether the converse of this statement holds. One can show, however, at least for n = 3 that if σ minimizes $I_{U_K^{\sigma}}$ (in other words, U_K^{σ} is conditionally positive definite), but fails to minimize I_K , then the global minimizer of I_K cannot be supported on the whole sphere.

Lemma 8.4.4. Let $K : (\mathbb{S}^{d-1})^3 \to \mathbb{R}$ be a continuous, symmetric, and rotationally invariant three-input kernel. Assume that U_K^{σ} is conditionally positive definite on the sphere \mathbb{S}^{d-1} (i.e. σ minimizes $I_{U_K^{\sigma}}$), but at the same time σ is not a minimizer of I_K . Let μ be a minimizer of I_K . Then $\operatorname{supp}(\mu) \subsetneq \mathbb{S}^{d-1}$. *Proof.* Assume, by contradiction, that $\operatorname{supp}(\mu) = \mathbb{S}^{d-1}$. Then, by Theorem 8.3.2, $U_K^{\mu^2}(x) = I_K(\mu)$ for every $x \in \mathbb{S}^{d-1}$, and therefore,

$$I_{U_K^{\sigma}}(\mu) = I_K(\mu,\mu,\sigma) = \int_{\mathbb{S}^{d-1}} U_K^{\mu^2}(x) d\sigma(x) = I_K(\mu).$$

On the other hand, obviously, $I_K(\sigma) = I_{U_K^{\sigma}}(\sigma)$. Since μ is a minimizer of I_K , and σ is not, we have $I_K(\mu) < I_K(\sigma)$. This implies that $I_{U_K^{\sigma}}(\mu) < I_{U_K^{\sigma}}(\sigma)$, which contradicts the conditional positive definiteness of U_K^{σ} .

8.5 Positive Definite Kernels

Corollary 8.4.2 of the previous section already provided a class of 3-positive definite functions. In this section we provide several other classes of kernels that are (conditionally) *n*-positive definite.

General Classes of (Conditionally) *n*-positive Definite Kernels

We start with some very natural examples, which show how to construct (conditionally) n-positive definite kernels from kernels with fewer inputs. In particular, we show that an n-input kernel can be constructed from m-input ones, m < n, by considering the sum or product over all m-element subsets of inputs. We first deal with the statement about the sum.

Proposition 8.5.1. Let $2 \le m \le n-1$, and suppose $H : \Omega^m \to \mathbb{R}$ is continuous, symmetric, and conditionally *m*-positive definite. Then

$$K(z_1,...,z_n) := \sum_{1 \le j_1 < j_2 < \cdots < j_m \le n} H(z_{j_1}, z_{j_2}, ..., z_{j_m})$$

is conditionally n-positive definite.

Proof. Let v be a finite signed Borel measure on Ω such that $v(\Omega) = 0$. Then for any fixed $z_1, ..., z_{n-2} \in \Omega$, since H is conditionally *m*-positive definite, we have

$$\begin{split} \int_{\Omega} \int_{\Omega} K(z_1, ..., z_{n-2}, x, y) d\mathbf{v}(x) d\mathbf{v}(y) &= \int_{\Omega} \int_{\Omega} \sum_{\Omega = 1 \le l_1 < \dots < l_m \le n-2} H(z_{l_1}, ..., z_{l_m}) d\mathbf{v}(x) d\mathbf{v}(y) \\ &+ \int_{\Omega} \int_{\Omega} \sum_{1 \le k_1 < \dots < k_{m-1} \le n-2} \left(H(z_{k_1}, ..., z_{k_{m-1}}, x) + H(z_{k_1}, ..., z_{k_{m-1}}, y) \right) d\mathbf{v}(x) d\mathbf{v}(y) \\ &+ \int_{\Omega} \int_{\Omega} \sum_{1 \le j_1 < \dots < j_{m-2} \le n-2} H(z_{j_1}, ..., z_{j_{m-2}}, x, y) d\mathbf{v}(x) d\mathbf{v}(y) \\ &= \sum_{1 \le j_1 < \dots < j_{m-2} \le n-2} \int_{\Omega} \int_{\Omega} H(z_{j_1}, ..., z_{j_{m-2}}, x, y) d\mathbf{v}(x) d\mathbf{v}(y) \ge 0, \end{split}$$

which shows that *K* is conditionally *n*-positive definite.

We can also prove an analogue of Proposition 8.5.1 for products of positive definite functions.

Proposition 8.5.2. Let $2 \le m \le n-1$ and assume that $H : \Omega^m \to \mathbb{R}$ is continuous, symmetric, and *m*-positive definite. If *H* is a nonnegative function or m = n-1, then

$$K(z_1,...,z_n) = \prod_{1 \le j_1 < \cdots < j_m \le n} H(z_{j_1},...,z_{j_m})$$

is n-positive definite.

Proof. Fix $z_1, \ldots, z_{n-2} \in \Omega$. We can write

$$K(z_1, \dots, z_{n-2}, x, y) = \prod_{1 \le j_1 < \dots < j_m \le n-2} H(z_{j_1}, \dots, z_{j_m})$$
(8.35)

$$\times \prod_{1 \le j_1 < \dots < j_{m-1} \le n-2} H(z_{j_1}, \dots, z_{j_{m-1}}, x)$$
(8.36)

$$\times \prod_{1 \le j_1 < \dots < j_{m-1} \le n-2} H(z_{j_1}, \dots, z_{j_{m-1}}, y)$$
(8.37)

×
$$\prod_{1 \le j_1 < \dots < j_{m-2} \le n-2} H(z_{j_1}, \dots, z_{j_{m-2}}, x, y).$$
 (8.38)

Observe that the product in line (8.35) is non-negative when $H \ge 0$ or if m = n - 1 (the product is empty in the latter case). The product of lines (8.36) and (8.37) is positive definite as a function of x and y: indeed, it has the form $F(x, y) = \phi(x)\phi(y)$ and hence

$$I_F(\mu) = \left(\int\limits_{\Omega} \phi(x) d\mu(x)\right)^2 \ge 0$$

for any $\mu \in \mathscr{M}(\Omega)$. Finally, every factor in the product in line (8.38) is positive definite as a function of *x* and *y*, because *H* is *m*-positive definite. Thus, Schur's product theorem (see Lemma 8.1.3) ensures that the whole product is positive definite as a function of *x* and *y*, therefore, *K* is *n*-positive definite.

Propositions 8.5.1 and 8.5.2 provide us with large classes of *n*-positive definite kernels. However, these constructions do not exhaust all such kernels. In the following subsection, we provide examples of three-positive definite kernels on the sphere, which are not obtained from two-input kernels by the methods described above.

Three-positive Definite Kernels on the Sphere

We use the same notation as in Section 8.4: for $x, y, z \in \mathbb{S}^{d-1}$, we set $t = \langle x, y \rangle$, $u = \langle x, z \rangle$, and $v = \langle z, y \rangle$. In Corollary 8.4.2, we showed that K = tuv is 3-positive definite on the sphere. Observe that this is a specific case of Proposition 8.5.2 above, since $\langle x, y \rangle$ is a positive definite function on \mathbb{S}^{d-1} . More generally, Proposition 8.5.2 implies that any kernel of the form K(x,y,z) = h(t)h(u)h(v) is 3-positive definite, as long as *h* is a positive definite function on the sphere.

The kernels considered in Lemmas 8.5.3 and 8.5.4 are closely related to the parallelepiped spanned by the vectors x, y, and $z \in \mathbb{S}^{d-1}$. Indeed, setting a = 2 in (8.39), one obtains negative volume squared of this parallelepiped: this kernel is not conditionally 3-positive definite according to Proposition 8.6.3, even though σ is a minimizer of the corresponding energy, as shown in Theorem 8.8.1. However, positive definiteness does hold for other values of the parameter a.

Lemma 8.5.3. *For a* < 1,

$$K(x, y, z) = t^{2} + u^{2} + v^{2} - auvt + \frac{1}{1 - a}$$
(8.39)

is 3-positive definite.

Proof. Due to rotational invariance, we need only check one value of z. Let $z = e_1$. We

have that

$$\begin{split} K(x,y,e_1) &= \langle x,y \rangle^2 + x_1^2 + y_1^2 - ax_1y_1 \langle x,y \rangle + \frac{1}{1-a} \\ &= \left(\langle x,y \rangle^2 - ax_1y_1 \langle x,y \rangle - (1-a)x_1^2y_1^2 \right) + (1-a)x_1^2y_1^2 + x_1^2 + y_1^2 + \frac{1}{1-a} \\ &= \left(\langle x,y \rangle^2 - ax_1y_1 \langle x,y \rangle - (1-a)x_1^2y_1^2 \right) \\ &+ \left(x_1^2 \sqrt{1-a} + \frac{1}{\sqrt{1-a}} \right) \left(y_1^2 \sqrt{1-a} + \frac{1}{\sqrt{1-a}} \right) \\ &= \sum_{j=2}^d \sum_{k=2}^d x_j y_j x_k y_k + (2-a) \sum_{m=2}^d x_1 y_1 x_m y_m \\ &+ \left(x_1^2 \sqrt{1-a} + \frac{1}{\sqrt{1-a}} \right) \left(y_1^2 \sqrt{1-a} + \frac{1}{\sqrt{1-a}} \right). \end{split}$$

We quickly see that for any finite signed Borel measure $v \in \mathscr{M}(\mathbb{S}^{d-1})$,

$$\int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} K(x, y, e_1) d\mathbf{v}(x) d\mathbf{v}(y) = \sum_{j=2}^d \sum_{k=2}^d \left(\int_{\mathbb{S}^{d-1}} x_j x_k d\mathbf{v}(x) \right)^2 + (2-a) \sum_{m=2}^d \left(\int_{\mathbb{S}^{d-1}} x_1 x_m d\mathbf{v}(x) \right)^2 + \left(\int_{\mathbb{S}^{d-1}} \left(x_1^2 \sqrt{1-a} + \frac{1}{\sqrt{1-a}} \right) d\mathbf{v}(x) \right)^2 \ge 0,$$

hence, *K* is 3-positive definite.

Lemma 8.5.4. For $a \le 1$, $K(x, y, z) = t^2 + u^2 + v^2 - auvt$ is conditionally 3-positive definite.

Proof. For a < 1, according to Lemma 8.5.3, $K + \frac{1}{1-a}$ is 3-positive definite. Thus, for any fixed $z \in \mathbb{S}^{d-1}$ and any $v \in \mathcal{M}(\mathbb{S}^{d-1})$ with $v(\mathbb{S}^{d-1}) = 0$,

$$I_{K_z}(\mathbf{v}) = I_{K_z + \frac{1}{1-a}}(\mathbf{v}) \ge 0,$$

i.e. *K* is conditionally 3-positive definite. Lemma 8.1.3 then gives the result for a = 1. \Box

8.6 Some Counterexamples

While our results provide new and less complicated means to determine minimizers for a wide range of kernels, it is clear that more general ideas are necessary to categorize all kernels on the sphere for which σ is a minimizer. In this subsection, we present naturally arising kernels on the sphere which are *not* conditionally 3-positive definite on the sphere, but yet the three-input energies generated by these kernels are minimized by the uniform measure σ .

The semidefinite programming methods of Bachoc and Vallentin [BV08], which we will discuss in Section 8.7 in the context relevant to this paper, are more computationally difficult than ours, and would likely be infeasible for non-polynomial kernels in the context relevant to this paper. At the same time, they apply to certain functions which are not covered by our methods from Section 8.4. In particular, Theorem 8.7.1 shows that the energies with kernels given by polynomials

$$S_{0,1,1}^d(x,y,z) = uv + vt + tu$$
(8.40)

and

$$S_{1,0,0}^d(x,y,z) = (t - uv) + (u - vt) + (v - tu)$$
(8.41)

are both minimized by σ . However, neither function is conditionally 3-positive definite, as we demonstrate below. This implies that the converse to Theorem 8.4.1 does not hold. In addition, the potential of both kernels with respect to σ is a positive definite two-input kernel, which provides evidence that this might indeed be the correct necessary and sufficient condition for σ to minimize the three-input energy (see the discussion before Theorem 8.4.3).

The former example (8.40) is particularly interesting, since the energy functional with this kernel is convex at the minimizer σ , which suggests that conditional *n*-positive defi-

niteness and convexity of the energy functional are perhaps not equivalent for $n \ge 3$, unlike the two-input case (see Proposition 3.1.6). We summarize these properties in the following proposition:

Proposition 8.6.1. Let $\Omega = \mathbb{S}^{d-1}$ and set

$$K(x, y, z) = S_{0,1,1}^d(x, y, z) = uv + vt + tu.$$

The kernel K satisfies the following:

- 1. the uniform measure σ minimizes the energy I_K ,
- 2. the energy functional I_K is convex at σ ,
- 3. $U_K^{\sigma}(x,y)$ is positive definite,
- 4. K is not conditionally 3-positive definite.

Proof. As mentioned above, part (1) follows from the semidefinite programming method as stated in Theorem 8.7.1. However, there is also a simple direct proof of this fact. Observe that by symmetry, for any $v \in \mathbb{P}(\mathbb{S}^{d-1})$,

$$I_{K}(\boldsymbol{\nu}) = 3 \int_{\mathbb{S}^{d-1}} \left(\int_{\mathbb{S}^{d-1}} \langle x, y \rangle d\boldsymbol{\nu}(x) \right)^{2} d\boldsymbol{\nu}(y) \ge 0 = I_{K}(\boldsymbol{\sigma}).$$
(8.42)

We now turn to parts (2) and (3). We first note that

$$U_K^{\boldsymbol{\sigma}}(x,y) = \int_{\mathbb{S}^{d-1}} \langle z,x \rangle \langle y,z \rangle d\boldsymbol{\sigma}(z) = \frac{1}{d} \langle x,y \rangle,$$

which follows from Corollary 2.5.2. Hence, the kernel $U_K^{\sigma}(x, y)$ is positive definite, i.e. (3) holds. Therefore σ minimizes the two-input energy with this kernel, i.e., for any $v \in \mathbb{P}(\mathbb{S}^{d-1})$,

$$I_{U_K^{\sigma}}(\mathbf{v}) = I_K(\mathbf{v},\mathbf{v},\mathbf{\sigma}) \ge I_{U_K^{\sigma}}(\mathbf{\sigma}) = I_K(\mathbf{\sigma}) = 0.$$

Observe also that $U_K^{\sigma^2}(x) = 0$ and thus $I_K(\sigma, \sigma, v) = 0$.

For an arbitrary $v \in \mathbb{P}(\mathbb{S}^{d-1})$ and $t \in [0,1]$, define $\sigma_t = (1-t)\sigma + tv$. Then

$$I_{K}(\sigma_{t}) = (1-t)^{3}I_{K}(\sigma) + 3(1-t)^{2}tI_{K}(\sigma,\sigma,\nu) + 3(1-t)t^{2}I_{K}(\nu,\nu,\sigma) + t^{3}I_{K}(\nu)$$

= 3(1-t)t²I_{K}(\nu,\nu,\sigma) + t^{3}I_{K}(\nu).

If $I_K(v) > 0$, we can choose t_v sufficiently small so that for all $t \in (0, t_v)$, we have $I_K(v, v, \sigma) \le \frac{1+t}{3t} I_K(v)$, since the right-hand side goes to $+\infty$ as $t \to 0$. Then

$$I_{K}(\sigma_{t}) \leq (1-t^{2})tI_{K}(\nu) + t^{3}I_{K}(\nu) = tI_{K}(\nu) = tI_{K}(\nu) + (1-t)I_{K}(\sigma).$$

It remains to consider the case $I_K(v) = 0$. According to (8.42), in this situation, $\int_{\mathbb{S}^{d-1}} \langle x, y \rangle dv(x) = 0 \text{ for } v\text{-a.e. } y \in \mathbb{S}^{d-1}, \text{ and therefore}$

$$\int_{\mathbb{S}^{d-1}}\int_{\mathbb{S}^{d-1}}\langle x,y\rangle d\boldsymbol{\nu}(x)d\boldsymbol{\nu}(y)=0.$$

But this implies that

$$I_{K}(\boldsymbol{\nu},\boldsymbol{\nu},\boldsymbol{\sigma}) = I_{U_{K}^{\sigma}}(\boldsymbol{\nu}) = \int_{\mathbb{S}^{d-1}} \int_{\mathbb{S}^{d-1}} \frac{1}{d} \langle x, y \rangle d\boldsymbol{\nu}(x) d\boldsymbol{\nu}(y) = 0.$$

Thus, when $I_K(v) = 0$, we have

$$I_{K}(\sigma_{t}) = 3(1-t)t^{2}I_{K}(\nu,\nu,\sigma) + t^{3}I_{K}(\nu) = 0 = (1-t)I_{K}(\sigma) + tI_{K}(\nu)$$

for all $t \in [0, 1]$. This finishes the proof that I_K is convex at σ .

Finally, we show that I_K is not conditionally 3-positive definite, i.e. part (4). Taking

 $\mu = \delta_{e_2} - \delta_{-e_1}$ and $z = e_1$, a straightforward computation shows that

$$I_{K_{\tau}}(\mu) = I_{K}(\delta_{e_{1}}, \mu, \mu) = -1 < 0,$$

which proves our claim.

The behavior of the kernel $S_{1,0,0}^d$ is somewhat different.

Proposition 8.6.2. Let $\Omega = \mathbb{S}^{d-1}$ and set

$$K(x, y, z) = S_{1,0,0}^d(x, y, z) = (t - uv) + (u - vt) + (v - tu).$$

The kernel K satisfies the following:

- 1. the uniform measure σ minimizes the energy I_K ,
- 2. the energy functional I_K is not convex at σ ,
- 3. K is not conditionally 3-positive definite,
- 4. $U_K^{\sigma}(x, y)$ is positive definite.

Proof. As in Proposition 8.6.1, part (1) follows from Theorem 8.7.1. For part (2), we see that since

$$I_{K}(\delta_{e_{1}}, \delta_{e_{1}}, \sigma) = \int_{\mathbb{S}^{d-1}} (1 - z_{1}^{2}) d\sigma(z) > 0 = I_{K}(\sigma) = I_{K}(\sigma, \sigma, \delta_{e_{1}}) = I_{K}(\delta_{e_{1}}),$$

we have that for all $t \in (0, 1)$,

$$I_{K}(t\delta_{e_{1}}+(1-t)\sigma)=3t^{2}(1-t)I_{K}(\delta_{e_{1}},\delta_{e_{1}},\sigma)>tI_{K}(\delta_{e_{1}})+(1-t)I_{K}(\sigma),$$

so I_K is not convex at σ , and therefore K is not conditionally 3-positive definite, according

to Corollary 8.2.7. Applying Corollary 2.5.2, we find that

$$U_{K}^{\sigma}(x,y) = \int_{\mathbb{S}^{d-1}} \langle x,y \rangle - \langle x,z \rangle \langle z,y \rangle d\sigma(z) = \frac{d-1}{d} \langle x,y \rangle$$

is indeed positive definite.

This shows that convexity of I_K at σ and the fact that σ is a minimizer of I_K are not equivalent for three-input energies, unlike in the classical two-input case (see Theorem 3.3.1).

Our remaining two functions arise from geometric problems and are the focus of Section 8.8, where we shall demonstrate that the corresponding energies are nevertheless minimized by the uniform measure σ (Theorems 8.8.1 and 8.8.4). These three-input kernels are related to the volume of the parallelepiped spanned by three unit vectors and the area of the triangle defined by three points on the sphere. We start with the former.

Proposition 8.6.3. Assume $d \ge 3$, and let V(x,y,z) be the volume of the parallelepiped spanned by the vectors $x, y, z \in \mathbb{S}^{d-1}$. Define the kernel $K(x,y,z) = -V^2(x,y,z)$. Then K is not conditionally 3-positive definite.

Proof. It is well-known that the volume squared of the parallelepiped spanned by three vectors is given by the determinant of their Gram matrix, i.e.

$$V^{2}(x, y, z) = \det \begin{pmatrix} 1 & u & v \\ u & 1 & t \\ v & t & 1 \end{pmatrix} = 1 - u^{2} - v^{2} - t^{2} + 2uvt.$$

Fixing $z = e_1$, we find that $K_{e_1}(x, y) = t^2 + y_1^2 + x_1^2 - 2tx_1y_1 - 1$. It is easy to check that $K_{e_1}(e_1, e_1) = K_{e_1}(e_1, y) = K_{e_1}(x, e_1) = 0$ and hence

$$K_{e_1}(x,y) + K_{e_1}(e_1,e_1) - K_{e_1}(e_1,y) - K_{e_1}(x,e_1) = K_{e_1}(x,y).$$
(8.43)

Taking $v = \delta_{e_2} + \delta_{e_3}$, one can compute

$$I_{K_{e_1}}(\mathbf{v}) = -2 < 0,$$

i.e. K_{e_1} is not positive definite. Lemma 2.2.6 and (8.43) then tell us that K_{e_1} is not conditionally positive definite and thus *K* is not conditionally 3-positive definite.

We now turn to area squared of a triangle and prove an analogous statement.

Proposition 8.6.4. Assume that $d \ge 2$. Let A(x, y, z) be the area of the triangle with vertices at $x, y, z \in \mathbb{S}^{d-1}$ and set $K(x, y, z) = -A^2(x, y, z)$. Then K is not conditionally 3-positive definite.

Proof. Using Heron's formula or linear algebra, see (8.53), one can compute that

$$A^{2}(x,y,z) = \frac{3}{4} - \frac{1}{2}(u+v+t) + \frac{1}{2}(uv+vt+tu) - \frac{1}{4}(u^{2}+v^{2}+t^{2}).$$

Fixing $z = e_1$, we find that

$$4K_{e_1}(x,y) = t^2 + x_1^2 + y_1^2 + 2t + 2x_1 + 2y_1 - 2x_1y_1 - 2tx_1 - 2ty_1 - 3.$$

The rest of the argument almost repeats the proof of Proposition 8.6.3: we have that

$$K_{e_1}(x,y) + K_{e_1}(e_1,e_1) - K_{e_1}(e_1,y) - K_{e_1}(x,e_1) = K_{e_1}(x,y),$$
(8.44)

as well as

$$I_{K_{e_1}}(\delta_{e_2}+\delta_{-e_1})=-2<0,$$

and an application of Lemma 2.2.6 finishes the proof.

8.7 Semidefinite Programming

The semidefinite programming method is a powerful and delicate tool that has been successfully applied to numerous optimization problems on the sphere [BV08, CW12, dLMd-OFV, Mus14]. Some intrinsic features of the method (the so-called three-point bounds or, more generally, *k*-point bounds [CW12, dLMdOFV]) naturally give rise to certain multi-input energies. Thus, it comes as no surprise that this method is also useful in our context.

We briefly recall some of the main notions of the method. We start with the fundamental class of polynomials constructed by Bachoc and Vallentin in [BV08]. The role of these polynomials in semidefinite programming is similar to that of the classical Gegenbauer polynomials in linear programming. The construction starts with a class of infinite matrices and associated polynomials of the form

$$(Y_k^d)_{i+1,j+1}(x,y,z) := Y_{k,i,j}^d(x,y,z) := \lambda_{i,j}^{k,d} P_i^{d+2k}(u) P_j^{d+2k}(v) Q_k^d(u,v,t),$$
(8.45)

where $k, i, j \in \mathbb{N}_0$, P_m^h is the Gegenbauer polynomial of degree m on \mathbb{S}^{h-1} , normalized so that $P_m^h(1) = 1$, i.e. $P_m^h = C_m^{(\frac{h-3}{2}, \frac{h-3}{2})}$,

$$\lambda_{i,j}^{k,d} := \frac{A_{d-1}}{A_{d-2}} \frac{A_{d+2k-2}}{A_{d+2k-1}} \left(\dim(\mathscr{H}_i^{d+2k}) \dim(\mathscr{H}_j^{d+2k}) \right)^{1/2}$$

(recall that A_{h-1} is the surface area of \mathbb{S}^{h-1} and \mathscr{H}_n^h is the space of spherical harmonics of degree *n* on \mathbb{S}^{h-1}), and

$$Q_k^d(u,v,t) = ((1-u^2)(1-v^2))^{\frac{k}{2}} P_k^{d-1} \left(\frac{t-uv}{\sqrt{(1-u^2)(1-v^2)}}\right).$$

Below we provide the upper left 3×3 , 2×2 , and 1×1 submatrices of infinite matrices Y_0^d , Y_1^d , and Y_2^d , respectively, which is all that will be needed for the geometric results in

Section 8.8:

$$\begin{pmatrix} 1 & \sqrt{d}v & \sqrt{\frac{(d+1)d}{2} - 1} \frac{dv^2 - 1}{d - 1} \\ \sqrt{d}u & duv & \sqrt{\frac{(d+1)d^2}{2} - d} u\frac{dv^2 - 1}{d - 1} \\ \sqrt{\frac{(d+1)d}{2} - 1} \frac{dv^2 - 1}{d - 1} & \sqrt{\frac{(d+1)d^2}{2} - d} \frac{du^2 - 1}{d - 1}v & \left(\frac{(d+1)d}{2} - 1\right)\frac{du^2 - 1}{d - 1}\frac{dv^2 - 1}{d - 1} \end{pmatrix} \\ \begin{pmatrix} \frac{d}{d - 1}(t - uv) & \frac{d\sqrt{d + 2}}{d - 1}u(t - uv) \\ \frac{d\sqrt{d + 2}}{d - 1}v(t - uv) & \frac{d(d + 2)}{d - 1}uv(t - uv) \end{pmatrix}, \begin{pmatrix} \frac{d(d + 2)}{(d - 1)(d + 1)}\frac{(d - 1)(t - uv)^2 - (1 - u^2)(1 - v^2)}{d - 2} \end{pmatrix} \end{pmatrix}$$

Note that the polynomials comprising these matrices are not symmetric. Averaging over all permutations π of the variables *x*, *y*, and *z*, one obtains the following symmetric matrices and associated polynomials

$$(S_k^d)_{i+1,j+1}(x,y,z) := S_{k,i,j}^d(x,y,z) := \frac{1}{6} \sum_{\pi} Y_{k,i,j}^d(\pi(x),\pi(y),\pi(z)).$$
(8.46)

These polynomials and matrices have a variety of nice properties, which we summarize below in the form pertinent to our discussion. In [BV08], all the results are stated for discrete energies of points set. However, the extension to energy integrals and measures, which we need, follows immediately from the weak^{*} density of discrete measures in $\mathbb{P}(\mathbb{S}^{d-1})$. The following statements hold:

• For any $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ and $e \in \mathbb{S}^{d-1}$, the infinite matrices

$$\int_{\mathbb{S}^{d-1}}\int_{\mathbb{S}^{d-1}}Y_k^d(x,y,e)d\mu(x)d\mu(y)$$

and

$$S_k^d(\mu) := \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} \int\limits_{\mathbb{S}^{d-1}} S_k^d(x, y, z) d\mu(x) d\mu(y) d\mu(z)$$

are positive semidefinite (to be more precise, all finite principal minors are positive semidefinite).

- For $(k, i, j) \neq (0, 0, 0)$, we have $I_{S^d_{k, i, j}}(\sigma) = 0$ and $I_{Y^d_{k, i, j}}(\sigma, \sigma, \delta_e) = 0$ for all $e \in \mathbb{S}^{d-1}$.
- For $k \ge 1$ and any $e \in \mathbb{S}^{d-1}$, one has $I_{S^d_{k,i,j}}(\delta_e) = I_{Y^d_{k,i,j}}(\delta_e) = 0$.

Consider an infinite, symmetric, positive semidefinite matrix A with finitely many nonzero entries. Then for any $k \ge 1$ and $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$, one easily sees that $\operatorname{tr}(S_k^d(\mu)A) \ge 0$, with equality if $\mu = \sigma$. Likewise, when k = 0, if A_0 is an infinite, positive semidefinite matrix Awith finitely many nonzero entries and such that all entries in the first row and first column are zeros, then $\operatorname{tr}(S_0^d(\mu)A_0) \ge 0$ for any probability measure μ , with equality if $\mu = \sigma$. (Requiring zeros in the first row and column guarantees that equality for $\mu = \sigma$; due to the fact that $S_{0,0,0}^d$ is a constant, equality might not be achieved otherwise.) This yields the following sufficient condition for σ to be a minimizer of a three-input energy integral I_K .

Theorem 8.7.1. Let $m \in \mathbb{N}_0$. For each $k \leq m$, let A_k be an infinite, symmetric, positive semidefinite matrix with finitely many nonzero entries, with the additional requirement that A_0 has only zeros in its first row and first column. Let

$$K(x, y, z) = C + \sum_{k=0}^{m} \operatorname{tr}(S_k^d(x, y, z), A_k),$$
(8.47)

where *C* is an arbitrary constant. Then σ minimizes I_K in $\mathbb{P}(\mathbb{S}^{d-1})$.

Under appropriate convergence assumptions, the theorem also holds with $m = \infty$. Notice that the fact that σ minimizes the energies with kernels (8.40) and (8.41) immediately follows from Theorem 8.7.1. In general, in order to apply Theorem 8.7.1, one needs to represent the kernel in the form (8.47) – a task which becomes more complex as *m* grows.

8.8 **Results in Probabilistic Geometry**

Riesz *s*-energies with the kernel $K(x,y) = ||x - y||^{-s}$, discussed in Section 2.1, are one of the most important classes of two-input energies. In particular, when s = -1, maximizing

the sum of distances between points on various spaces (as well as corresponding distance integrals) is a classical optimization problem from metric geometry [AS74, Bjö56, FT56]. One can construct interesting multi-input analogues of Riesz energies by replacing the distance with other geometric characteristics which depend on n points, such as area and volume.

For n = 3, some of the most natural examples include the area of the triangle generated by three points or the volume of the tetrahedron (or the parallelepiped) spanned by three vectors. This can be generalized to higher values of n by considering volumes of various simplices or polytopes generated by n points or vectors.

It is reasonable to conjecture that on the sphere, energy integrals with these three-input kernels (namely, the area of the triangle and the volume of the parallelepiped) are *maximized* by the uniform measure σ . Probabilistically, this can be reformulated in the following way: assume that three random points are chosen on the sphere \mathbb{S}^{d-1} independently according to a probability distribution μ . The conjecture then states that the expected value of these geometric quantities is maximized when the distribution μ is uniform, i.e. $\mu = \sigma$. The question was posed in this form in [Rom19].

This conjecture is supported, among other reasons, by the fact that for the classical case n = 2, the analogous kernels $|\sin(\arccos\langle x, y \rangle)| = \sqrt{1 - t^2}$ and $||x - y|| = \sqrt{2 - 2t}$ (i.e. the area of the parallelogram spanned by vectors *x* and *y* and the Euclidean distance between *x* and *y*, respectively) are both negative definite kernels on the sphere (up to an additive constant), and hence the corresponding two-input energies are maximized by σ .

In this section, we verify the conjecture above for the slightly different, yet closely related kernels V^2 and A^2 : the *squares* of the aforementioned volume and area. In these cases, the kernels are multivariate polynomials, which substantially simplifies the analysis. For both kernels, we provide two proofs: one based on the semidefinite programming methods as outlined in Section 8.7, the other a more direct proof based on geometry and linear algebra.

Volume of the Tetrahedron/Parallelepiped

Let V(x, y, z) denote the three-dimensional volume of the parallelepiped spanned by the vectors $x, y, z \in \mathbb{S}^{d-1}$. (Observe that the volume of the tetrahedron with vertices at x, y, z, and the origin is given by $\frac{1}{6}V(x, y, z)$.) The square of the volume V(x, y, z) is given by the determinant of the Gram matrix. Thus we consider the kernel

$$K(x,y,z) = V^{2}(x,y,z) = \det \begin{pmatrix} 1 & t & u \\ t & 1 & v \\ u & v & 1 \end{pmatrix} = 1 - t^{2} - u^{2} - v^{2} + 2tuv,$$
(8.48)

where, as before, we set $t = \langle x, y \rangle$, $u = \langle x, z \rangle$, $v = \langle z, y \rangle$.

As shown by Proposition 8.6.3, the kernel $-V^2$ is not conditionally 3-positive definite. Nevertheless, we shall show that σ is a minimizer of I_{-V^2} , i.e. a maximizer of $I_K = I_{V^2}$.

One can check directly that

$$K(x, y, z) = \frac{(d-1)(d-2)}{d^2} - \frac{2(d-2)}{d^2(d+2)} S_{0,2,2} - \frac{4(d-1)(d-2)}{d^2(d+2)} S_{1,1,1} - \frac{(3d-4)(d+1)(d-2)}{d^2(d+2)} S_{2,0,0},$$
(8.49)

hence, Theorem 8.7.1 implies that σ is a maximizer of I_K .

Theorem 8.8.1. Assume that $d \ge 3$ and $\Omega = \mathbb{S}^{d-1}$. Let $K(x, y, z) = V^2(x, y, z) = 1 - t^2 - u^2 - v^2 + 2uvt$. Then σ is a maximizer of I_K over $\mathbb{P}(\mathbb{S}^{d-1})$.

Moreover, since *K* in this case is a polynomial of degree two in every variable, and has no linear terms, any isotropic measure on the sphere, i.e., any measure $\mu \in \mathbb{P}(\mathbb{S}^{d-1})$ satisfying $\int_{\mathbb{S}^{d-1}} xx^T d\mu(x) = \frac{1}{d}I_d$, is also a maximizer, and in fact this classifies all maximizers.

This statement can be generalized to a higher number of inputs. As discussed in Section 6.5, we may consider optimization for probability measures on \mathbb{R}^d with the additional normalizing condition: $\int_{\mathbb{R}^d} ||x||^2 d\mu(x) = 1$. As before, we will denote the space of such mea-

sures by $\mathbb{P}^*(\mathbb{R}^d)$, and we say that $\mu \in \mathbb{P}(\mathbb{R}^d)$ is an isotropic measure if $\int_{\mathbb{R}^d} xx^T d\mu(x) = \frac{1}{d}I_d$. One can easily see that any isotropic measure $\mu \in \mathbb{P}(\mathbb{R}^d)$ satisfies the normalizing condition as well, i.e. $\mu \in \mathbb{P}^*(\mathbb{R}^d)$.

Let $3 \le n \le d$ and consider the *n*-input kernel $K(x_1, ..., x_n)$ defined as the determinant of the Gram matrix of the set of vectors $\{x_1, ..., x_n\} \subset \mathbb{R}^d$. Observe that for n = 3, this coincides with the 3-input kernel (8.48). The following result contains Theorem 8.8.1 as a special case. The case n = d has been proved by Rankin [Ran56] and the general case is due to Cahill and Casazza [CC].

Theorem 8.8.2. Let $d \ge 3$ and let the n-input kernel K be as defined above. Then the set of maximizing measures of I_K in $\mathbb{P}^*(\mathbb{R}^d)$ is the set of isotropic measures on \mathbb{R}^d . As a corollary, any isotropic measure on \mathbb{S}^{d-1} , in particular, the uniform surface measure σ , maximizes I_K over $\mathbb{P}(\mathbb{S}^{d-1})$.

Proof. We start with the discrete case, which was considered in [Ran56]. Let $\mu \in \mathbb{P}^*(\mathbb{R}^d)$ be an equal-weight discrete measure, i.e. $\mu = \frac{1}{N} \sum \delta_{x_i}$, with support consisting of $N \ge d$ vectors x_1, \ldots, x_N . Let D be a matrix formed by these vectors as columns. The normalization condition then implies that $\operatorname{tr}(DD^T) = \operatorname{tr}(D^TD) = \sum ||x_i||^2 = N$.

The proof of the case n = d is particularly concise and elegant, so we include it below.

We denote the set of all *d*-subsets of $\{1, ..., N\}$ by $[N]_d$ and set $[d] = \{1, ..., d\}$. For two sets of indices $S_1 \subset [d]$ and $S_2 \subset [N]$, let D_{S_1,S_2} stand for the minor of *D* defined by rows with indices in S_1 and columns with indices in S_2 . Observe that for $S = \{i_1, ..., i_d\} \in [N]_d$, the kernel *K* satisfies

$$K(x_{i_1},\ldots,x_{i_d}) = \det\left((D^T)_{[d],S}D_{S,[d]}\right) = \det^2(D_{[d],S}).$$

Therefore, the energy can be expressed as

$$I_{K}(\mu) = E_{K}(\{x_{i}\}_{i=1}^{N}) = \frac{d!}{N^{d}} \sum_{S \in [N]_{d}} \det^{2}(D_{[d],S}) = \frac{d!}{N^{d}} \sum_{S \in [N]_{d}} \det(D_{[d],S}) \det((D^{T})_{S,[d]})$$

By the Cauchy–Binet formula, the sum above is just $det(DD^T)$, i.e.

$$I_K(\mu) = \frac{d!}{N^d} \det(DD^T).$$

The matrix DD^T is positive definite, hence all eigenvalues are nonnegative, and their sum equals $tr(DD^T) = N$ due to the normalizing condition. Since the determinant of DD^T is the product of eigenvalues, by the geometric-arithmetic mean inequality, it satisfies

$$\det(DD^{T}) \le \left(\frac{\operatorname{tr}(DD^{T})}{d}\right)^{d} = \left(\frac{N}{d}\right)^{d}.$$
(8.50)

Hence, the energy is bounded above by

$$I_K(\mu) = \frac{d!}{N^d} \det(DD^T) \le \frac{d!}{N^d} \left(\frac{N}{d}\right)^d = \frac{d!}{d^d}.$$
(8.51)

Equality in (8.50)–(8.51) is possible if and only if all eigenvalues of DD^T are equal to $\frac{N}{d}$, i.e. $DD^T = \frac{N}{d}I_d$, which is equivalent to the fact that μ is an isotropic measure, since for the discrete measure $\mu = \frac{1}{N}\sum \delta_{x_i} \in \mathbb{P}(\mathbb{S}^{d-1})$ we would have

$$\int_{\mathbb{S}^{d-1}} x x^T d\mu(x) = \frac{1}{N} D D^T = \frac{1}{d} I_d.$$

Observe that this is equivalent to saying that the set $\{x_1, \ldots, x_N\} \subset \mathbb{S}^{d-1}$ is a projective 1-design.

Since the right-hand side of (8.51) does not depend on N, one may easily pass from discrete to arbitrary measures μ , thus finishing the proof of the theorem.

Area of the Triangle

We now turn to exploring the area A(x, y, z) of the triangle with vertices x, y, and z. As in the previous section, rather than considering the spherical case directly, we shall consider probability measures μ on the whole space \mathbb{R}^d which satisfy the normalizing condition $\int_{\mathbb{R}^d} ||x||^2 d\mu(x) = 1$, i.e. measures in the class $\mathbb{P}^*(\mathbb{R}^d)$. It is a standard geometrical fact that

$$A^{2}(x,y,z) = \frac{1}{4} \left(\|y-x\|^{2} \cdot \|z-x\|^{2} - \langle y-x,z-x\rangle^{2} \right).$$
(8.52)

We shall keep the notation $t = \langle x, y \rangle$, $u = \langle x, z \rangle$, $v = \langle z, y \rangle$ despite the fact that the vectors $x, y, z \in \mathbb{R}^d$ do not necessarily lie on the sphere \mathbb{S}^{d-1} . A straightforward computation shows that

$$A^{2}(x, y, z) = \frac{1}{4} (\|x\|^{2} \|y\|^{2} + \|y\|^{2} \|z\|^{2} + \|z\|^{2} \|x\|^{2}) + \frac{1}{2} (uv + vt + tu) - \frac{1}{2} (t\|z\|^{2} + v\|x\|^{2} + u\|y\|^{2}) - \frac{1}{4} (u^{2} + v^{2} + t^{2}).$$

$$(8.53)$$

We are now ready to prove that the expectation of the area of the triangle *squared* is maximized by isotropic measures with barycenter at zero – in particular, the uniform surface measure σ on the sphere \mathbb{S}^{d-1} is a maximizer.

Theorem 8.8.3. Suppose $d \ge 2$, and let $K : (\mathbb{R}^d)^3 \to \mathbb{R}$ be defined by $K(x, y, z) = A^2(x, y, z)$. Then μ is a maximizer of $I_K(\mu)$ over $\mathbb{P}^*(\mathbb{R}^d)$ if and only if μ is isotropic and has center of mass at the origin.

Observe that for discrete measures, this condition yields exactly the weighted spherical 2-designs.

Proof. Fix an arbitrary measure $\mu \in \mathbb{P}^*(\mathbb{R}^d)$. First of all, observe that the normalizing condition implies that the first line in the representation (8.53) contributes the constant $\frac{3}{4}$ to the energy. Invoking the normalizing condition again, we observe that

$$I_{t||z||^{2}}(\mu) = \int_{\mathbb{R}^{d}} \int_{\mathbb{R}^{d}} \int_{\mathbb{R}^{d}} \langle x, y \rangle ||z||^{2} d\mu(x) d\mu(y) d\mu(z)$$

$$= \int_{\mathbb{R}^{d}} \int_{\mathbb{R}^{d}} \langle x, y \rangle d\mu(x) d\mu(y) = \left\| \int_{\mathbb{R}^{d}} x \mu(x) \right\|^{2}.$$
(8.54)

Furthermore, applying the Cauchy–Schwarz inequality and the normalization condition, we obtain

$$I_{tu}(\mu) = \iint_{\mathbb{R}^d} \iint_{\mathbb{R}^d} \langle x, y \rangle \langle z, x \rangle d\mu(x) d\mu(y) d\mu(z) = \iint_{\mathbb{R}^d} \left\langle x, \iint_{\mathbb{R}^d} y d\mu(y) \right\rangle^2 d\mu(x)$$

$$\leq \iint_{\mathbb{R}^d} ||x||^2 \cdot \left\| \iint_{\mathbb{R}^d} y d\mu(y) \right\|^2 d\mu(x) = \left\| \iint_{\mathbb{R}^d} y d\mu(y) \right\|^2 = I_{t||z||^2}(\mu).$$
(8.55)

This inequality implies that the contribution of the two middle terms in the representation (8.53) is non-positive, i.e.

$$I_{\frac{1}{2}(uv+vt+tu)-\frac{1}{2}(t\|z\|^{2}+v\|x\|^{2}+u\|y\|^{2})}(\mu) \leq 0,$$

and therefore,

$$I_{K}(\mu) \leq \frac{3}{4} - \frac{1}{4} I_{u^{2} + v^{2} + t^{2}}(\mu).$$
(8.56)

Finally, we have a well-known estimate for the frame energy (discussed in Section 6.1)

$$I_{t^2}(\mu) = \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \langle x, y \rangle^2 d\mu(x) d\mu(y) \ge \frac{1}{d},$$
(8.57)

with the equality holding if and only if μ is isotropic. This can be inferred from Lemmas 6.1.3 and 6.5.1, but for completeness, we include a simple proof of (8.57):

$$\begin{split} \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \langle x, y \rangle^2 d\mu(x) d\mu(y) &= \int_{\mathbb{R}^d} \int_{\mathbb{R}^d} \sum_{i,j=1}^d x_i y_i x_j y_j d\mu(x) d\mu(y) \\ &= \sum_{i,j=1}^d \left(\int_{\mathbb{R}^d} x_i x_j d\mu(x) \right)^2 \ge \sum_{i=1}^d \left(\int_{\mathbb{R}^d} x_i^2 d\mu(x) \right)^2 \\ &\ge \frac{1}{d} \left(\sum_{i=1}^d \int_{\mathbb{R}^d} x_i^2 d\mu(x) \right)^2 = \frac{1}{d} \left(\int_{\mathbb{R}^d} ||x||^2 d\mu(x) \right)^2 = 1, \end{split}$$

where we have dropped the off-diagonal terms and used the Cauchy–Schwarz inequality, which implies that equality holds if and only if $\int xx^T d\mu(x) = \frac{1}{d}I_d$, i.e. μ is isotropic.

Putting together (8.56) and (8.57), we find that

$$I_K(\mu) \leq \frac{3}{4} - \frac{1}{4}I_{u^2 + v^2 + t^2}(\mu) \leq \frac{3}{4} - \frac{3}{4d} = \frac{3}{4}\frac{d-1}{d}.$$

The equality holds if and only if it holds in inequalities (8.55) and (8.57). In the latter case, it means that μ is isotropic. In the former, it means that, for μ -almost every $x \in \text{supp}(\mu)$, the vector x is collinear to $\int y d\mu(y)$. However, if $\int y d\mu(y) \neq 0$ that would imply that the support of μ is contained in a one-dimensional subspace, which is impossible for an isotropic measure. Therefore, $\int y d\mu(y) = 0$, in other words, the center of mass of μ has to be at the origin. This finishes the proof of the theorem.

Theorem 8.8.3 immediately implies the characterization of the maximizers among probability measures on the sphere.

Theorem 8.8.4. Suppose $d \ge 2$, and let $K : (\mathbb{S}^{d-1})^3 \to \mathbb{R}$ be defined by $K(x, y, z) = A^2(x, y, z)$. Then μ is a maximizer of $I_K(\mu)$ over $\mathbb{P}(\mathbb{S}^{d-1})$ if and only if μ is isotropic and has center of mass at the origin. In particular, the uniform measure σ maximizes I_K .

We want to point out that this statement can also be proved using semidefinite programming. Representation (8.53) for *x*, *y*, $z \in \mathbb{S}^{d-1}$ yields

$$K(x,y,z) = A^{2}(x,y,z) = \frac{3}{4} - \frac{u+v+t}{2} + \frac{ut+uv+tv}{2} - \frac{u^{2}+t^{2}+v^{2}}{4}.$$
 (8.58)

This kernel can be rewritten as

$$K = \frac{3(d-1)}{4d} - \frac{3(d+1)(d-2)}{4d(d+2)}S_{2,0,0} - \frac{3(d-1)}{2d}S_{1,0,0} - \frac{3(d-1)}{2d(d+2)}S_{1,1,1} - \frac{3}{2d(d+2)}S_{0,2,2},$$

and Theorem 8.7.1 then tells us that σ maximizes I_K . The fact that μ is isotropic and has

center of mass at the origin is easily seen to be equivalent to the relation $\int_{\mathbb{S}^{d-1}} p(x)d\mu(x) = \int_{\mathbb{S}^{d-1}} p(x)d\sigma(x)$ for each polynomial *p* of degree two. Since *K* is a polynomial of degree two in each variable, the conclusion of Theorem 8.8.4 follows.

Discrete Maximizers

While the kernels given by the volume of the parallelepiped and the area of the triangle are far more complicated to deal with than their squares, the results of Theorems 8.8.1 and 8.8.4 can nevertheless be used to show that a regular simplex is optimal for discrete energies with these kernels when the number of points is d + 1. We have the following general statement.

Theorem 8.8.5. Let $d \ge n-1$ and let $B : (\mathbb{S}^{d-1})^n \to [0,\infty)$ be a symmetric, rotationally invariant polynomial of degree no more than two in each of its variables, with the property that if $x_i = x_j$ for some $i \ne j$, then $B(x_1, x_2, ..., x_n) = 0$. Assume further that σ is a maximizer of I_B over $\mathbb{P}(\mathbb{S}^{d-1})$.

Let the function $f : [0, \infty) \to \mathbb{R}$ be concave, increasing, and right continuous at 0, and define the kernel $K(x_1, ..., x_n) = f(B(x_1, ..., x_n))$. If N = d + 1, then the vertices of regular Nsimplices inscribed in \mathbb{S}^{d-1} with centers at the origin maximize the discrete energy $E_K(\omega_N)$ over all N-point configurations on the sphere.

Proof. Let $\omega_N = \{z_1, ..., z_N\}$ be an arbitrary point configuration on \mathbb{S}^{d-1} . Then we have

$$E_B(\omega_N) = \frac{1}{N^n} \sum_{m_1=1}^N \cdots \sum_{m_n=1}^N B(z_{m_1}, ..., z_{m_n}) \le I_B(\sigma),$$

with equality occurring if the point configuration is a spherical 2-design. Since B is zero if two of its inputs are the same, we can restrict the sum to n-tuples with distinct entries.

Combining this with the fact that f is increasing and concave, we have

$$\begin{split} E_{K}(\boldsymbol{\omega}_{N}) &\coloneqq \frac{1}{N^{n}} \sum_{j_{1}=1}^{N} \cdots \sum_{j_{n}=1}^{N} f(B(z_{j_{1}}, ..., z_{j_{n}})) \\ &\leq \frac{N(N-1) \cdots (N-n+1)}{N^{n}} f\left(\sum_{\substack{z_{j_{1}}, ..., z_{j_{n}} \in \boldsymbol{\omega}_{N} \\ j_{1}, ..., j_{n} \text{ distinct}}} \frac{B(z_{j_{1}}, ..., z_{j_{n}})}{N(N-1) \cdots (N-n+1)}\right) \\ &= \frac{N(N-1) \cdots (N-n+1)}{N^{n}} f\left(\frac{N^{n} E_{B}(\boldsymbol{\omega}_{N})}{N(N-1) \cdots (N-n+1)}\right) \\ &\leq \frac{N(N-1) \cdots (N-n+1)}{N^{n}} f\left(\frac{N^{n} I_{B}(\boldsymbol{\sigma})}{N(N-1) \cdots (N-n+1)}\right). \end{split}$$

The second inequality is achieved if ω_N is a spherical 2-design, in particular, if ω_N is a regular simplex. The first one becomes an equality if

$$B(y_1,...,y_n) = \frac{N^n E_B(\omega_N)}{N(N-1)\cdots(N-n+1)}$$

for all distinct $y_1, ..., y_n \in \omega_N$. Since *B* is rotationally invariant, the simplex satisfies this condition as well. Hence, the regular simplex is a maximizer of E_K for N = d + 1.

Theorem 8.8.5 immediately applies to the energies considered in Theorems 8.8.1 and 8.8.4.

Corollary 8.8.6. Let K(x,y,z) denote the area of the triangle with vertices x,y,z or the volume of the parallelepiped spanned by x, y, and z. If N = d + 1, then the regular N-simplices inscribed in \mathbb{S}^{d-1} with centers at the origin maximize $E_K(\omega_N)$ over all N-point configurations on the sphere.

Proof. From Theorems 8.8.4 and 8.8.1, we see that in both cases, $K^2(x, y, z)$ satisfies the conditions of Theorem 8.8.5. Setting $f(x) = \sqrt{x}$ in Theorem 8.8.5 finishes the proof. \Box

Bibliography

- [AZ14] M. Aigner and G. Ziegler, Proofs from The Book, Fifth, Springer-Verlag Berlin Heidelberg, 2014.
- [AS74] R. Alexander and K. Stolarsky, Extremal Problems of Distance Geometry Related to Energy Integrals, Transactions of the American Mathematical Society 193 (1974), 1–31.
- [AGHP92] N. Alon, O. Goldreich, J. Hastad, and R. Peralta, Simple construction of almost k-wise independent random variables, Random Structures and Algorithms 3(3) (1992), 289–304.
 - [And96] N.N. Andreev, *An extremal property of the icosahedron*, East Journal of Approximation **2(4)** (1996), 459–462.
 - [And99] _____, A spherical code, Russian Mathematical Surveys 54 (1999), 255–256.
- [ABB⁺14] D.M. Appleby, I. Bengtsson, S. Brierley, A. Ericsson, M. Grassl, and J.A. Larsson, Systems of imprimitivity for the Clifford Group, Quantum Information & Computation 14(3-4) (2014), 339–360.
 - [AT43] B.M. Axilrod and E. Teller, Interaction of the van der Waals Type Between Three Atoms, The Journal of Chemical Physics 11(6) (1943), 299.
 - [BV08] C. Bachoc and F. Vallentin, New Upper Bounds for Kissing Numbers from Semidefinite Programming, Journal of the American Mathematical Society 21(3) (2008), 909–924.
 - [Bae02] J.C. Baez, *The octonions (with Erratum)*, Bulletin of the American Mathematical Society **39/42** (2002/2005), 145–205/213.
 - [Baj92] B. Bajnok, Construction of spherical t-designs, Geometriae Dedicata 43 (1992), 167-179.
- [BCLR13] D. Balagué, J. Carrillo, T. Laurent, and G. Raoul, Nonlocal interactions by repulsive-attractive potentials: Radial ins/stability, Physica D: Nonlinear Phenomena 260 (2013), 5–25.

- [BCHL03] R. Balan, P.G. Casazza, C. Heil, and Z. Landau, *Deficits and excesses of frames*, Advances in Computational Mathematics 18(2) (2003), 93–116.
 - [BB09] E. Bannai and E. Bannai, A survey on spherical designs and algebraic combinatorics on spheres, European Journal of Combinatorics 30 (2009), 1392–1425.
 - [BD79] E. Bannai and R. Damerell, *Tight spherical designs I*, Journal of the Mathematical Society of Japan **31(1)** (1979), 199–207.
 - [BD80] _____, *Tight spherical designs II*, Journal of the London Mathematical Society **21** (1980), 13–30.
 - [BH85] E. Bannai and S. Hoggar, On Tight t-designs in compact symmetric spaces of rank one, Proceedings of the Japan Academy, Series A Mathematical Sciences 61(3) (1985), 78–82.
 - [BH89] _____, *Tight t-designs and square free integers*, European Journal of Combinatorics **10** (1989), 113–135.
- [BMV04] E. Bannai, A. Munemasa, and B. Venkov, *The nonexistence of certain tight spherical designs*, Algebra i Analiz **16** (2004), 1–23.
 - [Bar20] A. Barg, *Stolarsky's Invariance Principle for Finite Metric Spaces*, Mathematika **67(1)** (2020), 158–186.
 - [BS21] A. Barg and M. Skriganov, Bounds for Discrepancies in the Hamming Space, Journal of Complexity (2021), available at https://doi.org/10.1016/j.jco.2021.101552.
- [BFSvdM19] L. Bäuml, F. Finster, D. Schiefeneder, and H. von der Mosel, Singular Support of Minimizers of the Causal Variational Principle on the Sphere, Calculus of Variations and Partial Differential Equations 58(205) (2019).
 - [Bec84a] J. Beck, *Some upper bounds in the theory of irregularities of distribution*, Acta Arithmetica **43(2)** (1984), 115–130.
 - [Bec84b] _____, Sums of distances between points on a sphere–an application of the theory of irregularities of distribution to discrete geometry, Mathematika **31(1)** (1984), 33–41.
 - [BC87] J. Beck and W. Chen, *Irregularities of Distribution*, Cambridge Tracts in Mathematics, vol. 89, Cambridge University Press, 1987.
 - [BF03] J. Benedetto and M. Fickus, *Finite normalized tight frames*, Advances in Computational Mathematics 18(2-4) (2003), 357–385.

- [BL98] J.J. Benedetto and S. Li, *The theory of multiresolution frames and applications to filter banks*, Applied and Computational Harmonic Analysis **5** (1998), 389–427.
- [BT93] J.J. Benedetto and A. Teolis, A Wavelet Auditory Model and Data Compression, Applied and Computational Harmonic Analysis 1 (1993), 3–28.
- [BCR84] C. Berg, J.P.R. Christensen, and P. Ressel, Harmonic Analysis on Semigroups: Theory of Positive Definite and Related Functions, Graduate Texts in Mathematics, vol. 100, Springer-Verlag New York, 1984.
- [BLT06] A.L. Bertozzi, M.A. Lewis, and C.M. Topaz, A nonlocal continuum model for biological aggregation, Bulletin of Mathematical Biology 68(7) (2006), 1601–1623.
- [Bes78] A.L. Besse, Manifolds all of whose geodesics are closed, A Series of Modern Surveys in Mathematics, vol. 93, Springer-Verlag Berlin Heidelberg, 1978.
- [BD19] D. Bilyk and F. Dai, Geodesic distance Riesz energy on the sphere, Transactions of the American Mathematical Society 372 (2019), 3141–3166.
- [BDM18] D. Bilyk, F. Dai, and R. Matzke, *Stolarsky principle and energy optimization on the sphere*, Constructive Approximation 48(1) (2018), 31–60.
- [BFG⁺a] D. Bilyk, D. Ferizović, A. Glazyrin, R. Matzke, J. Park, and O. Vlasiuk, Potential Theory with Multivariate Kernels, available at https://arxiv.org/pdf/2104.03410.pdf.
- [BFG⁺b] _____, Optimal measures for multivariate geometric potentials (in preparation).
- [BGM⁺a] D. Bilyk, A. Glazyrin, R. Matzke, J. Park, and O. Vlasiuk, *Energy on Spheres and Discreteness of Minimizing Measures*, Journal of Functional Analysis (to appear), available at https://arxiv.org/pdf/1908.10354.
- [BGM⁺b] _____, Optimal Measures for p-frame Energies on Spheres, available at https://arxiv. org/pdf/1908.00885.
 - [BL17] D. Bilyk and M. Lacey, One bit sensing, discrepancy, and Stolarsky principle, Sbornik: Mathematics 208(5-6) (2017), 744–763.
 - [BL] _____, Random tessellations, restricted isometric embeddings, and one bit sensing, available at https://arxiv.org/pdf/1512.06697.pdf.
 - [BM19] D. Bilyk and R. Matzke, On the Fejes Tóth problem on the sum of acute angles, Proceedings of American Mathematical Society 147 (2019), 51–59.

- [BMV] D. Bilyk, R. Matzke, and O. Vlasiuk, *Positive Definiteness and the Stolarsky Invariance Principle (In Preparation).*
- [Bjö56] G. Björck, Distributions of positive mass, which maximize a certain generalized energy integral, Arkiv För Matematik **3** (1956), 255–269.
- [BC14] A. Blanchet and G. Carlier, From Nash to Cournot-Nash equilibria via the Monge-Kantorovich problem, Philosophical Transactions of the Royal Society A: Mathematical, Physical, and Engineering Sciences 372 (2014), (2028:20130398).
- [Blü91] M. Blümlinger, Slice discrepancy and irregularities of distribution on spheres, Mathematika 38(1) (1991), 105–116.
- [Boc41] S. Bochner, *Hilbert distances and positive definite functions*, Annals of Mathematics **42** (1941), 647–656.
- [BRV13] A. Bondarenko, D. Radchenko, and M. Viazovska, Optimal asymptotic bounds for spherical designs, Annals of Mathematics 178(2) (2013), 443–452.
- [BRV15] _____, Well-Separated Spherical Designs, Constructive Approximation 41 (2015), 93–112.
- [BV10] A.V Bondarenko and M.S. Viazovska, Spherical Designs via Brouwer Fixed Point Theorem, SIAM Journal on Discrete Math 24 (2010), 207-217.
- [BHS19] S. Borodachov, D. Hardin, and E. Saff, *Discrete Energy of Rectifiable Sets*, Springer Monographs in Mathematics, Springer-Verlag New York, 2019.
- [BL88] J. Bourgain and J. Lindenstrauss, *Projection Bodies*, Geometric aspects of functional analysis, Lecture Notes in Mathematics 1317 (1988), 250–270.
- [BCDC⁺19] C. Boyer, A. Chambolle, Y. De Castro, V. Duval, F. de Gournay, and P. Weiss, On Representer Theorems and Convex Regularization, SIAM Journal on Optimization 29(2) (2019), 1260– 1281.
 - [Bra06] J. Brauchart, About the second term of the asymptotics for optimal Riesz energy on the sphere in the potential-theoretical case, Integral Transforms and Spectral Functions 17(5) (2006), 321–328.
 - [BD13] J.S. Brauchart and J. Dick, A simple proof of Stolarsky's invariance principle, Proceedings of the American Mathematical Society 141 (2013), 2085–2096.
 - [BHS12] J.S. Brauchart, D. Hardin, and E. Saff, *Discrete energy asymptotics on a Riemannian circle*, Uniform Distribution Theory 7(2) (2012), 77–108.

- [BMZ07] H.P. Büchler, A. Micheli, and P. Zoller, *Three-body Interactions with Cold Polar Molecules*, Nature Physics **3** (2007), 726–731.
 - [CC] J. Cahill and P.G. Casazza, Optimal Parseval frames: Total coherence and total volume, available at https://arxiv.org/abs/1910.01733.
- [CFP17] J. Carrillo, A. Figalli, and F.S. Patacchini, Geometry of minimizers for the interaction energy with mildly repulsive potentials, Annales de l'Institut Henri Poincaré C, Analyse non linéaire 34 (2017), 1299–1308.
- [CMV03] J. Carrillo, R.J. McCann, and C. Vallani, Kinetic equilibration rules for granular media and related equations: entropy dissipation and mass transportion estimates, Revista Matemática Iberoamericana 19 (2003), 971-1018.
 - [CK03] P.G. Casazza and J. Kovacevic, *Equal-norm tight frames with erasures*, Advances in Computational Mathematics 18(2-4) (2003), 387–430.
- [CFTW06] P.G. Casazza, M. Fickus, J.C. Tremain, and E. Weber, *The Kadison-Singer problem in mathematics and engineering: a detailed account*, Operator Theory, Operator Algebras, and Applications, Contemperary Mathematics **414** (2006), 299-355.
- [CRPW12] V. Chandrasekaran, B. Recht, P. Parrilo, and A. Willsky, *The convex geometry of linear inverse problems*, Foundations of Computational Mathematics **12** (2012), 805–849.
- [CGG⁺20] X. Chen, V. Gonzales, E. Goodman, S. Kang, and K. Okoudjou, *Universal Optimal Configurations for the p-Frame Potentials*, Advances in Computational Mathematics **46(4)** (2020).
 - [Cho58] G. Choquet, Universal Optimal Configurations for the p-Frame Potentials, Séminaire Brelot-Choquet-Deny. Théorie du potentiel 3(4) (1958), 1–7 (French).
 - [Chr03] O. Christensen, An Introduction to Frames and Riesz Bases, Birkhäuser, Boston, MA, 2003.
- [CMY86] J. Cleary, S.A. Morris, and D. Yost, Numerical Geometry-Numbers for Shapes 93(4) (1986), 260-275.
- [CCEK07] H. Cohn, J. Conway, N. Elkies, and A. Kumar, *The D*₄ root system is not universally optimal, Experimental Mathematics 16 (2007), 313–320.
 - [CK07] H. Cohn and A. Kumar, Universally Optimal Distributions of Points on Spheres, Journal of the American Mathematical Society 20(1) (2007), 99–148.
- [CKM⁺17] H. Cohn, A. Kumar, S. Miller, D. Radchenko, and M. Viazovska, *The sphere packing problem in dimension 24*, Annals of Mathematics **185(3)** (2017), 1017–1033.

- [CKM16] H. Cohn, A. Kumar, and G. Minton, Optimal Simplices and Codes in Projectives Space, Geometry & Topology 20 (2016), 1289–1357.
 - [CW12] H. Cohn and J. Woo, *Three-Point Bounds for Energy Minimization*, Journal of the American Mathematical Society 25(4) (2012), 929–958.
 - [CS99] J.H. Conway and N.J.A. Sloane, Sphere packings, lattices and groups, 3rd, Springer, New York, NY, 1999.
 - [DG03] S.B. Damelin and P.J. Grabner, *Energy functionals, numerical integration and asymptotic equidistribution on the sphere*, Journal of Complexity **19** (2003), 231–246.
- [DGM86] I. Daubechies, A. Grossmann, and Y. Meyer, *Painless Nonorthogonal Expansions*, Journal of Mathematical Physics 27(5) (1986), 1271–1283.
 - [Dav99] G. David, Analytic Capacity, Calderón-Zygmund operators, and Rectifiability, Publicacions matemàtiques 43(1) (1999), 3–25.
 - [Dav63] P.J. Davis, Interpolation and Approximation, Blaisdell Publishing Company, New York, 1963.
 - [DX13] F. Dai and Y. Xu, Approximation theory and harmonic analysis on spheres and balls, Springer Monographs in Mathematics, Springer, New York, NY, 2013.
- [dLMdOFV] D. de Laat, F.C. Machado, F.M. de Oliveira Filho, and F. Vallentin, k-point semidefinite programming bounds for equiangular lines, available at https://arxiv.org/pdf/1812. 06045.
 - [Del73] P. Delsarte, An algebraic approach to the association schemes of coding theory, Philips Research Reports Supplements, vol. 10, N. V. Philips' Gloeilampenfabrieken, Eindhoven, Netherlands, 1973.
 - [Del76] _____, Association schemes and t-designs in regular semilattices, Journal of Combinatorial Theory, Series A 20(2) (1976), 230–243.
 - [DGS75] P. Delsarte, J.M. Goethals, and J.J. Seidel, *Bounds for systems of lines, and Jacobi Polynomials*, Philips Research Reports **30** (1975), 91-105.
 - [DGS77] _____, Spherical codes and designs, Geometriae Dedicata 6 (1977), 363–388.
 - [DL93] R.A. DeVore and G.G. Lorentz, *Constructive Approximation*, Grundlehren der mathematischen Wissenschaften, vol. 303, Springer-Verlag, Berlin Heidelberg, 1993.

- [DT97] M. Drmota and R. Tichy, Sequences, discrepancies and applications, Lecture Notes in Mathematics, vol. 1651, Springer-Verlag, Berlin Heidelberg, 1997.
- [Dou64] R.G. Douglas, On Extremal Measures and Subspace Density, Michigan Mathematical Journal 11(3) (1964), 243–246.
- [DS52] R.J. Duffin and A.C. Schaeffer, A class of nonharmonic Fourier Series, Transactions of the American Mathematical Society 72 (1952), 341–366.
- [EO12] M. Ehler and K.A. Okoudjou, *Minimization of the probabilistic p-frame potential*, Journal of Statistic Planning and Inference 142(3) (2012), 645–659.
- [FN08] B. Farkas and B. Nagy, Transfinite Diameter, Chebyshev Constant and Energy on Locally Compact Spaces, Potential Analysis 28 (2008), 241–260.
- [Fas11] G.E. Fasshauer, *Positive Definite Kernels: Past, Present and Future*, Dolomites Research Notes on Approximation 4 (2011), 21-63.
- [FT56] L. Fejes Tóth, On the sum of distances determined by a point set, Acta Mathematica Academiae Scientiarum Hungarica 7 (1956), 387–401.
- [FT59] _____, Über eine Punktverteilung auf der Kugel, Acta Mathematica Academiae Scientiarum Hungarica 10 (1959), 13–19 (German).
- [FS13] F. Finster and D. Schiefeneder, On the Support of Minimizers of Causal Variational Principles, Archive for Rational Mechanics and Analysis 210(2) (2013), 321–364.
- [FH77] C. FitzGerald and R. Horn, On fractional Hadamard powers of positive definite matrices, Journal of Mathematical Analysis and Applications 61(3) (1977), 633–642.
- [FT15] L.C. Flatley and F. Theil, Face-Centered Cubic Crystallization in Atomistic Configurations, Archive for Rational Mechanics and Analysis 218 (2015), 363–416.
- [FVZ16] F. Fodor, V. Vigh, and T. Zarnócz, *On the angle sum of line*, Archiv der Mathematik **106** (2016), 91–100.
- [Föp12] L. Föppl, Stabile Anordnungen von Elektronen im Atom, Journal f
 ür die reine und angewandte Mathematik 141 (1912), 251–302.
- [Fre53] H. Freudenthal, Zur Ebenen Oktavengeometrie, Nederlandse Akademie van Wetenschappen Proceedings, Series A 56 (1953), 195–200 (German).

- [Fug60] B. Fuglede, On the theory of potentials in locally compact spaces, Acta Mathematica 103 (1960), 139–215.
- [Gan67] R. Gangolli, Positive definite kernels on homogeneous spaces and certain stochastic processes related to Lévy's Brownian motion of several parameters, Annales de l'Institut Henri Poincaré, Section B 3(2) (1967), 121–226.
- [GMS74] E.N. Gilbert, F.J. MacWilliams, and N.J.A. Sloane, *Codes which detect deception*, Bell System Technical Journal 53(3) (1974), 405–424.
 - [Gla] A. Glazyrin, *Moments of isotropic measures and optimal projective codes*, available at https: //arxiv.org/pdf/1904.11159.pdf.
 - [GP20] A. Glazyrin and J. Park, *Repeated Minimizers of the p-frame Energies*, SIAM Journal on Discrete Mathematics 34(4) (2020), 2411—2423.
- [Gro96] H. Groemer, *Geometric applications of Fourier series and spherical harmonics*, Encyclopedia of Mathematics and its Applications, vol. 61, Cambridge University Press, Cambridge, 1996.
- [HS04] D.P. Hardin and E.B. Saff, *Discretizing manifolds via minimum energy points*, Notices of the American Mathematical Society 51(10) (2004), 1186–1194.
- [HS05] _____, Minimal Riesz Energy point Configurations for Rectifiable d-dimensional Manifolds, Adv. Math. 193(1) (2005), 174–204.
- [HBZO19] H. He, K. Basu, Q. Zhao, and A. Owen, Permutaiton p-value approximation via generalized Stolarsky invariance, Annals of Statistics 47(1) (2019), 583–611.
 - [Hel78] S. Helgason, *Differential Geometry, Lie Groups, and Symmetric Spaces*, Academic Press, 1978.
 - [Hel84] _____, Groups and geometric analysis: Integral geometry, invariant differential operators, and spherical functions, Academic Press, 1984.
- [HNW11] A. Hinrichs, P. Nickolas, and R. Wolf, A note on the metric geomety of the unit ball, Mathematische Zeitschrift 268 (2011), 887-896.
- [Hog82a] S.G. Hoggar, *t-designs in projective spaces*, European Journal of Combinatorics **3** (1982), 233–254.
- [Hog82b] _____, *Parameters of t-designs in* \mathbb{FP}^{d-1} , European Journal of Combinatorics **3** (1982), 233–254.

- [Hog89] _____, *Tight 4 and 5-designs in projective spaces*, Graphs and Combinatorics **5** (1989), 87–94.
- [HP06] D.D. Holm and V. Putkaradze, Formation of clumps and patches in self-aggregation of finitesized particles, Physica D Nonlinear Phenomena 220(2) (2006), 183–196.
- [Jia08] M. Jiang, *On the sum of distances along a circle*, Discrete Mathematics **308(10)** (2008), 2038–2045.
- [KKLS21] K. Kang, H.K. Kim, T. Lim, and G. Seo, Uniqueness and characterization of local minimizers for the interaction energy with mildly repulsive potentials, Calculus of Variations and Partial Differential Equations 60 (2021).
 - [Kar83] A.F. Karr, Extreme points of certain sets of probability measures, with applications, Mathematics of Operations Research 8(1) (1983), 74–85.
 - [KP02] S.G. Krantz and H.R. Parks, A Primer of Real Analytic Functions, 2nd ed., Birkhäuser, Boston, MA, 2002.
 - [Kel70] J. Kelly, *Combinatorial Inequalities*, Combinatorial Structures and their Applications, Proceedings of the Calgary International Conference on Combinatorial Structures and their Applications (1970), 201–208.
 - [Knu97] D.E. Knuth, *Fundamental algorithms*, Third, The art of computer programming, vol. 1, Addison-Wesley, Reading, MA, 1997.
 - [Kol05] A. Koldobsky, Fourier analysis in convex geometry, Mathematical Surveys and Monographs, vol. 116, American Mathematical Society, Providence, RI, 2005.
- [KSUB11] T. Kolokolnikov, H. Sun, D. Uminsky, and A.L. Bertozzi, *Stability of ring patterns arising from two-dimensional particle interactions*, Physical Review E, Statistical, Non Linear, and Soft Matter Physics 84 (2011), 523–538.
 - [KY94] A.V. Kolushov and V.A. Yudin, *On the Korkin-Zolotarev construction*, Diskretnaya Matematika **6** (1994), 155-157.
 - [KY97] _____, *Extremal dispositions of points on the sphere*, Analysis Mathematica **23** (1997), 25–34.
 - [KM93] J. Korevaar and J.L.H. Meyers, Spherical Faraday cage for the Case of Equal Point Charges and Chebyshev-type quadrature on the Sphere, Integral Transforma and Spectral Function 1(2) (1993), 105-117.

- [KMR01] V. Kozlov, V.G. Maz'ya, and J. Rossman, Spectral Problems Associated with Corner Singularities of Solutions of Elliptic Equations, Mathematical surveys and monographs, vol. 85, American Mathematical Society, Providence, RI, 2001.
 - [KS98] A. Kuijlaars and E. Saff, Asymptotics for minimal discrete energy on the sphere, Transactions of the American Mathematical Society 350(2) (1998), 523–538.
 - [Lan72] N.S. Landkof, Foundations of Modern Potential Theory, Grundlehren der mathematischen Wissenschaften, vol. 180, Springer-Verlag, Berlin Heidelberg, 1972.
 - [Lar62] H Larcher, Solution of a geometric problem by Fejes Tóth, Michigan Mathematics Journal 9 (1962), 45–51.
 - [LS73] P.W.H. Lemmens and J.J. Seidel, *Equiangular lines*, Journal of Algebra 24 (1973), 494–512.
 - [Lev79] V.I. Levenshtein, On bounds for packing in n-dimensional Euclidean space, Doklady Akademii Nauk 25(6) (1979), 1299–1303.
 - [Lev98] _____, Universal bounds for codes and designs, Handbook of coding theory, Vol. I, II (1998), 499–648.
 - [Lev92] _____, Designs as maximum codes in polynomial metric spaces, Acta Applicandae Mathematicae 29 (1992), 1–82.
 - [Lev98] _____, On designs in compact metric spaces and a universal bound on their size, Discrete Mathematics **192(1-3)** (1998), 251-271.
 - [LM21] T. Lim and R.J. McCann, On Fejes Tóth's Conjectured Maximizer for the Sum of Angles Between Lines, Applied Mathematics & Optimization 30 (2021).
 - [LMa] _____, Isodiametry, variance, and regular simplices from particle interactions, available at https://arxiv.org/abs/1907.13593.
 - [LMb] _____, Maximizing powers of the Angle Between Pairs of Points in Projective Space (Appendix by D. Bilyk, A. Glazyrin, R. Matzke, J. Park, and O. Vlasiuk), available at https: //arxiv.org/abs/2007.13052.
 - [Lio08] P.L. Lions, Équations aux dérivées partielles et applications, L'annuaire du Collége de France 118 (2008), 95-104, available at https://www.college-de-france.fr/media/ pierre-louis-lions/UPL40948_lions_cours0708.pdf.
- [LYZ00] E. Lutwak, D. Yang, and G.Y. Zhang, L_p affine isoperimetric inequalities, Journal of Differential Geometry 56 (2000), 111–132.
- [LZ97] E. Lutwak and G.Y. Zhang, *Blaschke-Santaló inequalities*, Journal of Differential Geometry 47(1) (1997), 111–132.
- [Lyu09] Y.I. Lyubich, On tight projective designs, Designs, Codes, and Cryptography 51 (2009), 21–31.
- [MS14] E. Mainini and U. Stefanelli, Crystallization in Carbon Nanostructures, Communications in Mathematical Physics 328 (2014), 545–571.
- [Mak02] A.A. Makhnev, On the nonexistence of strongly regular graphs with the parameters (486, 165, 36, 66), Ukrainian Mathematical Journal 54 (2002), 1137–1146.
 - [MT] D.B. Massey and L.D. Tráng, Notes on Real and Complex Analytic and Semianalytic Singularities.
- [Mat99] J. Matoušek, Geometric discrepancy: An illustrated guide, Algorithms and Combinatorics, vol. 18, Springer-Verlag, Berlin Heidelberg, 1999.
- [MMV96] P. Mattila, M.S. Melnikov, and J. Verdera, *The Cauchy integral, analytic capacity, and uniform rectifiability*, Annals of Mathematics(2) **144(1)** (1996), 127–136.
 - [Mer09] J. Mercer, Functions of positive and negative type and their connection with the theory of integral equations, Philosphical Transactions of the Royal Society A 209(441–458) (1909), 415–446.
- [Mim90] Y. Mimura, *A construction of spherical 2-designs*, Graphs and Combinatorics **6** (1990), 369–373.
- [MEK99] A. Mogilner and L. Edelstein-Keshet, A non-local model for a swarm, Journal of Mathematical Biology 38(6) (1999), 534–570.
- [MEKBS03] A. Mogilner, L. Edelstein-Keshet, L. Bent, and A. Spiros, *Mutual interactions, potentials, and individual distance in a social aggregation*, Journal of Mathematical Biology 47 (2003), 353–389.
 - [Mül66] C. Müller, *Spherical Harmonics*, Lecture Notes in Mathematics, vol. 17, Springer-Verlag, Berlin-New York, 1966.
 - [Mus08] O.R. Musin, *The kissing number in four dimensions*, Annals of Mathematics **168(1)** (2008), 1–32.
 - [Mus14] _____, Multivariate positive definite functions on spheres, Discrete Geometry and Algebraic Combinatorics, Contemporary Mathematics 625 (2014), 177–190.

- [Nes00] Y. Nesterov, Squared functional systems and optimization problems, High performance optimization, Applied Optimization 33 (2000), 405–440.
- [Neu81] A. Neumaier, Combinatorial Configurations in terms of Distances, (Eindvohen University of Technology) Memorandum (1981), 81-90.
- [Nie65] F. Nielsen, On Summen af Afstandene Mellem n Punkter pa en Kugleflade, Nordisk Matematisk Tidskrift 13 (1965), 45-50 (Danish).
- [OS79] A.M. Odlyzko and N.J.A. Sloane, *New bounds on the number of unit spheres that can touch a unit sphere in n dimensions*, Journal of Combinatorial Theory, Series A **26** (1979), 210–214.
- [Pet14] F. Petrov, Maximum Sum of Angles between n Lines, Mathoverflow post (2014), available at https://mathoverflow.net/q/173712.
- [PS76] G. Pólya and G. Szegö, Problems and theorems in analysis, (Revised and enlarged English translation by C. E. Billigheimer), Vol. 2, Springer-Verlag, New York-Heidelberg, 1976.
- [PS31] _____, Über den transfiniten Durchmesser (Kapazitätskonstante) von ebenen und räumlichen Punktmengen, Journal für die reine und angewandte Mathematik 165 (1931), 4–49 (German).
- [Qi05] L. Qi, *Eigenvalues of a real supersymmetric tensor*, Journal of Symbolic Computation **40** (2005), 1302–1324.
- [Ran55] R.A. Rankin, *The Closest Packing of Spherical Caps in n Dimensions*, Glasgoow Mathematical Journal 2(3) (1955), 139–144.
- [Ran56] _____, On the Minimal Points of Positive Definite Quadratic Forms, Mathematika 3(1) (1956), 15–24.
- [Rao47] C.R. Rao, Factorial Experiments Derivable from Combinatorial Arrangements of Arrays, Supplement to the Journal of the Royal Statistical Society 9(1) (1947), 128–139.
- [RBKSC04] J.M. Renes, R. Blume-Kohout, A.J. Scott, and C.M. Caves, Symmetric information complete quantum measurements, Journal of Mathematical Physics 45 (2004), 2171–2182.
 - [Ric57] H. Richter, Parameterfreie Abschätzung und Realisierung von Erwartungswerten, Blätter der Deutsche Gesellschaft für Versicherungsmathematik 3 (1957), 147–162 (German).
 - [Rog58] W.W. Rogosinski, *Moments of non-negative mass*, Proceedings of the Royal Society of London Ser. A 245 (1958), 1–27.

- [Rog62] _____, Non-negative linear functionals, moment problems, and extremum problems in polynomial spaces, Studies in mathematical analysis and related topics; Essays in Honor of George Pólya (1962), 316–324.
- [Rom19] P.C. Romeo, Math Stack Exchange post (2019), available at https://math. stackexchange.com/questions/3114400.
- [Ros51] P.C. Rosenbloom, Quelques classes de problèmes extrémaux, Bulletin de la Société Mathématique de France 79 (1951), 1–58.
- [San15] F. Santambrogio, Optimal Transport for Applied Mathematicians: Calculus of Variations, PDEs, and Modeling, Progress in Nonlinear Differential Equations and Their Applications, vol. 87, Birkhäuser Basel, 2015.
- [Sch93] R. Schneider, Convex Bodies: the Brunn-Minkowsi theory, Encyclopedia of Mathematics and its Applications, vol. 44, Cambridge University Press, Cambridge, 1993.
- [Sch42] I. Schoenberg, Positive Definite Functions on Spheres, Duke Mathematical Journal 9 (1942), 96–108.
- [SvdW53] K. Schütte and B.L. van der Waerden, Das Problem der dreizehn Kugeln, Mathematishe Annalen 125 (1953), 325–334 (German).
 - [Sch] R.E. Schwartz, The phase transition in five point energy minimization, available at https: //arXiv:1610.03303v3.
 - [Sch20] _____, Five Point Energy Minimization: A Synopsis, Constructive Approximation 51 (2020), 537–564.
 - [SG10] A.J. Scott and M. Grassl, Symmetric informationally complete positive operator-valued measures: a new computer study, Journal of Mathematical Physics 51 (2010), 042203.
 - [Sei73] J.J. Seidel, A survey of two-graphs, Colloquio Internazionale sulle Teorie Combinatorie (Rome, 1973), Tomo I (1973), 481–511.
 - [SZ84] P.D. Seymour and T. Zaslavsky, Averaging sets: a Generalization of Mean Values and Spherical Designs, Advances in Mathematics 52(3) (1984), 213–240.
 - [Sha01] O. Shatalov, *Isometric Embeddings* $\ell_2^m \to \ell_p^m$ and Cubature Formulas Over Classical Fields, Ph.D. Thesis, Technion-Israel Institute of Technology (2001).
 - [Sid74] V.M. Sidel'nikov, New Estimates for the Closest Packing of Spheres in n-dimensional Euclidean space, Matematicheskii Sbornik 24 (1974), 148–158.

- [Skr17] M. Skriganov, *Point distributions in compact metric spaces*, Mathematika **63(3)** (2017), 1157–1171.
 - [Skr] _____, Point distributions in compact metric spaces, III. Two-point homogeneous spaces, available at https://arxiv.org/pdf/1701.04545.pdf.
- [Skr20] _____, Stolarsky's invariance principle for projective spaces, Journal of Complexity 56 (2020), 101428.
- [Spe60] G. Sperling, Lösung einer elementargeometrischen Frage von Fejes Tóth, Archiv der Mathematic 11 (1960), 69–71 (German).
- [Sta81] D. Stanton, A Partially ordered set and q-Krawtchouk polynomials, Journal of Combinatorial Theory, Series A 30 (1981), 276–284.
- [Sta86] _____, *t-designs in classical association schemes*, Graphs and Combinatorics **2** (1986), 283–286.
- [SW85] F.H. Stillinger and T.A. Weber, Computer simulation of local order in condensed phases of silicon, Physical Review B 31 (1985), 5262–5271.
- [Sti01] J. Stillwell, *The story of the* 120-*cell*, Notices of the American Mathematical Society **48** (2001), 17–24.
- [Sti93] D.R. Stinson, Combinatorial Designs and Cryptography, Surveys in Combinatorics, 1993, London Mathematical Lecture Note Series 187 (1993), 257–287.
- [Sto73] K.B. Stolarsky, Sums of distances between points on a sphere. II., Proceedings of the American Mathematical Society 41 (1973), 575–582.
- [Sto75] _____, Spherical distributions of N points with maximal distance sums are well spaced, Proceedings of the American Mathematical Society **48** (1975), 203–206.
- [SH03] T. Strohmer and R.W. Heath, Grassmannian Frames with Applications to Coding and Communications, Applied and Computational Harmonic Analysis 14(3) (2003), 257–275.
- [Sze75] G. Szegő, Orthogonal Polynomials, Forth, American Mathematical Society, Providence, RI, 1975.
 - [Szö] F. Szöllősi, All complex equiangular tight frames in dimension 3, available at https:// arxiv.org/pdf/1402.6429.pdf.

- [Tay95] M. Taylor, Cubature for the Sphere and the Discrete Spherical Harmonic Transform, SIAM Journal on Numerical Analysis 32(2) (1995), 667–670.
- [Tch57] V. Tchakaloff, Formules de cubatures mécaniques à coefficients non négatifs, Bulletin des Sciences Mathématiques 81 (1957), 123–134 (French).
- [Tor10] S. Torquato, *Reformulation of the covering and quantizer problems as ground states of inter*acting particles, Physical review E, Statistical, nonlinear, and soft matter physics 82(5) (2010), 056109.
- [Tos06] T. Tossavainen, On the zeros of finite sums of exponential functions, Australian Mathematical Society Gazette 33(1) (2006), 47–50.
- [Tyl87] D.E. Tyler, *A distribution-free M-estimator of multivaritate scatter*, Annals of Statistics **15(1)** (1987), 234–251.
- [UFW17] M. Unser, J. Fageot, and J.P. Ward, *Splines are universal solutions of linear inverse problems* with generalized TV regularization, SIAM Review **59(4)** (2017), 769–793.
- [Ven01] B.B. Venkov, Réseaux euclidiens, designs sphériques et formes modulaires: Réseaux et "designs" sphériques, Monographs of L'Enseignement Mathématique 37 (2001), 87–111 (French).
- [Via17] M. Viazovska, The sphere packing problem in dimension 8, Annals of Mathematics 185(3) (2017), 991–1015.
 - [Vil] N.J. Vilenkin, Special Functions and the Theory of Group Representations, Translations of Mathematics Monographs, vol. 22, American Mathematical Society, Providence, RI.
 - [Vla] O. Vlasiuk, Discreteness of the minimizers of weakly repulsive interaction energies on Riemannian manifolds, available at Preprint: https://arxiv.org/abs/2003.01597.
- [VBUKB12] J.H. Von Brecht, D. Uminsky, T. Kolokolnikov, and A.L. Bertozzi, *Predicting pattern for*mation in particle interactions, Mathematical Models and Methods in Applied Sciences 22 (2012), 87–111.
 - [WV91] G. Wagner and B. Volkmann, *On Averaging Sets*, Monatshefte für Mathematik **111** (1991), 69–78.
 - [Wag92] G. Wagner, On means of distances on the surface of a sphere. II. Upper bounds, Pacific Journal of Mathematics 154(2) (1992), 381–396.

- [Wan52] H.C. Wang, Two-Point Homogeneous Spaces, Annals of Mathematics 55 (1952), 177–191.
- [Wel74] L. Welch, Lower bounds on the maximum cross correlation of signals, IEEE Transactions on Information Theory 20(3) (1974), 397–399.
- [Win98] G. Winkler, *Extreme points of moment sets*, Mathematics of Operations Research **13** (1998), 581–587.
- [Wol11] J. Wolf, *Spaces of constant curvature*, 6th, American Mathematical Society, Providence, RI, 2011.
- [Wol07] _____, Harmonic Analysis on Commutative Spaces, Mathematical Surveys and Monographs, vol. 142, American Mathematical Society, Providence, RI, 2007.
- [Woź91] H. Woźniakowski, Average case complexity of multivariate integration, Bull. Amer. Math. Soc. 24(1) (1991), 887–896.
- [WS15] L. Wu and D. Slepčev, *Nonlocal interaction equations in environments with heterogeneities and boundaries*, Communications in Partial Differential Equations **40(7)** (2015), 1241–1281.
- [XX21] Z. Xu and Z. Xu, *The Minimizers of the p-frame Potential*, Applied and Computational Harmonic Analysis 52 (2021), 366–379.
- [Yud92] V.A. Yudin, *Minimum potential energy of a point system of charges*, Diskretnaya Matematika 4(2) (1992), 115–121.
- [Yud97] _____, Lower bounds for spherical designs, Izvestiya Mathematics 61(3) (1997), 673–683.
- [Zau11] V.G. Zauner, Quantum Designs: Foundations of a Noncommutative Design Theory, International Journal of Quantum Information 9(1) (2011), 445–507.
- [Zel09] V.G. Zelevinsky, *Three-Body Forces and Many-Body Dynamics*, Physics of Atomic Nuclei 72 (2009), 1107–1115.
- [ZST16] G. Zhang, F.G. Stillinger, and S. Torquato, *The Perfect Glass Paradigm: Disordered Hyper-uniform Glasses Down to Absolute Zero*, Scientific Reports 6, No.36963 (2016).
- [ZD83] S. Zhen and G.J. Davies, Calculation of the Lennard-Jones n m potential Energy Parameters for Metals, Physica Status Solidi (A) 78(2) (1983), 595–605.
- [ZCC⁺18] P. Zimmermann, A. Casamayou, N. Cohan, G. Connan, T. Dumont, L. Fousse, F. Maltey, M. Meulien, M. Mezzarobba, C. Pernet, N.M. Thiéry, E. Bray, J. Cremona, M. Forets, A. Ghitza, and H. Thomas, *Computational Mathematics with SageMath*, Society for Industrial and Applied Mathematics, Philadelphia, PA 1 (2018).

[Zuh62] S. Zuhovickii, On the approximation of real functions in the sense of P. L. Čebyšev, American Mathematical Society Translations: Series 2 19 (1962), 221-252.